UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

|--|

CURRENT REPORT

Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): May 2, 2014 (April 28, 2014)

SLM CORPORATION

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 001-13251 (Commission File Number) 52-2013874 (I.R.S. Employer Identification No.)

300 Continental Drive, Newark, Delaware (Address of principal executive offices)

19713 (Zip Code)

Registrant's telephone number, including area code: (302) 283-8000

ck the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following isions:
Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

ITEM 1.01. ENTRY INTO A MATERIAL DEFINITIVE AGREEMENT.

Agreement and Plan of Merger

On April 30, 2014 (the "Distribution Date"), the previously announced spin-off of Navient Corporation ("Navient") from SLM Corporation ("Sallie Mae" or the "Company") was completed. The spin-off was effected through the distribution by the Company of all the shares of common stock of Navient (the "Distribution") to the holders of shares of the Company's common stock, on a one-to-one basis, as of the close of business on April 22, 2014, the record date for the Distribution. As a result of the Distribution, Navient is an independent, publicly traded company that operates the education loan management, servicing and asset recovery business owned by the Company immediately prior to the Distribution.

In connection with the spin-off, an internal corporate reorganization was effected to implement the separation of the loan management, servicing and asset recovery business from our consumer banking business. As part of the reorganization, on April 29, 2014, the Company (formerly named New BLC Corporation) replaced SLM Corporation ("Existing SLM") as the parent holding company of the Company pursuant to a holding company merger (the "Merger"). In accordance with Section 251(g) of the Delaware General Corporation Law (the "DGCL"), by action of the Existing SLM board of directors and without a shareholder vote, Existing SLM was merged into Navient, LLC, a wholly owned subsidiary of the Company, with Navient, LLC surviving ("SLM SurvivorCo"). Immediately following the Merger, the Company changed its name to "SLM Corporation."

In connection with the Merger, the Company, Existing SLM and Navient, LLC entered into an Agreement and Plan of Merger, dated April 28, 2014 (the "Agreement and Plan of Merger"). Pursuant to the Agreement and Plan of Merger and as a result of the Merger:

- all issued and outstanding shares of Existing SLM common stock were converted into shares of the Company's common stock, on a one-for-one basis;
- each series of issued and outstanding shares of Existing SLM preferred stock was converted into the same series of identical shares of the Company's preferred stock, on a one-for-one basis;
- the charter and bylaws of the Company immediately after the effective time of the Merger were substantially identical to the charter and bylaws of Existing SLM immediately prior to the effective time of the Merger;
- the directors of Existing SLM immediately prior to the effective time of the Merger became the directors of the Company at the effective time of the Merger; and
- SLM SurvivorCo retained all of the liabilities of Existing SLM, including its outstanding unsecured public debt and derivative contracts.

Upon consummation of the Merger, the Company's common stock was deemed to be registered under Section 12(b) of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), pursuant to Rule 12g-3(a) promulgated thereunder. For purposes of Rule 12g-3(a), the Company is the successor issuer to Existing SLM.

The foregoing description of the Merger and the Agreement and Plan of Merger is qualified in its entirety by reference to the full text of the Agreement and Plan of Merger, which attached is hereto as Exhibit 2.1 and incorporated by reference herein.

Separation and Distribution Agreement

Following the Merger, through a series of internal transactions, all of the assets and liabilities related to the Company's consumer banking business, including Sallie Mae Bank, the student education loans it holds, a new private education loan servicing company and the Upromise Rewards business were distributed to the Company by SLM SurvivorCo (the successor by merger to Existing SLM). SLM SurvivorCo also distributed the capital stock of Navient to the Company. SLM SurvivorCo, which continues to hold substantially all of the assets and liabilities related to Navient's loan management, servicing and asset recovery business, was then contributed by the Company to Navient and is its wholly owned subsidiary.

On April 28, 2014, the Company entered into a Separation and Distribution Agreement with Existing SLM and Navient. The following description of the Separation and Distribution Agreement is qualified in its entirety by reference to the full text of that agreement, which is attached hereto as Exhibit 2.2 and incorporated by reference herein.

The Separation and Distribution Agreement provides for the internal corporate reorganization by which the assets and liabilities related to (i) the consumer banking business, including Sallie Mae Bank, the student education loans it holds, a new private education loan servicing company and the Upromise Rewards business, were distributed to the Company and (ii) the loan management, servicing and asset recovery business were contributed to Navient. Following the internal corporate reorganization, and pursuant to the Separation and Distribution Agreement, the Company effected the Distribution, pursuant to which to all of the issued and outstanding shares of common stock of Navient were distributed on the Distribution Date, on a one-to-one basis, to the holders of shares of our common stock as of the close of business on April 22, 2014, the record date for the Distribution. The Separation and Distribution Agreement also provides for, among other things:

- the settlement of accounts between the Company and Navient;
- the obligation of each party to indemnify the other against liabilities retained or assumed by that party pursuant to the Separation and the Distribution Agreement and in connection with claims of third parties;
- the allocation among the parties of rights and obligations under insurance policies;
- the agreement of the Company and Navient (i) not to engage in certain competitive business activities for a period of five years, (ii) as to the effect of the non-competition provisions on post-spin M&A activities of the parties and (iii) regarding "first look" opportunities;
- the creation of a governance structure, including a separation oversight committee, by which matters related to the separation and other transactions contemplated by the Separation and Distribution Agreement will be monitored and managed; and
- · confidentiality obligations of the parties.

The Separation and Distribution Agreement further provides for the conditions to the Distribution (the "Conditions"), including (1) the receipt of a private letter ruling from the Internal Revenue Service (the "IRS") to the effect that Merger and the Distribution will generally not result in the recognition of any gain or loss under the Internal Revenue Code, as amended (the "Code") to the Company, Navient or their stockholders (the "PLR Condition")(described in more detail below), (2) receipt of a tax opinion from Baker Botts L.L.P., tax counsel to the Company and Navient, regarding the satisfaction of certain requirements for tax-free treatment under Section 355 of the Code (on which the IRS will not rule)(the "Section 355 Opinion"), (3) receipt of an insolvency opinion from an independent appraisal firm, (4) the SEC having declared the Company's Form 10 registration statement relating to the Distribution effective under the Exchange Act and (5) the acceptance of the Navient common stock for listing on the NASDAQ Global Select Market, subject to official notice of distribution. Under the Separation and Distribution Agreement, any of the Conditions may be waived by the Company.

All of the Conditions were satisfied prior to the Distribution, except for the PLR Condition.

Waiver of PLR Condition

The PLR Condition contemplated that the Company would receive a private letter ruling from the IRS to the effect that, among other things, (i) the Merger would qualify as a "reorganization" within the meaning of Section 368(a)(1)(F) of the Internal Revenue Code and would not be integrated with the rest of the separation and distribution of Navient and (ii) the separation and distribution would qualify as a reorganization for U.S. federal income tax purposes under Sections 355 and 368(a)(1)(D) of the Code and would generally not result in the recognition of any gain or loss to Sallie Mae, Navient or their stockholders. As of the date hereof, the Company

anticipates eventual receipt of the private letter ruling contemplated by the PLR Condition. However, due to unforeseen delays in the review process the PLR Condition was not satisfied on the Distribution Date. The Company determined to waive the PLR Condition and completed the Distribution on April 30, 2014.

In lieu of the private letter ruling, the Company completed the Distribution based upon a tax opinion of Baker Botts L.L.P., which acted as tax advisor to the Company in connection with the separation and distribution of Navient. The tax opinion covers the matters described above for which the private letter ruling is being sought as well as the matters contemplated by the Section 355 Opinion.

The foregoing description of the tax opinion of Baker Botts L.L.P. is qualified in its entirety by reference to the full text of their opinion, including the qualifications and disclaimers set forth therein, which is attached hereto as Exhibit 8.1 and incorporated by reference herein.

Ancillary Agreements

The Company and Navient have entered into several agreements ancillary to the Separation and Distribution Agreement, including the following agreements:

- Transition Services Agreement, dated April 28, 2014, between the Company and Navient;
- Tax Sharing Agreement, dated April 28, 2014, between the Company and Navient;
- · Employees Matters Agreement, dated April 28, 2014, among the Company, Navient and Existing SLM; and
- · Loan Servicing and Administration Agreement, dated April 28, 2014, between Sallie Mae Bank and Navient.

Set forth below is a summary of each of these agreements. Each summary is qualified in its entirety by reference to the full text of the agreement being summarized, which is attached hereto as Exhibits 10.1, 10.2, 10.3 and 10.4, respectively, and incorporated by reference herein.

Transition Services Agreement

The Transition Services Agreement provides for the provision of various services, on an interim transitional basis, by Navient and the Company to each other. The transitional services include access to certain information technology development, hosting and related support services, provision of certain short-term loan servicing functions, customer communications services, access to shared facilities, support services related to third-party transition service obligations and certain student loan trust and other administrative support services. In general, the agreed-upon charges for such services are generally intended to allow the servicing party to recover all out-of-pocket costs and a predetermined profit based on a mark-up of such costs.

Under the Transition Services Agreement, the separation oversight committee will be responsible for monitoring and managing all matters related to the transition and the provision of services by either party to the other. The Transition Services Agreement will terminate on the expiration of the term of the last service provided under it, and in any event no later than 24 months after the Distribution Date.

Tax Sharing Agreement

The Tax Sharing Agreement governs the respective rights, responsibilities and obligations of the Company and Navient after the Distribution relating to taxes, including with respect to the payment of taxes, the preparation and filing of tax returns and the conduct of tax contests. Under the Tax Sharing Agreement, each party is generally liable for taxes attributable to its business. The Tax Sharing Agreement also addresses the allocation of tax liabilities that are incurred as a result of the Merger, separation and distribution. In addition, the Tax Sharing Agreement restricts the parties from taking certain actions that could prevent the Merger, separation and distribution from qualifying for the tax treatment described in the tax opinion of Baker Botts L.L.P.

Employee Matters Agreement

The Employee Matters Agreement governs compensation and employee benefit obligations of the Company and Navient with respect to current and former employees of each company, and generally allocates liabilities and responsibilities relating to employee compensation and benefit plans and programs. The Employee Matters Agreement provides for the treatment of outstanding Existing SLM equity awards as a result of the Distribution. As a general rule, Existing SLM equity awards will be handled as follows: (i) awards granted prior to February 4, 2014 will be divided into both Sallie Mae and Navient equity awards; and (ii) awards granted on or after February 4, 2014 will become solely equity awards of the holder's post-distribution employer. The exercise price of awards will be set so as to maintain the intrinsic value of the original Existing SLM award immediately prior to the Distribution. Sallie Mae and Navient awards will be subject to substantially the same terms, vesting conditions and other restrictions that applied to the original Existing SLM awards they replace.

The Employee Matters Agreement also sets forth general principles relating to various employee matters, including with respect to the transfer of employees between Sallie Mae and Navient, the assumption and retention of liabilities and related assets, expense reimbursements, workers' compensation, leaves of absence and other matters with respect to such employees, the provision of comparable benefits and employee service credits, the sharing of employee information and the duplication or acceleration of benefits to certain employees.

Loan Servicing and Administration Agreement

The Loan Servicing and Administration Agreement governs the terms by which Navient provides servicing, administration and collection services for Sallie Mae Bank's portfolio of FFELP Loans, as well as servicing history information with respect to private education loans previously serviced by Navient and access to certain promissory notes in Navient's possession. The loan servicing and administration agreement has a fixed term with a renewal option in favor of Sallie Mae Bank. In general, Navient will not be able to resign as the servicer or administrator under the agreement except where continued service as the servicer or administrator would violate applicable law, including any rule, regulation or order of the FRB; however, no such resignation shall be effective until a replacement servicer and administrator shall have been appointed. Fees charged by Navient under the loan servicing and administration agreement will be customary for agreements of this type between third parties and paid monthly.

The Company also entered into a joint marketing agreement, a key systems agreement, a data sharing agreement and a sublease agreement with Navient, none of which is material to the Company. For a description of these agreements, see the section captioned "Certain Relationships and Related Party Transactions" in the information statement filed as Exhibit 99.1 to Navient's Registration Statement on Form 10 (File No. 001-36228).

ITEM 5.02 DEPARTURE OF DIRECTORS OR CERTAIN OFFICERS; ELECTION OF DIRECTORS; APPOINTMENT OF CERTAIN OFFICERS; COMPENSATORY ARRANGEMENTS OF CERTAIN OFFICERS.

As previously announced, in connection with the holding company merger (the "Holding Company Reorganization") and pursuant to the terms of the Agreement and Plan of Merger, on April 29, 2014, Raymond J. Quinlan assumed the role of Executive Chairman and Chief Executive Officer of Sallie Mae. As a result of such appointment, Mr. Quinlan is now the principal executive officer of Sallie Mae. Prior to the Holding Company Reorganization, John F. Remondi held the position of President and Chief Executive Officer of Sallie Mae; however, as a result of the Holding Company Reorganization, Mr. Remondi resigned from all positions held with the Company. Following the Distribution, Mr. Remondi will continue to hold the position of Chief Executive Officer of Navient.

Additionally, on April 29, 2014, in connection with the Holding Company Reorganization and pursuant to the terms of the Agreement and Plan of Merger, Steven J. McGarry assumed the role of Executive Vice President and Chief Financial Officer of Sallie Mae. As a result of such appointment, Mr. McGarry is now the principal financial officer and principal accounting officer of Sallie Mae.

Further, in connection with the spin-off of Navient on April 30, 2014, directors Anne Torre Bates, William M. Diefenderfer, III, Diane Suitt Gilleland, Barry A. Munitz, John F. Remondi, Steven L. Shapiro and Barry L. Williams, each tendered their resignation from the Board of Directors of Sallie Mae. Each of the aforementioned individuals is currently a director of Navient and following the completion of the spin-off Sallie Mae and Navient have no common directors. Also on April 30, 2014, immediately following the aforementioned resignations, the remaining members of the Board of Directors (Earl A. Goode, Ronald F. Hunt, Howard H. Newman, Frank C. Puleo, Raymond J. Quinlan, Wolfgang Schoellkopf, and Anthony P. Terracciano) unanimously approved the appointment of Paul G. Child, Joseph A. DePaulo, Carter Warren Franke, Marianne Keler, Jed H. Pitcher, William N. Shiebler and Robert S. Strong to fill the vacancies on the Board of Directors which were created as a result of the aforementioned resignations. Each of the director appointees currently serves as a director of Sallie Mae Bank. Biographical and other information concerning each of the director appointees is set forth in Sallie Mae's most recent definitive proxy statement on Schedule 14A filed with the SEC on April 30, 2014, and such information is incorporated herein by reference. In contemplation of Messrs. Child, Pitcher, Shiebler and Strong and Mses. Franke and Keller being appointed to the Board of Directors, on April 29, 2014, each individual received a bonus payment of \$30,000 in recognition of their increased services to Sallie Mae Bank and their contributions toward the completion of the Distribution.

Following the aforementioned director resignations and appointments, the Board of Directors is in the process of reconstituting its committee memberships and realigning its committee structure to align with its new focus as a standalone consumer banking business. As of the date of this filing it is anticipated that the following new directors will serve on the following committees:

- Audit Committee Jed H. Pitcher and Paul G. Child;
- Compensation, Nominations and Governance Committee Jed H. Pitcher and William N. Shiebler;
- Risk Committee William N. Shiebler, Paul G. Child and Marianne Keler; and
- Preferred Stock Committee Jed H. Pitcher, William N. Shiebler and Robert S. Strong.

Finally, on May 1, 2014, the Company issued the remaining portion of its 2014 long-term incentive awards for its senior executives, including Laurent C. Lutz and Joseph A. DePaulo. As previously announced in the Company's Current Report on Form 8-K filed with SEC on February 19, 2014, the Company previously issued two-thirds of their 2014 long-term incentive award in the form of restricted stock units of the Company's common stock, with the remaining one-third to be issued after the Distribution of Navient. The remaining one-third of the 2014 long-term incentive award was issued in the form of restricted stock units with substantially similar terms as the February 2014 restricted stock unit grant.

ITEM 5.03 AMENDMENTS TO ARTICLES OF INCORPORATION OR BYLAWS; CHANGE IN FISCAL YEAR

On April 28, 2014, the Company filed a certificate of amendment to its amended and restated certificate of incorporation changing its name from New BLC Corporation to SLM Corporation, effective immediately after the effective time of the Merger. The certificate of amendment is attached hereto as Exhibit 3.1.

ITEM 8.01 OTHER EVENTS.

On April 30, 2014, the Company issued a press release announcing the completion of the aforementioned Distribution. A copy of the press release is attached hereto as Exhibit 99.1 and incorporated by reference herein.

ITEM 9.01 FINANCIAL STATEMENTS AND EXHIBITS.

(d) Exhibits

Exhibit Number	Description
2.1*	Agreement and Plan of Merger by and among SLM Corporation, New BLC Corporation and Navient, LLC, dated as of April 28, 2014*
2.2*	Form of Separation and Distribution Agreement by and among SLM Corporation, New BLC Corporation and Navient Corporation, dated as of April 28, 2014*
3.1*	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of New BLC Corporation, filed on April 28, 2014*
8.1*	Opinion of Baker Botts L.L.P. regarding certain tax matters*
10.1*	Transition Services Agreement by and between New Corporation and SLM Corporation, dated as of April 28, 2014*
10.2*	Employee Matters Agreement between SLM Corporation, New BLC Corporation and New Corporation, dated as of April 28, 2014*
10.3*	Tax Sharing Agreement between Navient Corporation and New BLC Corporation, dated as of April 28, 2014*
10.4*	Loan Servicing and Administration Agreement by and among Navient Corporation and Sallie Mae Bank, dated as of April 28, 2014*
99.1*	Press Release of Sallie Mae, dated as of April 30, 2014*

^{*} Filed herewith.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

SLM CORPORATION

Date: May 2, 2014

By: /s/ Laurent C. Lutz

Laurent C. Lutz

Executive Vice President, General Counsel and Secretary

EXHIBIT INDEX

(d) Exhibits

Exhibit Number	Description
2.1*	Agreement and Plan of Merger by and among SLM Corporation, New BLC Corporation and Navient, LLC, dated as of April 28, 2014*
2.2*	Form of Separation and Distribution Agreement by and among SLM Corporation, New BLC Corporation and Navient Corporation, dated as of April 28, 2014*
3.1*	Certificate of Amendment to the Amended and Restated Certificate of Incorporation of New BLC Corporation, filed on April 28, 2014*
8.1*	Opinion of Baker Botts L.L.P. regarding certain tax matters*
10.1*	Transition Services Agreement by and between New Corporation and SLM Corporation, dated as of April 28, 2014*
10.2*	Employee Matters Agreement between SLM Corporation, New BLC Corporation and New Corporation, dated as of April 28, 2014*
10.3*	Tax Sharing Agreement between Navient Corporation and New BLC Corporation, dated as of April 28, 2014*
10.4*	Loan Servicing and Administration Agreement by and among Navient Corporation and Sallie Mae Bank, dated as of April 28, 2014*
99.1*	Press Release of Sallie Mae, dated as of April 30, 2014*

^{*} Filed herewith.

Execution Copy

AGREEMENT AND PLAN OF MERGER

by and among

SLM CORPORATION (Existing SLM)

NEW BLC CORPORATION (SLM BankCo)

and

NAVIENT, LLC (Merger Sub)

Dated as of April 28, 2014

TABLE OF CONTENTS

		Page
ARTICLE I THE MEI	RGER	3
SECTION 1.01.	The Merger	3
SECTION 1.02.	Effective Time	3
SECTION 1.03.	Effects of the Merger	3
SECTION 1.04.	Charter Documents	4
SECTION 1.05.	Directors and Officers	4
SECTION 1.06.	Additional Actions	4
ARTICLE II EFFECT	OF THE MERGER	5
SECTION 2.01.	Effect on Capital	5
SECTION 2.02.	No Surrender of Certificates; Book-Entry Shares	6
SECTION 2.03.	Treatment of Equity Compensation Arrangements	6
ARTICLE III MISCEI	LLANEOUS	7
SECTION 3.01.	Termination	7
SECTION 3.02.	Amendment	7
SECTION 3.03.	Interpretation	7
SECTION 3.04.	Parties in Interest	7
SECTION 3.05.	Counterparts	8
SECTION 3.06.	Entire Agreement	8
SECTION 3.07.	GOVERNING LAW	8
SECTION 3.08.	Assignment	8
SECTION 3.09.	Severability	8

i

This **AGREEMENT AND PLAN OF MERGER** (this "<u>Agreement</u>"), dated as of April 28, 2014, is entered into by and among SLM CORPORATION, a Delaware corporation ("<u>Existing SLM</u>"), NEW BLC CORPORATION, a Delaware corporation and a wholly-owned subsidiary of Existing SLM ("<u>SLM BankCo</u>"), and NAVIENT, LLC, a Delaware limited liability company and a wholly-owned subsidiary of SLM BankCo ("<u>Merger Sub</u>").

WHEREAS, as of the date hereof, the authorized capital stock of Existing SLM consists of (i) 1,125,000,000 shares of common stock, par value \$0.20 per share (the "Existing SLM Common Stock"), of which, as of the close of business on April 14, 2014, 422,739,239 shares were issued and outstanding and (ii) 20,000,000 shares of "blank check" preferred stock, par value \$0.20 per share (the "Existing SLM Preferred Stock");

WHEREAS, as of the date hereof, there are two authorized series of the Existing SLM Preferred Stock: (i) 3,450,000 shares of 6.97% Cumulative Redeemable Preferred Stock, Series A, par value \$0.20 per share (the "Existing SLM Series A Preferred Stock"), of which, as of the close of business on April 27, 2014, 3,300,000 shares were issued and outstanding; and (ii) 4,000,000 shares of Floating Rate Non-Cumulative Preferred Stock, Series B, par value \$0.20 per share (the "Existing SLM Series B Preferred Stock"), of which, as of the close of business on April 27, 2014, 4,000,000 shares were issued and outstanding;

WHEREAS, SLM BankCo is a newly formed Delaware corporation and Merger Sub is a newly formed Delaware limited liability company, each having been formed for the purpose of participating in the transactions herein contemplated;

WHEREAS, immediately prior to the Effective Time, the authorized capital stock of SLM BankCo shall consist of (i) 1,125,000,000 shares of common stock, par value \$0.20 per share (the "<u>SLM BankCo Common Stock</u>"), of which 100 shares will be issued and outstanding and (ii) 20,000,000 shares of "blank check" preferred stock, par value \$0.20 per share, of which there are two authorized series: (i) 3,450,000 shares of 6.97% Cumulative Redeemable Preferred Stock, Series A, par value \$0.20 per share (the "<u>SLM BankCo Series A Preferred Stock</u>"), of which there will be no shares issued and outstanding; and (ii) 4,000,000 shares of Floating Rate Non-Cumulative Preferred Stock, Series B, par value \$0.20 per share (the "<u>SLM BankCo Series B Preferred Stock</u>"), of which there will be no shares issued and outstanding.

WHEREAS, as of the date hereof, SLM BankCo owns 100 common units in Merger Sub (the "Merger Sub Common Units"), representing 100% of the issued and outstanding limited liability company interests of Merger Sub;

WHEREAS, Existing SLM desires to create a new holding company structure by merging Existing SLM with and into Merger Sub (the "Merger"), with Merger Sub being the surviving entity in the Merger (the "Surviving Company"), and in connection therewith, following the Effective Time (as defined below), Merger Sub, as the Surviving Company, will be a direct, wholly-owned subsidiary of SLM BankCo;

WHEREAS, immediately following the Effective Time, the designations, rights, powers and preferences, and the qualifications, limitations and restrictions of (i) the SLM BankCo Common Stock will be the same as those of Existing SLM Common Stock, (ii) the SLM BankCo Series A Preferred Stock will be the same as those of Existing SLM Series B Preferred Stock will be the same as those of Existing SLM Series B Preferred Stock;

WHEREAS, the certificate of incorporation of SLM BankCo and the by-laws of SLM BankCo immediately following the Effective Time will be identical to the Restated Certificate of Incorporation of Existing SLM (the "Existing SLM Charter") and the By-Laws of Existing SLM (the "Existing SLM By-Laws"), in each case, in effect immediately prior to the Effective Time (other than with respect to certain modifications permitted by Section 251(g) of the General Corporation Law of the State of Delaware, as amended (the "DGCL"));

WHEREAS, the limited liability company agreement of the Surviving Company will contain provisions identical to the Existing SLM Charter as in effect immediately prior to the Effective Time (other than references to members rather than shareholders or stockholders, references to units or limited liability company interests rather than stock or shares, references to managers and members rather than directors, a reduction in the number of classes of units or limited liability company interests that the Surviving Corporation is authorized to issue and with such other modifications as are permitted by Section 251(g) of the DGCL);

WHEREAS, the directors of Existing SLM immediately prior to the Effective Time will be the directors of SLM BankCo immediately following the Effective Time;

WHEREAS, the officers of SLM BankCo immediately prior to the Effective Time will continue as and be the officers of SLM BankCo immediately following the Effective Time;

WHEREAS, for U.S. federal income tax purposes, it is intended that (i) the Merger qualify as a "reorganization" within the meaning of Section 368(a)(1) of the Internal Revenue Code of 1986, as amended (the "Code"), and the rules and regulations promulgated thereunder and (ii) this Agreement shall constitute a "plan of reorganization" within the meaning of Sections 354 and 361 of the Code;

WHEREAS, the board of directors of each of Existing SLM and SLM BankCo and the board of managers of Merger Sub have (i) determined that it is in the best interests of their respective companies and respective stockholders and members, and declared it advisable, to enter into this Agreement and (ii) approved the execution, delivery and performance of this Agreement and the consummation of the Merger in accordance with Section 251(g) of the DGCL and Section 18-209(b) of the Delaware Limited Liability Company Act (the "DLLCA") and upon the terms set forth in this Agreement.

NOW, THEREFORE, in consideration of the covenants and agreements contained in this Agreement the parties hereto agree as follows:

ARTICLE I THE MERGER

SECTION 1.01. <u>The Merger</u>. At the Effective Time and subject to the conditions set forth in this Agreement, and in accordance with the provisions of Section 251(g) of the DGCL and Section 18-209 of the DLLCA, Existing SLM shall be merged with and into Merger Sub, the separate corporate existence of Existing SLM shall cease, and Merger Sub shall continue as the Surviving Company.

SECTION 1.02. <u>Effective Time</u>. Upon the terms set forth in this Agreement, the parties hereto shall cause the Merger to be consummated by filing a certificate of merger (the "<u>Certificate of Merger</u>") in substantially the form attached hereto as <u>Exhibit A</u> with the Secretary of State of the State of Delaware in accordance with the relevant provisions of the DGCL and the DLLCA (the time of such filing, or such later time as may be specified in the Certificate of Merger, being referred to as the "<u>Effective Time</u>").

SECTION 1.03. Effects of the Merger. The Merger shall have the effects set forth herein and in the applicable provisions of the DGCL and the DLLCA. Without limiting the generality of the foregoing, at the Effective Time all the assets and property of every kind and every interest in the assets and property, wherever located, and the rights, privileges, immunities, powers, franchises and authority of Merger Sub and Existing SLM shall vest in the Surviving Company, and all obligations and liabilities of Merger Sub and Existing SLM, both known and unknown, shall become the obligations and liabilities of the Surviving Company. Without limiting the generality of the foregoing, all outstanding obligations and liabilities of Existing SLM for outstanding Indebtedness shall become the obligations and liabilities of the Surviving Company. The term "Indebtedness" for this purpose means:

- 1. all obligations of Existing SLM for borrowed money and all obligations of Existing SLM evidenced by bonds, debentures, notes, indentures, loan agreements or other similar instruments;
- 2. all direct or indirect contingent obligations of Existing SLM under letters of credit, bankers' acceptance, bank guaranties, surety bonds and similar instruments;
- 3. all obligations of Existing SLM under any and all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, interest rate swaps or options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing; and
- 4. all obligations of Existing SLM, contingent or otherwise, guaranteeing any of the foregoing obligations or liabilities of any other person or entity.

SECTION 1.04. Charter Documents.

(a) At the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub or Existing SLM, the amended and restated limited liability company agreement of Merger Sub immediately prior to the Effective Time shall be the limited liability company agreement of the Surviving Company until thereafter changed or amended in accordance with the DLLCA and the terms of such limited liability company agreement.

(b) In accordance with Section 251(g) of the DGCL, SLM BankCo agrees to file (and Existing SLM as the sole stockholder of SLM BankCo hereby approves the filing of) a Restated Certificate of Incorporation of SLM BankCo with the Secretary of State of the State of Delaware prior to the Effective Time which Restated Certificate of Incorporation of SLM BankCo (and any other necessary filings) will be identical to the Existing SLM Charter in effect immediately prior to the Effective Time, except that references to the name of Existing SLM shall be replaced with references to the name of SLM BankCo and with such other modifications as may permitted by Section 251(g) of the DGCL. SLM BankCo further agrees to adopt by-laws effective at the Effective Time identical to the Existing SLM By-Laws in effect immediately prior to the Effective Time except that references to the name of Existing SLM shall be replaced with references to the name of SLM BankCo. SLM BankCo agrees to file (and Existing SLM as the sole stockholder of SLM BankCo hereby approves the filing of) a certificate of amendment to its certificate of incorporation to change its name, as of or immediately after the Effective Time, to "SLM Corporation."

SECTION 1.05. Directors and Officers.

(a) <u>Surviving Company</u>. The parties hereto shall take all actions necessary so that (i) the officers of Merger Sub immediately prior to the Effective Time shall be the officers of the Surviving Company and (ii) the managers of the Surviving Company shall consist solely of John Remondi, Somsak Chivavibul, Tim Hynes, John Kane, Steve O'Connell, Sheila Ryan-Macie, Andrew Wachtel Ted Morris, Pat Lawicki and Jerry Maher, in each case to hold office in accordance with the limited liability company agreement of the Surviving Company until their respective successors are duly elected or appointed and qualified or until their earlier resignation or removal.

(b) <u>SLM BankCo</u>. The parties hereto shall take all actions necessary so that (i) the directors of Existing SLM immediately prior to the Effective Time shall be the directors of SLM BankCo immediately after the Effective Time, and (ii) the officers of SLM BankCo immediately prior to the Effective Time shall continue as the officers of SLM BankCo immediately after the Effective Time, in each case to hold office in accordance with the certificate of incorporation and by-laws of SLM BankCo until their respective successors are duly elected or appointed and qualified or until their earlier resignation or removal.

SECTION 1.06. <u>Additional Actions</u>. Subject to the terms of this Agreement, the parties hereto shall take all such reasonable and lawful action as may be necessary or appropriate in order to effectuate the Merger and to comply with the requirements of Section 251(g) of the DGCL and Section 18-209 of the DLLCA. If, at any time after the Effective Time, the Surviving Company shall consider or be advised that any deeds, bills of sale, assignments, assurances or any other actions or things are necessary or desirable to vest, perfect or confirm, of record or otherwise, in the Surviving Company its right, title or interest in, to or under any of the rights, properties or assets of either Merger Sub or Existing SLM acquired or to be acquired by the Surviving Company as a result of, or in connection with, the Merger or

otherwise to carry out this Agreement, the officers and managers of the Surviving Company shall be authorized to secure and deliver, in the name and on behalf of each of Merger Sub and Existing SLM, all such deeds, bills of sale, assignments and assurances and to take and do, in the name and on behalf of each of Merger Sub and Existing SLM or otherwise, all such actions and things as may be necessary or desirable to vest, perfect or confirm any and all right, title and interest in, to and under such rights, properties or assets in the Surviving Company or otherwise to carry out this Agreement.

ARTICLE II EFFECT OF THE MERGER

SECTION 2.01. Effect on Capital. At the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub, Existing SLM, SLM BankCo or any holder of any of the following securities:

- (a) <u>Conversion of Existing SLM Common Stock</u>. Each share of Existing SLM Common Stock issued and outstanding immediately prior to the Effective Time (other than any Existing SLM Common Stock to be cancelled pursuant to Section 2.01(b)) shall be converted automatically into one validly issued, fully paid and nonassessable share of SLM BankCo Common Stock having the same designations, rights, powers and preferences, and the qualifications, limitations and restrictions thereof as the shares of Existing SLM Common Stock being so converted.
- (b) <u>Cancellation of Treasury Stock</u>. Each share of Existing SLM Common Stock that is owned by Existing SLM immediately prior to the Effective Time shall automatically be canceled and shall cease to exist, and no consideration shall be delivered in exchange therefor.
- (c) <u>Conversion of Existing SLM Series A Preferred Stock</u>. Each share of Existing SLM Series A Preferred Stock issued and outstanding immediately prior to the Effective Time shall be converted automatically into one validly issued, fully paid and nonassessable share of SLM BankCo Series A Preferred Stock having the same designations, rights, powers and preferences, and the qualifications, limitations and restrictions thereof as the shares of Existing SLM Series A Preferred Stock being so converted.
- (d) <u>Conversion of Existing SLM Series B Preferred Stock</u>. Each share of Existing SLM Series B Preferred Stock issued and outstanding immediately prior to the Effective Time shall be converted automatically into one validly issued, fully paid and nonassessable share of SLM BankCo Series B Preferred Stock having the same designations, rights, powers and preferences, and the qualifications, limitations and restrictions thereof as the shares of Existing SLM Series B Preferred Stock being so converted.
- (e) <u>Capital of Merger Sub</u>. Each Merger Sub Common Unit issued and outstanding immediately prior to the Effective Time shall remain outstanding as one Common Unit of the Surviving Company.
- (f) <u>Capital Stock of SLM BankCo</u>. Each share of SLM BankCo Common Stock issued and outstanding that is owned by Existing SLM immediately prior to the Effective Time shall automatically be cancelled and shall cease to exist, and no consideration shall be delivered in exchange therefor.

(g) No Appraisal Rights. In accordance with the DGCL and the DLLCA, no appraisal rights shall be available to any holders of Existing SLM Common Stock, Existing SLM Series A Preferred Stock, Existing SLM Series B Preferred Stock or the Merger Sub Common Units in connection with the Merger.

SECTION 2.02. No Surrender of Certificates; Book-Entry Shares. At the Effective Time, the designations, rights, powers and preferences, and qualifications, limitations and restrictions, of each of the SLM BankCo Common Stock, SLM BankCo Series A Preferred Stock and SLM BankCo Series B Preferred Stock will, in each case, be identical with those of Existing SLM Common Stock, Existing SLM Series A Preferred Stock and Existing SLM Series B Preferred Stock, respectively, immediately prior to the Effective Time. Accordingly, each outstanding certificate that, immediately prior to the Effective Time, evidenced outstanding Existing SLM Common Stock, Existing SLM Series A Preferred Stock or Existing SLM Series B Preferred Stock shall, from the Effective Time, be deemed and treated for all corporate purposes to evidence the ownership of the same number of shares of SLM BankCo Common Stock, SLM BankCo Series A Preferred Stock or SLM Series B Preferred Stock or Existing SLM Series B Preferred Stock shall, from the Effective Time, evidenced outstanding Existing SLM Common Stock, Existing SLM Series A Preferred Stock or Existing SLM Series B Preferred Stock shall, from the Effective Time, be deemed and treated for all corporate purposes to evidence the ownership of the same number of shares of SLM BankCo Common Stock, SLM BankCo Series A Preferred Stock or SLM BankCo Series B Preferred Stock, respectively.

SECTION 2.03. Treatment of Equity Compensation Arrangements.

(a) Options. At the Effective Time, each option, whether or not exercisable, to purchase shares of Existing SLM Common Stock which is then outstanding and unexercised (an "Option") shall cease to represent a right to acquire shares of Existing SLM Common Stock and shall be converted automatically into an option to acquire, under the same terms and conditions including exercise price, as were applicable to such Option immediately prior to the Effective Time, shares of SLM BankCo Common Stock, and SLM BankCo shall assume each Option and each plan or agreement pursuant to which such Option was granted. The number of shares of SLM BankCo Common Stock purchasable upon exercise of such Option shall be equal to the number of shares of Existing SLM Common Stock that were purchasable under such Option immediately prior to the Effective Time.

(b) <u>RSUs</u>. At the Effective Time, each outstanding restricted stock unit ("<u>RSU</u>") representing the right to acquire shares of Existing SLM Common Stock shall be automatically converted into an RSU representing the right to acquire, under the same terms and conditions, the same number of shares of SLM BankCo Common Stock and SLM BankCo shall assume each RSU and each plan or agreement pursuant to which such RSU was granted.

(c) Other Rights. At the Effective Time, SLM BankCo shall assume all plans providing for the right to acquire Existing SLM Common Stock and all other rights to acquire, or to receive a distribution of, Existing SLM Common Stock, and such rights shall automatically be converted into a right to acquire, or to receive a distribution of, SLM BankCo Common Stock under the same terms and conditions.

(d) <u>Reservation of Shares</u>. At or prior to the Effective Time, SLM BankCo shall reserve for issuance the number of shares of SLM BankCo Common Stock necessary to satisfy SLM BankCo's obligations under Section 2.03. The terms of each Option, RSU or other right to acquire SLM BankCo Common Stock shall, in accordance with its terms, be subject to adjustment as appropriate to reflect any change in capitalization (as provided for in the applicable plan or agreement) with respect to SLM BankCo Common Stock at or subsequent to the Effective Time.

ARTICLE III MISCELLANEOUS

SECTION 3.01. <u>Termination</u>. This Agreement may be terminated at any time prior to the Effective Time by Existing SLM. In the event of termination of this Agreement, this Agreement shall forthwith become void and have no effect, and none of Existing SLM, SLM BankCo, Merger Sub nor their respective stockholders, directors or officers shall have any liability with respect to such termination or abandonment.

SECTION 3.02. Amendment. This Agreement may not be amended except by an instrument in writing signed by each of the parties hereto.

SECTION 3.03. Interpretation. When a reference is made in this Agreement to an Article or a Section, such reference shall be to an Article of or a Section of this Agreement unless otherwise indicated. The table of contents and headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement. Whenever the words "includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation". The words "hereof", "herein" and "hereunder" and words of similar import when used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The word "or" is not exclusive. All terms defined in this Agreement shall have the defined meanings when used in any certificate or other document made or delivered pursuant hereto unless otherwise defined therein. The definitions contained in this Agreement are applicable to the singular as well as the plural forms of such terms and to the masculine as well as to the feminine and neutral genders of such term. Any law defined or referred to herein means such law as from time to time amended, modified or supplemented. References to a person are also to its permitted successors and assigns.

SECTION 3.04. <u>Parties in Interest</u>. This Agreement shall be binding upon and inure to the benefit of each party hereto and their respective successors and permitted assigns, and nothing in this Agreement, express or implied, is intended to confer upon any other person any rights or remedies of any nature whatsoever under or by reason of this Agreement.

SECTION 3.05. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties (including by facsimile or other electronic image scan transmission).

SECTION 3.06. Entire Agreement. This Agreement constitutes the entire agreement, and supersedes all prior agreements and understandings, both written and oral, among the parties with respect to the subject matter hereof and thereof.

SECTION 3.07. GOVERNING LAW. THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF DELAWARE, WITHOUT REGARD TO ITS CONFLICT OF LAWS PRINCIPLES.

SECTION 3.08. <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, in whole or in part, by operation of law or otherwise by any of the parties without the prior written consent of the other parties, and any assignment without such consent shall be null and void. Subject to the preceding sentence, this Agreement will be binding upon, inure to the benefit of, and be enforceable by, the parties and their respective successors and assigns.

SECTION 3.09. <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect.

[signature page follows]

IN WITNESS WHEREOF, SLM BankCo, Existing SLM and Merger Sub have caused this Agreement to be signed by their respective officers hereunto duly authorized, all as of the date first written above.

SLM CORPORATION

By: /s/ John F. Remondi

Name: John F. Remondi

Title: President and Chief Executive Officer

NEW BLC CORPORATION

By: /s/ Raymond Quinlan
Name: Raymond Quinlan
Title: Chief Executive Officer

NAVIENT, LLC

By: New BLC Corporation, as sole member

By: /s/ John F. Remondi
Name: John F. Remondi
Title: Chief Executive Officer

CERTIFICATE OF MERGER

OF

SLM CORPORATION

(a Delaware corporation)

WITH AND INTO

NAVIENT, LLC

(a Delaware limited liability company)

Pursuant to Section 251(g) of the Delaware General Corporation Law and Section 18-209 of the Delaware Limited Liability Company Act, the undersigned certifies as follows:

FIRST: The name, state of domicile, jurisdiction of formation or organization and type of entity of each of the constituent entities are as follows: (i) SLM Corporation, a Delaware corporation ("Existing SLM"), and (ii) Navient, LLC, a Delaware limited liability company ("Navient").

SECOND: An Agreement and Plan of Merger (the "Merger Agreement") by and among Existing SLM, Navient and New BLC Corporation, a Delaware corporation ("SLM BankCo"), providing for the merger of Existing SLM with and into Navient, with Navient being the surviving limited liability company (the "Merger"), has been approved, adopted, certified, executed and acknowledged by each of SLM BankCo, Existing SLM and Navient in accordance with Section 251(g) of the General Corporation Law of the State of Delaware and Section 18-209 of the Delaware Limited Liability Company Act.

THIRD: The name of the surviving company is Navient, LLC, a Delaware limited liability company (the "Surviving Company").

FOURTH: The amended and restated limited liability company agreement of Merger Sub will be the limited liability company agreement of the Surviving Company, until such time as it may be amended, modified or otherwise changed in accordance with applicable law.

FIFTH: The Merger will become effective on April 29, 2014 at 9:00 a.m. Eastern Time.

SIXTH: The executed Merger Agreement is on file at the principal place of business and office of the Surviving Company, the address of which is 300 Continental Drive, Newark, Delaware 19713.

SEVENTH: A copy of the Merger Agreement will be furnished by the Surviving Company, on request and without cost, to any stockholder or member of the constituent entities to the Merger.

IN WITNESS WHEREOF, Navient, LLC has caused this Certificate of Merger to be duly executed on April 28, 2014.

NAVIENT, LLC

By: New BLC Corporation,

as Sole Member and authorized person

By:

Name: Eric Watson

Title: Vice President and Secretary

Execution Copy

SEPARATION AND DISTRIBUTION AGREEMENT

by and among

SLM CORPORATION (Existing SLM),

NEW BLC CORPORATION (SLM BankCo)

and

NAVIENT CORPORATION (Navient)

Dated as of April 28, 2014

TABLE OF CONTENTS

	Page
ARTICLE I DEFINITIONS	1
ARTICLE II THE SEPARATION	22
2.1 The Reorganization and Separation 2.2 Navient Assets; SLM BankCo Assets 2.3 Navient Liabilities; SLM BankCo Liabilities 2.4 Transfer Documents 2.5 Treatment of Shared Contracts 2.6 Approvals and Notifications 2.7 Misallocations; Waiver of Bulk Sales Laws 2.8 Novation of Navient Liabilities 2.9 Novation of SLM BankCo Liabilities 2.10 Release Under Certain Agreements 2.11 Bank Accounts; Cash Balances 2.12 Ancillary Agreements 2.13 Disclaimer of Representations and Warranties 2.14 Intellectual Property 2.15 Financial Information Certifications 3.16 Misdirected Communications	22 26 29 33 34 35 39 40 41 42 43 44 45 51
2.17 Parent Guarantee	52
ARTICLE III THE DISTRIBUTION	52
 3.1 Sole and Absolute Discretion; Cooperation 3.2 Actions Prior to the Distribution 3.3 Conditions to the Distribution 3.4 The Distribution 	52 53 54 56
ARTICLE IV MUTUAL RELEASES; INDEMNIFICATION	56
 4.1 Release of Pre-Distribution Claims 4.2 Indemnification by Navient 4.3 Indemnification by SLM BankCo 4.4 Indemnification Obligations Net of Insurance Proceeds and Other Amounts 4.5 Procedures for Indemnification of Third Party Claims 4.6 Additional Matters 4.7 Remedies Cumulative 4.8 Survival of Indemnities 	56 59 61 61 62 64 65
ARTICLE V CERTAIN OTHER MATTERS	
 5.1 Insurance Matters 5.2 Late Payments 5.3 Amendment to Existing SLM Holdco Operating Agreement 5.4 Non-Competition Matters 	66 68 68 68

5.5 5.6 5.7 5.8 5.9 5.10		76 77 77 78 78 80
ARTICLE	VI EXCHANGE OF INFORMATION; CONFIDENTIALITY	80
6.1 6.2 6.3 6.4 6.5 6.6 6.7 6.8 6.9 6.10	Agreement for Exchange of Information; Archives Ownership of Information Compensation for Providing Information Record Retention Limitations of Liability Other Agreements Providing for Exchange of Information In-House Lawyers and Other Legal Staff Production of Witnesses; Records; Cooperation Confidentiality Protective Arrangements	80 80 81 81 81 81 84 85
ARTICLE	VII GOVERNANCE AND DISPUTE RESOLUTION	86
7.1 7.2 7.3 7.4	Program Managers; Separation Oversight Committee Escalation Process; Good Faith Negotiation Dispute Resolution Continuation of Services and Commitments	86 88 90 93
ARTICLE	VIII FURTHER ASSURANCES AND ADDITIONAL COVENANTS	93
8.1	Further Assurances	93
ARTICLE	IX TERMINATION	95
9.1 9.2	Termination Effect of Termination	95 95
ARTICLE	X MISCELLANEOUS	95
10.11	Governing Law Assignability Third Party Beneficiaries Notices Severability Force Majeure Publicity	95 95 96 96 97 97 98 98 98
10.13	Specific Performance; Resolution of Certain Disputes Amendments	98 99

10.15 Interpretation	99
10.16 Limitations of Liability	100
10.17 Performance	100
10.18 Schedules.	100

EXHIBITS

Exhibit A [reserved]

Exhibit B Existing SLM Holdco Operating Agreement

Exhibit C Navient By-laws

Exhibit D Navient Certificate of Incorporation

Exhibit E Plan of Merger

Exhibit F SMI Certificate of Incorporation

SCHEDULES

Schedule 1.1(a)NotesSchedule 1.1(b)Securitization TrustsSchedule 1.2(b)SLM BankCo ContractsSchedule 1.5(a)Navient Registrable IPSchedule 1.5(b)SLM BankCo Registrable IPSchedule 1.6Private Loan Servicing AssetsSchedule 1.7Private Loan Servicing Liabilities

Schedule 1.8(a) SLM BankCo Systems Schedule 1.8(b) Excluded Systems

Schedule 1.8(c) Private Loan Servicing Business Systems

Schedule 1.9(a) SLM BankCo Licensed Systems
Schedule 1.9(b) Navient Licensed Systems
Schedule 2.1(m) Equity Interests to be Distributed

Schedule 2.2(a)(ii)

Schedule 2.2(a)(v)(A)

Schedule 2.2(a)(v)(B)

Schedule 2.2(a)(v)(B)

Schedule 2.2(a)(vii)

Navient Entities

Navient Real Property

Navient Leases

Other Navient Assets

Schedule 2.2(b)(i) Assets to be Transferred to SLM BankCo

Schedule 2.2(b)(iii)

SLM BankCo Entities

Schedule 2.2(b)(vi)(A)

SLM BankCo Real Property

Schedule 2.2(b)(vi)(B)

SLM BankCo Leases

Schedule 2.3(a)(v)

Discontinued Businesses

Schedule 2.3(a)(i)(B) Extended Servicing Liability Matters

Schedule 2.3(b)(i) Pre-Separation Consumer Banking Business Liabilities

Schedule 2.5(a) Shared Contracts

Schedule 2.10(a) Agreements to be Terminated

Schedule 2.10(b) Exceptions to Agreements to be Terminated

Schedule 7.1(a) Program Managers

Schedule 7.1(b) Separation Oversight Committee Schedule 7.3 Dispute Resolution Procedures

SEPARATION AND DISTRIBUTION AGREEMENT

This SEPARATION AND DISTRIBUTION AGREEMENT, dated as of April 28, 2014 (this "<u>Agreement</u>"), is by and among SLM Corporation, a Delaware corporation ("<u>Existing SLM</u>"), New BLC Corporation, a Delaware corporation ("<u>SLM BankCo</u>"), and Navient Corporation, a Delaware corporation ("<u>Navient</u>"). Capitalized terms used herein and not otherwise defined shall have the respective meanings assigned to them in <u>Article I.</u>

RECITALS

WHEREAS, the Existing SLM Board has determined that it is in the best interests of Existing SLM and its stockholders to separate into two separate publicly-traded companies;

WHEREAS, in furtherance of the foregoing, the Existing SLM Board has determined that it is appropriate and desirable for Existing SLM to undergo an internal corporate reorganization as a first step in the separation of its education loan management business from its consumer banking business (the "Separation"), and to thereafter make a distribution, on a pro rata basis, to holders of shares of Existing SLM Common Stock on the Record Date of all the outstanding shares of Navient Common Stock owned by SLM BankCo (the "Distribution");

WHEREAS, for U.S. federal income tax purposes, the Separation and the Distribution, taken together, are intended to qualify as a transaction that is tax-free under Sections 355 and 368(a)(1)(D) of the Code;

WHEREAS, it is appropriate and desirable to set forth the principal corporate transactions to be undertaken to effect the Separation and the Distribution and certain other agreements that will govern various matters relating to the Separation and the Distribution and the relationship of SLM BankCo, Navient and their respective Subsidiaries following the Separation and the Distribution; and

WHEREAS, prior to the date hereof certain assets, functions and employees have been consolidated within Sallie Mae Bank pursuant to a pre-existing transition plan, and it is the intention of the Parties that such assets, functions and employees remain within Sallie Mae Bank following the Separation and the Distribution.

NOW, THEREFORE, in consideration of the mutual agreements, provisions and covenants contained in this Agreement, the Parties, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

For the purpose of this Agreement, the following terms shall have the following meanings:

"AAA" shall have the meaning set forth in Section 7.3(c)(ii).

"Acquired Excluded Business" shall have the meaning set forth in Section 5.4(g)(i)(B).

"Action" shall mean any dispute, suit, countersuit, mediation, arbitration or investigation (whether criminal, civil or administrative) by or before any federal, state, local, foreign or international Governmental Authority having authority over a Party or any member of its Group or any mediation or arbitration tribunal.

"Affiliate" shall mean, when used with respect to a specified Person, a Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified Person. For the purpose of this definition, "control" (including with correlative meanings, "controlled by" and "under common control with"), when used with respect to any specified Person shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or other interests, by contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment, undertaking or otherwise. It is expressly agreed that, from and after the Effective Time, for purposes of this Agreement and the Ancillary Agreements, (a) no member of the Navient Group shall be deemed an Affiliate of any member of the SLM BankCo Group and (b) no member of the SLM BankCo Group shall be deemed an Affiliate of any member of the Navient Group, except that the foregoing shall not affect the treatment of SLM BankCo and SMI as "affiliates" for purposes of federal customer data privacy laws, including the Gramm-Leach-Bliley Act, during the Preferred Stock Period.

"Agent" shall mean Computershare Trust Company, N.A., or such other trust company or bank duly appointed by Existing SLM to act as distribution agent, transfer agent and registrar for the shares of Navient Common Stock in connection with the Distribution.

"Agreement" shall have the meaning set forth in the Preamble.

"Ancillary Agreements" shall mean the Transition Services Agreement, the Tax Sharing Agreement, the Employee Matters Agreement, the Loan Servicing Agreement, the Joint Marketing Agreement, the Key Systems Agreement, the Data Sharing Agreement, the Sublease Agreement, any contract, letter or instrument that expressly identifies itself as an "Ancillary Agreement" within the meaning hereof and the Transfer Documents.

"Applicable Deadline" shall have the meaning set forth in Section 7.3(c)(i).

"Applicable Non-Compete Period" shall have the meaning set forth in Section 5.4(f).

"Approvals or Notifications" shall mean any consents, waivers, approvals, permits or authorizations to be obtained from, notices, registrations or reports to be submitted to, or other filings to be made with, any third Person, including any Governmental Authority.

- "Arbitrable Dispute" shall have the meaning set forth in Section 7.3(c).
- "Arbitration Act," means the United States Arbitration Act, 9 U.S.C. §§ 1 et seq.
- "Arbitration Demand Notice" shall have the meaning set forth in Section 7.3(c).
- "Assets" shall mean, with respect to any Person, the assets, properties, claims and rights (including goodwill) of such Person, wherever located (including in the possession of vendors or other third Persons or elsewhere), of every kind, character and description, whether real, personal or mixed, tangible, intangible or contingent, in each case whether or not recorded or reflected or required to be recorded or reflected on the books and records or financial statements of such Person, including rights and benefits pursuant to any contract, license, permit, indenture, note, bond, mortgage, agreement, concession, franchise, instrument, undertaking, commitment, understanding or other arrangement.
 - "ATLAS Development Projects" shall have the meaning set forth in Section 5.9(b).
- "ATLAS Software" shall mean the Software, Technology and related Intellectual Property that comprises the ATLAS System, excluding any Third Party Software.
- "ATLAS System" shall mean the private credit originations engine that supports the primary business functions to (i) enable loan origination and (ii) conduct the core integration of all components necessary to complete loan originations, including all equipment, Software, and associated attachments, features, accessories, peripherals and cabling, and all additions, modifications, substitutions, upgrades or enhancements to such system as of the Effective Time.
 - "Auction Rate Securities and Reset Rate Notes" shall mean all outstanding notes of each series listed on Schedule 1.1(a).
 - "Cash Contribution" shall have the meaning set forth in Section 2.2(b)(iv).
- "Certificate of Merger" shall mean the Certificate of Merger to be filed pursuant to Section 251(g) of the DGCL in furtherance of the Merger, substantially in the form of Exhibit A to the Plan of Merger.
 - "Change in Law or Interpretation" shall have the meaning set forth in Section 2.3(a)(i)(A).
- "Change of Control" shall mean, with respect to Navient or SLM BankCo, as applicable, the occurrence after the Effective Time of any of the following: (a) the sale, conveyance, transfer or other disposition (however accomplished), in one or a series of related transactions, of all or substantially all of the assets of such party or such party's Group to a third Person that is not an Affiliate of such party prior to such transaction or the first of such related transactions; (b) the consolidation, merger or other business combination of a party with or into

any other entity, immediately following which the then-current stockholders of the party, as such, fail to own in the aggregate at least a majority of the voting power in the election of directors of all the outstanding voting securities of the surviving party in such consolidation, merger or business combination or of its ultimate publicly-traded parent entity; (c) a transaction or series of transactions in which any Person or "group" (as such term is used in Sections 13(d) and 14(d) of the Exchange Act) acquires at least a majority of the voting power in the election of directors of all the outstanding voting securities of such party (other than (i) a reincorporation, holding company merger or similar corporate transaction in which each of such party's stockholders owns, immediately thereafter, interests in the new parent company in substantially the same percentage as such stockholder owned in such party immediately prior to such transaction, or (ii) in connection with a transaction described in clause (b), which shall be governed by such clause (b)); or (d) a majority of the board of directors of such party ceasing to consist of individuals who have become directors as a result of being nominated or elected by a majority of such party's directors.

"Citibank Loan Agreement" shall mean that certain Term Loan Agreement dated as of September 17, 2010 by and among Bull Run 1 LLC, as Borrower, SLM, as Guarantor, Citibank, N.A., as the Administrative Agent, the Syndicate Agent, the Collateral Agent, and a Lender, and additional lenders that may become a party thereto.

"Code" shall mean the Internal Revenue Code of 1986, as amended.

"College Planning Tools" means the Software, Technology and related Intellectual Property owned by SLM BankCo and the other members of the SLM BankCo Group as of the Effective Time that comprises the following IT systems: (i) College Planning Calculator (f/k/a Education Investment Planner); (ii) the content hosted on the College Ahead (f/k/a College Answer) website; and (iii) the content hosted on 529.com website. For the avoidance of doubt, "College Planning Tools" shall not include the uniform resource locators (URLs) associated with (ii) and (iii).

"Contribution" shall have the meaning set forth in Section 2.1(k).

"Covered Personnel" shall have the set forth in Section 5.5.

"<u>Data Sharing Agreement</u>" shall mean the Data Sharing Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Deficient Loans" shall have the meaning set forth in Section 4.2(f).

"Delayed Navient Asset" shall have the meaning set forth in Section 2.6(c).

"Delayed Navient Liability" shall have the meaning set forth in Section 2.6(c).

"Delayed SLM BankCo Asset" shall have the meaning set forth in Section 2.6(h).

"Delayed SLM BankCo Liability" shall have the meaning set forth in Section 2.6(h).

"DGCL" shall mean the General Corporation Law of the State of Delaware.

"Disclosure Document" shall mean (i) any registration statement (including the Form 10 and any Form S-8, Form S-1 or Form S-3) filed with the SEC by or on behalf of any Party or any of its controlled Affiliates, and also includes any information statement (including the Information Statement), prospectus, offering memorandum, offering circular, periodic report or similar disclosure document, whether or not filed with the SEC or any other Governmental Authority, which in each case is filed or provided primarily to describe the Separation, the Distribution or the Navient Group or primarily relates to transactions contemplated under this Agreement or any Ancillary Agreement, (ii) any registration statement, periodic report or other disclosure document filed with the SEC or any other Governmental Authority by any Party prior to the Effective Time and (iii) any documents incorporated by reference in the filings described in clauses (i) and (ii) above.

"Dispute" shall have the meaning set forth in Section 7.2(a)(i).

"Dispute Notice" shall have the meaning set forth in Section 7.2(a)(i).

"Distribution" shall have the meaning set forth in the Recitals.

"Distribution Date" shall mean April 30, 2014.

"DLLCA" shall mean the Delaware Limited Liability Company Act.

"D&O Tail Policy" shall mean the six year directors' and officers' liability insurance policy purchased by Existing SLM with respect to liabilities arising prior to the Effective Time.

"<u>DSLP Loans</u>" shall mean educational loans provided by the William D. Ford Federal Direct Loan Program to students and parent borrowers directly through the U.S. Department of Education rather than through a bank or other lender.

"Effective Time" shall mean 4:00 p.m., Eastern Standard Time, on the Distribution Date.

"Effective Time Sallie Mae" shall mean SLM BankCo and its Subsidiaries (including Navient and Existing SLM Holdco), as constituted immediately prior to the Effective Time.

"Eligible Offering" shall have the meaning set forth in Section 5.6(b).

"Employee Matters Agreement" shall mean the Employee Matters Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Equity Distribution" shall have the meaning set forth in Section 2.1(m).

"Equity Interests" of any Person shall mean any membership interests, capital stock or other equity interest or any securities convertible into or exchangeable for equity interests, or any other rights, warrants or options (whether or not currently exercisable) to acquire any of the foregoing securities

"Exchange Act" shall mean the U.S. Securities Exchange Act of 1934, as amended, together with the rules and regulations promulgated thereunder.

"Excluded SMI Projects" has the meaning set forth in the Transition Services Agreement.

"Excluded Systems" shall mean the Effective Time Sallie Mae IT systems listed on Schedule 1.8(b).

"Existing SLM" shall have the meaning set forth in the Preamble; provided, that all references to Existing SLM in this Agreement shall be to Existing SLM as it exists prior to the Merger Effective Time.

"Existing SLM Common Stock" shall mean the common stock, par value \$0.20 per share, of Existing SLM.

"Existing SLM Board" shall mean the Board of Directors of Existing SLM and any authorized committee thereof, as constituted prior to the Merger Effective Time.

"Existing SLM Holdco" shall mean the limited liability company that is the surviving Person in the Merger.

"Existing SLM Holdco Operating Agreement" shall mean the limited liability company agreement of Existing SLM Holdco, substantially in the form of Exhibit B.

"Existing SLM Lawyers" shall have the meaning set forth in Section 6.7(a).

"Existing SLM Series A Preferred Stock" shall mean the 6.97% Cumulative Redeemable Preferred Stock, Series A, par value \$0.20 per share, of Existing SLM.

"Existing SLM Series B Preferred Stock" shall mean the Floating Rate Non-Cumulative Preferred Stock, Series B, par value \$0.20 per share, of Existing SLM.

"Extended Servicing Cut-off Date" shall mean the third anniversary of the Operational Servicing Date.

"FFELP Loans" shall mean education loans to students and parents of students that are guaranteed or reinsured in accordance with the Federal Family Education Loan Program, initiated by Title IV of the Higher Education Act of 1965 and terminated (as to new loans) under the Health Care and Education Reconciliation Act of 2010.

"FHLB-DM Facility" shall mean (a) the Advances, Pledge and Security Agreement between HICA Education Loan Corporation and the Federal Home Loan Bank of Des Moines, dated January 15, 2010, and (b) the Affiliate Collateral Pledge and Security Agreement between SLM Education Credit Finance Corporation, HICA Education Loan Corporation and the Federal Home Loan Bank of Des Moines, dated January 15, 2010.

"First Look Period" shall have the meaning set forth in Section 5.6(b).

"Force Majeure" shall mean a force not reasonably within the control of the party (including acts of God, acts of war, riots, insurrections, illegality of performance, strikes or other industrial disturbances, breakage or accident to machinery or equipment, fires, earthquakes, hurricanes, floods and other disasters). For the avoidance of doubt, the following events shall not constitute Force Majeure events: (a) acts or omissions of subcontractors; and (b) delays or failures to the extent they could have been avoided or their impact mitigated through the use of commercially reasonable business continuity measures.

"Form 10" shall mean the Registration Statement on Form 10 filed by Navient with the SEC to effect the registration of shares of Navient Common Stock pursuant to the Exchange Act in connection with the Distribution, as such Registration Statement may be amended (including by post-effective amendment) or supplemented from time to time prior to the Distribution Date.

"Governmental Approvals" shall mean any notices, reports or other filings to be made, or any consents, registrations, approvals, permits or authorizations to be obtained from, any Governmental Authority.

"Governmental Authority" shall mean any nation or government, any state, municipality or other political subdivision thereof, and any entity, body, agency, commission, department, board, bureau, court, tribunal or other instrumentality, whether federal, state, county, local, domestic, foreign or multinational, exercising executive, legislative, judicial, regulatory, administrative or other similar functions of, or pertaining to, government and any executive official thereof.

"Government Education Loan" shall mean a loan issued expressly for post-secondary educational expenses to a borrower that is made, insured, guaranteed or subsidized by a Governmental Authority, other than a FFELP Loan or a DSLP Loan.

"Group" shall mean either the Navient Group and/or the SLM BankCo Group, as the context requires.

"Historical P/C Policies" means any historical occurrence-based property and casualty policies of insurance held by Effective Time Sallie Mae or its predecessors at any time prior to the Effective Time.

"Indemnifying Party" shall have the meaning set forth in Section 4.4(a).

"Indemnitee" shall have the meaning set forth in Section 4.4(a).

"Indemnity Payment" shall have the meaning set forth in Section 4.4(a).

"Indentures" shall mean (i) that certain Indenture, dated as of October 1, 2000, between Existing SLM (formerly USA Education, Inc.) and The Bank of New York Mellon, as successor to J.P. Morgan Chase Bank, National Association, formerly Chase Manhattan Bank, as amended or supplemented from time to time, and (ii) that certain Indenture, dated as of April 25, 2006, between Existing SLM and Deutsche Trustee Company Limited, as amended or supplemented from time to time.

"Information" shall mean information, whether or not patentable or copyrightable, in written, oral, electronic or other tangible or intangible forms, stored in any medium, including studies, reports, records, books, contracts, instruments, surveys, discoveries, ideas, concepts, know-how, techniques, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, data, computer data, disks, diskettes, tapes, computer programs or other software, marketing plans, customer names, communications by or to attorneys (including attorney-client privileged communications), memos and other materials prepared by attorneys or under their direction (including attorney work product), hard copy audit, external audit and internal control records and other technical, financial, employee or business information or data.

"Information Statement" shall mean the information statement forming part of the Form 10, which is to be delivered to Record Holders in connection with the Distribution, as such information statement may be amended or supplemented from time to time prior to the Distribution Date.

"Initial Cash Contribution" shall mean \$425 million.

"Insurance Proceeds" shall mean those monies (a) received by an insured from an insurance carrier; or (b) paid by an insurance carrier on behalf of the insured.

"Intellectual Property" shall mean all of the following, whether arising under the Laws of the United States or of any foreign or multinational jurisdiction: (a) patents, patent applications (including patents issued thereon) and statutory invention registrations, including

reissues, divisions, continuations, continuations in part, substitutions, renewals, extensions and reexaminations of any of the foregoing, and all rights in any of the foregoing provided by international treaties or conventions, (b) trademarks, service marks, trade names, service names, trade dress, logos and other source or business identifiers, including all goodwill associated with any of the foregoing, and any and all common law rights in and to any of the foregoing, registrations and applications for registration of any of the foregoing, all rights in and to any of the foregoing provided by international treaties or conventions, and all reissues, extensions and renewals of any of the foregoing, (c) Internet domain names, (d) copyrightable works, copyrights, moral rights, mask work rights, database rights and design rights, in each case, other than Software, whether or not registered, and all registrations and applications for registration of any of the foregoing, and all rights in and to any of the foregoing provided by international treaties or conventions, (e) confidential and proprietary information, including trade secrets, invention disclosures, processes and know-how, in each case, other than Software, and (f) intellectual property rights arising from or in respect of any Technology.

"IRS" shall mean the United States Internal Revenue Service.

"IRS Ruling" shall have the meaning set forth in Section 3.3(a)(iii).

"IT Transition" shall have the meaning set forth in Section 5.8.

"IT Transition Milestones" shall have the meaning set forth in the Transition Services Agreement.

"Joint E&O Policy" shall mean the errors and omissions liability insurance policy jointly purchased by SLM BankCo and Navient.

"Joint F.I. Bond Policy" shall mean the Financial Institutions Bond policy jointly purchased by SLM BankCo and Navient.

"Joint Legal Materials" shall have the meaning set forth in Section 6.7(d).

"Joint Marketing Agreement" shall mean the Joint Marketing Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Joint Materials" shall have the meaning set forth in Section 5.9(b).

"Key Systems Agreement" shall mean the Key Systems Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Large New Government Education Loan" shall mean a Government Education Loan created pursuant to a Law adopted or enacted after the Effective Time and as to which

originations are reasonably expected to equal or exceed \$50 billion per annum. For the avoidance of doubt, "Large New Government Education Loan" does not include a loan that consolidates or refinances an education loan.

"Law" shall mean any national, supranational, federal, state, provincial, local or similar law (including common law), statute, code, order, ordinance, rule, regulation, treaty (including any income tax treaty), license, permit, authorization, approval, consent, decree, injunction, binding judicial or administrative interpretation or other requirement, in each case, enacted, promulgated, issued or entered by a Governmental Authority.

"Legacy Indebtedness" shall mean (a) all unsecured notes issued by Existing SLM pursuant to the Indentures, (b) all Auction Rate Securities and Reset Rate Notes, (c) all advances under the FHLB-DM Facility, (d) any indebtedness for borrowed money under the Citibank Loan Agreement and (e) any other indebtedness of SLM Holdco or any of its Subsidiaries as of the Effective Time.

"Legacy Swaps" shall mean all rate swap transactions, basis swaps, credit derivative transactions, forward rate transactions, interest rate swaps or options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swap transactions, cross currency rate swap transactions, currency options, spot contracts, or any other similar transactions or any combination of any of the foregoing entered into by Existing SLM and in effect as of or before the Effective Time, including any of the foregoing entered into in connection with or by reference to any Legacy Indebtedness or Securitization Trust Notes.

"Legal Materials" shall have the meaning set forth in Section 6.7(d).

"<u>Liabilities</u>" shall mean all debts, guarantees, assurances, commitments, liabilities, responsibilities, losses, remediation, deficiencies, damages, fines, penalties, settlements, sanctions, costs, expenses, interest and obligations of any nature or kind, whether absolute or contingent, matured or unmatured, accrued or not accrued, asserted or unasserted, liquidated or unliquidated, foreseen or unforeseen, known or unknown, reserved or unreserved, or determined or determinable, including those arising under any Law, claim (including any Third Party Claim), demand, action, or order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Authority or arbitration tribunal, and those arising under any contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment or undertaking or any fines, damages or equitable relief that is imposed, in each case, including all costs and expenses relating thereto.

"linked" shall have the meaning set forth in Section 2.11(a).

"Loan Servicing Agreement" shall mean the Amended and Restated Loan Servicing and Administration Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Losses" shall mean actual realized losses, costs, damages, amounts paid in settlement or restitution, fines, civil money and other penalties and expenses (including reasonable legal and accounting fees and reasonable expenses and costs of investigation and litigation), whether or not involving a Third Party Claim.

"M&A Out Date" shall have the meaning set forth in Section 5.4(g).

"Mediation Request" shall have the meaning set forth in Section 7.3(b).

"Merger" shall have the meaning set forth in Section 2.1(i).

"MergerSub" shall mean Navient, LLC, a Delaware limited liability company.

"Merger Effective Time" shall mean the date and time when the Merger shall become effective in accordance with the DGCL and the

"NASDAQ" shall mean the NASDAQ Global Select Market.

DLLCA.

"Navient" shall have the meaning set forth in the Preamble.

"Navient Accounts" shall have the meaning set forth in Section 2.11(a).

"Navient Assets" shall have the meaning set forth in Section 2.2(a).

"Navient Balance Sheet" shall mean the unaudited pro forma consolidated balance sheet of Navient, including the notes thereto, as of December 31, 2013, as set forth under the caption "Unaudited Pro Forma Condensed Consolidated Financial Statements" in the Information Statement.

"Navient Board" shall mean the Board of Directors of Navient and any authorized committee thereof.

"Navient Business" shall mean all businesses, operations and activities conducted at any time from and after the Effective Time by the Navient Group.

"Navient Bylaws" shall mean the Amended and Restated Bylaws of Navient, substantially in the form of Exhibit C.

"Navient Certificate of Incorporation" shall mean the Amended and Restated Certificate of Incorporation of Navient, substantially in the form of Exhibit D.

"Navient Common Stock" shall mean the common stock, par value \$0.01 per share, of Navient authorized by the Navient Certificate of Incorporation.

- "Navient Contracts" shall mean any contracts and agreements, other than SLM BankCo Contracts, to which Effective Time Sallie Mae is a party or by which Effective Time Sallie Mae's Assets are bound at or prior to the Effective Time, whether or not in writing, including the following:
 - (a) any customer, marketing, co-marketing, supply or vendor contracts or agreements entered into prior to the Effective Time that relates to the Pre-Separation Education Loan Management Business;
 - (b) (i) any indenture, servicing agreement or administration agreement or other transaction documents entered into by the Securitization Trusts, (ii) any trust agreements governing the Securitization Trusts, (iii) the Excess Distribution Certificates issued by the Securitization Trusts and (iv) any underwriting agreement, pricing agreement, purchase agreement (and the master securitization terms thereto), sale agreement or interim trust agreement entered into by any member of the Navient Group or the SLM BankCo Group in respect of the Securitization Trusts, but shall exclude purchase agreements and the like entered into by any member of the SLM BankCo Group;
 - (c) each FFELP Loan and each Private Education Loan that is not owned by Sallie Mae Bank;
 - (d) subject to Section 2.14, any license agreement related exclusively to the Pre-Separation Education Loan Management Business;
 - (e) any guarantee, indemnity, representation, warranty or other Liability of Effective Time Sallie Mae in respect of any other Navient Contract, any Navient Liability or the Pre-Separation Education Loan Management Business;
 - (f) any employment, change of control, retention, consulting, indemnification, termination, severance or other similar agreement with any employee, independent contractor or consultant of the Navient Group that is in effect as of the Effective Time;
 - (g) any interest rate, currency, commodity or other swap, collar, cap or other hedging or similar agreements or arrangements related exclusively to the Pre-Separation Education Loan Management Business or entered into by or on behalf of any member (including any division or business unit) of the Navient Group;
 - (h) the Indentures and the Legacy Indebtedness;
 - (i) the Legacy Swaps; and

(j) any other documents or agreements not entered into, retained by or expressly assumed by a member of the SLM BankCo Group pursuant to this Agreement or any Ancillary Agreement.

"Navient Counsel" shall have the meaning set forth in Section 6.7(a).

"Navient Entities" shall have the meaning set forth in Section 2.2(a)(ii).

"Navient Excluded Persons" shall mean such Persons (and their respective successors) as may be agreed by Navient and SLM BankCo in writing for purposes of Section 5.4(b)(iii), Section 5.4(b)(iii), and Section 5.4(b)(ii).

"Navient Group" shall mean Navient, each Subsidiary of Navient and each Affiliate of Navient, as constituted as of and after the Effective Time.

"Navient Indemnitees" shall have the meaning set forth in Section 4.3.

"Navient Intellectual Property" shall mean (a) the Navient Names, Navient Marks and Navient Patents, (b) all Other IP that is not used or held for use exclusively in the Pre-Separation Consumer Banking Business, (c) the Intellectual Property that is an integral part of the Excluded Systems, (d) SMI's undivided joint ownership interest in the ATLAS Software and (e) the SMI Licensed IP.

"Navient Liabilities" shall have the meaning set forth in Section 2.3(a).

"Navient Licensed Systems" means the Software, Technology and related Intellectual Property owned by Navient as of the Effective Time that comprises the IT systems identified on Schedule 1.9(b).

"Navient Names and Navient Marks" shall mean the names, marks, trade dress, logos, monograms, domain names and other source or business identifiers of any word or element owned, held or licensed by Effective Time Sallie Mae, including those listed on Schedule 1.5(a) (in block letters or otherwise), either alone or in combination with other words or elements, and all names, marks, trade dress, logos, monograms, domain names and other source or business identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words or elements, together with the goodwill associated with any of the foregoing, in each case other than names, marks, trade dress, logos, monograms, domain names and other source or business identifiers that are SLM BankCo Names and SLM BankCo Marks.

"Navient Licensed Materials" shall have the meaning set forth in Section 2.14(a).

"Navient Patents" means the United States patents identified on Schedule 1.5(a), including all provisionals, divisions, reexaminations, continuations and continuations-in-part, and all reissues and extensions thereof.

"Navient Program Manager" shall have the meaning set forth in Section 7.1(a)(i).

"Navient Restricted Business" shall have the meaning set forth in Section 5.4(b).

"Navient Software" shall mean all Software, other than the SLM BankCo Software, owned or licensed by Effective Time Sallie Mae, including the Software that is an integral part of the Excluded Systems; provided, that Navient Software expressly excludes ATLAS Software and the Software that comprises the SLM BankCo Licensed Systems.

"Navient Technology" shall mean all Technology, other than the SLM BankCo Technology, owned or licensed by Effective Time Sallie Mae, including the Technology that is an integral part of the Excluded Systems; provided, that Navient Technology expressly excludes the Technology that is an integral part of each of the ATLAS Software and the SLM BankCo Licensed Systems.

"Navient Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"Non-assignable Contract" shall have the meaning set forth in Section 2.5(a).

"Non-Competition Period" shall have the meaning set forth in Section 5.4(b).

"Non-Solicitation Period" shall have the meaning set forth in Section 5.5.

"Offering Party" shall have the meaning set forth in Section 5.6(b).

"Operating Agreement Amendment" shall have the meaning set forth in Section 5.3.

"Operating Agreement Amendment Proposal" shall have the meaning set forth in Section 5.3.

"Operational Servicing Date" shall mean the first day on which SLM BankCo and Navient are operating under separate loan servicing clients for Private Education Loans on the loan servicing system of their common servicing system vendor.

"Other IP" shall mean all Intellectual Property that is owned by, licensed by or to, or sublicensed by Effective Time Sallie Mae.

"Parties" shall mean (i) prior to the Effective Time, Existing SLM (including for this purpose only, Existing SLM Holdco after the Merger Effective Time), Navient and SLM BankCo and (ii) from and after the Effective Time, Navient and SLM BankCo.

"Permitted Government Education Loan" shall mean a Government Education Loan that is made by a private lender which retains significant credit risk. For the avoidance of doubt, "Permitted Government Education Loan" does not include a loan that consolidates or refinances an education loan.

"Person" shall mean an individual, a general or limited partnership, a corporation, a trust, a joint venture, an unincorporated organization, a limited liability entity, any other entity and any Governmental Authority.

"Plan of Merger" shall mean the Agreement and Plan of Merger by and among Existing SLM, SLM BankCo and MergerSub, substantially in the form of $\underline{\text{Exhibit E}}$.

"Preferred Stock Period" shall have the meaning set forth in the SMI Certificate of Incorporation.

"Pre-Separation Consumer Banking Business" shall mean the businesses, operations and activities conducted immediately prior to the Effective Time by Sallie Mae Bank, Upromise and Sallie Mae Business Services and their Subsidiaries.

"Pre-Separation Education Loan Management Business" shall mean (a) all businesses, operations and activities conducted at any time prior to the Merger Effective Time by or on behalf of Existing SLM and its Subsidiaries and Affiliates, other than the Pre-Separation Consumer Banking Business, and (b) all businesses, operations and conducted at any time prior to the Merger Effective Time by or on behalf of Existing SLM and its current and former Subsidiaries and Affiliates that have been sold, divested or discontinued.

"Prime Rate" shall mean the latest U.S. prime rate as published by the U.S. edition of the Wall Street Journal under the section "Borrowing Benchmarks—Money Rates—Prime Rate" (or any replacement to that section).

"Private Education Loan" shall mean (A) a loan provided by a private educational lender that (i) is not a Government Education Loan or a FFELP Loan and (ii) is issued expressly for post-secondary educational expenses to a borrower, regardless of whether the loan is provided through the educational institution that the subject student attends or directly to the borrower from the private educational lender. For the avoidance of doubt, "Private Education Loan" does not include (x) an extension of credit under an open end consumer credit plan, a reverse mortgage transaction, a residential mortgage transaction or any other loan that is secured by real property or a dwelling or (y) a loan that consolidates or refinances an education loan.

"Private Loan Servicing Assets" shall mean (i) the Private Loan Servicing Registrable IP, (ii) the Intellectual Property, Software and Technology owned by SMI and that is used or held for use exclusively in the Pre-Separation Consumer Banking Business, including the Intellectual Property, Software and Technology that is an integral part of the SLM BankCo Systems, but excluding the ATLAS Software and the Intellectual Property, Software and Technology that is an integral part of the Excluded Systems, (iii) all of the Equity Interests of Sallie Mae Business Services (iv) those Assets of SMI identified on Schedule 1.6, which are to be transferred to Private ServiceCo, and (v) the assignment of a perpetual license to the Software, Technology and related Intellectual Property that comprises the third-party Software component of the ATLAS System.

"Private Loan Servicing Business" shall mean the business, operations and activities conducted at any time after the Effective Time by Private Service Co.

"Private Loan Servicing Business Systems" shall mean the Effective Time Sallie Mae IT systems listed on Schedule 1.8(c).

"Private Loan Servicing Registrable IP" shall mean the Registrable IP, SLM BankCo Patents, SLM BankCo Name and SLM BankCo Marks set forth on Schedule 1.5(b) to be transferred to Private ServiceCo.

"Private Loan Servicing Liabilities" shall mean all of Liabilities of SMI that are to be transferred to Private ServiceCo and that are not Navient Liabilities, as set forth on Schedule 1.7.

"Private ServiceCo" shall mean SMB Servicing Company, Inc., a Delaware corporation.

"Program Managers" shall have the meaning set forth in Section 7.1(a)(ii).

"RE 2" shall have the meaning set forth in Section 2.1(d).

"RE 3" shall have the meaning set forth in Section 2.1(d).

"RE 5" shall have the meaning set forth in Section 2.1(g).

"Record Date" shall mean 4:00 p.m., Eastern Standard Time, on April [30], 2014.

"Record Holders" shall mean the holders of record of issued and outstanding shares of Existing SLM Common Stock as of the Record Date.

"Registrable IP" shall mean all patents, patent applications, statutory invention registrations, registered trademarks, registered service marks, registered Internet domain names and copyright registrations.

"Release" shall mean any release, spill, emission, discharge, leaking, pumping, pouring, dumping, injection, deposit, disposal, dispersal, leaching or migration of Hazardous Materials into the environment (including, ambient air, surface water, groundwater and surface or subsurface strata).

"<u>Representatives</u>" shall mean, with respect to any Person, any of such Person's directors, officers, employees, agents, consultants, advisors, accountants, attorneys or other representatives.

"Sallie Mae Bank" shall mean Sallie Mae Bank, a Utah industrial bank and insured depository institution.

"Sallie Mae Board" shall mean (i) the Existing SLM Board prior to the Merger Effective Time and (ii) the SLM BankCo Board from and after the Merger Effective Time up to the Effective Time.

"Sallie Mae Business Services" shall mean Sallie Mae Business Services, LLC, a Delaware limited liability company.

"SEC" shall mean the United States Securities and Exchange Commission.

"Securitization Trust Notes" shall mean all notes backed by FFELP Loans or Private Education Loans issued by the Securitization Trusts.

"Securitization Trusts" shall mean any Delaware statutory trust established by any member of the Navient Group or the SLM BankCo Group prior to the Effective Time for the purpose of issuing notes backed by FFELP Loans or Private Education Loans, including those listed on Schedule 1.1(b).

"Security Interest" shall mean any mortgage, security interest, pledge, lien, charge, claim, option, right to acquire, voting or other restriction, right-of-way, covenant, condition, easement, encroachment, restriction on transfer or other encumbrance of any nature whatsoever.

"Separation" shall have the meaning set forth in the Recitals.

"Separation Oversight Committee" shall have the meaning set forth in Section 7.1(b)(i).

"Separation Oversight Period" shall have the meaning set forth in Section 7.1(a)(iii).

"Servicing Liability Cut-off Date" shall mean the third anniversary of the Effective Time.

"Shared Contract" shall have the meaning set forth in Section 2.5(a).

"Shared Policies" shall have the meaning set forth in Section 5.1(c).

"SLM BankCo" shall have the meaning set forth in the Preamble.

"SLM BankCo Accounts" shall have the meaning set forth in Section 2.11(a).

"SLM BankCo Assets" shall have the meaning set forth in Section 2.2(b).

"SLM BankCo Balance Sheet" shall mean the consolidated balance sheet of SLM BankCo, including the notes thereto, as of December 31, 2013, to be filed as an exhibit on Form 8-K by SLM BankCo within four business days after the Distribution Date.

"SLM BankCo Board" shall mean the board of directors of SLM BankCo and any authorized committee thereof, as it is constituted from and after the Merger Effective Time.

"SLM BankCo Business" shall mean the businesses, operations and activities conducted at any time from and after the Effective Time by the SLM BankCo Group.

"SLM BankCo Bylaws" shall mean the Amended and Restated Bylaws of SLM BankCo in effect as of the Effective Time.

"SLM BankCo Certificate of Incorporation" shall mean the Amended and Restated Certificate of Incorporation of SLM BankCo in effect as of the Effective Time.

"SLM BankCo Contracts" shall mean the contracts and agreements (whether or not in writing) listed on Schedule 1.2(b) or entered into, assumed by or assigned to a member of the SLM BankCo Group as of the Effective Time.

"SLM BankCo Common Stock" shall mean the common stock, par value \$0.20 per share, authorized by the SLM BankCo Certificate of Incorporation.

"SLM BankCo Counsel" shall have the meaning set forth in Section 6.7(a).

"SLM BankCo Entities" shall have the meaning set forth in Section 2.2(b)(iii).

"SLM BankCo GAAP Equity" shall mean the consolidated equity of SLM BankCo, as determined under generally accepted accounting principles followed by SLM BankCo, as of the Distribution Date after giving effect to the Separation and the Distribution but before giving effect to the Initial Cash Contribution.

"SLM BankCo Group" shall mean SLM BankCo, each Subsidiary of SLM BankCo and each Affiliate of SLM BankCo, as constituted as of and after the Effective Time.

"SLM BankCo Indemnitees" shall have the meaning set forth in Section 4.2.

"SLM BankCo Intellectual Property" shall mean (a) the SLM Name, SLM BankCo Marks and SLM BankCo Patents, and (b) all Other IP that is used or held for use exclusively in the Pre-Separation Consumer Banking Business, including the Intellectual Property that is an integral part of the SLM BankCo Systems, but excluding the Intellectual Property that is an integral part of the Excluded Systems.

"SLM BankCo Liabilities" shall have the meaning set forth in Section 2.3(b).

"SLM BankCo Licensed Systems" means the Software, Technology and related Intellectual Property owned by SLM BankCo as of the Effective Time that comprises the IT systems identified on Schedule 1.9(a).

"SLM BankCo Patents" means the United States patents identified on Schedule 1.5(b), including all provisionals, divisions, reexaminations, continuations and continuations-in-part, and all reissues and extensions thereof.

"SLM BankCo Program Manager" shall have the meaning set forth in Section 7.1(a)(ii).

"SLM BankCo Restricted Business" shall have the meaning set forth in Section 5.4(c).

"SLM BankCo Series A Preferred Stock" shall mean the 6.97% Cumulative Redeemable Preferred Stock, Series A, par value \$0.20 per shares, of SLM BankCo, as authorized by the SLM BankCo Certificate of Incorporation.

"SLM BankCo Series B Preferred Stock" shall mean the Floating Rate Non-Cumulative Preferred Stock, Series B, par value \$0.20 per shares, of SLM BankCo, as authorized by the SLM BankCo Certificate of Incorporation.

"SLM BankCo Software" shall mean all Software that is owned or licensed by Effective Time Sallie Mae that is used or held for use exclusively in the Pre-Separation Consumer Banking Business, including the Software that is an integral part of the SLM BankCo Systems, but excluding the Software that is an integral part of the Excluded Systems and the Software that comprises the Navient Licensed Systems.

"SLM BankCo Systems" shall mean the Effective Time Sallie Mae IT systems listed on Schedule 1.8(a).

"SLM BankCo Technology" shall mean all Technology that is owned or licensed by Effective Time Sallie Mae that is used or held for use exclusively in the Pre-Separation Consumer Banking Business, including the Technology that is an integral part of the SLM BankCo Systems, but excluding the Technology that is an integral part of each of the Excluded Systems or the Navient Licensed Systems.

"SLM BankCo Transfer Documents" shall have the meaning set forth in Section 2.4(a).

"SLM BankCo Names and SLM BankCo Marks" shall mean the names, marks, trade dress, logos, monograms, domain names and other source or business identifiers of Effective Time Sallie Mae using or containing "Sallie Mae," "SLM," or any other word or element listed on Schedule 1.5(b) (in block letters or otherwise), either alone or in combination with other words or elements, and all names, marks, trade dress, logos, monograms, domain names and other source or business identifiers confusingly similar to or embodying any of the foregoing either alone or in combination with other words or elements, together with the goodwill associated with any of the foregoing.

"SMI" shall mean Sallie Mae, Inc., a Delaware corporation (it being acknowledged that Sallie Mae, Inc. will change its name to Navient Solutions, Inc. on or about the Effective Time).

Exhibit F.

 $\label{eq:smicro} \begin{tabular}{l} \bf ``SMI Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation of SMI, substantially in the form of the amended and the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation of SMI, substantially in the form of the amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation of SMI, substantially in the form of the Amended and Restated Certificate of Incorporation of SMI, substantially in the form of the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation'' shall mean the Amended and Restated Certificate of Incorporation of SMI, substantially in the Incorporation of SMI and Incorpor$

" $\underline{SMI\ Board}" shall\ mean\ the\ board\ of\ directors\ of\ SMI\ and\ any\ authorized\ committee\ thereof.$

"SMI Licensed IP" shall have the meaning set forth in Section 2.1(k)(iv).

"SMI Preferred Stock" shall mean the preferred stock, par value \$0.01 per share, of SMI authorized by the SMI Certificate of

Incorporation.

"SMI Preferred Stock Issuance" shall have the meaning set forth in Section 2.1(n).

"Software" shall mean any and all (a) computer programs, including any and all software implementation of algorithms, models and methodologies, whether in source code, object code, human readable form or other form, (b) databases and compilations, including any and all data and collections of data, whether machine readable or otherwise, (c) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, (d) screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons and icons and (e) documentation, including user manuals and other training documentation, relating to any of the foregoing.

"Sublease Agreement" shall mean the Sublease Agreement relating to the real property in Reston, Virginia to be subleased by Navient (or one of its Subsidiaries) to SLM BankCo (or one or more of its Subsidiaries) to be entered into by and between SLM BankCo and Navient or one or more of their respective Subsidiaries in connection with the Separation and the Distribution.

"Subsidiary" shall mean, with respect to any Person, any corporation, limited liability company, joint venture, trust or partnership of which such Person (a) beneficially owns, either directly or indirectly, more than fifty percent (50%) of (i) the total combined voting power of all classes of voting securities of such Person, (ii) the total combined equity interests or (iii) the capital or profit interests, in the case of a partnership, or (b) otherwise has the power to vote, either directly or indirectly, sufficient securities to elect a majority of the board of directors or similar governing body.

"Taxes" shall have the meaning set forth in the Tax Sharing Agreement.

"Tax Return" shall have the meaning set forth in the Tax Sharing Agreement.

"<u>Tax Sharing Agreement</u>" shall mean the Tax Sharing Agreement to be entered into by and between SLM BankCo and Navient or their respective Subsidiaries in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Technology" shall mean all technology, designs, formulae, algorithms, procedures, methods, discoveries, processes, techniques, ideas, know-how, research and development, technical data, tools, materials, specifications, processes, inventions (whether patentable or unpatentable and whether or not reduced to practice), apparatus, creations, improvements, works of authorship in any media, confidential, proprietary or nonpublic information, and other similar materials, and all recordings, graphs, drawings, reports, analyses and other writings, and other tangible embodiments of the foregoing in any form whether or not listed herein, in each case, other than Software.

"Third Party Claim" shall have the meaning set forth in Section 4.5(a).

"Threatened Action" shall mean any subpoena, written findings of an examination or other notice of intent, formal or informal, sufficient to specifically allege or cite unfair, deceptive, abusive or discriminatory practices or for any other reason to create the material risk of an Action or demand for damages, fines, civil money penalties or restitution by or before any federal, state, local, foreign or international Governmental Authority having authority over a Party or any member of its Group or any mediation or arbitration tribunal.

"Transfer Documents" shall have the meaning set forth in Section 2.4(b).

"<u>Transition Services Agreement</u>" shall mean the Transition Services Agreement to be entered into by and between SLM BankCo and Navient in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.

"Trigger Event" shall have the meaning set forth in Section 4.2(e).

"Unreleased Navient Liability" shall have the meaning set forth in Section 2.8(b).

"Unreleased SLM BankCo Liability" shall have the meaning set forth in Section 2.9(b).

"Upromise" shall mean Upromise, Inc., a Delaware corporation.

ARTICLE II THE SEPARATION

The Reorganization and Separation. Prior to the Distribution, the Parties shall take or cause to be taken the following actions (to the extent not taken prior to the date hereof):

- (a) Formation of Navient; Charter Documents. The Parties acknowledge that Existing SLM caused Navient to be formed as a wholly owned Subsidiary of Existing SLM on November 7, 2013, for the purpose of holding the Navient Assets following the Separation and the Distribution. By no later than the Merger Effective Time, Existing SLM and Navient shall cause to be taken such action as may be required under the DGCL and Navient's charter documents to cause Navient's certificate of incorporation to become the Navient Certificate of Incorporation and Navient's bylaws to become the Navient Bylaws.
- (b) Formation of Private ServiceCo. The Parties acknowledge that SMI caused Private ServiceCo to be formed as a wholly owned Subsidiary of SMI on September 12, 2013, for the purpose of holding the Private Loan Servicing Business following the Separation and the Distribution.
- (c) Formation of SLM BankCo; Charter Documents. The Parties acknowledge that Existing SLM caused SLM BankCo to be formed as a wholly owned Subsidiary of Existing SLM on December 2, 2013, for the purpose of holding the SLM BankCo Assets following the Separation and the Distribution.
- (d) Formation of RE 3. The Parties acknowledge that Existing SLM caused RE Services 2, LLC, a wholly owned subsidiary of Existing SLM ("RE 2"), to form RE Services 3, LLC, a Delaware limited liability company ("RE 3"), as a wholly owned Subsidiary of RE 2 on October 7, 2013, for the purpose of holding RE 2's building located at 8320 Craig Street, Indianapolis, Indiana.
- (e) Contribution to RE 3. The Parties acknowledge that Existing SLM caused RE 2 to contribute the building located at 8320 Craig Street, Indianapolis, Indiana, to RE 3 on April 16, 2014.
- (f) Distribution of RE 3 to Existing SLM. On April 28, 2014, RE 2 distributed 100% of the membership interests of RE 3 to Existing SLM, such that RE 3 is a wholly owned Subsidiary of Existing SLM.

- (g) Contribution of RE 3 and RE 5 to Sallie Mae Bank. On April 28, 2014, Existing SLM contributed 100% of the membership interests in RE 3 and 100% of the stock of RE Services 5, Inc., a Delaware corporation ("RE 5"), to Sallie Mae Bank, such that each of RE 3 and RE 5 is a wholly owned Subsidiary of Sallie Mae Bank.
- (h) Formation of MergerSub. The Parties acknowledge that SLM BankCo caused MergerSub to be formed as a wholly owned Subsidiary of SLM BankCo on April 7, 2014.
- (i) Merger of Existing SLM and MergerSub. On the Distribution Date but prior to the Effective Time, Existing SLM shall merge with and into MergerSub (the "Merger") in accordance with Section 251(g) of the DGCL and Section 18-209 of the DLLCA, the Certificate of Merger and the Plan of Merger. As a result of the Merger and in accordance with the Plan of Merger: (i) MergerSub shall (x) continue as the surviving entity, (y) be governed by the Existing SLM Holdco Operating Agreement and (z) remain a wholly owned Subsidiary of SLM BankCo; (ii) each issued and outstanding share of Existing SLM Common Stock shall be converted into one share of SLM BankCo Common Stock; (iii) each issued and outstanding share of Existing SLM Series A Preferred Stock shall be converted into one share of SLM BankCo Series A Preferred Stock; and (iv) each issued and outstanding share of Existing SLM Series B Preferred Stock shall be converted into one share of SLM BankCo Series B Preferred Stock. The directors of Existing SLM immediately prior to the Merger Effective Time shall be the directors of SLM BankCo as of the Merger Effective Time, and the officers of SLM BankCo immediately prior to the Merger Effective Time shall remain the officers of SLM BankCo as of the Merger Effective Time.
- (j) Name Change of SLM BankCo. SLM BankCo shall file an amendment to the SLM BankCo Certificate of Incorporation to change its name to "SLM Corporation" as of or immediately following the Merger Effective Time, and Existing SLM shall take such action as may be required under applicable Law to permit such filing.
- (k) Contribution of Private Loan Servicing Business. Following the Merger Effective Time, Existing SLM Holdco shall cause SMI to contribute to Private ServiceCo all of the Private Loan Servicing Assets and Private Loan Servicing Liabilities, and Private ServiceCo shall accept such Private Servicing Loan Assets and assume such Private Loan Servicing Liabilities, as follows (the "Contribution"):
 - (i) Transfer and Assignment of Private Loan Servicing Assets. Existing SLM Holdco shall cause SMI to contribute, assign, transfer, convey and deliver to Private ServiceCo, and Private ServiceCo shall accept, from SMI, all of the direct or indirect right, title and interest in and to all of the Private Loan Servicing Assets.

- (ii) Acceptance and Assumption of Private Loan Servicing Liabilities. Existing SLM Holdco shall cause SMI to assign to Private ServiceCo, and Private ServiceCo shall assume and agree faithfully to timely perform, discharge and fulfill in accordance with their respective terms, all of the Private Loan Servicing Liabilities.
- (iii) Contribution of Joint Ownership Interest in ATLAS Software. Existing SLM Holdco shall cause SMI to contribute, assign, transfer, convey and deliver to Private ServiceCo an undivided joint ownership interest in and to the ATLAS Software, subject to SMI's retention of an undivided joint ownership interest and associated rights in and to the ATLAS Software. Subject to the agreement as joint owners to the restrictions set forth in Section 5.9(a), and the rights to certain derivative works as set forth in Section 5.9(a), each joint owner of the ATLAS Software shall otherwise have full rights of ownership with respect to the ATLAS Software, including, but not limited to the right to reproduce, use, develop, improve, perform, display, manufacture, create derivative works of, and otherwise fully exploit the ATLAS Software without restriction and in parallel with each other without the further consent of, or accounting to, the other joint owner. For the avoidance of doubt, subject to Section 5.9(b), all derivative works prepared from the ATLAS Software or any part thereof after the Effective Time shall be and remain the exclusive property of the member of the Navient Group or the SLM BankCo Group, as applicable, developing such derivative work of the ATLAS Software. Each joint owner of the ATLAS Software shall have all right and ability to obtain registrations of the Registrable IP with respect to the derivative works of the ATLAS Software created by it without the consent of, or accounting to, the other joint owner.
- (iv) Contribution of License to SMI Licensed IP. Existing SLM Holdco shall cause SMI to contribute, assign, transfer, convey and deliver to Private ServiceCo a nonexclusive, perpetual, royalty-free, transferable, assignable, sublicenseable, worldwide right and license to all Intellectual Property, Technology and Software, in each case (1) owned by SMI and (2) used or held for use in connection with or integral to the Private Loan Servicing Business Systems (the "SMI Licensed IP"), including, but not limited to the right to reproduce, use, develop, improve, perform, display, distribute, sell, offer to sell, nonexclusively license and sublicense, create derivative works from, and otherwise fully exploit without restriction, the SMI Licensed IP. For the avoidance of doubt, SMI Licensed IP expressly excludes (x) the ATLAS Software (as joint ownership of the same will be granted to Private ServiceCo in accordance with paragraph (ii) above), and (y) Private Loan Servicing Registrable IP (as the same will be contributed to Private ServiceCo in accordance with paragraph (i) above).
- (l) Distribution of Private ServiceCo Stock. Following the Contribution and the Merger Effective Time and prior to the Effective Time, Existing SLM Holdco shall cause SMI to distribute all of the Equity Interests of Private ServiceCo to Existing SLM Holdco, such that Private ServiceCo becomes a wholly owned Subsidiary of Existing SLM Holdco.

- (m) Distribution of Equity Interests of Certain Subsidiaries to SLM BankCo. Following the Contribution and the Merger Effective Time and prior to the Effective Time, Existing SLM Holdco shall distribute all of the Equity Interests of each of the Subsidiaries listed on Schedule 2.1(m) to SLM BankCo and Existing SLM shall withdraw as the sole member of The Sallie Mae Fund and SLM BankCo shall become the sole member of The Sallie Mae Fund (collectively, the "Equity Distribution"), such that each of such Subsidiaries becomes a wholly owned Subsidiary of SLM BankCo.
- (n) Issuance of SMI Preferred Stock. Existing SLM shall cause SMI to file the SMI Certificate of Incorporation with the Delaware Secretary of State prior to its issuance of the SMI Preferred Stock hereunder. Following the Equity Distribution and prior to the Effective Time, SLM BankCo shall purchase from SMI 200 shares of SMI Preferred Stock for a purchase price of \$800 the "SMI Preferred Stock Issuance").
- (o) Contribution of Existing SLM Holdco to Navient. Following the Equity Distribution and the SMI Preferred Stock Issuance and prior to the Effective Time, SLM BankCo shall contribute all of the Equity Interests of Existing SLM Holdco to Navient, such that Existing SLM Holdco becomes a wholly owned Subsidiary of Navient.
- (p) Transfer of Other Assets and Assumption of Other Liabilities. Prior to the Effective Time and to the extent not effected pursuant to the preceding provisions of this Section 2.1:
 - (i) Transfer and Assignment of any Navient Assets held by SLM BankCo. SLM BankCo shall, and shall cause its Subsidiaries to, contribute, assign, transfer, convey and deliver to Navient or a Subsidiary of Navient designated by Navient, and Navient and such Subsidiary shall accept from SLM BankCo and its Subsidiaries, all of SLM BankCo's and such Subsidiaries' respective direct or indirect right, title and interest, in and to Navient Assets, if any, owned or held for use by any member of the SLM BankCo Group immediately prior to the Effective Time.
 - (ii) Acceptance and Assumption of Navient Liabilities. SLM BankCo shall, and shall cause its Subsidiaries to, assign to Navient or a Subsidiary of Navient designated by Navient, and Navient and such Subsidiary shall accept, assume and agree faithfully to timely perform, discharge and fulfill in accordance with their respective terms, Navient Liabilities, if any, held by any member of the SLM BankCo Group immediately prior to the Effective Time.

- (iii) Transfer and Assignment of any SLM BankCo Assets held by Navient. Navient shall, and shall cause its applicable Subsidiaries to, contribute, assign, transfer, convey and deliver to SLM BankCo, or a Subsidiary of SLM BankCo designated by SLM BankCo, and SLM BankCo or such Subsidiary shall accept from Navient and its Subsidiaries, all of Navient's and such Subsidiaries' respective direct or indirect right, title and interest in and to the SLM BankCo Assets, if any, owned or held for use by Navient or any of its Subsidiaries immediately prior to the Effective Time.
- (iv) Acceptance and Assumption of SLM BankCo Liabilities. Navient shall, and shall cause its Subsidiaries to, assign to SLM BankCo or a Subsidiary of SLM BankCo designated by SLM BankCo, and SLM BankCo and such Subsidiary shall accept, assume and agree faithfully to timely perform, discharge and fulfill in accordance with their respective terms, SLM BankCo Liabilities, if any, held by any member of the Navient Group immediately prior to the Effective Time.
- (q) Contribution of Private ServiceCo to Sallie Mae Bank. SLM BankCo shall contribute all of the outstanding capital stock of Private ServiceCo to Sallie Mae Bank, such that Private ServiceCo becomes a wholly owned Subsidiary of Sallie Mae Bank.

2.2 Navient Assets; SLM BankCo Assets.

- (a) Navient Assets. For purposes of this Agreement, "Navient Assets" shall mean all Assets of Effective Time Sallie Mae other than the SLM BankCo Assets, including the following (without limitation or duplication):
 - (i) all Navient Contracts and all rights, interests or claims of Effective Time Sallie Mae therein; <u>provided</u> that Navient Contracts that are Shared Contracts shall be governed by <u>Section 2.5</u> and be treated as Navient Assets to the extent so treated therein;
 - (ii) all issued and outstanding Equity Interests of any existing or previously formed, owned or existing Subsidiary of Effective Time Sallie Mae, other than the SLM BankCo Entities (such Subsidiaries, the "Navient Entities"), including the Subsidiaries listed on Schedule 2.2(a)(ii);

- (iii) except with respect to the license rights granted pursuant to <u>Section 2.14</u>, all rights, interests and claims of Effective Time Sallie Mae to any Navient Names and Navient Marks, Navient Intellectual Property, Navient Patents, Navient Software and Navient Technology, and SMI's undivided joint ownership interest in the ATLAS Software;
- (iv) all Information or other rights and interests with respect to Information of Effective Time Sallie Mae that is exclusively related to the Navient Assets, the Navient Liabilities, the Pre-Separation Education Loan Management Business or the Navient Entities and, subject to the provisions of any applicable Ancillary Agreements, a non-exclusive right to all Information of Effective Time Sallie Mae that is related to (but not exclusively related to) the Navient Assets, the Navient Liabilities, the Pre-Separation Education Loan Management Business or the Navient Entities;
- (v) (A) the owned real property listed on Schedule 2.2(a)(v)(A) and (B) the leases governing the leased real property listed on Schedule 2.2(a)(v)(B);
 - (vi) any and all Assets of any member of the Navient Group, as of the Effective Time, that are not SLM BankCo Assets;
- (vii) the Assets listed on Schedule 2.2(a)(vii) that are owned by the identified member of the SLM BankCo Group immediately prior to the Effective Time; and
- (viii) except as contemplated by Section 2.6(b), any and all other Assets (other than Intellectual Property, Software and Technology), owned or held by Effective Time Sallie Mae that have historically been used or held for use primarily in the Pre-Separation Education Loan Management Business that are not SLM BankCo Assets. The intention of this clause (viii) is to rectify any inadvertent omission of transfer or conveyance of any Asset to a member of the Navient Group that, had the Parties given specific consideration to such Asset as of the date hereof, would have been designated a Navient Asset. No Asset shall be deemed a Navient Asset solely as a result of this clause (viii) if such Asset is within the category or type of Asset expressly covered by the terms of this Agreement or an Ancillary Agreement as being a SLM BankCo Asset unless Navient can establish that the omission of the transfer or conveyance of such Asset to a member of the Navient Group was inadvertent.

All rights of the Navient Group in respect of Existing SLM insurance policies are set forth in <u>Section 5.1</u> and shall not otherwise be included in the Navient Assets.

- (b) SLM BankCo Assets. For the purposes of this Agreement, "SLM BankCo Assets" shall mean (without duplication):
- (i) all Assets (other than SLM BankCo Contracts) that are expressly contemplated by this Agreement or any Ancillary Agreement (or the schedules hereto or thereto) to be Assets to be transferred to or retained by any member of the SLM BankCo Group, including the Private Loan Servicing Assets and the other Assets set forth on Schedule 2.2(b)(i);

- (ii) all SLM BankCo Contracts and all rights, interests or claims of Effective Time Sallie Mae therein, <u>provided</u> that SLM BankCo Contracts that are Shared Contracts shall be governed by <u>Section 2.5</u> and be treated as SLM BankCo Assets to the extent so treated therein:
- (iii) all issued and outstanding Equity Interests in the entities listed on <u>Schedule 2.2(b)(iii)</u> (such Subsidiaries, the "<u>SLM BankCo</u> Entities");
- (iv) any cash or cash equivalents held at a member of the SLM BankCo Group and an amount (the "<u>Cash Contribution</u>"), in cash, equal to the difference between \$1.710 billion and the SLM BankCo GAAP Bank Equity. Prior to the Effective Time, Navient shall deliver, or cause another member of the Navient Group to deliver, for application against the Cash Contribution, an amount equal to the Initial Cash Contribution. The Parties shall true up the difference between the Cash Contribution and the Initial Contribution promptly after the final determination of the SLM BankCo GAAP Equity by SLM BankCo after the Effective Time, and pending payment such difference shall be deemed held in accordance with <u>Section 2.11(e)</u>;
- (v) except with respect to the license rights granted pursuant to <u>Section 2.14</u> and notwithstanding the separation and migration activities to be conducted pursuant to <u>Section 5.8</u>, any and all rights, interests and claims of Effective Time Sallie Mae to any of the SLM BankCo Names and SLM BankCo Marks, SLM BankCo Intellectual Property, SLM BankCo Patents, SLM BankCo Software or SLM BankCo Technology, and Private ServiceCo's undivided joint ownership interest in the ATLAS Software;
- (vi) (A) the owned real property listed on Schedule 2.2(b)(vi)(A) and (B) the leases governing the leased real property listed on Schedule 2.2(b)(vi)(B);
- (vii) any and all Assets of any member of the SLM BankCo Group, as of the Effective Time, that are not expressly identified in this Agreement or in any Ancillary Agreement as Navient Assets;
- (viii) all Information or other rights or interests with respect to Information of Effective Time Sallie Mae that is exclusively related to the SLM BankCo Assets, the SLM BankCo Liabilities, the Pre-Separation Consumer Banking Business or the SLM BankCo Entities and, subject to the provisions of any applicable Ancillary Agreement, a non-exclusive right to all Information that is related to (but not exclusively related to) the SLM BankCo Assets, the SLM BankCo Liabilities, the Pre-Separation Consumer Banking Business or the SLM BankCo Entities; and

(ix) except as contemplated by Section 2.6(b), any and all other Assets (other than Intellectual Property, Software and Technology), owned or held by Effective Time Sallie Mae that have historically been used or held for use primarily in the Pre-Separation Consumer Banking Business. The intention of this clause (ix) is to rectify any inadvertent omission of transfer or conveyance of any Assets to a member of the SLM BankCo Group that, had the Parties given specific consideration to such Asset as of the date hereof, would have been designated a SLM BankCo Asset. No Asset shall be deemed to be a SLM BankCo Asset solely as a result of this clause (ix) if such Asset is within the category or type of Asset expressly covered by the terms of this Agreement or an Ancillary Agreement to be a Navient Asset unless SLM BankCo can establish that the omission of the transfer or conveyance of such Asset to a member of the SLM BankCo Group was inadvertent.

All rights of the SLM BankCo Group in respect of Existing SLM insurance policies are set forth in Section 5.1 and shall not otherwise be included in the SLM BankCo Assets.

2.3 Navient Liabilities; SLM BankCo Liabilities.

- (a) *Navient Liabilities*. For the purposes of this Agreement, "Navient Liabilities" shall mean all Liabilities of Effective Time Sallie Mae other than SLM BankCo Liabilities, including the following Liabilities (without duplication):
 - (i) all Liabilities relating to, arising out of or resulting from the operation of the Pre-Separation Education Loan Management Business and the Pre-Separation Consumer Banking Business at any time prior to the Effective Time, other than Liabilities relating to, arising out of or resulting from the Pre-Separation Consumer Banking Business that are expressly allocated to or assumed by a member of the SLM BankCo Group pursuant to this Agreement or any Ancillary Agreement, other than the servicing and collections activities separately described in Section 2.3(a)(ii):
 - (ii) all Liabilities relating to, arising out of or resulting from:
 - (A) servicing, operations and collections activities performed by Existing SLM or any of its Subsidiaries (including SMI) prior to the Effective Time with respect to Private Education Loans and FFELP Loans that are SLM BankCo Assets or Navient Assets as of the Effective Time; provided, that such Liabilities arise from an Action or Threatened Action involving any member of the SLM BankCo Group with respect to which written notice is provided to Navient in accordance with Section 10.5 by or before the Servicing Liability Cut-off Date (but Navient will not be liable for Liabilities arising from any Action brought directly by any member of the SLM BankCo Group against any member of the Navient

Group with respect to which (i) there is not and has not been any related Action or Threatened Action by a third party against any member of either Group and (ii) a member of the Navient Group is the prevailing party in such Action); and *provided, further*, that (x) Liabilities that would otherwise be Navient Liabilities pursuant to this Section 4.3(a)(ii)(A) shall not be Navient Liabilities if they are due to or result from Laws (excluding for this purpose any interpretation of any Law) that are adopted or become effective after the Effective Time ("New Law"), and (y) in the case of a change in the interpretation of, or position held by, a Governmental Authority after the Effective Time with respect to any Law from the interpretation or position held by such Governmental Authority prior to the Effective Time (a "Change in Interpretation"), such Change in Interpretation will not diminish Navient's liability as provided above (including for restitution, disgorgement and similar monetary obligations, which will be Navient Liabilities), but SLM BankCo (and not Navient) will be responsible for the costs of implementing changes in systems and processes that the parties reasonably agree are required to be undertaken by Navient and/or SLM BankCo with respect to servicing, operations or collections activities on behalf of any member of the SLM BankCo Group attributable to such Change in Interpretation but only to the extent such Change in Interpretation would not otherwise require any member of the Navient Group to implement such changes in systems and processes; and

(B) servicing activities identified on Schedule 2.3(a)(ii)(B) performed by Navient or any of its Subsidiaries (including SMI or its successor) during the period commencing at the Effective Time and ending on the Operational Servicing Date pursuant to the Transition Services Agreement with respect to Private Education Loans owned by SLM BankCo or Navient as of and after the Effective Time; provided, that such Liabilities arise from an Action or Threatened Action involving any member of the SLM BankCo Group with respect to which written notice is provided to Navient in accordance with Section 10.5 by or before the Extended Servicing Liability Cut-off Date (but Navient will not be liable for Liabilities arising from any Action brought directly by any member of the SLM BankCo Group against any member of the Navient Group with respect to which (i) there is not and has not been any related Action or Threatened Action by a third party against any member of either Group and (ii) Navient is the prevailing party in such Action); and provided, further, that (x) Liabilities that would otherwise be Navient Liabilities pursuant to this Section 4.3(a)(ii)(B) shall not be Navient Liabilities if they are due to or result from a New Law that relates to such servicing activities by a member of the Navient Group that occurred prior to the effective date of such New Law and (y) in the case of a Change in

Interpretation affecting such servicing activities, such Change in Interpretation will not diminish Navient's liability as provided above (including for restitution, disgorgement and similar monetary obligations, to the extent they exceed the amount required to correct the borrowers' accounts, which will be Navient Liabilities), but SLM BankCo (and not Navient) will be responsible for the costs of implementing changes in systems and processes that the parties reasonably agree are required to be undertaken by Navient and/or SLM BankCo with respect to servicing, operations or collections activities on behalf of any member of the SLM BankCo Group attributable to such Change in Interpretation but only to the extent such Change in Interpretation would not otherwise require any member of the Navient Group to implement such changes in systems and processes.

- (iii) all Liabilities relating to, arising out of or resulting from the operation of the Navient Business at any time from and after the Effective Time;
- (iv) any and all Liabilities of any Navient Entity that are not expressly provided by this Agreement or any Ancillary Agreement (or the schedules hereto or thereto) to be a SLM BankCo Liability;
- (v) all agreements, obligations and Liabilities of any member of the Navient Group under the Navient Contracts or any Non-Assignable Contracts;
- (vi) all agreements, obligations and Liabilities of any member of the Navient Group under this Agreement or any of the Ancillary Agreements (in each case, subject to the limitations set forth in this Agreement or the Ancillary Agreement, as applicable);
- (vii) except to the extent set forth in <u>Section 2.3(b)(iii)</u>, all Liabilities relating to, arising out of or resulting from any of the businesses and operations terminated, divested, sold or discontinued by Existing SLM or any of its current or former Subsidiaries of Affiliates prior to the Merger Effective Time;
- (viii) (A) all Liabilities relating to, arising out of or resulting from the Legacy Indebtedness, (B) all Liabilities of any member of the SLM BankCo Group or the Navient Group relating to, arising out of or resulting from the Securitization Trust Notes and (C) all Liabilities relating to, arising out of or resulting from the Legacy Swaps;
- (ix) all Liabilities arising out of claims made by any current or former director, officer, employee, or agent of any Party against such Party or any member of the SLM BankCo Group or the Navient Group (or their respective directors, officers and employees) to the extent relating to, arising out of or

resulting from (A) the Pre-Separation Education Loan Management Business or the other business, operations, activities or Liabilities referred to in clauses (i) through (vi) above, inclusive, (B) the Pre-Separation Consumer Banking Business or (C) the Separation and the Distribution;

- (x) all Liabilities arising out of claims made by any current or former common stockholder of Existing SLM or SLM BankCo against any Party (or their respective directors, officers or employees) relating to, arising out of or resulting from (i) actions or omissions of Existing SLM or SLM BankCo at or prior to the Effective Time or (ii) the Separation and the Distribution; and
- (xi) all Liabilities arising out of common claims made within three years after the Effective Time by current or former holders of common stock and preferred stock of Existing SLM or SLM BankCo against any Party (or their respective directors, officers or employees) relating to, arising out of or resulting from (i) actions or omissions of Existing SLM or SLM BankCo at or prior to the Effective Time or (ii) the Separation and the Distribution. For the avoidance of doubt, the claims referred to in this Section 4.3(a)(xi) are those which do not depend, to any material extent, on whether the claimant is a holder of common stock or preferred stock.

Notwithstanding anything to the contrary herein, any Liability allocated to SLM BankCo under the Tax Sharing Agreement shall not be a Navient Liability.

- (b) SLM BankCo Liabilities. For the purposes of this Agreement, "SLM BankCo Liabilities" shall mean (without duplication):
- (i) those Liabilities related to, arising out of or resulting from operation of the Pre-Separation Consumer Banking Business prior to the Effective Time identified on Schedule 2.3(b)(i):
- (ii) the support activities to be provided by SLM BankCo relating to third party transition services agreements, as contemplated by the Transition Services Agreement;
- (iii) all agreements, obligations and Liabilities of any member of the SLM BankCo Group under the SLM BankCo Contracts or any Non-Assignable Contracts;
- (iv) all agreements, obligations and Liabilities of any member of the SLM BankCo Group under this Agreement or any of the Ancillary Agreements (in each case, subject to the limitations set forth in this Agreement or the Ancillary Agreement, as applicable);

- (v) the operation of the SLM BankCo Business at any time from and after the Effective Time; and
- (vi) except to the extent provided in $\underline{Section~4.2(e)}$, all Liabilities arising out of claims made by any current or former preferred stockholder of Existing SLM or SLM BankCo against any Party (or their respective directors, officers or employees) relating to, arising out of or resulting from (i) actions or omissions of Existing SLM or SLM BankCo at or prior to the Effective Time or (ii) the Separation and the Distribution.

Notwithstanding anything to the contrary herein, any Liability allocated to Navient under the Tax Sharing Agreement shall not be a SLM BankCo Liability.

2.4 Transfer Documents.

(a) SLM BankCo Transfer Documents. In furtherance of the contribution, assignment, transfer, conveyance and delivery of Navient Assets and the assignment and assumption of Navient Liabilities in accordance with Section 2.1, on or before the date that such Navient Assets are contributed, assigned, transferred, conveyed or delivered or such Navient Liabilities are assigned and assumed: (i) SLM BankCo shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such assignments, bills of sale, contribution agreements, distribution agreements, quitclaim deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent necessary to evidence the contribution, assignment, transfer, conveyance and delivery of all of SLM BankCo's and its Subsidiaries' right, title and interest in and to the Navient Assets to Navient and its Subsidiaries, and (ii) Navient shall execute and deliver, and shall cause its applicable Subsidiaries to execute and deliver, such assignments and assumptions of contracts and other instruments of assignment and assumption as and to the extent necessary to evidence the valid and effective assignment to and assumption by Navient and its Subsidiaries of the Navient Liabilities. All of the foregoing documents contemplated by this Section 2.4(a) shall be referred to collectively herein as the "SLM BankCo Transfer Documents."

(b) Navient Transfer Documents. In furtherance of the contribution, assignment, transfer, conveyance and delivery of SLM BankCo Assets and the assignment and assumption of SLM BankCo Liabilities set forth in Section 2.1, on or before the date that such SLM BankCo Assets are contributed, assigned, transferred, conveyed or delivered or such SLM BankCo Liabilities are assigned and assumed: (i) Navient shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such assignments, bills of sale, contribution agreements, distribution agreements, quitclaim deeds, stock powers, certificates of title, assignments of contracts and other instruments of transfer, conveyance and assignment as and to the extent

necessary to evidence the contribution, assignment, transfer, conveyance and delivery of all of Navient's and its Subsidiaries' right, title and interest in and to the SLM BankCo Assets to SLM BankCo and its Subsidiaries and (ii) SLM BankCo shall execute and deliver, and shall cause its Subsidiaries to execute and deliver, such assignments and assumptions of contracts and other instruments of assignment and assumption as and to the extent necessary to evidence the valid and effective assignment to and assumption by SLM BankCo and its Subsidiaries of the SLM BankCo Liabilities. All of the foregoing documents contemplated by this Section 2.4(b) shall be referred to collectively herein as the "Navient Transfer Documents" and, together with the SLM BankCo Transfer Documents, the "Transfer Documents."

2.5 Treatment of Shared Contracts.

(a) Subject to applicable Law (including, in the case of the SLM BankCo Group, banking regulations and the advice of the SLM BankCo Group's counsel in respect thereof), and without limiting the generality of the obligations set forth in this Article II, unless the Parties otherwise agree or the benefits of any contract, agreement, arrangement, commitment or understanding described in this Section 2.5 are expressly conveyed to a member of a Group pursuant to this Agreement or an Ancillary Agreement: (i) any contract, agreement, arrangement, commitment or understanding that is listed on Schedule 2.5(a) shall be assigned in part to one or more members of each Group, if so assignable, or appropriately duplicated, novated or amended prior to, on or after the Effective Time, so that each Party or one or more other members of its Group shall, as of the Effective Time, be entitled to the rights and benefits, and shall assume the related portion of any Liabilities, inuring to its respective businesses, in each case, in accordance with the allocation of benefits and burdens set forth on Schedule 2.5(a); and (ii) (A) any contract, agreement, arrangement, commitment or understanding that is an SLM BankCo Asset or SLM BankCo Liability but that, prior to the Effective Time, inured in part to the benefit or burden of any member of the Navient Group (other than any such contract, agreement, arrangement, commitment or understanding covering substantially the same services or arrangements that are covered by a contract, agreement, arrangement, commitment or understanding entered into by a member of the Navient Group in connection with the Separation), and (B) any contract, agreement, arrangement, commitment or understanding that is a Navient Asset or a Navient Liability but that, prior to the Effective Time, inured in part to the benefit or burden of any member of the SLM BankCo Group (other than any such contract, agreement, arrangement, commitment or understanding covering substantially the same services or arrangements that are covered by a contract, agreement, arrangement, commitment or understanding entered into by a member of the SLM BankCo Group in connection with the Separation), shall be assigned in part to the applicable member(s) of the other Group, if so assignable, or appropriately duplicated, novated or amended prior to, on or after the Effective Time, such that SLM BankCo and Navient or the members of its respective Group shall, as of the Effective Time, be entitled to the rights and benefits, and shall

assume the related portion of any Liabilities, that inured to its respective businesses immediately prior to the Effective Time (any contract, agreement, arrangement, commitment or understanding referred to in clause (i) or (ii) above, a "Shared Contract").

(b) Notwithstanding the foregoing, (1) in no event shall any member of either Group be required to assign (or amend) any Shared Contract in its entirety or to assign a portion of any contract, agreement, arrangement, commitment or understanding which is not assignable (or cannot be amended) by its terms (including any terms imposing consents or conditions on an assignment where such consents or conditions have not been obtained or fulfilled) and (2) if any contract, agreement, arrangement, commitment or understanding cannot be so partially assigned by its terms or otherwise, or cannot be duplicated, novated or amended or if such assignment or amendment would impair the benefit the parties thereto derive therefrom (each, a "Non-assignable Contract"), then, each of Navient and SLM BankCo shall, and shall cause their respective Subsidiaries to, take such other reasonable and permissible actions (including by providing prompt notice to SLM BankCo or Navient, as the case may be, with respect to any relevant claim of Liability or other relevant matters arising in connection with a Non-assignable Contract so as to allow SLM BankCo or Navient or other members of its respective Group the ability to exercise any applicable rights thereunder) to cause the appropriate member of the Navient Group or the SLM BankCo Group, as the case may be, to receive the rights and benefits of that portion of each Non-assignable Contract that relates to the Pre-Separation Education Loan Management Business or the Pre-Separation Consumer Banking Business, as the case may be (in each case, to the extent so related), as if such Non-assignable Contract had been assigned to Navient or SLM BankCo or a member of its Group or duplicated, novated or amended, as the case may be, pursuant to this Section 2.5, and to bear the burden of the corresponding Liabilities (including any Liabilities that may arise by reason of such arrangement), as if such Liabilities had been assumed by a member of the applicable Group pursuant to this Section 2.5 as of the

(c) Nothing in this <u>Section 2.5</u> shall require any member of any Group to make any material payment (except to the extent advanced, assumed or agreed in advance to be reimbursed by any member of the other Group), incur any material obligation or grant any material concession for the benefit of any member of the other Group in order to effect any transaction contemplated by this <u>Section 2.5</u>.

2.6 Approvals and Notifications.

(a) Approvals and Notifications for Navient Transfers. To the extent that the transfer or assignment of any Navient Asset, the assignment and assumption of any Navient Liability, the Separation or the Distribution requires any Approvals or Notifications, the Parties shall use their commercially reasonable efforts to obtain or

make such Approvals or Notifications prior to the Effective Time or as soon as reasonably practicable thereafter; <u>provided</u>, <u>however</u>, that, except to the extent expressly provided in this Agreement or any of the Ancillary Agreements or as otherwise agreed between SLM BankCo and Navient, neither SLM BankCo nor Navient (nor any other member of its respective Group) shall be obligated to pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any Person in order to obtain or make such Approvals or Notifications.

- (b) Delayed Navient Transfers. If and to the extent that the valid, complete and perfected transfer or assignment to the Navient Group of any Navient Asset or assignment to or assumption by the Navient Group of any Navient Liability would be a violation of applicable Law or require any Approvals or Notifications in connection with the Separation or the Distribution that have not been obtained or made by the Effective Time, then, unless the Parties mutually shall otherwise determine, the transfer or assignment to the Navient Group of such Navient Assets or the assignment to and assumption by the Navient Group of such Navient Liabilities, as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal impediments are removed or such Approvals or Notifications have been obtained or made. Notwithstanding the foregoing, any such Navient Assets or Navient Liabilities shall continue to constitute Navient Assets and Navient Liabilities for all other purposes of this Agreement.
- (c) Treatment of Delayed Navient Assets and Delayed Navient Liabilities. If any transfer or assignment of any Navient Asset or any assumption of any Navient Liability intended to be transferred, assigned or assumed hereunder, as the case may be, is not consummated on or prior to the Effective Time, whether as a result of the provisions of Section 2.6(b) or for any other reason (any such Navient Asset, a "Delayed Navient Asset" and any such Navient Liability, a "Delayed Navient Liability"), then, insofar as reasonably possible and subject to applicable Law (including banking regulations and the advice of the SLM BankCo Group's counsel in respect thereof), the member of the SLM BankCo Group retaining such Delayed Navient Asset or such Delayed Navient Liability, as the case may be, shall thereafter hold such Delayed Navient Asset or Delayed Navient Liability in trust for the use and benefit of the member of the Navient Group to whom such Delayed Navient Asset is to be transferred or assigned, or which is to assume such Delayed Navient Liability (at the expense of such member of the Navient Group). In addition, the member of the SLM BankCo Group retaining such Delayed Navient Asset or such Delayed Navient Liability shall, insofar as reasonably possible and to the extent permitted by applicable Law (including banking regulations and the advice of the SLM BankCo Group's counsel in respect thereof), treat such Delayed Navient Asset or Delayed Navient Liability as it did prior to the Effective Time in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the member of the Navient Group to whom such Delayed Navient Asset is to be transferred or assigned, or

which will assume such Delayed Navient Liability, as the case may be, in order to place such member of the Navient Group in a substantially similar position as if such Delayed Navient Asset or Delayed Navient Liability had been transferred, assigned or assumed as contemplated hereby and so that all the benefits and burdens of ownership relating to such Delayed Navient Asset or Delayed Navient Liability, as the case may be, including use, risk of loss, potential for gain, profit and dominion, control and command over such Delayed Navient Asset or Delayed Navient Liability, as the case may be, and all costs and expenses related thereto, shall inure from and after the Effective Time to the Navient Group. Each of SLM BankCo and Navient shall, and shall cause the members of its Group to, (i) treat for all Tax purposes the Delayed Navient Assets and the Delayed Navient Liabilities as Assets owned by, and/or Liabilities of, as applicable, Navient, or its Subsidiaries, as applicable, not later than the Effective Time, and (ii) neither report nor take any Tax position (on a Tax Return or otherwise) inconsistent with such treatment (unless required by applicable Law).

- (d) Transfer of Delayed Navient Assets and Delayed Navient Liabilities. If and when the Approvals or Notifications, the absence of which caused the deferral of the transfer or assignment of any Delayed Navient Asset or the deferral of the assignment and assumption of any Delayed Navient Liability pursuant to Section 2.6(b), are obtained or made, and, if and when any other legal impediments for the transfer or assignment of any Delayed Navient Asset or the assignment and assumption of any Delayed Navient Liability have been removed, the transfer or assignment of the applicable Delayed Navient Asset or the assignment and assumption of the applicable Delayed Navient Liability, as the case may be, shall be effected in accordance with the terms of this Agreement or any applicable Ancillary Agreement.
- (e) Costs for Delayed Navient Assets and Delayed Navient Liabilities. No member of the SLM BankCo Group retaining a Delayed Navient Asset or Delayed Navient Liability due to the deferral of the transfer or assignment of such Delayed Navient Asset or the deferral of the assignment and assumption of such Delayed Navient Liability, as the case may be, shall be obligated, in connection with the foregoing, to expend any money unless the necessary funds are advanced (or otherwise made available) by Navient or another member of the Navient Group, other than reasonable out-of-pocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by Navient or the member of the Navient Group entitled to such Delayed Navient Asset or Delayed Navient Liability.
- (f) Approvals and Notifications for SLM BankCo Transfers. To the extent that the transfer or assignment of any SLM BankCo Asset or the assignment and assumption of any SLM BankCo Liability requires any Approvals or Notifications, the Parties shall use their commercially reasonable efforts to obtain or make such Approvals or Notifications prior to the Effective Time or as soon as reasonably practicable thereafter; provided, however, that, except to the extent expressly provided in this Agreement or in any Ancillary Agreement or as otherwise agreed between SLM BankCo

and Navient, neither SLM BankCo nor Navient nor any member of their respective Groups shall be obligated to contribute capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any Person in order to obtain or make such Approvals or Notifications.

- (g) Delayed SLM BankCo Transfers. If and to the extent that the valid, complete and perfected transfer or assignment to the SLM BankCo Group of any SLM BankCo Asset or assignment to and assumption by the SLM BankCo Group of any SLM BankCo Liability would be a violation of applicable Law or require any Approval or Notification that has not been obtained or made by the Effective Time, then, unless the parties hereto mutually shall otherwise determine, the transfer or assignment to the SLM BankCo Group of such SLM BankCo Assets or the assignment to and assumption by the SLM BankCo Group of such SLM BankCo Liabilities, as the case may be, shall be automatically deemed deferred and any such purported transfer, assignment or assumption shall be null and void until such time as all legal impediments are removed or such Approval or Notification has been obtained or made. Notwithstanding the foregoing, any such SLM BankCo Assets or SLM BankCo Liabilities shall continue to constitute SLM BankCo Assets and SLM BankCo Liabilities for all other purposes of this Agreement.
- (h) Treatment of Delayed SLM BankCo Assets and Delayed SLM BankCo Liabilities. If any transfer or assignment of any SLM BankCo Asset or any assignment to and assumption of any SLM BankCo Liability intended to be transferred, assigned or assumed hereunder, as the case may be, is not consummated on or prior to the Effective Time whether as a result of the provisions of Section 2.6(g) or for any other reason (any such SLM BankCo Asset, a "Delayed SLM BankCo Asset" and any such SLM BankCo Liability, a "Delayed SLM BankCo Liability"), then, insofar as reasonably possible, the member of the Navient SLM BankCo Group to which such Delayed SLM BankCo Asset is to be transferred or assigned, or which is to assume such Delayed SLM BankCo Liability, as the case may be, and subject to applicable Law (including banking regulations and the advice of the SLM BankCo Group's counsel in respect thereof), the member of the Navient Group retaining such Delayed SLM BankCo Asset or such Delayed SLM BankCo Liability, as the case may be, shall thereafter hold such Delayed SLM BankCo Asset or Delayed SLM BankCo Liability in trust for the use and benefit of the member of the SLM BankCo Group to whom such Delayed SLM BankCo Asset is to be transferred or assigned, or which is to assume such Delayed SLM BankCo Liability (at the expense of such member of the SLM BankCo Group). In addition, the member of the Navient Group retaining such Delayed SLM BankCo Asset or such Delayed SLM BankCo Liability as it did prior to the Effective Time in the ordinary course of business in accordance with past practice and take such other actions as may be reasonably requested by the member of the SLM Group to whom such Delayed SLM

BankCo Asset is to be transferred or assigned, or which will assume such Delayed SLM BankCo Liability, as the case may be, in order to place such member of the SLM BankCo Group in a substantially similar position as if such Delayed SLM BankCo Asset or Delayed SLM BankCo Liability had been transferred, assigned or assumed as contemplated hereby and so that all the benefits and burdens of ownership relating to such Delayed SLM BankCo Asset or Delayed SLM BankCo Liability, as the case may be, including use, risk of loss, potential for gain, profits and dominion, control and command over such Delayed SLM BankCo Asset or Delayed SLM BankCo Liability, as the case may be, and all costs and expenses related thereto, shall inure from and after the Effective Time to the SLM BankCo Group. Each of SLM BankCo and Navient shall, and shall cause the members of its Group to, (i) treat for all Tax purposes the Delayed SLM BankCo Assets and the Delayed SLM BankCo Liabilities as Assets owned by, and/or Liabilities of, as applicable, SLM BankCo, or its Subsidiaries, as applicable, not later than the Effective Time, and (ii) neither report nor take any Tax position (on a Tax Return or otherwise) inconsistent with such treatment (unless required by applicable Law).

- (i) Transfer of Delayed SLM BankCo Assets and Delayed SLM BankCo Liabilities. If and when the Approvals or Notifications, the absence of which caused the deferral of the transfer or assignment of any Delayed SLM BankCo Asset or the deferral of the assignment and assumption of any Delayed SLM BankCo Liability, are obtained or made, and, if and when any other legal impediments for the transfer or assignment of any Delayed SLM BankCo Asset or the assignment and assumption of any Delayed SLM BankCo Liability have been removed, the transfer or assignment of the applicable Delayed SLM BankCo Asset or the assignment and assumption of the applicable Delayed SLM BankCo Liability, as the case may be, shall be effected in accordance with the terms of this Agreement and/or any applicable Ancillary Agreement.
- (j) Costs for Delayed SLM BankCo Assets and Delayed SLM BankCo Liabilities. Any member of the Navient Group retaining a Delayed SLM BankCo Asset or Delayed SLM BankCo Liability due to the deferral of the transfer or assignment of such Delayed SLM BankCo Asset or the deferral of the assignment and assumption of such Delayed SLM BankCo Liability, as the case may be, shall not be obligated, in connection with the foregoing, to expend any money unless the necessary funds are advanced (or otherwise made available) by a member of the SLM BankCo Group, other than reasonable out-of-pocket expenses, attorneys' fees and recording or similar fees, all of which shall be promptly reimbursed by a member of the SLM BankCo Group.

Misallocations; Waiver of Bulk Sales Laws.

(k) Misallocations. In the event that at any time or from time to time (whether prior to, at or after the Effective Time), one Party hereto (or any member of such Party's respective Group) shall receive or otherwise possess any Asset that is

allocated to the other Party (or any member of such Party's Group) pursuant to this Agreement or any Ancillary Agreement, such Party shall promptly transfer, or cause to be transferred, such Asset to the Party so entitled thereto (or to a member of such Party's Group), and such Party (or member of its Group) shall accept such Asset. Prior to any such transfer, the Party (or member of its Group) receiving or possessing such Asset shall hold such Asset in trust for such other Party (or member of its Group). In the event that at any time or from time to time (whether prior to, at or after the Effective Time), one Party hereto (or any member of its Group) shall receive or otherwise assume any Liability that is allocated to the other Party (or a member of its Group) pursuant to this Agreement or any Ancillary Agreement, such Party (or member of its Group) shall promptly transfer, or cause to be transferred, such Liability to the other Party (or member of its Group) to which such Liability is allocated, and such Party (or member of its Group) shall accept, assume and agree to faithfully and timely perform such Liability in accordance with its terms.

(l) Waiver of Bulk-Sale and Bulk-Transfer Laws. Navient hereby waives compliance by each and every member of the SLM BankCo Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer of any or all of the Navient Assets to the members of the Navient Group. SLM BankCo hereby waives compliance by each and every member of the Navient Group with the requirements and provisions of any "bulk-sale" or "bulk-transfer" Laws of any jurisdiction that may otherwise be applicable with respect to the transfer of any or all of the SLM BankCo Assets to the members of the SLM BankCo Group.

2.7 Novation of Navient Liabilities.

(a) Each of SLM BankCo and Navient, at the request of the other, shall use its commercially reasonable efforts to obtain, or to cause to be obtained, as soon as reasonably practicable, any consent, substitution, approval or amendment required to novate or assign to Navient or one or more other members of the Navient Group designated by Navient all Navient Liabilities (including Navient Contracts) and obtain in writing the unconditional release of each member of the SLM BankCo Group that is a party to any such Navient Liabilities, so that, in any such case, the members of the Navient Group shall be solely responsible for such Navient Liabilities; provided, however, that, except as otherwise expressly provided in this Agreement or any of the Ancillary Agreements, neither SLM BankCo nor Navient shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

(b) If SLM BankCo or Navient is unable to obtain, or to cause to be obtained, any such required consent, substitution, approval, amendment or release and the applicable member of the SLM BankCo Group continues to be bound by a Navient Liability (each, an "Unreleased Navient Liability"), Navient shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the SLM BankCo Group, as the case may be, (i) pay, perform and discharge fully all the obligations or other Liabilities of such member of the SLM BankCo Group that constitute Unreleased Navient Liabilities from and after the Effective Time and (ii) use its commercially reasonable efforts to effect such payment, performance or discharge prior to any demand for such payment, performance or discharge is permitted to be made by the obligee thereunder on any member of the SLM BankCo Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased Navient Liabilities shall otherwise become assignable or able to be novated, SLM BankCo shall promptly assign, or cause to be assigned, and Navient or the applicable Navient Group member shall assume, such Unreleased Navient Liabilities without exchange of further consideration.

2.8 Novation of SLM BankCo Liabilities.

(a) Each of SLM BankCo and Navient, at the request of the other, shall use its commercially reasonable efforts to obtain, or to cause to be obtained, as soon as reasonably practicable, any consent, substitution, approval or amendment required to novate or assign to SLM BankCo or one or more other members of the SLM BankCo Group designated by SLM BankCo all SLM BankCo Liabilities and obtain in writing the unconditional release of each member of the Navient Group that is a party to any such arrangements, so that, in any such case, the members of the SLM BankCo Group shall be solely responsible for such SLM BankCo Liabilities; provided, however, that, except as otherwise expressly provided in this Agreement or any of the Ancillary Agreements, neither SLM BankCo nor Navient shall be obligated to contribute any capital or pay any consideration in any form (including providing any letter of credit, guaranty or other financial accommodation) to any third Person from whom any such consent, substitution, approval, amendment or release is requested.

(b) If SLM BankCo or Navient is unable to obtain, or to cause to be obtained, any such required consent, substitution, approval, amendment or release and the applicable member of the Navient Group continues to be bound by such SLM BankCo Liability (each, an "Unreleased SLM BankCo Liability"), SLM BankCo shall, to the extent not prohibited by Law, as indemnitor, guarantor, agent or subcontractor for such member of the Navient Group, as the case may be, (i) pay, perform and discharge fully all the obligations or other Liabilities of such member of the Navient Group that constitute Unreleased SLM BankCo Liabilities from and after the Effective Time and (ii) use its commercially reasonable efforts to effect such payment, performance or discharge prior to any demand for such payment, performance or discharge is permitted to be made by the obligee thereunder on any member of the Navient Group. If and when any such consent, substitution, approval, amendment or release shall be obtained or the Unreleased SLM BankCo Liabilities shall otherwise become assignable or able to be novated, Navient shall promptly assign, or cause to be assigned, and SLM BankCo Liabilities without exchange of further consideration.

2.9 Release Under Certain Agreements.

(a) Except as set forth in Section 2.10(b), in furtherance of the releases and other provisions of Section 4.1, Navient and each member of the Navient Group, on the one hand, and SLM BankCo and each member of the SLM BankCo Group, on the other hand, hereby agree with respect to any and all agreements, arrangements, commitments or understandings, whether or not in writing, including each of the agreements set forth on Schedule 2.10(a), between or among Navient and/or any member of the Navient Group, on the one hand, and SLM BankCo and/or any member of the SLM BankCo Group, on the other hand, effective as of the Effective Time, each member of the SLM BankCo Group and each member of the Navient group is hereby released from all liabilities, obligations, rights and benefits accruing under such agreements from and after the Effective Time. No such released agreement, arrangement, commitment or understanding (including any provision thereof which purports to survive release or termination) shall be of any further force or effect as against the released Party or the other members of its Group as to any period after the Effective Time. At or prior to the Effective Time, each of Navient and SLM BankCo shall deliver to the other Party an agreement executed by each member of such Party's Group evidencing such releases. Each Party shall, at the reasonable request of the other Party, take, or cause to be taken, such further actions as may be necessary to effect the foregoing.

(b) The provisions of Section 2.10(a) shall not apply to any of the following agreements, arrangements, commitments or understandings (or to any of the provisions thereof): (i) this Agreement and the Ancillary Agreements (and each other agreement or instrument expressly contemplated by this Agreement or any Ancillary Agreement to be entered into by any of the Parties or any of the members of their respective Groups which is to continue after the Effective Time); (ii) any agreements, arrangements, commitments or understandings listed or described on Schedule 2.10(b); (iii) any agreements, arrangements, commitments or understandings to which any Person other than the Parties hereto and their respective Affiliates is a party; (iv) any intercompany accounts payable or accounts receivable accrued as of the Effective Time that are reflected in the books and records of the Parties or otherwise documented in writing in accordance with past practices, which shall be settled in the manner contemplated by Section 2.10(c); (v) any agreements, arrangements, commitments or understandings to which any non-wholly owned Subsidiary of SLM BankCo or Navient, as the case may be, is a party (it being understood that directors' qualifying shares or similar interests will be disregarded for purposes of determining whether a Subsidiary is wholly owned); and (vi) any Shared Contracts.

(c) All outstanding intercompany accounts receivable and accounts payable between any member of the SLM BankCo Group, on the one hand, and any member of the Navient Group, on the other hand, shall, prior to the Effective Time, be repaid, settled or otherwise eliminated by means of cash payments, a dividend, capital contribution, a combination of the foregoing, or otherwise as determined by Existing SLM in its sole and absolute discretion; provided, however, that any intercompany accounts payable or accounts receivable accrued as of the Effective Time that has not been invoiced prior to the Effective Time shall be invoiced and paid in accordance with past practices and provided, further, that no indebtedness (i) between SMI and its Subsidiaries, on the one hand, and Private ServiceCo, on the other hand, and (ii) between a member of the SLM BankCo Group, on the one hand, and a member of the Navient Group, on the other hand, will be settled or cancelled in connection with the Separation and Distribution other than the settlement of intercompany loans and intercompany open account balances attributable to the normal business operations of SMI or any member of the SLM BankCo Group prior to the Effective Time.

2.10 Bank Accounts; Cash Balances.

- (a) SLM BankCo and Navient each agrees to take, or cause the respective members of their respective Groups to take, as of the Effective Time (or such earlier time as SLM BankCo and Navient may agree), all actions necessary to amend all contracts or agreements governing each bank and brokerage account owned by Navient or any other member of the Navient Group (collectively, the "Navient Accounts") and all contracts or agreements governing each bank or brokerage account owned by SLM BankCo or any other member of the SLM BankCo Group (collectively, the "SLM BankCo Accounts") such that each such Navient Account and SLM BankCo Account, if currently linked (whether by automatic withdrawal, automatic deposit or any other authorization to transfer funds from or to, hereinafter "linked") to any SLM BankCo Account or Navient Account, respectively, is delinked from such SLM BankCo Account or Navient Account, respectively.
- (b) It is intended that following consummation of the actions contemplated by Section 2.11(a) there will be in place a centralized cash management process pursuant to which the Navient Accounts will be managed centrally and funds collected will be transferred into one or more centralized accounts maintained by Navient.
- (c) It is intended that following consummation of the actions contemplated by Section 2.11(a) there will continue to be in place a centralized cash management process pursuant to which the SLM BankCo Accounts will be managed centrally and funds collected will be transferred into one or more centralized accounts maintained by SLM BankCo.

- (d) With respect to any outstanding checks issued or payments initiated by SLM BankCo, Navient, or any of their respective Subsidiaries prior to the Effective Time, such outstanding checks and payments shall be honored following the Effective Time by the Person or Group owning the account on which the check is drawn or from which the payment was initiated, respectively.
- (e) As between SLM BankCo and Navient (and the members of their respective Groups) all payments made to and reimbursements received by either Party (or member of its Group) after the Effective Time that relate to a business, Asset or Liability of the other Party (or member of its Group) shall be held by such Party (or member of its Group) in trust for the use and benefit of the Party (or member of its Group) entitled thereto and, promptly following receipt by such Party (or member of its Group) of any such payment or reimbursement, such Party shall pay over, or shall cause the applicable member of its Group to pay over, to the other Party (or the member of its Group) the amount of such payment or reimbursement without right of set-off.
- 2.11 <u>Ancillary Agreements</u>. Effective at or prior to the Effective Time, each of SLM BankCo and Navient (and the applicable members of their respective Groups) will execute and deliver each Ancillary Agreement to which it is a party.
- 2.12 Disclaimer of Representations and Warranties. EACH OF SLM BANKCO (ON BEHALF OF ITSELF AND EACH MEMBER OF THE SLM BANKCO GROUP) AND NAVIENT (ON BEHALF OF ITSELF AND EACH MEMBER OF THE NAVIENT GROUP) UNDERSTANDS AND AGREES THAT, EXCEPT AS EXPRESSLY SET FORTH HEREIN OR IN ANY ANCILLARY AGREEMENT, NO PARTY TO THIS AGREEMENT, ANY ANCILLARY AGREEMENT OR ANY OTHER AGREEMENT OR DOCUMENT CONTEMPLATED BY THIS AGREEMENT IS REPRESENTING OR WARRANTING IN ANY WAY AS TO THE ASSETS, BUSINESSES OR LIABILITIES TRANSFERRED OR ASSUMED AS CONTEMPLATED HEREBY OR THEREBY, AS TO ANY CONSENTS OR APPROVALS REQUIRED IN CONNECTION THEREWITH, AS TO THE VALUE OR FREEDOM FROM ANY SECURITY INTERESTS OF, OR ANY OTHER MATTER CONCERNING, ANY ASSETS OF SUCH PARTY, OR AS TO THE ABSENCE OF ANY DEFENSES OR RIGHT OF SETOFF OR FREEDOM FROM COUNTERCLAIM WITH RESPECT TO ANY CLAIM OR OTHER ASSET, INCLUDING ANY ACCOUNTS RECEIVABLE, OF ANY PARTY, OR AS TO THE LEGAL SUFFICIENCY OF ANY ASSIGNMENT, DOCUMENT OR INSTRUMENT DELIVERED HEREUNDER TO CONVEY TITLE TO ANY ASSET OR THING OF VALUE UPON THE EXECUTION, DELIVERY AND FILING HEREOF OR THEREOF. EXCEPT AS MAY EXPRESSLY BE SET FORTH HEREIN OR IN ANY ANCILLARY AGREEMENT, ALL SUCH ASSETS ARE BEING TRANSFERRED ON AN "AS IS," "WHERE IS" BASIS (AND, IN THE CASE OF ANY REAL PROPERTY, BY MEANS OF A QUITCLAIM OR SIMILAR FORM OF DEED OR CONVEYANCE) AND THE RESPECTIVE TRANSFEREES SHALL BEAR THE ECONOMIC AND LEGAL RISKS THAT (I) ANY CONVEYANCE WILL PROVE TO BE INSUFFICIENT TO VEST IN THE TRANSFEREE GOOD AND MARKETABLE TITLE, FREE AND CLEAR OF ANY SECURITY INTEREST, AND (II) ANY NECESSARY APPROVALS OR NOTIFICATIONS ARE NOT OBTAINED OR MADE OR THAT ANY REQUIREMENTS OF LAWS OR JUDGMENTS ARE NOT COMPLIED WITH.

2.13 Intellectual Property.

- (a) License Grants.
- (i) By Navient. As of the Effective Time and to the extent not otherwise licensed by SMI to Private ServiceCo pursuant to Section 2.1(k)(iv), Navient and the other members of the Navient Group hereby grant to SLM BankCo and the other members of the SLM BankCo Group a non-exclusive, perpetual, royalty-free, transferable, assignable, sublicenseable, worldwide right and license to all (x) Other IP constituting Navient Intellectual Property, (y) Navient Software and (z) Navient Technology, in each case that is owned by Navient or the other members of the Navient Group and is used or held for use in the Pre-Separation Consumer Banking Business, including Intellectual Property, Software and Technology that are an integral part of the Private Loan Servicing Business Systems, but excluding Intellectual Property, Software and Technology that are an integral part of the Excluded Systems (the "Navient Licensed Materials"), and including, but not limited to the right to reproduce, use, develop, improve, perform, display, distribute, sell, offer to sell, nonexclusively license and sublicense, create derivative works from, and otherwise fully exploit without restriction, the Navient Licensed Materials. Further, as of the Effective Time, Navient and the other members of the Navient Group hereby grant to SLM BankCo and the other members of the SLM BankCo Group a non-exclusive, perpetual, royalty-free, non-transferable, non-assignable, non-sublicenseable, worldwide right and license to all Navient Licensed Systems, including the right to reproduce, use, develop, improve, perform, and create derivative works from such Navient Licensed Systems solely for SLM BankCo's and the other members of the SLM BankCo Group's internal business purposes. The Navient Licensed Materials and Navient Licensed Systems shall be provided on an "AS IS" basis and without warranty of any kind.
- (ii) By SLM BankCo. As of the Effective Time, SLM BankCo and the other members of the SLM BankCo Group hereby grant to Navient and the other members of the Navient Group a non-exclusive, perpetual, royalty-free, transferable, assignable, sublicenseable, worldwide right and license to all SLM BankCo Licensed Systems, and including, but not limited to the right to reproduce, use, develop, improve, perform, display, distribute, sell, offer to sell, nonexclusively license and sublicense, create derivative works from, and otherwise fully exploit without restriction, the SLM BankCo Licensed Systems. Further, SLM BankCo and the other members of the SLM BankCo Group hereby

grant to Navient and the other members of the Navient Group a non-exclusive, perpetual, royalty-free, transferable, assignable, sublicenseable, worldwide right and license to all enhancements of, modifications to, and derivative works of the SLM BankCo Licensed Systems developed by SLM BankCo and the other members of the SLM BankCo Group as of the end of the Preferred Stock Period. For the avoidance of doubt, neither SLM BankCo nor any other member of the SLM BankCo Group shall have any obligation to license, nor shall Navient nor any member of the Navient Group have any rights to, enhancements of, modifications to, or derivative works of the SLM BankCo Licensed Systems made by SLM BankCo or any other member of the SLM BankCo Group as of any date later than the end of the Preferred Stock Period. The SLM BankCo Licensed Systems (including any enhancements, modifications or derivative works) shall be provided on an "AS IS" basis and without warranty of any kind.

(iii) College Planning Tools. As of the Effective Time, SLM BankCo and the other members of the SLM BankCo Group hereby grant to Navient and the other members of the Navient Group a non-exclusive, perpetual, royalty-free, non-transferable, non-assignable, non-sublicenseable, worldwide right and license to the College Planning Tools, including the right to reproduce, use, develop, improve, perform, and create derivative works from such College Planning Tools solely for Navient's and the other members of the Navient Group's internal business purposes; provided, that Navient may (x) sell, assign or transfer the College Planning Tools pursuant to a transaction contemplated by Section 5.4(g)(i)(D); (y) license the College Planning Tools to a Governmental Authority (other than a private or state institution of higher learning (e.g., school, college, university)) pursuant to an agreement with such Governmental Authority if the grant of such license is required by applicable Law or published procurement policy of such Governmental Authority, and (z) grant a perpetual, nontransferable, nonassignable, nonsublicenseable license to the College Planning Tools in connection with the sale or disposition of all or substantially all of the assets comprising a line of business then-using the College Planning Tools (or an entity comprising such line of business) to the acquiror of such entity or assets comprising such line of business solely for use by such acquiror to perform the applicable line of business. For the avoidance of doubt, neither SLM BankCo nor any other member of the SLM BankCo Group shall have any obligation to license, nor shall Navient nor any member of the Navient Group have any rights to, enhancements of, modifications to, or derivative works of the College Planning Tools made by SLM BankCo or any other member of the SLM BankCo Group as of any date later than the Effective Time. The College Planning Tools shall be provided on an "AS IS" basis and without warranty of any kind.

(b) Covenant to Enforce.

(i) If either of SLM BankCo or Navient becomes aware of potential infringement by a third party of the ATLAS Software (including Joint Materials developed pursuant to Section 5.9(b)) or the SLM BankCo Licensed Systems, then that Party will notify the other Party as soon as possible, and the Parties agree to discuss and determine how best to secure the cessation of such infringement. If the Parties agree to begin an action for infringement against a third party, then prior to asserting such action, the Parties shall negotiate in good faith to determine the allocation of the expenses and disbursements paid in connection with such action and the allocation of all monies actually received as a result of the infringement action. If the Parties cannot agree to begin such an action or cannot agree on the allocation of expenses and disbursements, then either Party will have the right to prosecute the infringement action, and that Party will bear all the expense and be entitled to retain all monies received from such action. In such a case, the non-enforcing Party shall not be obligated to provide any assistance to the enforcing Party in connection with such lawsuit, and in the event that the non-enforcing Party provides prior written consent to being named as a party to the lawsuit in order for the enforcing Party to sustain jurisdiction or standing of the action, the enforcing Party shall solely bear the expenses of the non-enforcing Party's participation. If during a joint infringement action either Party decides to discontinue its participation in the action, then if the other Party wishes to continue it shall pay all future expenses associated with such action and will retain all monies or consideration from such action after first reimbursing any actionrelated expenses incurred by the discontinuing Party at a rate equal to the percentage of total expenses contributed by the discontinuing Party up to the date of discontinuance. Neither SLM BankCo nor any of its Affiliates will join or attempt to join Navient or any Navient Affiliate, and Navient and its Affiliates will not be joined, to any litigation alleging infringement by a third party of the ATLAS Software (including Joint Materials) unless Navient has provided prior written consent. Neither Navient nor any of its Affiliates will join or attempt to join SLM BankCo or any SLM BankCo Affiliate, and SLM BankCo and its Affiliates will not be joined, to any litigation alleging infringement by a third party of the ATLAS Software (including Joint Materials), unless SLM BankCo has provided prior written consent.

(ii) If at any time, SLM BankCo (on its behalf or on the behalf of any member of the SLM BankCo Group) brings to the attention of Navient any unlicensed infringement of the Navient Licensed Materials or the Navient Licensed Systems (as they exist as of the Effective Time and excluding any modifications or enhancements made by SLM BankCo or the members of the SLM BankCo Group) that is affecting or could reasonably be expected to affect SLM BankCo's or the other members of the SLM BankCo Group's rights hereunder, including SLM BankCo and the other members of the SLM BankCo Group's rights to use the Navient Licensed Materials or the Navient Licensed

Systems, then Navient will (i) use commercially reasonable efforts to secure cessation of the infringement and (ii) if commercially reasonable under the circumstances (taking into consideration, among other things, costs and expected financial outcome), bring and diligently prosecute a lawsuit against the infringer. Navient in its own name will bring and prosecute such suit. SLM BankCo will join in such suit with Navient to the extent required by Law, but may otherwise join in such suit at its option. Further, if at any time, Navient (on its behalf or on the behalf of any member of the Navient Group) brings to the attention of SLM BankCo any unlicensed infringement of the SLM BankCo Licensed Systems (as they exist as of the Effective Time and excluding any modifications or enhancements made by Navient or the other members of the Navient Group's rights hereunder, including Navient's and the other members of the Navient Group's rights hereunder, including Navient's and the other members of the Navient Group's rights to use the SLM BankCo Licensed Systems, then SLM BankCo will (i) use commercially reasonable efforts to secure cessation of the infringement and (ii) if commercially reasonable under the circumstances (taking into consideration, among other things, costs and expected financial outcome), bring and diligently prosecute a lawsuit against the infringer. SLM BankCo in its own name will bring and prosecute such suit. Navient will join in such suit with SLM BankCo to the extent required by Law, but may otherwise join in such suit at its option.

- (iii) If the Parties join in any suit covered by paragraph (ii), the Parties shall jointly control such suit and cooperate with each other with respect to such suit. If either Navient or SLM BankCo fails to jointly diligently prosecute any such suit, the other Party may solely prosecute such suit and the Party failing to jointly bring or prosecute shall use commercially reasonable efforts to cooperate with the other Party if it determines to proceed with such suit.
- (iv) Navient and SLM BankCo will be entitled to any separate recoveries from such suit as may be awarded to each of them, respectively.
- (v) If Navient, as licensor of the Navient Licensed Materials and the Navient Licensed Systems, or SLM BankCo, as licensor of the SLM BankCo Licensed Systems (each, a "Licensor") fails to (x) secure cessation of the infringement or bring and diligently prosecute suit against the infringer within 60 days after the receipt by Navient, as licensee of the SLM BankCo Licensed Systems, or SLM BankCo, as licensee of the Navient Licensed Materials and the Navient Licensed Systems (each, a "Licensee"), of the notice by the Licensor pursuant to paragraph (ii) above, or (y) diligently prosecute any suit joined by the Licensee pursuant to paragraph (ii) above, then, subject to paragraph (iii) above, the Licensee shall be entitled to bring and prosecute actions or claims against such infringer on behalf of the Licensor and its Affiliates.

(vi) Each Party will bear its own costs and expenses (including legal fees and expenses) incurred in connection with any suits brought pursuant to this paragraph (b), including suits brought by either Party pursuant to subparagraphs (iii) and (v) above.

(c) Covenants Not to Sue.

- (i) Navient covenants and agrees on behalf of itself and the other members of the Navient Group to take no action, directly or indirectly, to file or to cause to be filed any suit or institute or assist in instituting any proceeding in any court or tribunal, wherever located, against SLM BankCo or any other member of the SLM BankCo Group for any claims or causes of action arising from or relating to the infringement of Navient Intellectual Property by the modifications, enhancements or derivative works of the ATLAS Software (including Joint Materials), the SMI Licensed Materials, Navient Licensed Materials or Navient Licensed Systems created in compliance with this Agreement, any Ancillary Agreement and Intellectual Property associated therewith, which Navient and the other members of the Navient Group has, may have or hereafter might have, for or because of any matter or thing done, omitted or suffered to be done from and after the Effective Time.
- (ii) SLM BankCo covenants and agrees on behalf of itself and the other members of the SLM BankCo Group to take no action, directly or indirectly, to file or to cause to be filed any suit or institute or assist in instituting any proceeding in any court or tribunal, wherever located, against Navient or any other member of the Navient Group for any claims or causes of action arising from or relating to the infringement of SLM BankCo Intellectual Property (excluding the Private Loan Servicing Registrable IP) by the modifications, enhancements or derivative works of the ATLAS Software (including Joint Materials) or the SLM BankCo Licensed Systems created in compliance with this Agreement, any Ancillary Agreement and Intellectual Property associated therewith, which SLM BankCo and the other members of the SLM BankCo Group has, may have or hereafter might have, for or because of any matter or thing done, omitted or suffered to be done from and after the Effective Time.

(d) Names and Marks.

(i) Except as provided in, contemplated by or required in connection with the provision of services pursuant to any Ancillary Agreement or as provided in this Section 2.14(d), as of the Effective Time (A) Navient shall not have any right to use or display the SLM BankCo Names or the SLM BankCo Marks in any form and (B) SLM BankCo shall not have the right to use or display the Navient Names or the Navient Marks in any form; provided, however, that (x) to the extent such SLM BankCo Names and SLM BankCo Marks were used or

displayed by any member of the Navient Group prior to the Effective Time, the members of the Navient Group shall, as soon as reasonably practicable, but in any event within one year after the Effective Time, at their expense, cease all use or display of all SLM BankCo Names and SLM BankCo Marks and shall remove any and all references to the SLM BankCo Names and SLM BankCo Marks on Assets owned or held by them (including on business cards, stationary, commercial signs and similar identifiers) that refer or pertain specifically to the SLM BankCo Names and SLM BankCo Marks, and (y) that Navient shall have the right to continue to use the SLM BankCo Names and SLM BankCo Marks in perpetuity (i) to the extent they are incorporated into historical memorabilia, awards, customer benefit programs, and the like prior to the Effective Time and (ii) in new material that refers to the SLM BankCo Names and SLM BankCo Marks in the context of the loans made to any customer that carry such trademarked/service-marked benefit or name. In addition, each party shall have the right to use the others' respective Names and Marks in perpetuity to the extent they are incorporated into (i) materials that speak generally to the history of the respective companies or (ii) the source code or system code of Software used by Effective Time Sallie Mae and where such Names and Marks are not visible to customers or other third-party users of such Software. Each Party hereby grants to the other Party's Group, for a period starting at the Effective Time and ending on the 180th day following the Operational Servicing Date, a worldwide, non-exclusive, non-transferable, royalty-free license to use the SLM Name and SLM BankCo Marks or Navient Name and Navient Marks, as applicable, as and to the extent used by the licensee's respective business as of the Effective Time. Each Party shall use the names and marks of the other Party as allowed hereunder only in connection with goods or services that are of a level of quality at least equal to the quality of comparable goods or services marketed by that Party (or members of its respective Group) before the Effective Time.

(ii) Notwithstanding the foregoing, nothing contained in this Agreement will prevent any Party (or any member of its respective Group) from using the other's names or marks in documents intended to be filed with Governmental Authorities or real property registration offices, systems or databases (including MERS), in materials intended for distribution to such Party's stockholders or in any other communication (including correspondence) in any medium that describes the current or former relationship between the Parties (or members of their respective Groups).

(e) Perfection of Intellectual Property in ATLAS Software; Enforcement of Contract Rights. Following the Effective Time, Private ServiceCo shall have the right, but not an obligation, to file, prosecute, and maintain, any patent application and resulting patent in the ATLAS Software and any copyright registration for the ATLAS Software, in any jurisdiction, and SMI shall have no right and no

obligation to file, prosecute, or maintain such Registrable IP in the ATLAS Software, except as described below. Notwithstanding the foregoing, each Party agrees to execute all reasonably necessary documents, including declarations, powers of attorney, and relevant ownership documents associated with perfection of such Registrable IP in the ATLAS Software. The costs of such patent or copyright filing, prosecution, and maintenance, if any, shall be borne entirely by Private ServiceCo. In the event Private ServiceCo decides to abandon, or otherwise allow to lapse any such Registrable IP in the ATLAS Software, Private ServiceCo agrees to offer to assign to SMI such filing, prosecution, and maintenance rights, with no obligation, on any such Registrable IP in the ATLAS Software. Private ServiceCo shall provide notice to SMI of its intent to abandon, or to allow a patent or copyright property to lapse, at least thirty (30) days prior to an original Governmental Authority -imposed deadline, but shall have no obligation to notify SMI of an intent not to file a patent application or copyright registration on the ATLAS Software. Private ServiceCo shall notify SMI of the issuance of any patent or registration of any copyright filed on the ATLAS Software within thirty (30) business days from such issuance or registration. This notification requirement shall apply to patents issued and copyrights registered by any jurisdiction. All patents and patent applications filed on the ATLAS Software and copyright registrations filed for the ATLAS Software shall be owned jointly by Private ServiceCo and SMI, regardless of which party files, prosecutes, and maintains the patent or copyright. Private ServiceCo also agrees to provide reasonable cooperation to Navient in seeking to enforce rights that arise pursuant to the contract to acquire the underlying source code that formed the basis of the ATLAS Software and the contract to assist in the development of such code, to the extent reasonably necessary to permit Navient full use of s

2.14 Financial Information Certifications.

(a) If requested by SLM BankCo, and in order to assist or enable the principal executive officer or officers, principal financial officer or officers and controller or controllers of SLM BankCo to make the certifications required of them under Section 302 and Section 906 of the Sarbanes-Oxley Act of 2002 ("SOX Certifications"), Navient shall provide certification statement as to the Navient Group for the fiscal quarter during which the Distribution Date occurs to those certifying officers of SLM BankCo, which certifications shall be in substantially the same form as had been provided by officers of Existing SLM in SOX Certifications delivered prior to the Distribution Date (provided that such certifications shall be made by Navient rather than individual officers), or as otherwise agreed upon between the Parties. Such certification statements shall also reflect any changes from prior SOX Certifications necessitated by the transactions contemplated by this Agreement.

(b) If requested by Navient, and in order to enable the principal executive officer or officers, principal financial officer or officers and controller or controllers of Navient to make the SOX Certifications, SLM BankCo shall provide a certification

statements as to the SLM BankCo Group for the fiscal quarter during which the Distribution Date occurs to those certifying officers of Navient, which certifications shall be in substantially the same form as had been provided by officers of Subsidiaries of Existing SLM to support SOX Certifications delivered by officers of Existing SLM prior to the Distribution Date (provided that such certification shall be made by SLM BankCo rather than individual officers), or as otherwise agreed upon between the Parties. Such certification statements shall also reflect any changes necessitated by the transactions contemplated by this Agreement.

- 2.15 <u>Misdirected Communications</u>. If, on or after the Effective Time, any member of a Group receives any mail, courier package, facsimile transmission, invoice, purchase order, written service request or other written document that is intended for a member of the other Group or that, in the case of the SLM BankCo Group, relates to the Pre-Separation Education Loan Management Business or the Navient Business or, in the case of the Navient Group, relates to the Pre-Separation Consumer Banking Business or the SLM BankCo Business, such documents shall be forwarded to the other Group, no later than five Business Days after the date of receipt thereof, as follows: (a) if to the Navient Group, by email addressed to john.marvel@navient.com or delivery at the following address: John Marvel, Senior Director Procurement, 11100 USA Parkway, Fishers IN 46037 and (b) if to the SLM BankCo Group, by email addressed to karen.wire@salliemae.com or delivery at the following address: Karen Wire, Senior Director Vendor Management, 11100 USA Parkway, Fishers IN 46037.
- 2.16 <u>Parent Guarantee</u>. Prior to or as of the Effective Time, Navient shall sign a full and unconditional guarantee of all unsecured notes of Existing SLM Holdco issued under the Indentures and outstanding as of the Effective Time, and shall take such further action as may be necessary pursuant to Rule 12h-5 under the Exchange Act and Rule 3-10 of Regulation S-X to eliminate any requirement that Existing SLM Holdco, after the Effective Time, continue to file periodic reports under the Exchange Act.

ARTICLE III THE DISTRIBUTION

3.1 Sole and Absolute Discretion; Cooperation.

(a) The Sallie Mae Board shall, in its sole and absolute discretion, determine the timing of the Distribution and the conditions to the consummation of the Distribution. In addition, Existing SLM may, at any time and from time to time until the Effective Time, delay the timing of, or cancel, the Distribution. For the avoidance of doubt, nothing in this Agreement shall in any way limit the right of the Sallie Mae Board to terminate the Distribution or cancel this Agreement and the Ancillary Agreements prior to the Effective Time.

- (b) Each of the Parties shall cooperate with the others to accomplish the Distribution and shall take any and all actions necessary or desirable to effect the Distribution, including in respect of the registration under the Exchange Act of shares of Navient Common Stock pursuant to the Form 10.
- 3.2 Actions Prior to the Distribution. Prior to the Effective Time and subject to the terms and conditions set forth herein, the Parties shall take, or cause to be taken, the following actions in connection with the Distribution:
 - (a) *Notice to NASDAQ*. Existing SLM shall, to the extent practicable, give the NASDAQ (i) not less than 15 days' advance notice of the Merger and (ii) not less than 10 days' advance notice of the Record Date in compliance with Rule 10b-17 under the Exchange Act.
 - (b) Certificate of Incorporation and Bylaws. Prior to the Effective Time, Existing SLM and Navient shall take all necessary actions so that, as of the Effective Time, the Navient Certificate of Incorporation and Navient Bylaws shall be the certificate of incorporation and bylaws of Navient.
 - (c) Navient Directors and Officers. Existing SLM and Navient shall take all necessary actions so that as of the Effective Time: (i) the directors and executive officers of Navient shall include those set forth in the Information Statement mailed to the Record Holders; and (ii) each individual referred to in clause (i) shall have resigned from his or her position, if any, as a member of the SLM BankCo Board and/or as an executive officer of SLM BankCo or the Sallie Mae Bank.
 - (d) Exchange Listing. Navient shall prepare and file, and shall use its reasonable best efforts to have approved, an application for the listing on the NASDAQ, subject to official notice of distribution, of the shares of Navient Common Stock to be distributed in the Distribution.
 - (e) Securities Law Matters. Navient shall file such amendments to the Form 10 or supplements to the Information Statement as may be necessary or advisable in order to cause the Form 10 to remain effective and the Information Statement to remain current in compliance with federal securities Laws. The Parties shall cooperate in preparing, filing with the SEC and causing to become effective registration statements or amendments thereof which are required to reflect the establishment of, or amendments to, any employee benefit and other plans necessary or advisable in connection with the transactions contemplated by this Agreement and the Ancillary Agreements (including the Employee Matters Agreement). Existing SLM and Navient will prepare, and Navient will, to the extent required under applicable Law, file with the SEC such documentation, including no-action letters, which Existing SLM may determine to be necessary or desirable to lawfully effectuate the Distribution, and Existing SLM and Navient shall each use its reasonable best efforts to obtain all

necessary approvals from the SEC with respect thereto as soon as practicable. The Parties shall take all such action, if any, as may be necessary or appropriate under the securities or blue sky laws of any State of the United States (and any comparable Laws under any foreign jurisdiction) in connection with the Distribution.

- (f) Mailing of Information Statement. Existing SLM shall cause the Information Statement to be mailed to the Record Holders.
- (g) *The Distribution Agent*. Existing SLM and SLM BankCo shall enter into a distribution agent agreement with the Agent or otherwise provide instructions to the Agent regarding the Distribution.

3.3 Conditions to the Distribution.

- (a) The consummation of the Distribution will be subject to the satisfaction, or waiver by the Sallie Mae Board in its sole and absolute discretion, of the following conditions:
 - (i) The SEC declaring effective the Form 10, with no order suspending the effectiveness of the Form 10 in effect and no proceedings for such purposes pending before or threatened by the SEC.
 - (ii) The Information Statement and such other information concerning Navient, its business, operations and management, the Distribution and such other matters as Existing SLM shall determine, in its sole and absolute discretion and as may otherwise be required by Law, shall have been mailed to the Record Holders.
 - (iii) Existing SLM shall have received a private letter ruling from the IRS (the "IRS Ruling") to the effect that, among other things, (A) the Merger (together with the conversion of the shares of Existing SLM common and preferred stock into shares of SLM BankCo common and preferred stock) will qualify as a "reorganization" within the meaning of Section 368(a)(1)(F) of the Code, and will not be integrated with the rest of the Separation and Distribution and (B) the Distribution, together with the contribution described in Section 2.1(o), will qualify as a transaction that is a "reorganization" for U.S. federal income tax purposes under Sections 355 and 368(a) (1)(D) of the Code, and such IRS Ruling shall not have been revoked or modified in any material respect.
 - (iv) Existing SLM and SLM BankCo shall have received an opinion from Baker Botts L.L.P. to the effect that the Distribution, together with the contribution described in Section 2.1(o), shall qualify as a transaction that is described in Sections 355(a) and 368(a)(1) (D) of the Code.

- (v) An independent appraisal firm acceptable to Existing SLM shall have delivered one or more opinions to the Existing SLM Board confirming the solvency and financial viability of Existing SLM and SLM BankCo before the consummation of the Distribution and each of SLM BankCo and Navient after consummation of the Distribution, and such opinions shall be acceptable to the Sallie Mae Board in its sole and absolute discretion and such opinions shall not have been withdrawn or rescinded.
 - (vi) The Separation shall have occurred as contemplated by Section 2.1.
- (vii) The actions and filings necessary or appropriate under applicable U.S. federal, U.S. state or other securities Laws or blue sky Laws and the rules and regulations thereunder shall have been taken or made, and, where applicable, have become effective or been accepted.
 - (viii) Each of the Ancillary Agreements shall have been duly executed and delivered by the parties thereto.
- (ix) No order, injunction or decree issued by any Governmental Authority of competent jurisdiction or other legal restraint or prohibition preventing the consummation of the Separation, the Distribution or any of the transactions related thereto shall be pending, threatened, issued or in effect.
- (x) The approval for listing on the NASDAQ of the shares of Navient Common Stock to be delivered in the Distribution shall have been obtained, subject to official notice of distribution.
- (xi) No other events or developments shall exist or shall have occurred that, in the judgment of the Sallie Mae Board, in its sole and absolute discretion, makes it inadvisable to effect the Separation, the Distribution or the transactions related thereto.
- (b) The foregoing conditions are for the sole benefit of Existing SLM and, following the Merger Effective Time, Effective Time Sallie Mae, and shall not give rise to or create any duty on the part of the Sallie Mae Board to waive or not waive any of such conditions or in any way limit the Sallie Mae Board's right to terminate this Agreement as set forth in Article IX. If the Sallie Mae Board waives any condition set forth in Section 3.3(a), it shall promptly issue a press release disclosing such fact and file a Current Report on Form 8-K with the SEC describing such waiver.

3.4 The Distribution.

(a) Subject to Section 3.3, prior to the Effective Time, Navient will deliver to the Agent authorizations for such number of outstanding shares of Navient Common Stock as is necessary to effect the Distribution, and shall cause the transfer agent for the Existing SLM Common Stock to instruct the Agent to credit on the stock transfer records of the Agent, on the Distribution Date, for the benefit of each holder of SLM BankCo Common Stock on the Record Date, or its designated transferee or transferees, the appropriate number of shares of Navient Common Stock by way of direct registration in book-entry form. The Distribution shall be deemed effective at the Effective Time.

(b) Subject to Sections 3.3, each record holder of shares of Existing SLM Common Stock on the Record Date will be entitled to receive in the Distribution one share of Navient Common Stock for each whole share of Existing SLM Common Stock held by such record holder on the Record Date. For the avoidance of doubt, holders of shares of Existing SLM Preferred Stock shall not be entitled to receive any shares of capital stock of Navient in the Distribution.

ARTICLE IV MUTUAL RELEASES; INDEMNIFICATION

4.1 Release of Pre-Distribution Claims.

(a) Except as provided in Sections 4.1(c) and 4.1(d) and any Ancillary Agreement, effective as of the Effective Time, Navient does hereby, for itself and each other member of the Navient Group, their respective successors and assigns, remise, release and forever discharge SLM BankCo and the other members of the SLM BankCo Group, their respective successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the SLM BankCo Group (in each case, in their respective capacities as such), and their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed at or before the Effective Time. Prior to or at the Effective Time, Navient shall deliver to SLM BankCo a release to the foregoing effect executed by each member of the Navient Group. For purposes of this Section 4.1(a), an Affiliate shall not include an entity in which Navient owns, directly or indirectly, less than 50% of the total combined equity interests.

- (b) Except as provided in Sections 4.1(c) and 4.1(d) and any Ancillary Agreement, effective as of the Effective Time, SLM BankCo does hereby, for itself and each other member of the SLM BankCo Group, their respective successors and assigns, remise, release and forever discharge Navient, the respective members of the Navient Group, their respective successors and assigns, and all Persons who at any time prior to the Effective Time have been shareholders, directors, officers, agents or employees of any member of the Navient Group (in each case, in their respective capacities as such), and their respective heirs, executors, administrators, successors and assigns, from any and all Liabilities whatsoever, whether at law or in equity (including any right of contribution), whether arising under any contract or agreement, by operation of law or otherwise, existing or arising from any acts or events occurring or failing to occur or alleged to have occurred or to have failed to occur or any conditions existing or alleged to have existed on or before the Effective Time. Prior to or at the Effective Time, SLM BankCo shall deliver to Navient a release to the foregoing effect executed by each member of the SLM BankCo Group. For purposes of this Section 4.1(b), an Affiliate shall not include an entity in which SLM BankCo owns, directly or indirectly, less than 50% of the total combined equity interests.
- (c) Nothing contained in Section 4.1(a) or 4.1(b) shall impair the right of any member of a Group to enforce this Agreement, any Ancillary Agreement or any agreements, arrangements, commitments or understandings that are specified in Section 2.10(b) or an applicable Schedule as not to terminate as of the Effective Time. Nothing contained in Section 4.1(a) or 4.1(b) shall release any Person from:
 - (i) any Liability provided in or resulting from any agreement between or among any members of the SLM BankCo Group and/or the Navient Group that is specified in <u>Section 2.10(b)</u> or an applicable Schedule as not to terminate as of the Effective Time, or any other Liability specified in <u>Section 2.10(b)</u> as not to terminate as of the Effective Time;
 - (ii) any Liability, contingent or otherwise, assumed, transferred, assigned or allocated to the Group of which such Person is a member in accordance with, or any other Liability of any member of any Group under, this Agreement or any Ancillary Agreement;
 - (iii) any Liability for unpaid amounts for products or services or refunds owing on products or services due on a value-received basis for work done by a member of one Group at the request or on behalf of a member of the other Group;
 - (iv) any Liability that a Party may have with respect to indemnification pursuant to this Agreement, any Ancillary Agreement or otherwise for claims brought against one or more Parties by third Persons, which Liability shall be governed by the provisions of this Article IV and Article V article V">Article

(v) any Liability the release of which would result in the release of any Person other than the Party or other members of its Group intended to be released pursuant to this <u>Section 4.1</u>.

In addition, nothing contained in Section 4.1(a) shall release (i) any member of the SLM BankCo Group from honoring its pre-existing obligations to indemnify, pursuant to its charter documents or otherwise, any director, officer or employee of Navient who was a director, officer or employee of any member of the SLM BankCo Group on or prior to the Effective Time, to the extent such director, officer or employee becomes a named defendant in any Action with respect to which such director, officer or employee was entitled to such indemnification pursuant to such pre-existing obligations; it being understood that, if the underlying obligation giving rise to such Action is a Navient Liability, Navient shall indemnify SLM BankCo for such Liability (including SLM BankCo's costs to indemnify the director, officer or employee) in accordance with the provisions set forth in this Article IV or (ii) any member of the Navient Group (including Existing SLM Holdco (as the successor to Existing SLM)) from honoring its pre-existing obligations to indemnify, pursuant to its charter documents or otherwise, any director, officer or employee of SLM BankCo who was a director, officer or employee of any member of the Navient Group on or prior to the Effective Time, to the extent such director, officer or employee becomes a named defendant in any Action with respect to which such director, officer or employee was entitled to such indemnification pursuant to such pre-existing obligations; it being understood that, if the underlying obligation giving rise to such Action is a SLM BankCo Liability, SLM BankCo shall indemnify Navient for such Liability (including Navient's costs to indemnify the director, officer or employee) in accordance with the provisions set forth in this Article IV.

- (d) Navient shall not make, and shall not permit any other member of the Navient Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification, against SLM BankCo or any other member of the SLM BankCo Group, or any other Person released pursuant to Section 4.1(a), with respect to any Liabilities released pursuant to Section 4.1(a). SLM BankCo shall not make, and shall not permit any other member of the SLM BankCo Group to make, any claim or demand, or commence any Action asserting any claim or demand, including any claim of contribution or any indemnification against Navient or any other member of the Navient Group, or any other Person released pursuant to Section 4.1(b), with respect to any Liabilities released pursuant to Section 4.1(b).
- (e) It is the intent of each of SLM BankCo and Navient, by virtue of the provisions of this Section 4.1, to provide for a full and complete release and discharge of all Liabilities existing or arising from all acts and events occurring or failing to occur or alleged to have occurred or to have failed to occur and all conditions existing or alleged to have existed at or before the Effective Time, between or among

Navient or any other member of the Navient Group, on the one hand, and SLM BankCo or any other member of the SLM BankCo Group, on the other hand (including any contractual agreements or arrangements existing or alleged to exist between or among any such members on or before the Effective Time), except as expressly set forth in Section 4.1(c). At any time, at the request of any other Party, each Party shall cause each member of its respective Group to execute and deliver further releases reflecting the provisions hereof.

4.2 <u>Indemnification by Navient</u>. Except as provided in <u>Section 4.4</u>, Navient shall, and shall cause the other members of the Navient Group to, indemnify, defend and hold harmless SLM BankCo, each member of the SLM BankCo Group and each of their respective directors, officers, employees and agents, in each case in their respective capacities as such, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>SLM BankCo Indemnitees</u>"), from and against any and all Losses of the SLM BankCo Indemnitees relating to, arising out of or resulting from, directly or indirectly, any of the following items (without duplication):

(a) Navient Liabilities;

- (b) any breach by Navient or any other member of the Navient Group of this Agreement or any of the Ancillary Agreements (in each case, subject to the limitations set forth in this Agreement or the Ancillary Agreement, as applicable);
- (c) except as set forth in Section 4.3(b) or the Tax Sharing Agreement, any Action or Threatened Action involving SLM BankCo or any of its Subsidiaries or their respective directors, executive officers, employees or agents that arises out of or primarily relates to the Separation or the Distribution:
- (d) except to the extent it relates to a SLM BankCo Liability, any guarantee, indemnification obligation, surety bond or other credit support agreement, arrangement, commitment or understanding for the benefit of any member of the Navient Group or provided in connection with any Navient Asset or Navient Liability by any member of the SLM BankCo Group that continues after the Effective Time;
- (e) the repurchase or redemption, in whole or in part, of the SLM BankCo Series A Preferred Stock or the SLM BankCo Series B Preferred Stock (or both) following the entry of a judgment or order adverse to SLM BankCo of a court in any Action, filed within three years of the Effective Time, arising out of (x) any actions or omissions of Existing SLM or SLM BankCo prior to the Effective Time or (y) the Separation and the Distribution (the "Trigger Event"). Navient's obligation to indemnify SLM BankCo for such Losses shall be limited to the remedy set forth below in this Section 4.2(e), which shall be subject to the following conditions precedent: (i) within 90 days of the Trigger Event, SLM BankCo shall have received an instruction, order or decree of a Governmental Authority having regulatory authority over SLM BankCo (a

"Regulatory Authority") that SLM BankCo increase its capital; and (ii) within 12 months after the Trigger Event, SLM BankCo shall have used its good faith reasonable efforts to raise such capital by means of a public or private sale of equity or debt securities in the capital markets. If the foregoing conditions are met, then Navient shall purchase from SLM BankCo, upon its request given within 12 months after the Trigger Event, preferred stock or debt securities of SLM BankCo (x) that would qualify as Tier 1 capital under Basel III, assuming SLM BankCo were a bank holding company subject to regulation under the Bank Holding Company Act of 1956, as amended, (y) the purchase price for which is equal to the lesser of (A) \$400 million and (B) the minimum amount of capital required by the Regulatory Authority to be raised by SLM BankCo and (z) having such other terms and conditions (including as to pricing) as Navient may reasonably request as are customary for similar securities of issuers having a similar credit profile as SLM BankCo and not inconsistent with the requirement of the preceding clause (A);

(f) the inability of any member of the SLM BankCo Group to sell, finance or securitize, on customary terms, loans that are SLM BankCo Assets as of the Effective Time due to inadequate or deficient servicing, originations, operations or collection activities with respect to such loans performed by Existing SLM or any of its Subsidiaries (including SMI) prior to the Effective Time ("Deficient Loans"). Navient's obligation to indemnify SLM BankCo for such Losses shall be limited to the remedy set forth below in this Section 4.2(f), which shall be subject to the condition that Navient receive from SLM BankCo written notice in accordance with Section 10.5 by or before the Servicing Liability Cut-Off Date setting forth in reasonable detail SLM BankCo's efforts to sell, finance or securitize the Deficient Loans and the basis on which SLM BankCo has determined such loans constitute Deficient Loans. If the foregoing condition is met, then Navient shall purchase, at SLM BankCo's request, such Deficient Loans at a price of 100% of the principal balance plus 100% of the accrued and unpaid interest thereon to the date of purchase.

(g) any untrue statement or alleged untrue statement of a material fact or omission or alleged omission to state a material fact required to be stated therein or necessary to make the statements therein not misleading, with respect to all information contained in the Form 10, the Information Statement or any other Disclosure Document.

To the extent that a reserve or other account (a "Reserve") is or was created by Effective Time Sallie Mae on behalf of any member of the SLM BankCo Group, at or prior to the Effective Time, in respect of potential Losses for which such member of the SLM BankCo Group is entitled to indemnification pursuant to this Section 4.2, then such member of the SLM BankCo Group shall first apply the amount in the Reserve against such Losses before seeking indemnification therefor from Navient hereunder.

4.3 <u>Indemnification by SLM BankCo</u>. SLM BankCo shall, and shall cause the other members of the SLM BankCo Group to, indemnify, defend and hold harmless Navient, each other member of the Navient Group and each of their respective directors, officers, employees or agents, in each case in their respective capacities as such, and each of the heirs, executors, successors and assigns of any of the foregoing (collectively, the "<u>Navient Indemnitees</u>"), from and against any and all Losses of the Navient Indemnitees relating to, arising out of or resulting from, directly or indirectly, any of the following items (without duplication):

(a) the SLM BankCo Liabilities;

- (b) subject to Section 4.2(e), any Action (including the Actions captioned William McCrady v. SLM Corporation et. al., C.A. No. 9285-VCL and James L. Myers v. SLM Corporation et. al., C.A. No. 9371-VCL) relating to claims made by any current or former preferred stockholder of Existing SLM or SLM BankCo against any Party (or their respective directors, officers or employees) relating to, arising out of or resulting from (i) actions or omissions of Existing SLM or SLM BankCo at or prior to the Effective Time or (ii) the Separation and the Distribution; and
- (c) any breach by SLM BankCo or any other member of the SLM BankCo Group of this Agreement or any of the Ancillary Agreements (in each case, subject to the limitations set forth in this Agreement or the Ancillary Agreement, as applicable).

4.4 Indemnification Obligations Net of Insurance Proceeds and Other Amounts.

- (a) The Parties intend that any Losses subject to indemnification or reimbursement payment owed or made pursuant to this Article IV or Article V will be net of applicable Insurance Proceeds. Accordingly, the amount which any Party (an "Indemnifying Party") is required to pay to any Person entitled to indemnification hereunder (an "Indemnitee") will be reduced by any Insurance Proceeds theretofore actually recovered by or on behalf of the Indemnitee on account of a covered claim under any applicable policy of insurance and in respect of the related liability. If an Indemnitee receives a payment (an "Indemnity Payment") required by this Agreement from an Indemnifying Party in respect of any Losses and subsequently receives Insurance Proceeds, then the Indemnitee will promptly pay to the Indemnifying Party the amount of the Insurance Proceeds for such Losses, not to exceed the amount of the Indemnity Payment previously paid by the Indemnifying Party. The Parties acknowledge and agree that payment by an Indemnifying Party of an Indemnity Payment shall not be subject to, or conditioned upon, the prior receipt of Insurance Proceeds or filing of an insurance claim by the Indemnitee or the completion by any insurance company of its processes and procedures in respect of the same claim
- (b) An insurer who would otherwise be obligated to pay any claim shall not be relieved of the responsibility with respect thereto, it being expressly understood and agreed that no insurer or any other third party shall be entitled to a "windfall" (i.e., a benefit they would not be entitled to receive in the absence of the indemnification provisions) by virtue of the indemnification provisions hereof.

(c) The Parties intend that, except to the extent otherwise provided in an Ancillary Agreement, any indemnification or reimbursement payment in respect of Losses pursuant to this Article IV or Article V shall be (i) reduced to take into account the present value amount of any projected Tax benefit to the Indemnitee or its Affiliates resulting from the Losses so indemnified or reimbursed and (ii) increased so that the amount of such payment, reduced by the present value amount of all Taxes projected to be payable with respect to the receipt thereof (including by reason of any resulting loss of Tax benefits, but taking into account all correlative Tax benefits resulting from the payment of such Taxes), shall equal the amount of the payment which the Person receiving such payment would otherwise be entitled to receive pursuant to this Agreement. For purposes of this Section 4.4(c), the present value amount of any projected Tax benefit and any projected Taxes payable shall be calculated based on reasonable reporting positions, reasonable projections of applicable tax items (including income, gains, losses, deductions, and credits) and reasonable projections of applicable Tax rates of the Indemnitee or its Affiliates, as the case may be, supported by, and subject to the advice of, the outside tax advisor of the Indemnitee or its Affiliates, as the case may be, and a mutually agreed discount rate equal to the sum of LIBOR and the Indemnitee's credit spread over the yield of U.S. treasury notes having a duration approximately equal to the required tax payments. For all Tax purposes, to the extent permitted by applicable Tax Law, the Parties will treat any indemnification or reimbursement payment in respect of a liability pursuant to this Article IV or Article V as a capital contribution made by SLM BankCo to Navient or as a distribution made by Navient to SLM BankCo, as the case may be, on the date recited above on which the parties entered into the Agreement.

4.5 Procedures for Indemnification of Third Party Claims.

(a) If an Indemnitee shall receive notice or otherwise learn of the assertion by a Person (including any Governmental Authority) who is not a member of the SLM BankCo Group or the Navient Group of the commencement by any such Person of any Action against such Indemnitee (collectively, a "Third Party Claim") with respect to which an Indemnifying Party may be obligated to provide indemnification to such Indemnitee pursuant to Section 4.2 or 4.3, or any other Section of this Agreement or any Ancillary Agreement, such Indemnitee shall give such Indemnifying Party written notice thereof within 15 business days after becoming aware of such Third Party Claim. Any such notice shall describe the Third Party Claim in reasonable detail and include copies of all notices and documents (including court papers) received by the Indemnitee relating to the Third Party Claim. Notwithstanding the foregoing, the failure of an Indemnitee to provide timely written notice in accordance with this Section 4.5(a) shall not relieve an Indemnifying Party of its indemnification obligations under this Agreement, except to the extent to which the Indemnifying Party is actually prejudiced by the Indemnitee's failure to provide notice in accordance with this Section 4.5(a).

(b) Subject to the terms and conditions of any applicable iinsurance policy in place after the Effective Time, an Indemnifying Party may elect to defend and, unless the Indemnifying Party has specified any reservations or exceptions, to seek to settle or compromise, at such Indemnifying Party's own expense and by such Indemnifying Party's own counsel, any Third Party Claim if (i) the Third Party Claim involves only money damages and does not seek an injunction or other equitable relief against an Indemnified Party; and (ii) the Third Party Claim is not made by a Governmental Authority; provided, however, that Navient shall be required, at the election of SLM BankCo, to assume the defense of a Third Party Claim made by a Governmental Authority that primarily relates to or arises out of servicing and collections activities constituting Navient Liabilities. Within 30 days after the receipt of notice from an Indemnitee in accordance with Section 4.5(a) (or sooner, if the nature of such Third Party Claim so requires), the Indemnifying Party shall notify the Indemnitee of its election, if available to it, whether the Indemnifying Party will assume responsibility for defending such Third Party Claim, which election shall specify any reservations or exceptions. After notice from an Indemnifying Party to an Indemnitee of its election to assume the defense of a Third Party Claim, such Indemnitee shall have the right to employ separate counsel and to participate in (but not control) the defense, compromise or settlement thereof, but the fees and expenses of such counsel shall be the expense of such Indemnitee except as set forth in the next sentence. In the event that the Indemnifying Party has elected to assume the defense of the Third Party Claim but has specified, and continues to assert, any reservations or exceptions in such notice, then, in any such case, the reasonable fees and expenses of one separate counsel (along with one local counsel in each applicable jurisdiction) for all Indemnitees shall be borne by the I

(c) If an Indemnifying Party elects not to assume responsibility for defending a Third Party Claim or does not have the right to elect to assume responsibility therefor pursuant to Section 4.5(b), or fails to timely notify an Indemnitee of its election as provided in Section 4.5(b), such Indemnitee may defend such Third Party Claim at the cost and expense of the Indemnifying Party. If an Indemnifying Party is not permitted to assume the defense of a Third Party Claim pursuant to Section 4.5(b), the Indemnitee shall have the right to employ, at the Indemnifying Party's expense, counsel, experts and vendors of its choice (subject to the approval of the Indemnifying Party, which approval shall not be unreasonably withheld or delayed) to represent the Indemnitee, and the Indemnifying Party shall only be obligated to indemnify the Indemnitee for reasonable costs and expenses in connection with the defense of such Third Party Claim. If an Indemnitee engages in discussions to compromise or settle a Third Party Claim, such Indemnitee shall in all cases promptly share with the Indemnifying Party such discussions and all other communications made in connection with such compromise or settlement discussions.

(d) In the case of a Third Party Claim, (i) no Indemnifying Party shall consent to entry of any judgment or enter into any settlement of the Third Party Claim without the written consent of the Indemnitee if (A) the effect thereof is to permit any injunction, declaratory judgment, other order or other non-monetary relief to be entered, directly or indirectly against any Indemnitee, (B) such settlement or judgment requires or contains any admission of liability by the Indemnitee, (C) such settlement does not include a full and unconditional release of the Indemnitee as to that Third Party Claim, or (D) such judgment or settlement imposes any obligation or restriction on the Indemnitee other than the payment of money and (ii) unless the Indemnifying Party has failed to assume the defense of the Third Party Claim that it is permitted to assume in accordance with the terms of this Agreement or in circumstances under which an Indemnifying Party is not permitted to assume the defense of a Third Party Claim pursuant to Section 4.5(b), no Indemnitee shall consent to entry of any judgment or enter into any settlement of the Third Party Claim without the written consent of the Indemnifying Party.

(e) The above provisions of this Section 4.5 and the provisions of Section 4.6 do not apply to Taxes (which are governed by the Tax Sharing Agreement). In the case of any conflict between this Agreement and the Tax Sharing Agreement in relation to any matters addressed by the Tax Sharing Agreement, the Tax Sharing Agreement shall prevail.

4.6 Additional Matters.

(a) Subject to the terms, conditions and limitations of any applicable insurance policy, indemnification payments in respect of any Losses for which an Indemnitee is entitled to indemnification under this Article IV shall be paid by the Indemnifying Party to the Indemnitee as such Losses are incurred upon written demand by the Indemnitee, including reasonably satisfactory documentation setting forth the basis for the amount of such indemnification payment, including documentation with respect to calculations made and consideration of any Insurance Proceeds that actually reduce the amount of such indemnification payment or reimbursement. The indemnity agreements contained in this Article IV shall remain operative and in full force and effect, regardless of (i) any investigation made by or on behalf of any Indemnitee prior to the Effective Time, (ii) the knowledge by the Indemnitee of Losses for which it might be entitled to indemnification hereunder and (iii) any termination of this Agreement.

(b) Any claim on account of Losses for which an Indemnitee is entitled to indemnification under this Article IV which does not result from a Third Party Claim shall be asserted by written notice given by the Indemnitee to its Indemnifying Party. The written notice shall describe the Losses asserted and the

Indemnity Payment due in reasonable detail. Such Indemnifying Party shall have a period of 15 business days (unless a shorter period is required under the circumstances) after the receipt of such notice within which to respond thereto. If such Indemnifying Party does not respond within such 15-business day period, such Indemnifying Party shall be deemed to have refused to accept responsibility to make payment. If such Indemnifying Party does not respond within such 15-business day period or rejects such claim in whole or in part, such Indemnitee shall be free to pursue such remedies as may be available to such party as contemplated by this Agreement and the Ancillary Agreements.

- (c) In the event of payment by or on behalf of any Indemnifying Party to any Indemnitee in connection with any Third Party Claim, such Indemnifying Party shall be subrogated to and shall stand in the place of such Indemnitee as to any events or circumstances in respect of which such Indemnitee may have any right, defense or claim relating to such Third Party Claim against any claimant or plaintiff asserting such Third Party Claim or against any other Person. Such Indemnitee shall cooperate with such Indemnifying Party in a reasonable manner, and at the reasonable cost and expense of such Indemnifying Party, in prosecuting any subrogated right, defense or claim.
- (d) In the event of an Action relating to or arising out of Losses subject to indemnification or reimbursement pursuant to Article IV or Article V in which the Indemnifying Party is not a named defendant, if either the Indemnitee or Indemnifying Party shall so request, the Parties shall endeavor to substitute the Indemnifying Party for the named defendant. If such substitution or addition cannot be achieved for any reason or is not requested, the named defendant shall allow the Indemnifying Party to manage the Action as set forth in this Section 4.6(d), and the Indemnifying Party shall fully indemnify the named defendant against all reasonable costs of defending the Action (including court costs, sanctions imposed by a court, attorneys' fees, experts fees and all other external expenses), the costs of any judgment or settlement and the cost of any interest or penalties relating to any judgment or settlement.
- 4.7 <u>Remedies Cumulative</u>. The remedies provided in this <u>Article IV</u> shall be cumulative and, subject to the provisions of <u>Article VIII</u>, shall not preclude assertion by any Indemnitee of any other rights or the seeking of any and all other remedies against any Indemnifying Party.
- 4.8 <u>Survival of Indemnities</u>. The rights and obligations of each of SLM BankCo and Navient and their respective Indemnitees under this <u>Article IV</u> shall survive the sale or other transfer by any party of any Assets or businesses or the transfer or assignment by it of any Liabilities and the termination or expiration of this Agreement.

ARTICLE V CERTAIN OTHER MATTERS

5.1 Insurance Matters.

- (a) Existing SLM, SLM BankCo and Navient agree to cooperate in good faith to provide for an orderly transition of insurance coverage. Following the Effective Time, SLM BankCo and Navient agree to cooperate with respect to insurance coverage matters as described in this Section 5.1.
- (b) From and after the Effective Time, other than as expressly provided in Section 5.1(c) with respect to any Shared Policies and other than with respect to their Joint E&O Policy and Joint F.I. Bond Policy, neither SLM BankCo nor Navient, nor any member of its respective Group, shall have any rights to or under any of the other Group's insurance policies, including not having access to and not having the right to make claims under any of the other Group's insurance policies. At the Effective Time, each of Navient and SLM BankCo shall have in effect insurance policies as reasonably necessary or customary for companies operating a similar business. Such insurance programs may include, but are not limited to, comprehensive general liability, umbrella/excess liability, above ground/underground storage tank liability, surety bonds, commercial auto liability, workers' compensation, employer's liability, professional liability, property, employee dishonesty/crime, directors' and officers' liability and fiduciary liability.
- (c) From and after the Effective Time, with respect to any losses, damages and liability incurred by any member of the Navient Group or the SLM BankCo Group for which such member is entitled to coverage under the D&O Tail Policy or Historical P/C Policies of insurance (collectively, the "Shared Policies"), each Party will provide the other Party (or any member of the other Party's Group) with access to, and may make claims under, the Shared Policies, but solely to the extent that such Shared Policies provide coverage therefor; provided that such access to, and the right to make claims under, such insurance policies, shall be subject to the terms, conditions, and exclusions of such insurance policies, including any limits on coverage or scope, any deductibles or retentions and other fees and expenses, and shall be subject to the following additional conditions:
 - (i) each of Navient and SLM BankCo shall provide, as promptly as practicable, notice to the other Party with respect to any claims to be made under any Shared Policy and shall report, as promptly as practicable, claims in accordance with the terms of the applicable insurance policy;
 - (ii) except with respect to claims made by the SLM BankCo Group that relate to Liabilities that are Navient Liabilities, SLM BankCo shall pay to the Navient Group an amount equal any deductibles, self-insured retention, fees and expenses incurred by any member of the Navient Group to the extent resulting from any access to or any claims made by any member of the SLM BankCo Group under any Shared Policy;

- (iii) except with respect to claims made by the Navient Group that relate to Liabilities that are SLM BankCo Liabilities, Navient shall pay to the SLM BankCo Group an amount equal to any deductibles, self-insured retention, fees and expenses incurred by any member of the SLM BankCo Group to the extent resulting from any access to or any claims made by any member of the Navient Group under any Shared Policy; and
- (iv) except to the extent for payments and reimbursements required pursuant to Section 5.1(c)(ii) and (iii), neither the Navient Group nor the SLM BankCo Group shall have any liability or indemnification obligation to repay or reimburse the other Party for any uninsured, uncovered, unavailable or uncollectible insurance amounts under the Shared Policies to the extent resulting from any access to or any claims made by any member of either Group under any Shared Policy; provided, however, that any such uninsured, uncovered, unavailable or uncollectible insurance amounts under the Shared Policies shall not relieve or reduce an Indemnifying Party of its indemnification obligations under this Agreement.
- (d) All payments and reimbursements pursuant to <u>Section 5.1(c)(ii)</u> and <u>(iii)</u> will be made on demand after receipt of an invoice therefor from the other Party. If a Party incurs costs to enforce the other Party's obligations under <u>Section 5.1(c)(ii)</u> or <u>(iii)</u>, as applicable, the other Party shall indemnify such Party for reasonable enforcement costs, including reasonable attorneys' fees.
- (e) Prior to the Effective Time Navient and SLM BankCo shall jointly purchase a Joint E&O Policy and Joint F.I. Bond Policy, covering acts and omissions and on terms substantially similar to those obtained for similar insurance policies by Existing SLM prior to the Effective Time, with a term ending on the one-year anniversary of the Effective Time. Unless waived by the Parties, Navient and SLM BankCo shall extend or purchase a new Joint E&O Policy and Joint F.I. Bond Policy, on no less favorable terms, for such period or periods as may be necessary for them to be in effect throughout the period of the IT Transition. The cost of each Joint E&O Policy and Joint F.I. Bond Policy shall be shared by the Parties (including extensions and replacements) based on their relative revenue for the four most recent calendar quarters (for which financial information is available) prior to the effective date of such policies; *provided*, the initial premium shall be based on the pro forma revenue for 2013 for Navient and SLM BankCo, after giving effect to the transactions contemplated by this Agreement.

- (f) Each of Navient and SLM BankCo shall have the right to erode, exhaust limits, settle, release or otherwise resolve disputes relating to claims made by such Party under any Shared Policy, Joint E&O Policy or Joint F.I. Bond Policy. Navient and SLM BankCo shall cooperate and share such information as is reasonably necessary in order to permit the Parties to manage and conduct their insurance matters with respect to each such policy.
- (g) This Agreement shall not be considered as an attempted assignment of any policy of insurance or as a contract of insurance and shall not be construed to waive any right or remedy of any member of the SLM BankCo Group or the Navient Group in respect of any insurance policy.
- 5.2 <u>Late Payments</u>. Except as expressly provided to the contrary in this Agreement or in any Ancillary Agreement, any undisputed amount not paid when due pursuant to this Agreement or any Ancillary Agreement (and any amounts billed or otherwise invoiced or demanded and properly payable that are not paid within 30 days of such bill, invoice or other demand) shall accrue interest at a rate per annum equal to the Prime Rate plus 2%.
- 5.3 Amendment to Existing SLM Holdco Operating Agreement. As promptly as practicable (and in any event within five Business Days) after the Effective Time, Navient, as the sole member of Existing SLM Holdco, shall approve an amendment to the Existing SLM Holdco Operating Agreement to eliminate the requirement, set forth in Article III thereof, for SLM BankCo stockholders to approve certain actions by or involving Existing SLM Holdco as required by Section 251(g) of the DGCL (the "Operating Agreement Amendment"). SLM BankCo shall use its reasonable best efforts to (a) include in the proxy statement for the 2014 annual meeting of common stockholders of SLM BankCo (or any earlier meeting of such stockholders as determined by the SLM BankCo Board) a proposal to approve the Operating Agreement Amendment (the "Operating Agreement Amendment Proposal"), along with a recommendation of the SLM BankCo Board that the common stockholders of SLM BankCo approve the Operating Agreement Amendment Proposal, and (b) solicit the approval of the common stockholders of SLM BankCo of the Operating Agreement Amendment Proposal. In the event that the Operating Agreement Amendment Proposal is not approved at such annual meeting (or earlier meeting), SLM BankCo shall use its reasonable best efforts to obtain the approval of the Operating Agreement Amendment Proposal at each subsequent annual or special meeting of common stockholders of SLM BankCo until such approval is obtained.

5.4 Non-Competition Matters.

(a) Certain Definitions. For purposes of this Section 5.4, (i) "making," "make" or "solicit" refers to the process of acting or seeking to act as a lender in providing funds, as opposed to acting as a service provider for others, (ii) "origination services" refers to providing loan origination services (e.g., application processing and underwriting) for unrelated third parties, as opposed to acting as a lender in providing funds, (iii) "customer" is as defined in the regulations promulgated under the Gramm-

Leach-Bliley Act and (iv) a "Loan Customer" is a customer whose education loan is owned or was securitized by an entity, and for clarification, a "Navient Loan Customer" includes a Loan Customer whose education loan is owned by a member of the Navient Group or was securitized by Effective Time Sallie Mae on or prior to the Effective Time or by any members of the Navient Group from and after the Effective Time and remains outstanding and a "BankCo Loan Customer" includes a Loan Customer whose education loan is owned by a member of the SLM BankCo Group or was securitized by any member of the BankCo Group from and after the Effective Time and remains outstanding.

(b) Navient Restricted Business. From the Effective Time through December 31, 2018 (the "Non-Competition Period"), except as set forth in Section 5.4(g)(i) below. Navient will not, and will cause its Subsidiaries not to, advise, engage in or carry on any Navient Restricted Business (as defined below) or own, operate, control, share any revenues of or have any profit or other equity interest in any business engaged in a Navient Restricted Business. The "Navient Restricted Business" shall mean any of the following activities in the United States: (i) making Private Education Loans directly as a lender; (ii) entering into agreements to purchase Private Education Loans that have not yet been made; (iii) entering into agreements to provide Private Education Loan origination services, other than to Navient Excluded Persons; (iv) entering into agreements to provide servicing for Private Education Loans, other than to Navient Excluded Persons, members of the Navient Group and securitization trusts created by Effective Time Sallie Mae or, after the Effective Time, members of the Navient Group; (v) subject to Section 5.4(d), making a loan to consolidate or refinance a BankCo Loan Customer's Private Education Loans unless the borrower is a Navient Loan Customer immediately prior to making such loan; (vi) making any of the other consumer loans included in the SLM BankCo business plan presented to the Existing SLM Board in October 2013 (other than loans that consolidate or refinance education loans, Government Education Loans, Permitted Government Education Loans or FFELP loans otherwise permitted to be made by Navient in accordance with this Agreement); or (vii) advising or consulting with any unrelated third party regarding any of the foregoing activities (other than, as to clause (iii) or (iv) above, with Navient Excluded Persons [or with respect to processing services such as loan origination services or loan servicing not otherwise prohibited by clauses (i)-(vi) above)]. For the avoidance of doubt, the Navient Restricted Business shall not include (x) the ownership of Equity Interests in SMI; (y) the provision of services to, or for the benefit of, SLM BankCo as set forth in the Transition Services Agreement or any other Ancillary Agreement or as otherwise requested by SLM BankCo; or (z) agreements or arrangements with a third party in effect as of the Effective Time (and renewals or replacements thereof with the same party, its successor or its Affiliates).

- (c) SLM BankCo Restricted Business. During the Non-Competition Period, SLM BankCo will not, and will cause its Subsidiaries not to, advise, engage in or carry on any SLM BankCo Restricted Business (as defined below) or own, operate, control, share any revenues of or have any profit or other equity interest in any business engaged in a SLM BankCo Restricted Business. The "SLM BankCo Restricted Business" shall mean any of the following activities in the United States: (i) making Government Education Loans or FFELP loans directly as a lender; (ii) providing origination services, acquiring, collecting, holding or providing servicing activities (including loan servicing, default prevention, portfolio management and cohort rate management services) with respect to Government Education Loans, FFELP Loans or DSLP Loans); (iii) engaging in any third party debt collection (other than in respect of education loans owned by a member of the SLM BankCo Group at or after the Effective Time); (iv) subject to Section 5.4(d), making a loan to consolidate or refinance a Navient Loan Customer's Government Education Loans, FFELP Loans or Private Education Loans unless the borrower is a BankCo Loan Customer immediately prior to making such loan; (v) except as otherwise provided below, engaging in services included in the Navient business plan presented to the Existing SLM Board in October 2013; or (vi) advising or consulting with any third party regarding any of the foregoing. For the avoidance of doubt, the SLM BankCo Restricted Business shall not include (x) the ownership of the SMI Preferred Stock; (y) the provision of services to, or for the benefit of, Navient as set forth in the Transition Services Agreement or any other Ancillary Agreement or as otherwise requested by Navient; or (z) servicing or holding FFELP Loans owned by Sallie Mae Bank at the Effective Time.
- (d) Consolidation Loans. During the Non-Competition Period, and subject to the terms of any applicable loan sale agreement between any member of the Navient Group and any member of the SLM BankCo Group:
 - (i) If any member of the Navient Group or Effective Time Sallie Mae (other than Sallie Mae Bank) purchased or purchases a loan from any member of the SLM BankCo Group, the SLM BankCo Group may not solicit a consolidation loan from the same borrower (a "customer") even if the SLM BankCo Group makes a new Private Education Loan or Permitted Government Education Loan after the date the prior loan was purchased by the Navient Group; provided, however, that if the SLM BankCo Group complies with the second and third sentences of this Section 5.4(d)(i), making a consolidation loan to such customer shall not constitute a breach of this Agreement. If the SLM BankCo Group makes a consolidation loan to such customer within three years of the date the loan(s) were acquired from Sallie Mae Bank or any other member of the SLM BankCo Group by the Navient Group or Effective Time Sallie Mae (other than Sallie Mae Bank), whether such acquisition occurred before or after the Effective Time), SLM BankCo shall present an option to the Navient Group to purchase such loan within 90 days of the date the consolidation loan was made, at a price of 100% of the principal balance plus 100% of the accrued and unpaid interest ("Par"). If Navient (or another member of the Navient Group) does not purchase the loan

within 60 days of SLM BankCo's request, then within the next 30 days SLM BankCo (or its designee within the SLM BankCo Group) will refund the premium paid on any of the underlying loan(s) that were owned by a member of the Navient Group based on the following schedule: if the consolidation loan is made (A) within the first year following the purchase, 88% of the premium will be refunded, (B) in the second year following purchase, 82% of the premium will be refunded, and (C) in the third year, 75% of the premium will be refunded.

(ii) If any member of the SLM BankCo Group sells a loan to any member of the Navient Group (including any entity that becomes a member of the Navient Group after the Effective Time), the Navient Group may not solicit or make a consolidation loan to that customer that includes other loans of such customer that are held by the SLM BankCo Group; provided, however, that if the Navient Group complies with the second, third and fourth sentences of this Section 5.4(d)(i), making a consolidation loan to such customer that includes such other loans shall not constitute a breach of this Agreement. If, within three years of the date any member of the Navient Group (including any entity that becomes a member of the Navient Group after the Effective Time) acquired such loan from a member of the SLM BankCo Group, whether such loan was acquired before or after the Effective Time, any member of the Navient Group makes a consolidation loan which includes a loan made to such customer that is held by the SLM BankCo Group, Navient (or its designee within the Navient Group) will present to SLM BankCo, within 90 days of the date the consolidation loan was made, the right (a) to buy such loan at a price of Par within 60 days of Navient's notification, in which case simultaneously with such purchase, SLM BankCo (or its designee within the SLM BankCo Group) shall refund the premium paid by the applicable member of the Navient Group per the schedule set forth in Section 5.4(d)(i), or (b) to propose a premium that Navient (or its designee within the Navient Group) would pay for the portion of the principal of the consolidation loan that was consolidated away from the SLM BankCo Group (the "BankCo Principal"). Within 15 days of being notified by Navient of the making of such consolidation loan, SLM BankCo will advise Navient in writing as to whether it (or its designee) will acquire the consolidation loan on the terms set forth in this paragraph, or SLM BankCo will propose in writing the premium that Navient (or its designee) would pay for the BankCo Principal. Within 45 days following such notification, Navient or another member of the Navient Group will either sell such consolidation loan to SLM BankCo on the terms set forth in this paragraph, or, at Navient's option, pay the premium on such BankCo Principal to SLM BankCo or its designee (in lieu of

(e) Permitted Government Education Loans. The Parties acknowledge that Permitted Government Education Loans have attributes of both a Private Education Loan and a Government Education Loan and that both Navient and SLM BankCo should be permitted to participate in any market for Permitted Government Education Loans, subject to their compliance with this <u>Section 5.4(e)</u>. During the Non-Competition Period, the Parties agree that:

- (i) Permitted Government Education Loans shall be treated as Private Education Loans for purposes of Section 5.4(b) and the definition of "Navient Restricted Business," unless prior to taking any of the actions specified in clauses (i)—(vi), inclusive, of Section 5.4(b) as to a particular Permitted Government Education Loan, Navient, or another member of the Navient Group, first complies with Navient's obligations under Section 5.6(b) (or, if Navient does not intend to market such loans in concert with or through one or more third parties, Navient first notifies SLM BankCo in writing of its plans in respect of such loans that would otherwise violate Section 5.4(b)), in which case such Permitted Government Education Loan shall not be deemed a Private Education Loan or otherwise subject to the restrictions or limitations applicable to the Navient Restricted Business.
- (ii) Permitted Government Education Loans shall be treated as Government Education Loans for purposes of Section 5.4(c) and the definition of "SLM BankCo Restricted Business," unless prior to taking any of the actions specified in clauses (i)—(vi), inclusive, of Section 5.4(c) as to a particular Permitted Government Education Loan, SLM BankCo, or another member of the SLM BankCo Group, first complies with SLM BankCo's obligations under Section 5.6(b) (or, if SLM BankCo does not intend to market such loans in concert with or through one or more third parties, SLM BankCo first notifies Navient in writing of its plans in respect of such loans that would otherwise violate Section 5.4(c)), in which case such Permitted Government Education Loan shall not be deemed a Government Education Loan or otherwise subject to the restrictions or limitations applicable to the SLM Bank.
- (f) Large New Government Education Loans. The Parties acknowledge that the SLM BankCo Group should be permitted to participate, along with the Navient Group, in any market for Large New Government Education Loans, subject to SLM BankCo's compliance with this Section 5.4(f). During the Non-Competition Period, Large New Government Education Loans shall be treated as Government Education Loans for purposes of Section 5.4(c) and the definition of "SLM BankCo Restricted Business," unless prior to taking any of the actions specified in clauses (i)—(vi), inclusive, of Section 5.4(c) as to a particular Large New Government Education Loan program, SLM BankCo, or another member of the SLM BankCo Group, first complies with SLM BankCo's obligations under Section 5.6(b) (or, if SLM BankCo does not intend to market such loans in concert with or through one or more third parties, SLM BankCo first notifies Navient in writing of its plans in respect of such loans, including any that would otherwise violate Section 5.4(c)), in which case Large New Government Education Loans originated under such Large New Government Education Loan program shall not be deemed a Government Education Loan or otherwise subject to the restrictions or limitations applicable to the SLM BankCo Restricted Business.

(g) Acquisitions and Divestitures. The Parties recognize that during the course of the Non-Competition Period each of them may engage in acquisition and divestiture transactions. The Parties further recognize that the restrictions set forth in the other paragraphs of this Section 5.4 (together with this Section 5.4(g), collectively, the "Business Restrictions") have been structured in the context of the anticipated businesses of the Navient Group and the SLM BankCo Group, and may have uncertain application to acquisitions and divestures of companies, businesses and assets. It is the intent of the Parties that the Business Restrictions should apply, for the remainder of the Non-Competition Period, to any companies, business lines or assets that are divested to third parties that close during the period beginning at the Effective Time and ending on May 1, 2017 (the "M&A Out Date"), but should no longer apply to any such divestitures that close after the M&A Out Date. The Parties further intend that the Business Restrictions should apply throughout the Non-Competition Period to any acquisitions of companies, business lines or assets by the Parties. Based on the foregoing, the Parties agree as follows:

(i) Navient Restrictions.

- (A) The operations or assets of any Person or business acquired by a member of the Navient Group (whether acquired pursuant to a merger, consolidation, acquisition of assets or Equity Interests or otherwise) from a third party other than a Navient Excluded Person (a "Newly Acquired Navient Business") after the Effective Time shall be subject to the Business Restrictions, to the same extent as they are applicable to Navient immediately prior to the acquisition of the Newly Acquired Navient Business, for the remainder of the Non-Competition Period. Each Subsidiary or Affiliate of Navient that is engaged in the Newly Acquired Navient Business after the Effective Time shall be treated as a member of the Navient Group for all purposes of this Section 5.4;
- (B) If Navient or any member of the Navient Group acquires a Navient Excluded Person or any product line, business or assets of a Navient Excluded Person (whether acquired pursuant to a merger, consolidation, acquisition of assets or Equity Interests or otherwise) (an "Acquired Excluded Business") after the Effective Time, then the Acquired Excluded Business shall be subject to the Business Restrictions, to the same extent as they are applicable to Navient immediately prior to the acquisition of the Acquired Excluded Business, but only as to business activities of the Acquired Excluded Business after the purchase thereof by Navient or a member of the Navient Group. Pre-existing businesses of the Acquired Excluded Business shall not violate the applicable Business

Restrictions so long as no new contracts, agreements or arrangements are entered into (other than renewals or replacements thereof of similar scope with the same parties or their Affiliates) that would violate the Business Restrictions. Each Subsidiary or Affiliate of Navient that is engaged in the Acquired Excluded Business after the Effective Time shall be treated as a member of the Navient Group for all purposes of this Section 5.4, subject to the provisions of this Section 5.4(g)(i)(B);

(C) If any member of the Navient Group sells or disposes of a company, product line, business or assets (constituting less than all or substantially all of the assets of the Navient Group) to a third party (a "Navient Disposed Business") after the Effective Time and on or prior to the M&A Out Date, then the Navient Disposed Business (but not the third party acquiror or its other Subsidiaries), following such sale or other disposition, shall remain subject to the Business Restrictions, to the same extent as they are applicable to Navient immediately prior to the sale or other disposition of the Navient Disposed Business, for the remainder of the Non-Competition Period. Each entity directly engaged in a Navient Disposed Business sold or otherwise disposed of on or prior to the M&A Out Date (but not the acquiror or its other Subsidiaries) shall be treated as a member of the Navient Group for the remainder of the Non-Competition Period, and Navient shall cause the acquiring third party to enter into an agreement, acceptable to SLM BankCo, that such third party will not take any action to circumvent the foregoing provisions of this Section 5.4(g)(i)(C). If a Navient Disposed Business is sold or otherwise disposed of after the M&A Out Date, (x) the Business Restrictions shall not apply to the Navient Disposed Business, and the acquiror shall take free of the Business Restrictions, and (y) to the extent SLM BankCo was restricted in its activities as to the product line, business or assets included in the Navient Disposed Business pursuant to this Section 5.4, then the SLM BankCo Group shall no longer be bound by such restrictions as they relate to such product line, business or assets; and

(D) If all or substantially all of the assets and businesses of the Navient Group (taken as a whole) are acquired (whether pursuant to a merger, consolidation, acquisition of assets or Equity Interests or otherwise) by a third party or if Navient undergoes a Change of Control (each, a "Navient Acquisition") after the Effective Time and on or prior to the M&A Out Date, then Navient shall cause the acquiring third party to enter into an agreement, acceptable to SLM BankCo, that subjects the operations and activities of such acquiring Person, any Person that controls the acquiror, and their respective Subsidiaries (collectively, and including the Navient Group, the "Navient Acquiror Group") to the

Business Restrictions, to the same extent as they apply to Navient immediately prior to the consummation of the Navient Acquisition, for the remainder of the Non-Competition Period. Each member of the Navient Acquirer Group (as to any Navient Acquisition that occurs on or prior to the M&A Out Date) shall be treated as a member of the Navient Group for the remainder of the Non-Competition Period. If a Navient Acquisition occurs after the M&A Out Date, the Business Restrictions shall not apply to the Navient Acquirer Group (including the Navient Group). If following a Navient Acquisition the Business Restrictions no longer apply to the Navient Group, then the Business Restrictions shall cease to apply for the remainder of the Non-Competition Period to the SLM BankCo Group at the same time.

- (ii) SLM BankCo Restrictions. During the Applicable Non-Competition Period:
 - (A) The operations or assets of any Person or business acquired by a member of the SLM BankCo Group (whether acquired pursuant to a merger, consolidation, acquisition of assets or Equity Interests or otherwise) from a third party (a "Newly Acquired SLM BankCo Business") after the Effective shall be subject to the Business Restrictions, to the same extent as they are applicable to SLM BankCo, for the remainder of the Non-Competition Period. Each Subsidiary or Affiliate of SLM BankCo that is engaged in the Newly Acquired SLM BankCo Business shall be treated as a member of the SLM BankCo Group for all purposes of this Section 5.4;
 - (B) If any member of the SLM BankCo Group sells or disposes of a company, product line, business or assets (constituting less than all or substantially all of the assets of the SLM BankCo Group) to a third party (a "SLM BankCo Disposed Business") after the Effective Time and on or prior to the M&A Out Date, then the SLM BankCo Disposed Business (but not the third party acquiror or its other Subsidiaries), following such sale or other disposition, shall remain subject to the Business Restrictions, to the same extent as they are applicable to SLM BankCo, for the remainder of the Non-Competition Period. Each entity directly engaged in a SLM BankCo Disposed Business sold or otherwise disposed of on or prior to the M&A Out Date (but not the acquiror or its other Subsidiaries) shall be treated as a member of the SLM BankCo Group for the remainder of the Non-Competition Period, and SLM BankCo shall cause the acquiring third party to enter into an agreement, acceptable to Navient, that such third party will not take any action to circumvent the foregoing provisions of this Section 5.4(g)(ii)(C). If a SLM BankCo Disposed Business is sold or otherwise disposed of on or after the M&A Out Date,

(x) the Business Restrictions shall not apply to the SLM BankCo Disposed Business, and the acquiror shall take free of the Business Restrictions, and (y) to the extent Navient was restricted in its activities as to the product line, business or assets included in the SLM BankCo Disposed Business pursuant to this <u>Section 5.4</u>, then the Navient Group shall no longer be bound by such restrictions as they relate to such product line, business or assets; and

(C) If all or substantially all of the assets and businesses of SLM BankCo (taken as a whole) are acquired (whether pursuant to a merger, consolidation, acquisition of assets or Equity Interests or otherwise) by a third party, or if SLM BankCo or Sallie Mae Bank undergoes a Change of Control (each, an "SLM BankCo Acquisition") after the Effective Time and on or prior to the M&A Out Date, then SLM BankCo shall cause the acquiring third party to enter into an agreement, acceptable to Navient, that subjects the operations and activities of such acquiring Person, any Person that controls such acquiring Person, and their respective Subsidiaries (collectively, and including the SLM BankCo Group, the "SLM BankCo Acquiror Group") to the Business Restrictions, to the same extent as they apply to SLM BankCo immediately prior to the SLM BankCo Acquisition, for the remainder of the Non-Competition Period. Each member of the SLM BankCo Acquiror Group (as to any SLM BankCo Acquisition that occurs on or prior to the M&A Out Date) shall be treated as a member of the SLM BankCo Group for the remainder of the Non-Competition Period. If a SLM BankCo Acquisition occurs after the M&A Out Date, the Business Restrictions shall not apply to the SLM BankCo Acquiror Group (including the SLM BankCo Group). If following a SLM BankCo Acquisition the Business Restrictions no longer apply to the SLM BankCo Group, then the Business Restrictions shall cease to apply to the Navient Group for the remainder of the Non-Competition Period at the same time. Sallie Mae Bank shall be deemed to constitute "all or substantially all" of the business and assets of the SLM BankCo Group (taken as a whole) for purposes of this Section 5.4(g)(ii).

5.5 Non-Solicitation Matters. During the Non-Solicitation Period, each of SLM BankCo and Navient agrees not to solicit (and to cause its Subsidiaries not to solicit) Covered Personnel of the other Party's Group; provided, however, that it is understood that the obligations in this Section 5.5 shall not prohibit (i) generalized solicitations by advertising and the like that are not directed to Covered Personnel, (ii) solicitations of individuals whose employment was terminated by the other Party or a member of its Group or (iii) the transfer of any Delayed Transfer Employees. "Covered Personnel" means any employee, personnel or individual consultant or independent contractor that is employed by, or provides services to, the SLM BankCo Group or the Navient Group, as applicable, as of the Effective Time. The "Non-Solicitation Period" shall mean (i) two years following the Distribution Date for Covered Personnel located in Newark, Delaware, Reston, Virginia Newton, Massachusetts or Fishers, Indiana and (ii) one year following the Distribution Date for Covered Personnel located in any other location.

5.6 Cooperation.

- (a) *Third Party TSA Obligations*. Each of Navient and SLM BankCo shall use commercially reasonable efforts to assist and coordinate with the other Party to perform the obligations identified in the Transition Services Agreement under certain third party transition services agreements and other matters that may arise in connection with the sale of the Campus Solutions and UII businesses.
- (b) First Look. For a period of two years following the Effective Time (the "First Look Period"), each of Navient and SLM BankCo (as applicable, the "Offering Party") will first share with the other Party products or services (excluding products and services offered by Effective Time Sallie Mae, contemplated by the business plan of such Party approved by the Existing SLM Board in October 2013 prior to the Effective Time or contemplated by the Joint Marketing Agreement) that (i) the Offering Party intends to market broadly to its customers or other businesses in concert with or through one or more third parties or to obtain from a third party (other than, in each case, a third party that is a Governmental Authority) and (ii) with respect to which the Offering Party reasonably determines that an opportunity exists for the other Party with a reasonably detailed summary of the Eligible Offering, including a description of the opportunity envisioned by the Offering Party to be provided to the other Party, which may include (in the sole discretion of the Offering Party) a good faith estimate of the compensation to be provided to the other Party in connection with such opportunity. Notwithstanding the foregoing, the First Look Period shall be deemed commensurate with the Non-Competition Period for purposes of (i) a Party's compliance with is obligations pursuant to Section 5.4(e) with respect to Permitted Government Education Loans and (ii) SLM BankCo's compliance with is obligations pursuant to Section 5.4(f) with respect to Large New Government Education Loans.
- 5.7 Non-Disparagement. During the Non-Competition Period, each of SLM BankCo and Navient agree (and will cause the members of its respective Group and its and their employees acting in their capacity as employees) not to make written statements that are disparaging to the other Party or any member of such other Party's Group, unless reasonably required in response to any legal process or subpoena, to comply with Law or otherwise in connection with the enforcement of this Agreement or any Ancillary Agreement. If a Party is acquired in a transaction contemplated by Section 5.4(g)(i)(D) or Section 5.4(g)(ii)(C), the provisions of this Section 5.7 shall apply to both Parties for the remainder of the Applicable Non-Competition Period and no longer.

5.8 IT Transition. Prior to the Effective Time, SLM BankCo and Navient will enter into the Transition Services Agreement to, among other things, (i) facilitate the separation and migration of SLM BankCo Information, SLM BankCo Intellectual Property, SLM BankCo Software, SLM BankCo Technology, SLM BankCo Systems, the ATLAS Software and any IT hardware included within the SLM BankCo Assets into an independent IT environment operated by or for the benefit of SLM BankCo, (ii) duplicate and transfer the Navient Licensed Materials and the SMI Licensed IP into such IT environment, and (iii) facilitate the transfer of operational responsibility to SLM BankCo for SLM BankCo's systems and capabilities responsible for the servicing and collection functions of the Pre-Separation Consumer Banking Business (the "IT Transition"). As part of the IT Transition and pursuant to the Transition Services Agreement, (x) Navient shall provide SLM BankCo with access to Navient's IT environment (including the Navient Licensed Materials, SMI Licensed IP, SLM BankCo Information, SLM BankCo Intellectual Property, SLM BankCo Software, the ATLAS Software and SLM BankCo Technology hosted thereon) to enable SLM BankCo to operate the Pre-Separation Consumer Banking Business in the ordinary course, (y) Navient shall host and operate the SLM BankCo Systems for the benefit of SLM BankCo, and (z) Navient and SLM BankCo shall perform the IT migration projects and related activities specified for each such party, respectively, in the Transition Services Agreement as required to transition the Navient Licensed Materials, SMI Licensed IP, SLM BankCo Information, SLM BankCo Intellectual Property, SLM BankCo Technology, SLM BankCo IT environment.

5.9 Restrictions on ATLAS; ATLAS Development Projects

(a) Each Party in its respective capacity as a joint owner of the ATLAS Software covenants and agrees that neither it, nor any member of the SLM BankCo Group or Navient Group, as applicable, shall sell, assign, license, sublicense, distribute, rent, lease or in any way transfer the ATLAS Software to any third party other than pursuant to a transaction contemplated by Section 5.4(g)(i)(D) or Section 5.4(g)(ii)(C) without the written consent of the other Party; provided, however, that (i) either Party may license the ATLAS Software to a Governmental Authority (other than a private or state institution of higher learning (e.g., school, college, university)) pursuant to an agreement with such Governmental Authority if the grant of such license is required by applicable Law or published procurement policy of such Governmental Authority, and (ii) either Party may grant a perpetual, nontransferable, nonassignable, nonsublicenseable license to the ATLAS Software in connection with the sale or disposition of all or substantially all of the assets comprising a line of business then-using the ATLAS Software (or an entity comprising such line of business) to the acquiror to perform the applicable line of business.

(b) Following the Effective Time, the Parties intend to undertake certain application development projects related to the ATLAS Software that are agreed by SLM BankCo and Navient in writing ("ATLAS Development Projects"). Upon successful testing of the ATLAS System following completion of the later of the ATLAS Development Projects or the end of the Preferred Stock Period, SLM BankCo shall grant to Navient an undivided joint ownership interest in all materials created pursuant to the ATLAS Development Projects ("Joint Materials"). All Joint Materials shall be provided on an "AS IS" basis and without warranty of any kind. During the pendency of the ATLAS Development Projects, SLM BankCo shall grant Navient "view only" access to the Joint Materials stored on the applicable SLM BankCo or Navient development systems. Further, during the pendency of the ATLAS Development Projects, SLM BankCo shall use commercially reasonable efforts to deliver a copy of then-current Joint Materials within 90 days from the receipt of Navient's request; provided, however, (i) that Navient may make such request only once per consecutive three-month period, and (ii) SLM BankCo's efforts to deliver the Joint Materials within the 90-day period time do not adversely affect the IT Transition. Upon any such interim delivery of Joint Materials to Navient, SLM BankCo shall grant Navient an undivided joint ownership interest in such Joint Materials as and when such copies are made, on an "AS IS" basis and without warranty of any kind. Each Party shall, without limitation, have and retain the right to make, have made, use, lease, import, offer for sale, or sell, have sold and practice methods used in the creation or provision of products or services that incorporate the Joint Materials to the extent that such actions do not infringe upon the intellectual property rights of the other Party. Each Party shall retain the right to grant non-exclusive licenses to any Intellectual Property in the Joint Materials without any payment or accounting to the other Party. For the avoidance of doubt, each Party shall solely own all derivative works created thereafter by such Party without accounting to the other, including derivative works in the Joint Materials. Each Party in its respective capacity as a joint owner of the Joint Materials acknowledges and agrees that the Joint Materials shall be expressly subject to the restrictions provided in paragraph (a) above, and neither Navient nor SLM BankCo, nor any member of the SLM BankCo Group or Navient Group, as applicable, shall sell, assign, license, sublicense, distribute, rent, lease or in any way transfer the Joint Materials to any third party other than pursuant to a transaction contemplated by Section 5.4(g)(i)(D) or Section 5.4(g)(ii)(C) without the written consent of the other Party; provided, however, that (x) either Party may license the Joint Materials to a Governmental Authority (other than a private or state institution of higher learning (e.g., school, college, university)) pursuant to an agreement with such Governmental Authority if the grant of such license is required by applicable Law or published procurement policy of such Governmental Authority, and (y) either Party may grant a perpetual, nontransferable, nonassignable, nonsublicenseable license to the Joint Materials in connection with the sale or disposition of all or substantially all of the assets comprising a line of business then-using the ATLAS Software (or an entity comprising such line of business) to the acquiror of such entity or assets comprising such line of business solely for use by such acquiror to perform the applicable line of business.

5.10 <u>Delivery of Navient Licensed Systems</u>. SLM BankCo and Navient each acknowledge and agree that the Intellectual Property, Software and Technology included in the Navient Licensed Systems will not be delivered to SLM BankCo on the Distribution Date. Upon SLM BankCo's request and until the end of the Preferred Stock Period, Navient shall use commercially reasonable efforts to provide such Intellectual Property, Software and Technology in accordance with the delivery timeframe requested by SLM BankCo. If Navient is unable to provide such Intellectual Property, Software and Technology in accordance with the requested timeframe without adversely affecting the Services provided under Schedule 1 to the Transition Services Agreement or the Excluded SMI Projects, then the Parties will create a new Project under such Schedule 1, and using the rates provided therein, to address such delivery efforts.

ARTICLE VI EXCHANGE OF INFORMATION; CONFIDENTIALITY

6.1 Agreement for Exchange of Information; Archives. Subject to Section 6.4 and to Section 6.9 and any other applicable confidentiality obligations, for a period of seven years following the Effective Time, each of SLM BankCo and Navient, on behalf of its respective Group, agrees to allow representatives of the other Group reasonable access during normal business hours to, [or to provide], any Information in the possession or under the control of such respective Group as of the Effective Time which the requesting Party reasonably needs (a) to comply with reporting, disclosure, filing or other requirements imposed on the requesting Party (including under applicable securities or Tax Laws) by a Governmental Authority having jurisdiction over the requesting party, (b) for use in any other judicial, regulatory, administrative, tax or other proceeding or in order to satisfy audit, accounting, claims, regulatory, tax or other similar requirements, in each case other than claims or allegations that one Party to this Agreement has against another Party or a member of its respective Group, (c) subject to the foregoing clause (b), to comply with its obligations under this Agreement or any Ancillary Agreement, or, (d) with respect to Information stored at a third party site, subject to the foregoing clause (b), for the requesting Party's ordinary business purposes; provided, however, that, in the event that any Party determines that any such provision of Information could be commercially detrimental, violate any Law or agreement or waive any privilege otherwise available under applicable Law, including attorney-client privilege, the Parties shall take all reasonable measures to permit the compliance with such obligations in a manner that avoids any such harm or consequence, but if such measures are not reasonably possible, then such Party shall not be required to provide such Information.

6.2 Ownership of Information. Any Information owned by one Group that is provided to a requesting Party pursuant to Section 6.1 or Section 6.8 shall be deemed to remain the property of the providing Party. Unless specifically set forth herein, nothing contained in this Agreement or any Ancillary Agreement shall be construed as granting or conferring rights of license or otherwise in any such Information.

- 6.3 <u>Compensation for Providing Information</u>. The Party requesting Information agrees to reimburse the other Party for the reasonable costs, if any, of creating, gathering and copying such Information, to the extent that such costs are incurred for the benefit of the requesting Party. Except as may be otherwise specifically provided elsewhere in this Agreement, any Ancillary Agreement or any other agreement between the Parties, such costs shall be computed in accordance with the providing Party's standard methodology and procedures.
- 6.4 Record Retention. To facilitate the possible exchange of Information pursuant to this Article VI and other provisions of this Agreement or any applicable Ancillary Agreement after the Effective Time, the Parties agree to use their reasonable best efforts to retain all Information in their respective possession or control on the Effective Time in accordance with the record retention policies of Effective Time Sallie Mae as in effect as of the Effective Time. No Party will destroy, or permit any of its Subsidiaries to destroy, any Information which another Party may have the right to access [or obtain] pursuant to this Agreement or any Ancillary Agreement prior to the end of the retention period set forth in such record retention policies without first notifying the other Party of the proposed destruction and giving the other Party a reasonable opportunity to take possession of such information prior to such destruction. Notwithstanding the foregoing, Section 6.1 of the Tax Sharing Agreement will govern the retention of Tax Records (as defined in the Tax Sharing Agreement) and Section 2.4 of the Employee Matters Agreement will govern the retention of Employee Records (as defined in the Employee Matters Agreement).
- 6.5 <u>Limitations of Liability</u>. No Party shall have any liability to any other Party if any Information is destroyed in accordance with the applicable record retention policies referenced in <u>Section 6.4</u> or after reasonable best efforts by such Party to comply with the provisions of <u>Section 6.4</u>.
- 6.6 Other Agreements Providing for Exchange of Information. The rights and obligations granted under this Article VI are subject to any specific limitations, qualifications or additional provisions on the sharing, exchange, retention or confidential treatment of Information set forth in any Ancillary Agreement.

6.7 In-House Lawyers and Other Legal Staff.

(a) In-house lawyers employed by Existing SLM and its Affiliates (the "Existing SLM Lawyers") have for many years provided legal services to and jointly represented Existing SLM and its Subsidiaries and Affiliates, including Subsidiaries conducting the Pre-Separation Education Loan Management Business and the Pre-Separation Consumer Banking Business. After the Effective Time, certain of those Existing SLM Lawyers will remain employees of one or more members of the Navient

Group and represent the Navient Group ("Navient Counsel"). Others will resign from their positions with members of the Navient Group, and become employees of one or more members of the SLM BankCo Group and provide legal services to and represent only the SLM BankCo Group ("SLM BankCo Counsel"). After the Effective Time, (i) SLM BankCo Counsel will represent only the SLM BankCo Group, (ii) Navient Counsel will represent only the Navient Group, and (iii) SLM BankCo Counsel and Navient Counsel will owe a duty of loyalty and other professional obligations only to their respective clients.

- (b) The Parties have previously been jointly represented by the Existing SLM Lawyers in various legal matters of common interest. This joint representation existed at all times before the Effective Time. The joint representation included in its scope all matters prior to the Effective Time in which a Party or another member of its Group was represented by any of the Existing SLM Lawyers. The Parties agree that a joint representation privilege applies to such joint representation.
- (c) After the Effective Time, the SLM BankCo Group and the Navient Group will both continue to jointly own and control all privileges relating to all documents and Information created prior to the Effective Time as a result of the representation of any Party or any member of its respective Group by the Existing SLM Lawyers, and the Parties agree that the Separation shall not waive or affect any applicable privileges, including the attorney-client privilege, the litigation work product doctrine, the common interest privilege and the joint-client/joint representation privilege.
- (d) In advance of the Effective Time, the Parties agree to cause the Existing SLM Lawyers to endeavor to separate all legal files, documents and electronic information created prior to the Effective Time (the "Legal Materials") relating primarily to the Pre-Separation Education Loan Management Business from those relating primarily to the Pre-Separation Consumer Banking Business and deliver them at or after the Effective Time into the possession of the appropriate Party or member of its Group to which they relate. All such Legal Materials not separated as of the Effective Time shall be deemed "Joint Legal Materials." Both Navient Counsel and SLM BankCo Counsel will have the right, after the Effective Time, (i) to access, review and duplicate all Joint Legal Materials in the possession of the other that relate to their respective legal matters and (ii) only with the consent of the other Party, to separate and take sole possession of Joint Legal Materials relating solely to either the Pre-Separation Education Loan Management Business or the Pre-Separation Consumer Banking Business. Both SLM BankCo Counsel and Navient Counsel shall maintain and continue their respective Group's compliance with all litigation holds applicable to any Legal Materials or Joint Legal Materials they possess or come to possess.

- (e) The Parties acknowledge that the Legal Materials and Joint Legal Materials are products of the joint representation by the Existing SLM Lawyers and are privileged from disclosure to others as a result of the attorney-client privilege, the litigation work product doctrine, the common interest/joint defense privilege, the joint-client/joint representation privilege and other applicable privileges and protections. Neither the SLM BankCo Group nor the Navient Group shall have any right, ability or authority to waive any such privilege or protection on behalf of the other, nor shall any waiver of any applicable privilege or protection by one be effective or applicable to the other.
- (f) Unless and until the Parties agree in writing to waive any and all claims of privilege over any portion of the Legal Materials or Joint Legal Materials, the Parties shall assert all applicable privileges to resist production of any Legal Materials or Joint Legal Materials requested by any third party. If any third party requests or demands, by subpoena or otherwise, any Legal Materials or Joint Legal Materials, the Party (which for purposes of this section includes any member of its Group) which has received the request or demand shall immediately notify the other Party in writing. Each Party will then take all reasonable steps necessary to preserve all applicable rights and privileges with respect to such Legal Materials or Joint Legal Materials and shall cooperate fully with the other in any proceedings relating to the disclosure of such Legal Materials or Joint Legal Materials. Each Party has standing to enforce claims of privilege or similar grounds for withholding disclosure in response to any request or demand for the production of Legal Materials or Joint Legal Materials. In the event that any Party is served with or otherwise subject to legal process (including a subpoena) requiring it to testify about, produce or otherwise divulge Legal Materials or Joint Legal Materials, the Party subject to such process will (i) promptly supply the other Party with a copy of such subpoena or process; (ii) assert all applicable privileges, protections and objections; (iii) not waive any such privilege; and (iv) make every other reasonable effort to prevent or limit disclosure of the Legal Materials or Joint Legal Materials.
- (g) Nothing contained in this Agreement shall limit the right of any Party to use or disclose (i) documents or information generated by any member of its Group after the Effective Time (unless such documents or information contain information from the Legal Materials or Joint Legal Materials) or (ii) documents or information that are now, or hereafter become, public information without violation of this Agreement.
- (h) The Parties acknowledge that the SLM BankCo Group and the Navient Group may have or develop interests adverse to each other following the Effective Time. Each Party hereby waives (i) any and all current and future objections to SLM BankCo Counsel, Navient Counsel and any outside counsel that represented Existing SLM or any of its Affiliates prior to the Effective Time from continuing to represent or in the future representing their respective clients in any matter, including matters in which the SLM BankCo Group and the Navient Group are adverse and disputes relating to this Agreement, and (ii) all current and future rights to seek disqualification, whether based on the possession or disclosure of confidential information or otherwise, of any of the

SLM BankCo Counsel, Navient Counsel and such outside counsel from any representation of their clients in any matter, including matters in which the SLM BankCo Group and the Navient Group are adverse and disputes relating to this Agreement. If a dispute arises between or among the Parties (or members of their respective Groups) in the future, no Party may assert privilege against the other as to any Legal Materials or Joint Legal Materials created before the Effective Time, and both Parties shall be free to make use of such materials for the purpose of advancing their interests in such dispute.

- (i) The Parties acknowledge that disclosure of any Legal Materials or Joint Legal Materials in violation of this Agreement, or in violation of the continuing duty to maintain the confidentiality of Legal Materials or Joint Legal Materials, will cause the Parties to suffer irreparable harm for which there is no adequate remedy at law, and that immediate injunctive relief without the necessity of posting a bond is an appropriate and necessary remedy for any such violation or threatened violation, in addition to other remedies and relief that might be available at law or equity.
- (j) SLM BankCo and Navient shall, and shall seek to cause SLM BankCo Counsel and Navient Counsel and their respective outside counsel to, cooperate with each other and take all necessary or reasonably desirable actions to effect the foregoing provisions. The provisions of this Section 6.7 shall survive the Effective Time and remain in effect forever.

6.8 Production of Witnesses; Records; Cooperation.

- (a) Subject to Section 6.7, after the Effective Time, except in the case of an adversarial suit or proceeding by one Party against another Party, each Party shall use its commercially reasonable efforts to make available to the other Party, upon written request, the former, current and future directors, officers, employees, other personnel and agents of the members of its respective Group as witnesses within its control or which it otherwise has the ability to make available, to the extent that any such person (giving consideration to business demands of such directors, officers, employees, other personnel and agents) may reasonably be required in connection with any suit or proceeding in which the requesting Party may from time to time be involved, regardless of whether such suit or proceeding involves a matter with respect to which indemnification may be sought hereunder. The requesting Party shall bear all costs and expenses in connection therewith.
- (b) If an Indemnifying Party chooses to defend or to seek to compromise or settle any Third Party Claim, the other Party shall make available to such Indemnifying Party, upon written request, the former, current and future directors, officers, employees, other personnel and agents of the members of its respective Group as witnesses within its control or which it otherwise has the ability to make available, to

the extent that any such person (giving consideration to business demands of such directors, officers, employees, other personnel and agents) may reasonably be required in connection with such defense, settlement or compromise, or such prosecution, evaluation or pursuit, as the case may be, and shall otherwise cooperate in such defense, settlement or compromise, or such prosecution, evaluation or pursuit, as the case may be.

- (c) Without limiting the foregoing, the Parties shall cooperate and consult to the extent reasonably necessary with respect to any Actions.
- (d) Without limiting any provision of this <u>Section 6.8</u>, each of the Parties agrees to cooperate, and to cause each member of its respective Group to cooperate, with each other in the defense of any infringement or similar claim with respect to any Intellectual Property and shall not claim to acknowledge, or permit any member of its respective Group to claim to acknowledge, the validity or infringing use of any Intellectual Property of a third Person in a manner that would hamper or undermine the defense of such infringement or similar claim.
- (e) The obligation of the Parties to provide witnesses pursuant to this <u>Section 6.8</u> is intended to be interpreted in a manner so as to facilitate cooperation and shall include the obligation to provide as witnesses inventors and other officers without regard to whether the witness or the employer of the witness could assert a possible business conflict (subject to the exception set forth in the first sentence of <u>Section 6.8(a)</u>).
- (f) In connection with any matter contemplated by this <u>Section 6.8</u>, except in the case of an adversarial suit or proceeding by one Party against the other Party, the Parties will enter into a mutually acceptable joint defense agreement so as to maintain to the extent practicable any applicable attorney-client privilege or work product immunity of any member of either Group.

6.9 Confidentiality.

(a) Subject to Section 6.10, until the seven-year anniversary of the Effective Time, each of SLM BankCo and Navient, on behalf of itself and each member of its respective Group, agrees to hold, and to cause its respective Representatives to hold, in strict confidence, with at least the same degree of care that applied to Existing SLM's confidential and proprietary information pursuant to policies in effect immediately prior to the Distribution Date, all Information concerning each such other Group or its customers ("Group Information") that is either in its possession (including Information in its possession prior to the date hereof) or furnished by any such other Group or its respective Representatives at any time pursuant to this Agreement, any Ancillary Agreement or otherwise, and shall not use any such Group Information other than for such purposes as shall be expressly permitted hereunder or thereunder, except,

in each case, to the extent that such Group Information has been (i) in the public domain through no fault of such Party or any member of such Group or any of their respective Representatives, (ii) later lawfully acquired from other sources by such Party (or any member of such Party's Group) which sources are not themselves bound by a confidentiality obligation or (iii) independently generated without reference to any proprietary or confidential Information of the other Party; provided, however, that with respect to Group Information that relates to a Group's customers, the obligations of each Party under this Section 6.9 shall continue for so long as such information is in the possession of such Party or any of its Subsidiaries, Affiliates or Representatives.

- (b) Each Party agrees not to release or disclose, or permit to be released or disclosed, any Group Information to any other Person, except its Representatives who need to know such Information (who shall be advised of their obligations hereunder with respect to such Information), except in compliance with Section 6.10 or with the written consent of the other Party. Without limiting the foregoing, when any Group Information is no longer needed for the purposes contemplated by this Agreement or any Ancillary Agreement, each Party will promptly after request of the other Party either return to the other party all Group Information in a tangible form (including all copies thereof and all notes, extracts or summaries based thereon) or certify to the other Party that it has destroyed such Group Information (and such copies thereof and such notes, extracts or summaries based thereon).
- 6.10 Protective Arrangements. In the event that any Party or any member of its Group either determines on the advice of its counsel that it is required to disclose any Group Information pursuant to applicable Law or receives any demand under lawful process or from any Governmental Authority to disclose or provide Group Information of any other Party (or any member of another Party's Group) that is subject to the confidentiality provisions hereof, such Party shall, promptly upon receipt of a request for disclosure of any Information, notify the other Party (to the extent legally permitted) prior to disclosing or providing such Group Information and shall cooperate at the expense of the requesting Party in seeking any reasonable protective arrangements requested by the requesting Party. Subject to the foregoing, the Person that received such request may thereafter disclose or provide only that portion of Group Information that is legally required by such Law (as so advised by counsel) or by lawful process or such Governmental Authority.

ARTICLE VII GOVERNANCE AND DISPUTE RESOLUTION

- 7.1 Program Managers; Separation Oversight Committee.
- (a) Program Managers.
- (i) Navient shall appoint an individual to act as its program manager (the "Navient Program Manager"), who will have overall responsibility for

implementing, coordinating, tracking and managing all activities (including billing and payment activities) under this Agreement and each Ancillary Agreement and have authority to act on behalf of Navient with respect to such matters, for a period of three years following the Effective Time. Navient hereby appoints and designates the individual set forth on <u>Schedule 7.1(a)</u> to act as the initial Navient Program Manager. Navient shall promptly notify SLM BankCo in writing of its removal and replacement of the Navient Program Manager.

- (ii) SLM BankCo shall appoint an individual to act as its program manager (the "SLM BankCo Program Manager," and together with the Navient Program Manager, the "Program Managers"), who will have overall responsibility for implementing, coordinating, tracking and managing all activities (including billing and payment activities) under this Agreement and each Ancillary Agreement and have authority to act on behalf of SLM BankCo with respect to such matters, for a period of three years following the Effective Time. SLM BankCo hereby appoints and designates the individual set forth on Schedule 7.1(a) to act as the initial SLM BankCo Program Manager. SLM BankCo shall promptly notify Navient in writing of the removal and replacement of the SLM BankCo Program Manager. SLM BankCo may remove and/or replace the SLM BankCo Program Manager in its sole discretion.
- (iii) During the period from the Effective Time until the two-year anniversary of the Effective Time (the "Separation Oversight Period"), each of the Navient Program Manager and the SLM BankCo Program Manager shall deliver to the Separation Oversight Committee, no less frequently than monthly, such reports regarding the activities under this Agreement and the Ancillary Agreements as the Navient Program Manager or the SLM BankCo Program Manager shall deem appropriate or as may otherwise be requested from time to time by the Separation Oversight Committee.
- (iv) Following the expiration of the Separation Oversight Period, (A) the Navient Program Manager will periodically report to and update the Chief Executive Officer of Navient with respect to matters relating to this Agreement and the Ancillary Agreements and (B) the SLM BankCo Program Manager will periodically report to and update the Chief Executive Officer of SLM BankCo with respect to matters relating to this Agreement and the Ancillary Agreements.
 - (b) Separation Oversight Committee.
- (i) The Parties have established a six member separation oversight committee (the "Separation Oversight Committee"), which shall consist of an equal number of members designated by each of SLM BankCo and Navient. The initial members of the Separation Oversight Committee are set forth on Schedule 7.1(b). SLM BankCo may remove and/or replace any SLM BankCo designee to the Separation Oversight Committee, in its sole discretion, and shall promptly notify Navient of any such removal and/or replacement. Navient may remove

and/or replace any Navient designee to the Separation Oversight Committee, in its sole discretion, and shall promptly notify SLM BankCo of any such removal and/or replacement. The Separation Oversight Committee shall be responsible, during the Separation Oversight Period, for monitoring and managing, at a strategic level, progress on all matters related to transactions and activities that are to occur after the Effective Time as contemplated by this Agreement and the Ancillary Agreements.

- (ii) The Separation Oversight Committee shall meet (in person or telephonically) no less frequently than monthly. The Separation Oversight Committee shall have the authority to (a) establish one or more subcommittees from time to time as it deems appropriate or as may be contemplated pursuant to any Ancillary Agreement, with each such subcommittee comprised of one or more members of the Separation Oversight Committee or one or more employees of either Party or any member of its respective Group, and each such subcommittee having such scope of responsibility as may be determined by the Separation Oversight Committee from time to time; (b) delegate to any such committee any of the responsibilities of the Separation Oversight Committee; (c) combine, modify the scope of responsibility of, and disband any such subcommittees and (d) modify or reverse any such delegations. The Separation Oversight Committee may establish general procedures for managing the responsibilities assigned to it under this Section 7.1 and may modify such procedures from time to time.
- (iii) All decisions by the Separation Oversight Committee or any subcommittee thereof shall be effective only if agreed to by a majority of the members designated by each of SLM BankCo and Navient.
- (iv) The Navient members of the of the Separation Oversight Committee shall (i) during the Separation Oversight Period, periodically report to and update the Chief Executive Officer of Navient on the progress of the separation and with respect to other matters relating to this Agreement and the Ancillary Agreements and (ii) during the Preferred Stock Period, periodically (but no less frequently than quarterly) report to and update the SMI Board on the progress of the IT Transition. During the Separation Oversight Period, the SLM BankCo members of the Separation Oversight Committee will periodically report to and update the Chief Executive Officer of SLM BankCo on the progress of the separation and with respect to other matters relating to this Agreement and the Ancillary Agreements.
 - 7.2 Escalation Process; Good Faith Negotiation.
 - (a) Escalation Process During the Separation Oversight Period.

- (i) The Navient Program Manager and the SLM BankCo Program Manager shall attempt in good faith to negotiate a resolution to any issues, disputes, controversies or claims arising out of or relating to this Agreement or any Ancillary Agreement (including matters relating to each Party's rights or the performance of obligations under this Agreement or any Ancillary Agreement or regarding whether any Assets are Navient Assets or SLM BankCo Assets, any Liabilities are Navient Liabilities or SLM BankCo Liabilities) (each, a "Dispute") within five business days of written notice of any such Dispute (a "Dispute Notice") being delivered by one Program Manager to the other Program Manager.
- (ii) During the Separation Oversight Period, the Navient Program Manager and the SLM BankCo Program Manager shall escalate to the Separation Oversight Committee any Dispute that cannot be resolved by the Program Managers within five business days following delivery of a Dispute Notice. The Separation Oversight Committee shall attempt in good faith to negotiate a resolution of each Dispute escalated by the Program Managers.
- (iii) If any Dispute is not resolved by the Separation Oversight Committee within 10 business days following escalation of such Dispute to the Separation Oversight Committee, (A) the Navient members of the Separation Oversight Committee shall consult with (x) as to any Dispute that arises out of or related to the IT

Transition, the SMI Board and (y) as to all other Disputes, the Navient Board (or a committee thereof appointed for such purpose), which in each case may direct the Navient members how to proceed with respect to such Dispute, and (B) the SLM BankCo members of the Separation Oversight Committee shall consult with the SLM BankCo Board (or a committee thereof appointed for such purpose), which may direct the SLM BankCo members how to proceed with respect to such Dispute.

- (iv) If any Dispute remains unresolved for any reason for a period of 15 business days following escalation of such Dispute to the Separation Oversight Committee, or if a Party reasonably concludes that the other Party is not willing to negotiate as contemplated by this <u>Section 7.2</u>, the Dispute may be submitted to the dispute resolution procedures in accordance with <u>Section 7.3</u> or <u>Section 10.13</u>.
 - (b) Escalation Process Following the Separation Oversight Period.
- (i) The Navient Program Manager and the SLM BankCo Program Manager shall attempt in good faith to negotiate a resolution to each Dispute within five business days of a Dispute Notice being delivered by one Program Manager to the other Program Manager.
- (ii) If any Dispute is not resolved by the Program Managers within five business days following delivery of a Dispute Notice, (A) the Navient Program Manager shall consult with the Navient Board (or a committee thereof appointed for such purpose),

which may direct the Navient Program Manager how to proceed with respect to such Dispute and (B) the SLM BankCo Program Manager shall consult with the SLM BankCo Board (or a committee thereof appointed for such purpose), which may direct the SLM BankCo Program Manager how to proceed with respect to such Dispute.

- (iii) If any Dispute remains unresolved for any reason for a period of 15 business days following delivery of a Dispute Notice or if a Party reasonably concludes that the other Party is not willing to negotiate as contemplated by this <u>Section 7.2</u>, the Dispute may be submitted to the dispute resolution procedures in accordance with <u>Section 7.3</u> or <u>Section 10.13</u>.
- (c) Status of Discussions. All negotiations contemplated by this <u>Section 7.2</u> shall be confidential and shall be treated by the Parties as compromise and settlement negotiations for purposes of applicable rules of evidence.

7.3 Dispute Resolution.

(a) Dispute Resolution. Except as otherwise specifically provided in Section 7.2 or Section 7.3(c) or in any Ancillary Agreement, or in the case where a Party elects to resolve a Specific Dispute as contemplated by Section 10.13, the procedures for binding mediation set forth in this Section 7.3 and Schedule 7.3 hall apply to all Disputes (whether sounding in contract, tort or otherwise) that may arise out of or relate to, or arise under or in connection with this Agreement or any Ancillary Agreement or the transactions contemplated hereby or thereby, between or among any members of the Navient Group and the SLM BankCo Group. Each Party agrees on behalf of itself and the members of its Group that the procedures set forth in Section 7.3(b) and Schedule 7.3 shall be the sole and exclusive remedy in connection with any such Dispute and irrevocably waives any right to commence any suit or proceeding in or before any Governmental Authority or mediation or arbitration tribunal, except as expressly provided in Section 7.3(c), Section 7.3(d) or Section 10.13 and except to the extent provided under the Arbitration Act in the case of judicial review of arbitration results or awards. EXCEPT AS EXPRESSLY PROVIDED IN SECTION 7.3(d) OR SECTION 10.13, EACH PARTY ON BEHALF OF ITSELF AND EACH MEMBER OF ITS GROUP IRREVOCABLY WAIVES ANY RIGHT TO ANY TRIAL IN A COURT THAT WOULD OTHERWISE HAVE JURISDICTION OVER ANY CLAIM, CONTROVERSY OR DISPUTE SET FORTH IN THE FIRST SENTENCE OF THIS SECTION 7.3(a). All dispute resolution proceedings pursuant to this Section 7.3 shall be confidential and shall not be disclosed by any Party (other than disclosure to its advisors or to the extent disclosure is permitted pursuant to Section 6.10) and shall be treated as compromise and settlement negotiations for purposes of applicable rules of evidence.

¹ Schedule 7.3 to provide detailed provisions relating to qualifications, selection, procedures, timing, hearings, discovery and other matters.

(b) <u>Binding Mediation</u>. Except as provided in <u>Section 7.3(c)</u> with respect to an Arbitrable Matter or <u>Section 10.13</u> with respect to a Specified Dispute, any Dispute not resolved pursuant to <u>Section 7.2</u> shall, at the written request of any Party (a "<u>Mediation Request</u>"), which Mediation Request shall be given to the other Party in the manner set forth in <u>Section 10.5</u>, be submitted to binding mediation. The mediation shall be held in the State of Delaware or such other place as the Parties may mutually agree. The procedures for binding mediation shall be as set forth in <u>Schedule 7.3</u>. Any mediator selected pursuant to this <u>Section 7.3(b)</u> shall meet the qualifications set forth in <u>Schedule 7.3</u> and be neutral and disinterested with respect to each of the Parties and the matter.

(c) <u>Binding Arbitration</u>. Any Dispute relating to the IT Transition, the ATLAS Development Projects, the Preferred Stock Period or exclusivity or marketing rights under the Joint Marketing Agreement (each, an "<u>Arbitrable Matter</u>") shall be submitted to binding arbitration in accordance with this <u>Section 7.3(c)</u>. If an Arbitrable Matter is not resolved in accordance with <u>Section 7.2</u>, either Navient or SLM BankCo may, unless the Applicable Deadline (as hereinafter defined) has occurred, make a written demand (the "<u>Arbitration Demand Notice</u>") that the Dispute be resolved by binding arbitration, which Arbitration Demand Notice shall be given to the other Party in the manner set forth in <u>Section 10.5</u>. If either Party shall deliver an Arbitration Demand Notice, the other Party may itself deliver an Arbitration Demand Notice to such first Party with respect to any related Dispute with respect to which the Applicable Deadline has not passed without the requirement of first delivering a Dispute Notice as contemplated by <u>Section 7.2</u>. If either Party delivers an Arbitration Demand Notice with respect to any Dispute that is the subject of any then pending arbitration proceeding or of a previously delivered Arbitration Demand Notice, all such Disputes shall be resolved in the arbitration proceeding for which an Arbitration Demand Notice was first delivered unless the arbitrator in his or her sole discretion determines that it is impracticable or otherwise inadvisable to do so.

(i) Except as may be expressly provided in any Ancillary Agreement, any Arbitration Demand Notice may be given until two years after the later of (i) the occurrence of the act or event giving rise to the Dispute and (ii) the date on which such act or event was, or should have been, in the exercise of reasonable due diligence, discovered by the Party asserting the Dispute (as applicable and as it may in a particular case be specifically extended by the Parties in writing, the "Applicable Deadline"). Any discussions, negotiations or mediation between the Parties pursuant to this Agreement or otherwise will not toll the Applicable Deadline unless expressly agreed in writing by the Parties. Each Party agrees, on behalf of itself and the members of its respective Group, that, if an Arbitration Demand Notice with respect to a Dispute is not given prior to the expiration of the Applicable Deadline, such Dispute will be barred. Subject to Section 7.3(d), upon delivery of an Arbitration Demand Notice pursuant to Section 7.3(c) prior to the Applicable Deadline, the Dispute shall be decided by one or more arbitrators in accordance with the rules set forth in this Section 7.3 and Schedule 7.3.

- (ii) Except as otherwise set forth in this Section 7.3 and Schedule 7.3, any arbitration hereunder will be conducted in accordance with the American Arbitration Association (the "AAA") Commercial Arbitration Rules and Procedures then prevailing; provided, however, that to the extent that the provisions of this Agreement and the prevailing rules of the AAA conflict, the provisions of this Agreement (including this Section 7.3 and Schedule 7.3) shall govern. Unless the Parties otherwise agree, any such arbitration shall be conducted by and before a single arbitrator. Any arbitrator selected pursuant to this Section 7.3(c) shall be neutral and disinterested with respect to each of the Parties and the matter.
- (iii) The arbitrator shall have full power and authority to determine issues of arbitrability but shall otherwise be limited to interpreting or construing the applicable provisions of this Agreement or any Ancillary Agreement, and will have no authority or power to limit, expand, alter, amend, modify, revoke or suspend any condition or provision of this Agreement or any Ancillary Agreement; it being understood, however, that the arbitrator will have full authority to implement the provisions of this Agreement or any Ancillary Agreement and to fashion appropriate remedies for breaches of this Agreement (including interim or permanent injunctive relief); provided, however, that the arbitrator shall not have (i) any authority in excess of the authority a court having jurisdiction over the Parties and the controversy or dispute would have absent these arbitration provisions or (ii) any right or power to award exemplary, punitive, special, indirect, consequential, remote or speculative damages (including in respect of lost profits or revenues) or treble damages, except to the extent such damages are expressly permitted by the terms of this Agreement or any Ancillary Agreement; provided, that this clause (ii) shall not limit the award of any such damages to the extent they are included in any Liabilities to third parties as to which the provisions of this Agreement and the Ancillary Agreements and follow applicable Law (it being understood and agreed that this sentence shall not give rise to a right of judicial review of the award of the arbitrator).
- (iv) If a Party fails or refuses to appear at and participate in an arbitration hearing after due notice, the arbitrator may hear and determine the controversy upon evidence produced by the appearing Party. Any decision rendered under such circumstances shall be as valid and enforceable as if the Parties had appeared and participated fully at all stages.
- (v) Subject to Sections 4.2 and 4.3, the fees of the arbitrator and all other arbitration costs shall be borne equally by each Party involved in the matter, except that each Party shall be responsible for its own attorney's fees and other costs and expenses, including the costs of witnesses selected by such Party.

- (vi) Any arbitration award shall be an award with a holding in favor of or against a Party and shall include findings as to facts, issues or conclusions of law (including with respect to any matters relating to the validity or infringement of patents or patent applications) and shall include a statement of the reasoning on which the award rests. The award must also be in adequate form so that a judgment of a court may be entered thereupon. Judgment upon any arbitration award hereunder may be entered in any court having jurisdiction thereof.
- (vii) Any arbitration proceedings hereunder shall take place in the State of Delaware, unless another location is otherwise agreed to in writing by the Parties.
- (d) Regardless of whether a Dispute Notice, Mediation Request or Arbitration Demand Notice has been delivered, prior to the time at which the mediator or arbitrator is appointed pursuant to this Section 7.3, either Party may seek one or more temporary restraining orders in a court of competent jurisdiction if necessary in order to preserve and protect the status quo. Neither the request for, nor the grant or denial of, any such temporary restraining order shall be deemed a waiver of the obligation to arbitrate as set forth herein, and the mediator or arbitrator may order the Parties to petition the court to dissolve, continue or modify any such order. Any such temporary restraining order shall remain in effect until the first to occur of the expiration of the order in accordance with its terms or the dissolution thereof.
- (e) Except as required by Law, the Parties shall hold, and shall cause their respective Subsidiaries, officers, directors, employees, agents and other representatives to hold, the existence, content and result of mediation or arbitration in confidence in accordance with the provisions of this Article VII and except as may be required in order to enforce any agreement or award (other than disclosure to its advisors or to the extent disclosure is permitted pursuant to Section 6.10). Each of the Parties shall request that the mediator or arbitrator, as applicable, comply with such confidentiality requirement.
- (f) The interpretation of the provisions of this Article VII, only insofar as they relate to the agreement to arbitrate and any procedures pursuant thereto, shall be governed by the Arbitration Act and other applicable U.S. federal law. In all other respects, the interpretation of this Agreement shall be governed as set forth in Section 11.2.
- 7.4 Continuation of Services and Commitments. Unless otherwise agreed in writing, the Parties shall, and shall cause their Subsidiaries to, continue to honor all commitments under this Agreement and each Ancillary Agreement to the extent required by such Agreements during the course of dispute resolution pursuant to the provisions of this Article VII.

ARTICLE VIII FURTHER ASSURANCES AND ADDITIONAL COVENANTS

8.1 Further Assurances.

(a) In addition to the actions specifically provided for elsewhere in this Agreement, each of the Parties hereto shall use its reasonable best efforts, prior to,

on and after the Effective Time, to take, or cause to be taken, all actions, and to do, or cause to be done, all things, reasonably necessary, proper or advisable under applicable Laws, regulations and agreements to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements.

- (b) Without limiting the foregoing, prior to, on and after the Effective Time, each Party hereto shall cooperate with the other Party, and without any further consideration, but at the expense of the requesting Party, execute and deliver, or use its reasonable best efforts to cause to be executed and delivered, all instruments, including instruments of conveyance, assignment and transfer, and to make all filings with, and to obtain all Approvals or Notifications of, any Governmental Authority or any other Person under any permit, license, agreement, indenture or other instrument (including any consents or Governmental Approvals), and to take all such other actions as such Party may reasonably be requested to take by the other Party from time to time, consistent with the terms of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements and the transfers of the Navient Assets and the SLM BankCo Assets and the assignment and assumption of the Navient Liabilities and the SLM BankCo Liabilities and the other transactions contemplated hereby and thereby. Without limiting the foregoing, each Party will, at the reasonable request, cost and expense of the other Party, take such other actions as may be reasonably necessary to vest in such other Party good and marketable title to the Assets allocated to such Party under this Agreement or any of the Ancillary Agreements, free and clear of any Security Interest, if and to the extent it is practicable to do so.
- (c) On or prior to the Effective Time, Existing SLM, SLM BankCo and Navient in their respective capacities as direct and indirect shareholders of their respective Subsidiaries, shall each ratify any actions which are reasonably necessary or desirable to be taken by Existing SLM, SLM BankCo, Navient or any of their respective Subsidiaries, as the case may be, to effectuate the transactions contemplated by this Agreement and the Ancillary Agreements.
- (d) Prior to the first anniversary of the Effective Time, if one or more of the Parties identifies any commercial or other service that is needed to assure a smooth and orderly transition of the businesses in connection with the consummation of the transactions contemplated hereby, and that is not otherwise governed by the provisions of this Agreement or any Ancillary Agreement, subject to applicable Law (including, in the case of the SLM BankCo Group, banking regulations and the advice of the SLM BankCo Group's counsel in respect thereof), the Parties will cooperate in determining whether there is a mutually acceptable arm's-length basis on which the other Party will provide such service.

ARTICLE IX TERMINATION

- 9.1 <u>Termination</u>. This Agreement may be terminated by the Sallie Mae Board at any time, in its sole and absolute discretion, prior to the Effective Time. After the Effective Time, this Agreement may not be terminated except by an agreement in writing signed by each of the Parties.
- 9.2 Effect of Termination. In the event of any termination of this Agreement prior to the Effective Time, no Party (or any of its directors or officers) shall have any liability or further obligation to any other Party.

ARTICLE X MISCELLANEOUS

10.1 Counterparts; Entire Agreement; Corporate Power.

- (a) This Agreement and each Ancillary Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the Parties and delivered to the other Parties.
- (b) This Agreement, the Ancillary Agreements and the Exhibits, Schedules and appendices hereto and thereto contain the entire agreement among the Parties with respect to the subject matter hereof and thereof and supersede all previous agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter, and there are no agreements or understandings among the Parties other than those set forth or referred to herein or therein.
- (c) SLM BankCo represents on behalf of itself and each other member of the SLM BankCo Group, and Navient represents on behalf of itself and each other member of the Navient Group, as follows:
 - (i) each such Person has the requisite corporate or other power and authority and has taken all corporate or other action necessary in order to execute, deliver and perform this Agreement and each Ancillary Agreement to which it is a party and to consummate the transactions contemplated hereby and thereby; and
 - (ii) this Agreement and each Ancillary Agreement to which it is a party has been duly executed and delivered by it and constitutes a valid and binding agreement of it enforceable in accordance with the terms thereof.

- (d) Each Party acknowledges that it and each other Party may execute certain of the Ancillary Agreements by facsimile, stamp or mechanical signature. Each Party expressly adopts and confirms each such facsimile, stamp or mechanical signature made in its respective name as if it were a manual signature, agrees that it will not assert that any such signature is not adequate to bind such Party to the same extent as if it were signed manually and agrees that at the reasonable request of any other Party at any time it will as promptly as reasonably practicable cause each such Ancillary Agreement to be manually executed (any such execution to be as of the date of the initial date thereof).
- (e) Notwithstanding any provision of this Agreement or any Ancillary Agreement, none of Existing SLM, SLM BankCo nor Navient shall be required to take or omit to take any act that would violate its fiduciary duties to any minority shareholders of any non-wholly owned Subsidiary of Existing SLM, SLM BankCo or Navient, as the case may be (it being understood that directors' qualifying shares or similar interests will be disregarded for purposes of determining whether a Subsidiary is wholly owned).
- 10.2 <u>Governing Law.</u> This Agreement and, unless expressly provided therein, each Ancillary Agreement (and any claims or disputes arising out of or related hereto or thereto or to the transactions contemplated hereby and thereby or to the inducement of any party to enter herein and therein, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall be governed by and construed and interpreted in accordance with the Laws of the State of Delaware irrespective of the choice of laws principles of the State of Delaware including all matters of validity, construction, effect, enforceability, performance and remedies.
- 10.3 <u>Assignability</u>. Except as set forth in any Ancillary Agreement, this Agreement and each Ancillary Agreement shall be binding upon and inure to the benefit of the parties hereto and thereto, respectively, and their respective successors and permitted assigns; <u>provided, however</u>, that no party hereto or thereto may assign its respective rights or delegate its respective obligations under this Agreement or any Ancillary Agreement, directly or indirectly (including by sale of assets or stock or by means of a merger or consolidation) without the express prior written consent of the other parties hereto or thereto, respectively. Notwithstanding the foregoing, no such consent shall be required for the assignment of a party's rights and obligations under this Agreement or the Ancillary Agreements (except as may be otherwise provided in any such Ancillary Agreement) in whole in connection with a Change of Control of a Party so long as the resulting, surviving or transferee Person assumes all the obligations of the relevant party thereto by operation of Law or pursuant to an agreement in form and substance reasonably satisfactory to the other Party. Nothing in this Section 10.3 is intended to, or shall be construed to, prohibit either Party or any of its Subsidiaries from being party to or undertaking a Change of Control.

10.4 Third Party Beneficiaries. Except for the indemnification rights under this Agreement of any SLM BankCo Indemnitee or Navient Indemnitee in their respective capacities as such, (a) the provisions of this Agreement and each Ancillary Agreement are solely for the benefit of the parties thereto and are not intended to confer upon any Person except the parties thereto any rights or remedies hereunder or thereunder, and (b) there are no Third Party beneficiaries of this Agreement or any Ancillary Agreement and neither this Agreement nor any Ancillary Agreement shall provide any third Person with any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement or any Ancillary Agreement.

10.5 Notices. All notices, requests, claims, demands or other communications under this Agreement and, to the extent applicable and unless otherwise provided therein, under each of the Ancillary Agreements shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed or by registered or certified mail (postage prepaid, return receipt requested) to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Section 10.5):

If to SLM BankCo, to:

SLM Corporation 300 Continental Drive Newark, Delaware 19713 Attn: General Counsel

If to Navient, to:

given.

Navient Corporation 300 Continental Drive Newark, Delaware 19713 Attn: General Counsel

Any Party may, by notice to the other Party in accordance with this <u>Section 10.5</u>, change the address to which such notices are to be

10.6 <u>Severability</u>. If any provision of this Agreement or any Ancillary Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof or thereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby. Upon such determination, the Parties shall negotiate in good faith in an effort to agree upon such a suitable and equitable provision to effect the original intent of the Parties.

- 10.7 <u>Force Majeure</u>. No Party shall be deemed in default of this Agreement or any Ancillary Agreement to the extent that any delay or failure in the performance of its obligations under this Agreement or any Ancillary Agreement results from any Force Majeure. In the event of any such excused delay, the time for performance shall be extended for a period equal to the time lost by reason of the delay.
- 10.8 <u>Publicity</u>. Prior to the Effective Time, each of Navient and SLM BankCo shall consult with each other prior to issuing any press releases or otherwise making public statements with respect to the Separation, the Distribution or any of the other transactions contemplated hereby or under any Ancillary Agreement and prior to making any filings with any Governmental Authority with respect thereto.
- 10.9 Expenses. Except as expressly set forth in this Agreement (including Sections 5.1, 6.3, 6.8(a), 6.10, 7.3 and 8.1(b) and Article IV) or in any Ancillary Agreement, all fees, costs and expenses incurred prior to the Effective Time in connection with the preparation, execution and delivery of this Agreement and any Ancillary Agreement, and with the consummation of the transactions contemplated hereby and thereby, will be borne by Navient.
- 10.10 <u>Headings</u>. The article, section and paragraph headings contained in this Agreement and in the Ancillary Agreements are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement or any Ancillary Agreement.
- 10.11 <u>Survival of Covenants</u>. Except as expressly set forth in this Agreement or any Ancillary Agreement, the covenants, representations and warranties contained in this Agreement and each Ancillary Agreement, and liability for the breach of any obligations contained herein, shall survive the Separation and the Distribution and shall remain in full force and effect.
- 10.12 Waivers of Default. Waiver by any Party of any default by the other Party of any provision of this Agreement or any Ancillary Agreement shall not be deemed a waiver by the waiving Party of any subsequent or other default, nor shall it prejudice the rights of the other Party. No failure or delay by any Party in exercising any right, power or privilege under this Agreement or any Ancillary Agreement shall operate as a waiver thereof, nor shall a single or partial exercise thereof prejudice any other or further exercise thereof or the exercise of any other right, power or privilege.
- 10.13 Specific Performance; Resolution of Certain Disputes. Subject to the provisions of Article VII, in the event of any actual or threatened default in, or breach of, any of the terms, conditions and provisions of this Agreement or any Ancillary Agreement, the Party or Parties who are, or are to be, thereby aggrieved shall have (i) the right to specific performance and injunctive or other equitable relief, including with respect to any Specified Dispute and the right to implead a Party into an action, suit or proceeding relating to a Third Party Claim, in respect of its or their rights under this Agreement or such Ancillary Agreement and (ii) the right to bring an action, suit or proceeding to resolve any Specified Dispute, in addition to any and all

other rights and remedies at law or in equity, and all such rights and remedies shall be cumulative. The Parties agree that the remedies at law for any breach or threatened breach, including monetary damages, are inadequate compensation for any loss and that any defense in any action for specific performance that a remedy at law would be adequate is waived. Any requirements for the securing or posting of any bond with such remedy are waived by each of the Parties. For purposes of this Agreement, a "Specified Dispute" shall mean (i) any dispute relating to indemnification for Losses resulting from Navient Liabilities described in Section 2.3(a)(ii)(A), Section 2.3(a)(ii)(B), Section 2.3(a)(ii)(B), Section 2.3(a)(ii)(B), Section 2.3(a)(ii) or Section 2.3(a)(ii) or Section 2.3(a)(ii) (any dispute relating to indemnification for Losses resulting from SLM BankCo Liabilities described in Section 2.3(b)(i) or Section 2.3(b)(ii) any indemnification claims made under Section 4.2(c), Section 4.2(e), or Section 4.3(b), (iv) any Dispute with respect to whether a Third Party Claim is a Navient Liability or an SLM BankCo Liability, (v) any dispute relating to a breach by the other Party of the provisions of Section 5.4 or Section 6.9 and (vi) any Dispute arising from a breach of the Data Sharing Agreement, the Key Systems Agreement or Sections 13 or 14 of the Joint Marketing Agreement or the confidentiality or data privacy provisions of any other Ancillary Agreement.

10.14 Amendments. No provisions of this Agreement or any Ancillary Agreement shall be deemed waived, amended, supplemented or modified by any Party, unless such waiver, amendment, supplement or modification is in writing and signed by the authorized representative of the Party against whom it is sought to enforce such waiver, amendment, supplement or modification.

10.15 Interpretation. In this Agreement and any Ancillary Agreement, (a) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other genders as the context requires; (b) the terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement (or the applicable Ancillary Agreement) as a whole (including all of the Schedules, Exhibits and Appendices hereto and thereto) and not to any particular provision of this Agreement (or such Ancillary Agreement); (c) Article, Section, Schedule, Exhibit and Appendix references are to the Articles, Sections, Schedules, Exhibits and Appendices to this Agreement (or the applicable Ancillary Agreement) unless otherwise specified; (d) the word "including" and words of similar import when used in this Agreement (or the applicable Ancillary Agreement) shall mean "including, without limitation,"; (e) the word "or" shall not be exclusive; (f) unless expressly stated to the contrary in this Agreement or in any Ancillary Agreement, all references to "the date hereof," "the date of this Agreement," "hereby" and "hereupon" and words of similar import shall all be references to April 28, 2014, regardless of any amendment or restatement hereof and (g) references to any agreement, instrument or other document shall mean such agreement, instrument or other document as amended, supplemented or modified from time to time to the extent permitted by the provisions thereof and not prohibited by this Agreement.

10.16 <u>Limitations of Liability</u>. Notwithstanding anything in this Agreement to the contrary, and except with respect to any breach of any covenant or agreement contained in <u>Sections 5.4, 5.5, 5.7</u> or 6.9 or as expressly set forth in any Ancillary Agreement, neither Navient or its Affiliates, on the one hand, nor SLM BankCo or its Affiliates, on the other hand, shall be liable under this Agreement to the other for any special, indirect, punitive, exemplary, remote, speculative or similar damages in excess of compensatory damages of the other arising in connection with the transactions contemplated hereby (other than any such liability with respect to a Third Party Claim).

10.17 <u>Performance</u>. SLM BankCo will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the SLM BankCo Group. Navient will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement or in any Ancillary Agreement to be performed by any member of the Navient Group. Each Party (including its permitted successors and assigns) further agrees that it will (a) give timely notice of the terms, conditions and continuing obligations contained in this Agreement and any applicable Ancillary Agreement to all of the other members of its Group and (b) cause all of the other members of its Group not to take any action or fail to take any such action inconsistent with such Party's obligations under this Agreement, any Ancillary Agreement or the transactions contemplated hereby or thereby.

10.18 <u>Schedules</u>. The Parties may update the Schedules to this Agreement after the date hereof to correct inaccuracies and for information that is not available prior to the date hereof; *provided*, that any update to a Schedule that increases the Liability of a Party shall be subject to the consent of such Party, such consent not to be unreasonably withheld.

IN WITNESS WHEREOF, the parties have caused this Separation and Distribution Agreement to be executed by their duly authorized representatives.

SLM CORPORATION [Existing SLM]

By: /s/ John F. Remondi

Name: John F. Remondi

Title: President and Chief Executive Officer

NEW BLC CORPORATION

[SLM BankCo]

By: /s/ Raymond Quinlan

Name: Raymond Quinlan Title: Chief Executive Officer

NAVIENT CORPORATION

[Navient]

By: /s/ John F. Remondi

Name: John F. Remondi Title: Chief Executive Officer

CERTIFICATE OF AMENDMENT OF AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

OF

NEW BLC CORPORATION

NEW BLC CORPORATION (the "Corporation"), a corporation organized and existing under and by virtue of the General Corporation Law of the State of Delaware (the "DGCL"), DOES HEREBY CERTIFY:

FIRST: The Amended and Restated Certificate of Incorporation of the Corporation, as amended to date, is hereby amended to reflect a change in the name of the Corporation by deleting in its entirety ARTICLE FIRST of the Amended and Restated Certificate of Incorporation and replacing said ARTICLE FIRST with the following:

"FIRST: The name of the corporation is SLM Corporation (hereinafter the "Corporation")."

SECOND: The sole director of the Corporation has adopted a resolution, by written consent, proposing and declaring advisable the amendment described herein in accordance with the provisions of Sections 141(f) and 242 of the DGCL.

THIRD: In lieu of a meeting and vote of the sole stockholder, the sole stockholder has given written consent to the amendment described herein in accordance with the provisions of Section 228 and Section 242 of the DGCL.

FOURTH: The foregoing amendment was duly adopted in accordance with the provisions of Sections 242 and 228 (by the written consent of the stockholders of the Corporation) of the DGCL.

IN WITNESS WHEREOF, the undersigned has executed this Certificate on this 28th day of April, 2014.

NEW BLC CORPORATION

s/s Laurent C. Lutz, Jr.

By: Laurent C. Lutz, Jr.

Title: Executive Vice President, General Counsel and

Corporate Secretary

BAKER BOTTS IIP

THE WARNER 1299 PENNSYLVANIA AVE., NW WASHINGTON, D.C. 20004-2400

TEL +1 202.639.7700 FAX +1 202.639.7890 BakerBotts.com ABU DHABI HOUSTON
AUSTIN LONDON
BEIJING MOSCOW
BRUSSELS NEW YORK
DALLAS PALO ALTO
DUBAI RIO DE JANEIRO
HONG KONG RIYADH

WASHINGTON

April 30, 2014

SLM Corporation 300 Continental Drive Newark, Delaware 19713

Navient Corporation 300 Continental Drive, Newark, Delaware 19713 Don J Lonczak TEL: 2026397744 FAX: 2025851001

don.lonczak@bakerbotts.com

Ladies and Gentlemen:

We have acted as U.S. federal income tax counsel to SLM Corporation (formerly New BLC Corporation), a Delaware corporation ("Parent"), and the former SLM Corporation ("Old SLM"), also a Delaware corporation, in connection with (i) the merger (the "Reorganization") of Old SLM with and into a wholly-owned subsidiary of Parent, (ii) the internal spin-off by Sallie Mae, Inc., a Delaware corporation and subsidiary of Parent ("Internal Distributing"), of its private education loan servicing business (the "Internal Spin-off"), (iii) the spin-off of Parent's education loan management business to its public shareholders (the "External Spin-off") and (iv) certain related internal restructuring transactions (collectively, the Reorganization, the Internal Spin-off, the External Spin-off and the related internal restructuring transactions are referred to herein as the "Transactions").

Parent and Old SLM have requested our opinion (the "<u>Tax Opinion</u>") regarding certain U.S. federal income tax consequences of the Transactions. Pursuant to such request, Parent, Old SLM and we have agreed that this Tax Opinion addresses, considers and provides conclusions only with respect to the specific U.S. federal income tax matters addressed herein. Additional issues that are not addressed in this Tax Opinion could affect the U.S. federal income tax treatment of the Transactions. At the request of Parent and Old SLM, we also have addressed this Tax Opinion to Navient Corporation ("<u>External Controlled</u>"). We note, however, that we do not represent, and we undertake no additional attorney-client responsibilities with respect to, External Controlled.

In accordance with Treasury Department Circular 230 and pursuant to the request of Parent and Old SLM, this Tax Opinion addresses, considers and provides conclusions only with respect to the U.S. federal income tax matters discussed herein. Additional issues that are not discussed in our Tax Opinion could affect the U.S. federal income tax treatment of the Transactions. Our Tax Opinion was not written, and cannot be used by any taxpayer, for the purpose of avoiding penalties that may be imposed on it with respect to any U.S. federal income tax issue not expressly discussed in our Tax Opinion.

Except where expressly stated otherwise, all "section" references herein are to the Internal Revenue Code of 1986, as amended (the "Code").

BAKER BOTTS IIP

April 30, 2014

Factual Assumptions

In rendering our Tax Opinion, we have examined, and relied (without independent verification) upon the accuracy and completeness of all the facts, information, covenants, representations, and warranties contained in, originals or copies, certified or otherwise identified to our satisfaction, of:

- (i) the separation and distribution agreement, dated as of the date hereof, among Parent, Old SLM and External Controlled, and all other agreements entered into by such parties or their affiliates relating to the Transactions;
- (ii) the information statement and other documents included as part of the registration statement on Form 10 (File No. 001-36228), as amended (the "Registration Statement"), filed with the Securities and Exchange Commission, or incorporated by reference therein; and
- (iii) such other documents and records as we have deemed necessary or appropriate as a basis for the conclusions set forth below.

In addition, as to certain facts that may be material to our Tax Opinion, we have relied upon certain statements and representations made on behalf of Parent, Old SLM, External Controlled, Internal Distributing and SMB Servicing Company, Inc. ("Internal Controlled") by officers and other representatives of Parent and Old SLM and upon statements and representations set forth in a representation letter, delivered by Parent to us, dated the date hereof (the "Representation Letter"). Our Tax Opinion assumes that such statements and representations are true and correct as of the effective time of the External Spinoff, without regard to any qualification for knowledge or belief.

We have concluded that none of the factual assumptions, representations, statements, or findings upon which our Tax Opinion is based are unreasonable, within the meaning of Circular 230, because we do not know, and have no reason to believe, that any such factual assumptions, representations, statements, or findings are incorrect or incomplete. Please be aware, however, that, for the most part, we have not conducted any independent review of such factual assumptions, representations, statements, or findings.

Our Tax Opinion is conditioned upon, among other things, the initial and continuing accuracy and completeness of all of the facts and information provided by, and the covenants, representations, and warranties made by, Parent, Old SLM, External Controlled, Internal Distributing, Internal Controlled and all relevant parties to the Transactions described herein. Any change to or inaccuracy in any of such facts, information, covenants, representations, or warranties could affect or negate one or more of the conclusions stated herein.

BAKER BOTTS IIP

April 30, 2014

Qualifications as to Legal Matters

In rendering our Tax Opinion, we have relied upon applicable provisions of the Code, the Treasury regulations promulgated thereunder, pertinent judicial authorities, administrative rulings and practice, and such other authorities as we have considered relevant, in each case, as in effect on the date hereof. It should be noted that the Code, Treasury regulations, judicial decisions, administrative interpretations and such other authorities are subject to change at any time and, in some circumstances, with retroactive effect. A change in any of the authorities upon which our Tax Opinion is based or the occurrence of events subsequent to the effective time of the External Spin-off could affect one or more of our conclusions as stated herein.

The analysis of the U.S. federal income tax consequences of the Transactions involved a close evaluation of the applicable legal standards in light of the facts and circumstances related to the Transactions. No assurances can be given that the Internal Revenue Service ("IRS") will not assert a position contrary to one or more of the conclusions set forth in our Tax Opinion or that a court will not agree with the IRS's contrary position.

Opinion

Based solely upon and subject to the foregoing, we are of the opinion that, under current law, for U.S. federal income tax purposes:

- 1. The Reorganization will constitute a reorganization within the meaning of section 368(a)(1)(F) and will not be integrated with the other Transactions.
- 2. The Internal Spin-off will constitute a reorganization within the meaning of sections 368(a)(1)(D) and 355, such that Internal Distributing, Internal Controlled and Parent will not be subject to U.S. federal income tax in respect of the Internal Spin-off (except with respect to any intercompany transactions or excess loss account required to be taken into account in connection with the Internal Spin-off under Treasury regulation section 1.1502-13 or 1.1502-19, respectively).
- 3. The External Spin-off will constitute a reorganization within the meaning of sections 368(a)(1)(D) and 355, such that Parent, External Controlled and Parent's shareholders will not be subject to U.S. federal income tax in respect of the External Spin-off (except with respect to any intercompany transactions or excess loss account required to be taken into account in connection with the External Spin-off under Treasury regulation section 1.1502-13 or 1.1502-19, respectively).

Except as expressly set forth above, we express no other opinion relating to the tax consequences of the Reorganization, the External Spin-off or the Internal Spin-off, including any potential tax consequences under state, local or non-U.S. tax law. Our Tax Opinion is expressed as of the date hereof, and we disclaim any undertaking to advise Parent or External Controlled of

BAKER BOTTS LLP

April 30, 2014

any changes of or to any of the matters stated or assumed herein or of any subsequent changes in or to applicable law. Our Tax Opinion is solely for the benefit of the addressees above and is not to be relied upon, used, circulated, quoted or otherwise referred to for any purpose without our prior written consent.

Sincerely,

/s/ Baker Botts L.L.P.

Baker Botts L.L.P.

Exhibit 10.1

Execution Copy

TRANSITION SERVICES AGREEMENT

BY AND BETWEEN

NAVIENT CORPORATION

AND

SLM CORPORATION

DATED AS OF APRIL 29, 2014

TABLE OF CONTENTS

		Page
	ARTICLE I DEFINITIONS	
	ARTICLE II SERVICES, DURATION AND SERVICES MANAGERS	
Section 2.01. Section 2.02. Section 2.03. Section 2.04. Section 2.05. Section 2.07. Section 2.08.	Services; Scope Duration of Services Pricing for Services Changes to Services. Excluded Services Services Managers Personnel	8 8 8 9 9
	ARTICLE III ADDITIONAL ARRANGEMENTS	
Section 3.01. Section 3.02. Section 3.03. Section 3.04. Section 3.05. Section 3.06.	Software and Software Licenses Computer-Based and Other Resources Access to Facilities Cooperation Security and Privacy IT Transition Completion Requirements; IT Transition Changes	11 12 12 13 13
	ARTICLE IV COSTS AND DISBURSEMENTS	
Section 4.01. Section 4.02. Section 4.03.	Costs and Disbursements Tax Matters No Right to Set-Off	18 19 20
	ARTICLE V STANDARD FOR SERVICE	
Section 5.01. Section 5.02. Section 5.03.	Standard for Service Disclaimer of Warranties Compliance with Laws and Regulations	20 21 21
	ARTICLE VI LIMITED LIABILITY AND INDEMNIFICATION	
Section 6.01. Section 6.02. Section 6.03. Section 6.04. Section 6.05. Section 6.06.	Consequential and Other Damages Limitation of Liability Obligation To Re-perform; Liabilities Indemnification Liability for Payment Obligations Exclusion of Other Remedies	22 22 22 23 23 23

	ARTICLE VII TERM AND TERMINATION	
Section 7.01.	Term and Termination	23
	Effect of Termination	25
Section 7.03.	Force Majeure	25
	ADTICLE VIII	
	ARTICLE VIII GENERAL PROVISIONS	
Section 8.01.	No Agency	25
	Subcontractors	25
	Treatment of Confidential Information	26
Section 8.04.	Further Assurances	27
Section 8.05.	Dispute Resolution	27
Section 8.06.		27
Section 8.07.	Severability	28
Section 8.08.	Entire Agreement	28
Section 8.09.	No Third-Party Beneficiaries	28
Section 8.10.	Governing Law	28
Section 8.11.	Amendment; Waiver	28
Section 8.12.	Rules of Construction	28
	Counterparts	29
Section 8.14.	Assignability	29
	Public Announcements	30
	Non-Recourse	30
	Audit Rights	30
	Title to Intellectual Property	30
Section 8.19.	Order of Precedence	31
EXHIBIT I: Se	ervices Managers	
	Disclosing Party Security Requirements	
	Key Employees	
	Excluded SMI Projects	
	: Comprehensive Information Technology Services	
	2: Short-Term Shared Loan Servicing Services	
	: Customer Experience Services	
	: Trust Administration Services	
	: Facilities Services	
	5: Third Party TSA Support Services	
	2: Government Relations	
	3: Services Business Development Support and IT Services	
SCHEDULE 9	Financial and HR Systems Support Services	

TRANSITION SERVICES AGREEMENT

This TRANSITION SERVICES AGREEMENT, dated as of April 29, 2014 (this "Agreement"), is by and between Navient Corporation, a Delaware corporation ("Navient"), and SLM Corporation, a Delaware corporation ("SLM BankCo"). Unless otherwise defined in this Agreement, all capitalized terms used in this Agreement shall have the meaning set forth in the Separation and Distribution Agreement, dated as of April 28, 2014, by and among Existing SLM, SLM BankCo and Navient (as amended, modified or supplemented from time to time in accordance with its terms, the "Separation and Distribution Agreement").

RECITALS

WHEREAS, the board of directors of Existing SLM Corporation has determined that it is in the best interests of Existing SLM and its stockholders to separate Existing SLM's Navient Business and SLM BankCo Business;

WHEREAS, Existing SLM, SLM BankCo and Navient have entered into the Separation and Distribution Agreement;

WHEREAS, in order to facilitate and provide for an orderly separation and transition under the Separation and Distribution Agreement, the Parties (as defined herein) desire to enter into this Agreement to set forth the terms and conditions pursuant to which each of the Parties and their affiliates shall provide to the other the Services (as defined herein) for a transitional period; and

WHEREAS, the Separation and Distribution Agreement requires execution and delivery of this Agreement by Navient and SLM BankCo on or prior to the Distribution Date.

NOW, THEREFORE, in consideration of the foregoing and the mutual agreements contained in this Agreement and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties, intending to be legally bound, hereby agree as follows:

ARTICLE I DEFINITIONS

The following capitalized terms used in this Agreement shall have the meanings set forth below:

"Additional Services" shall have the meaning set forth in Section 2.04(a).

"Agreement" shall have the meaning set forth in the Preamble.

"Banking Agency Regulation P" shall have the meaning set forth in Section 3.05(c).

"Cause" shall mean a person's indictment for embezzlement or fraud, conviction of a felony, pleading guilty or *nolo contendere* to a felony or breach of fiduciary duty, or deliberate disregard of the Code of Business Conduct of Navient.

- "CFPB Regulation P" shall have the meaning set forth in Section 3.05(c).
- "CIO" shall have the meaning set forth in Section 3.06(b).
- "Completion Criteria" shall have the meaning set forth in Section 3.06(a).
- "Confidential Information" shall have the meaning set forth in Section 8.03.
- "cost," as used in any Schedule to this Agreement, shall mean the Provider's fully loaded cost inclusive of all standard overhead allocations, unless otherwise expressly provided in any such Schedule.
 - "Disaster Recovery/Business Continuity Plan" shall have the meaning set forth in Section 3.08(b).
 - "Disclosing Party" shall have the meaning set forth in Section 3.05(a).
 - "Disclosing Party Customer Information" shall have the meaning set forth in Section 3.05(c).
 - "Disclosing Party Security Requirements" shall have the meaning set forth in Section 3.05(b).
 - "DSS" shall have the meaning set forth in Section 3.05(g).
 - "Force Majeure" shall have the meaning set forth in the Separation and Distribution Agreement.
 - "FTC Final Rule" shall have the meaning set forth in Section 3.05(c).
 - "GLB Requirements" shall have the meaning set forth in Section 3.05(c).
 - "Group" shall have the meaning set forth in the Separation and Distribution Agreement.
 - "Information Security Program Requirements" shall have the meaning set forth in Section 3.05(c).
 - "Interest Payment" shall have the meaning set forth in Section 4.01(d).
 - "IT Transition" shall have the meaning set forth in Section 3.06(a).
 - "IT Transition Milestones" shall have the meaning set forth in Section 3.06(a).
 - " $\underline{\text{Key Employees}}$ " shall mean the individuals set forth on $\underline{\text{Exhibit III}}$ hereto.
 - "Milestone Dates" shall have the meaning set forth in Section 3.06(b).
 - "Navient" shall have the meaning set forth in the Preamble.

- "Navient Business Plan" shall mean the business plan of the Navient Group for the two-year period following the date of this Agreement, in the form and on the terms approved by the Board of Directors of SLM Corporation in October 2013.
 - "Navient Program Manager" shall have the meaning set forth in the Separation and Distribution Agreement.
 - "Navient Services" shall have the meaning set forth in Section 2.01.
 - "Navient Services Manager" shall have the meaning set forth in Section 2.07(a).
 - "Non-assignable Contract" shall have the meaning set forth in the Separation and Distribution Agreement.
- "Party" shall mean Navient and SLM BankCo individually, and "Parties" shall mean Navient and SLM BankCo collectively, and, in each case, their permitted successors and assigns.
 - "PCI" shall have the meaning set forth in Section 3.05(g).
 - "Preferred Stock" shall have the meaning set forth in the SMI Amended and Restated Charter.
 - "Preferred Stock Period" shall have the meaning set forth in the SMI Amended and Restated Charter.
 - "Provider" shall mean the Party or its Subsidiary or Affiliate providing a Service under this Agreement or obligated or designated to complete a task.
 - "Provider Indemnified Party" shall have the meaning set forth in Section 6.04.
 - "Receiving Party" shall have the meaning set forth in Section 3.05(a).
 - "Receiving Party Personnel" shall have the meaning set forth in Section 3.05(b).
 - "Recipient" shall mean the Party or its Subsidiary or Affiliate to or for whom a Service or task under this Agreement is being provided or performed.
 - "Recipient Indemnified Party" shall have the meaning set forth in Section 6.05.
 - "Redemption Date" has the meaning set forth in Section 3.06(h),
 - "Redemption Price" has the meaning set forth in in the SMI Amended and Restated Charter.
 - "Reimbursement Charge(s)" shall have the meaning set forth in Section 4.01(b).
 - $\label{eq:continuous} \begin{tabular}{ll} {\bf 'Representatives''} shall have the meaning set forth in the Separation and Distribution Agreement. \\ \end{tabular}$

- "Restricted Access Navient Customer Information" shall have the meaning set forth in Section 3.05(h).
- "Schedule(s)" shall have the meaning set forth in Section 2.01.
- "Security Incident" shall have the meaning set forth in Section 3.05(d).
- "Security Audit" shall have the meaning set forth in Section 3.05(e).
- "Separation and Distribution Agreement" shall have the meaning set forth in the Preamble.
- "Separation Projects" shall mean the separation and migration projects listed on Annex A to Statement of Work C to Schedule 1.
- "Service Charge(s)" shall have the meaning set forth in Section 4.01(a).
- "Service Decrease" shall have the meaning set forth in Section 2.04(b).
- "Service Extension" shall have the meaning set forth in Section 7.01(c).
- "Service Increase" shall have the meaning set forth in Section 2.04(a).
- "Service Locations" shall have the meaning set forth in Section 3.08(a).
- "Services" shall have the meaning set forth in Section 2.01.
- "Shared Contract" shall have the meaning set forth in the Separation and Distribution Agreement.
- "SLM BankCo" shall have the meaning set forth in the Preamble.
- "SLM BankCo Business Plan" shall mean the business plan of the SLM BankCo Group for the two-year period following the date of this Agreement, in the form and on the terms approved by the Board of Directors of SLM Corporation in October 2013.
 - "SLM BankCo Program Manager" shall have the meaning set forth in the Separation and Distribution Agreement.
 - "SLM BankCo Services" shall have the meaning set forth in Section 2.01.
 - "SLM BankCo Services Manager" shall have the meaning set forth in Section 2.07(b).
 - "SMI" shall mean Sallie Mae, Inc., a Delaware corporation and Subsidiary of Navient.
- "SMI Amended and Restated Charter" shall mean the Amended and Restated Certificate of Incorporation of SMI filed with the Secretary of State of the State of Delaware on April 28, 2014.

- "SMI Board" shall have the meaning set forth in the Separation and Distribution Agreement.
- "Special Preferred Director" shall have the meaning set forth in the SMI Amended and Restated Charter.
- "Subcontractors" shall have the meaning set forth in Section 3.05(c).
- "Taxes" shall have the meaning set forth in the Tax Matters Agreement.
- "Termination Charges" shall have the meaning set forth in Section 7.01(b)(i)(A).
- "Transfer Taxes" shall have the meaning set forth in Section 4.02.

ARTICLE II SERVICES, DURATION AND SERVICES MANAGERS

Section 2.01. <u>Services; Scope</u>. Subject to the terms and conditions of this Agreement, (a) Navient shall provide, or cause one or more of its Subsidiaries to provide, to the SLM BankCo Group the services for which Navient is the Provider as set forth in <u>Schedules 1, 2, 3, 4, 5, 6, 7, 8</u> and <u>9</u> to this Agreement (the "<u>Navient Services</u>") and (b) SLM BankCo shall provide, or cause one or more of its Subsidiaries to provide, to the Navient Group the services for which SLM BankCo is the Provider as set forth in <u>Schedules 1, 2, 3, 4, 5, 6, 7, 8</u> and <u>9</u> to this Agreement (the "<u>SLM BankCo Services</u>," and, collectively with the Navient Services, any Additional Services, any Service Increases, any Service Decreases and any New Services, the "<u>Services</u>"). The description and scope of the Services shall be as set forth on <u>Schedules 1, 2, 3, 4, 5, 6, 7, 8</u> and <u>9</u> (each a "<u>Schedule</u>", and collectively, the "<u>Schedules</u>"). All of the Services shall be for the sole use and benefit of the Recipient and its respective Affiliates.

Section 2.02. <u>Duration of Services</u>. Subject to the terms of this Agreement, each of Navient and SLM BankCo shall provide or cause to be provided to the respective Recipient(s) each Service until the earlier to occur of, with respect to each such Service, (a) the expiration of the duration of the term for such Service (or, subject to the terms of <u>Section 7.01(c)</u>, the expiration of any Service Extension) as set forth on the applicable Schedule or (b) the date on which such Service is terminated under <u>Section 7.01(b)</u>; <u>provided</u>, <u>however</u>, that the duration of the Services shall not extend beyond the two-year anniversary of the date of this Agreement.

Section 2.03. <u>Pricing for Services</u>. Subject to the terms of this Agreement, the Service Charge for each Service to be provided by a Provider shall be as reflected on the Schedules. With respect to each Service or category of Services, the applicable Schedule shall set forth (i) the Recipient that will be invoiced the Service Charge for such Service or category of Services and (ii) the Provider that will be paid such Service Charge. Service Charges and any applicable Reimbursement Charges shall be invoiced and paid in accordance with <u>Article IV</u>.

Changes to Services. (a) Service Increases; Additional Services. After the date of this Agreement, if a Recipient requests that a Provider (i) increase the volume, amount or frequency, as applicable, of any Service provided by a Provider (any such increase, a "Service Increase") by up to 50% of the volume, amount or frequency contemplated by the SLM BankCo

Business Plan or the Navient Business Plan, as applicable, or (ii) provide a service that is materially different from and in addition to the Services then included on the Schedules hereto (each, an "Additional Service") but was provided, immediately prior to the Effective Time, by a member of the Navient Group to the Pre-Separation Consumer Banking Business, in the case of a request by SLM BankCo, or by a member of the SLM BankCo Group to the Pre-Separation Education Loan Business, in the case of a request by Navient (other than because the Parties agreed such service would not be provided), then in each case such Provider shall be obligated to perform such Service Increase or Additional Service. If the Recipient requests that the Provider perform a Service Increase that exceeds 150% of the Services to be performed in the manner contemplated by the SLM BankCo Business Plan or Navient Business Plan, as applicable, then the Provider shall use commercially reasonable efforts to cooperate with the Recipient to provide such Service Increase.

(b) Service Decreases. After the date of this Agreement, a Recipient may request the Provider to decrease the volume, amount, level or frequency, as applicable, of any Service provided by a Provider (any such decrease, a "Service Decrease") by providing a written notice to Provider; provided, however, that, after delivery of a notice of Service Decrease, the Provider shall have no obligation to agree to a subsequent request of the Recipient for a Service Increase with respect to such decreased Services.

Section 2.04. (c) *Mutual Agreement; Supplements to Schedules.* In connection with any request for Additional Services, Service Increases or Service Decreases in accordance with this Section 2.04, the Navient Services Manager and the SLM BankCo Services Manager shall in good faith negotiate the terms of a supplement to the applicable Schedule, which terms shall be consistent with the terms of, and the pricing methodology used for, similar Services provided under this Agreement. Upon the mutual written agreement of the Parties, the supplement to the applicable Schedule shall describe in reasonable detail the nature, description, scope, expiration and duration, Service Charges and other terms applicable to such Additional Services, Service Increases or Service Decrease in a manner similar to that in which the Services are described in the existing Schedules. Each supplement to the applicable Schedule, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such supplement and the Additional Services or Service Increases set forth therein shall be deemed "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement. Excluded Services. It is not the intent of the Provider to render, nor of the Recipient to receive from the Provider, professional advice or opinions, whether with regard to Tax, legal, treasury, finance, employment or other business and financial matters, technical advice, including environmental matters. The Recipient shall not rely on, or construe, any Service rendered by or on behalf of the Provider as such professional advice or opinions or technical advice, and the Recipient shall require any Provider to render services that are prohibited from being provided by the Provider or to the Recipient by any applicable Law or if the provision of such services by the Provider or to the Recipient would violate any applicable Law.

Section 2.06. <u>Services Managers</u>. (a) Navient hereby appoints and designates the individuals set forth on <u>Exhibit I</u> to act as the initial services managers for the corresponding Services Schedules on <u>Exhibit I</u> (each, a"<u>Navient Services Manager</u>"), each of whom will be

directly responsible for coordinating and managing the delivery of the Navient Services under the applicable Service Schedule and have authority to act on Navient's behalf with respect to matters relating to the provision of the identified Services under this Agreement. Each Navient Services Manager will report to the Navient Program Manager and will work with the personnel of the Navient Group to periodically address issues and matters raised by SLM BankCo relating to the provision of Services under this Agreement. Navient shall notify SLM BankCo of the appointment of a different Navient Services Manager, if necessary, in accordance with Section 8.06.

(b) SLM BankCo hereby appoints and designates the individuals set forth on <u>Exhibit1</u> to act as the initial services managers for the corresponding Services Schedules on <u>Exhibit1</u> (each, a "<u>SLM BankCo Services Manager</u>"), each of whom will be directly responsible for coordinating and managing the delivery of the SLM BankCo Services under the applicable Services Schedule and have authority to act on SLM BankCo's behalf with respect to matters relating to the provision of the identified Services under this Agreement. Each SLM BankCo Services Manager will report to the SLM BankCo Program Manager and will work with the personnel of the SLM BankCo Group to periodically address issues and matters raised by Navient relating to the provision of Services under this Agreement. SLM BankCo shall notify Navient of the appointment of a different SLM BankCo Services Manager, if necessary, in accordance with Section 8.06.

Section 2.07. <u>Personnel</u>. (a) The Provider of any Service will make available to the Recipient of such Service such personnel as may be necessary to provide such Service on the understanding that such personnel shall remain employed and/or engaged by the Provider. The Provider will have the right, in its reasonable discretion, to (i) designate which personnel it will assign to perform such Service, and (ii) remove and replace such personnel at any time; <u>provided, however</u>, that any such removal or replacement shall not be the basis for any increase in any Service Charge or Reimbursement Charge payable hereunder or relieve the Provider of its obligation to provide any Service hereunder; <u>provided, further</u>, that the Provider will use its commercially reasonable efforts to limit the disruption to the Recipient in the transition of the Services to different personnel.

(b) In the event that the provision of any Service by the Provider requires the cooperation and services of the personnel of the Recipient, the Recipient will make available to the Provider such personnel (who shall be appropriately qualified for purposes of so supporting the provision of such Service by the Provider) as may be necessary for the Provider to provide such Service on the understanding that such personnel shall remain employed and/or engaged by the Recipient. The Recipient will have the right, in its reasonable discretion, to (i) designate which personnel it will make available to the Provider in connection with the provision of such Service, and (ii) remove and replace such personnel at any time; provided, however, that any resulting increase in costs to the Provider shall be borne by the Recipient and any resulting adverse effect to the provision of such Service by the Provider will not be deemed a breach of this Agreement by the Provider, further, that the Recipient will use its commercially reasonable efforts to limit the disruption to the Provider in the transition of such personnel. If the Provider, in its reasonable discretion and following discussions with the Recipient, requests the Recipient to remove and/or replace any such personnel from their roles in respect of the Services being provided by the Provider, the Recipient shall comply with such request.

- (c) No Provider shall be liable under this Agreement for any Liabilities incurred by the Recipient Indemnified Parties that are primarily attributable to, or that are a consequence of, any actions or inactions of the personnel of the Recipient, except for any such actions or inactions undertaken pursuant to the direction of the Provider.
- (d) Nothing in this Agreement shall grant the Provider, or its employees, agents and third-party providers that are performing the Services, the right directly or indirectly to control or direct the operations of the Recipient or any member of its Group. Such employees, agents and third-party providers shall not be required to report to the management of the Recipient nor be deemed to be under the management or direction of the Recipient. The Recipient acknowledges and agrees that, except as may be expressly set forth herein as a Service or otherwise expressly set forth in the Separation and Distribution Agreement, another Ancillary Agreement or any other applicable agreement, no Provider or any member of its Group shall be obligated to provide, or cause to be provided, any service or goods to any Recipient or any member of its Group.

ARTICLE III ADDITIONAL ARRANGEMENTS

Section 3.01. Assistance Regarding Software and Software Licenses. Navient shall use commercially reasonable efforts to assist SLM BankCo in its efforts to obtain licenses (or other appropriate rights) to use, duplicate and distribute, as necessary and applicable, certain computer software necessary for Navient to provide, and SLM BankCo to receive, Navient Services or for Navient to transition such Navient Services to SLM BankCo; provided, however, that Navient shall not be required to pay any fees or other payments or incur any obligations or liabilities to enable SLM BankCo to obtain any such license or rights (except and to the extent that SLM BankCo advances such fees or payments to Navient); provided, further, that Navient shall not be required to seek broader rights or more favorable terms for SLM BankCo than those applicable to Effective Time Sallie Mae or as may be applicable to Navient from time to time hereafter; provided, further, that SLM BankCo shall bear only those costs that relate solely and directly to obtaining such licenses (or other appropriate rights) in the ordinary course. The Parties acknowledge and agree that there can be no assurance that Navient's efforts will be successful or that SLM BankCo will be able to obtain such licenses or rights on acceptable terms or at all and, where Navient enjoys rights under any enterprise or site license or similar license, the Parties acknowledge that such license typically precludes partial transfers or assignments or operation of a service bureau on behalf of unaffiliated entities. In the event that SLM BankCo is unable to obtain such software licenses, the Parties shall work together using commercially reasonable efforts to obtain an alternative software license to allow Navient to provide, and SLM BankCo to receive, such Navient Services (or allow Navient to transition such Navient Services to SLM BankCo), and the Parties shall negotiate in good faith an amendment to the applicable Schedule to reflect any such new arrangement, which amended Sched

(a) If and to the extent requested by Navient, SLM BankCo shall use commercially reasonable efforts to assist Navient in its efforts to obtain licenses (or other appropriate rights) to use, duplicate and distribute, as necessary and applicable, certain computer software necessary for SLM BankCo to provide, and Navient to receive, SLM BankCo Services or for SLM BankCo

to transition such SLM BankCo Services to Navient; provided, however, that SLM BankCo shall not be required to pay any fees or other payments or incur any obligations or liabilities to enable Navient to obtain any such license or rights (except and to the extent that Navient advances such fees or payments to SLM BankCo); provided, further, that SLM BankCo shall not be required to seek broader rights or more favorable terms for Navient than those applicable to Effective Time Sallie Mae or as may be applicable to SLM BankCo from time to time hereafter; provided, further, that Navient shall bear only those costs that relate solely and directly to obtaining such licenses (or other appropriate rights) in the ordinary course. The Parties acknowledge and agree that there can be no assurance that SLM BankCo's efforts will be successful or that Navient will be able to obtain such licenses or rights on acceptable terms or at all and, where SLM BankCo enjoys rights under any enterprise or site license or similar license, the Parties acknowledge that such license typically precludes partial transfers or assignments or operation of a service bureau on behalf of unaffiliated entities. In the event that Navient is unable to obtain such software licenses, the Parties shall work together using commercially reasonable efforts to obtain an alternative software license to allow SLM BankCo to provide, and Navient to receive, such SLM BankCo Services (or allow SLM BankCo to transition such SLM BankCo Services to Navient), and the Parties shall negotiate in good faith an amendment to the applicable Schedule to reflect any such new arrangement, which amended Schedule shall not require Navient to pay for any fees, Taxes, expenses or costs relating to the software license that Navient was unable to obtain pursuant to the provisions of this Section 3.01(b).

(b) In the event that there are any costs associated with obtaining software licenses in accordance with this Section 3.01 that (i) would not be payable in the ordinary course, including in the form of a "transfer fee" or other similar fees or expenses payable by the Recipient or the Provider, and (ii) would not have been payable by the Recipient or the Provider absent the need for a consent or waiver in connection with the license that the Recipient is seeking to obtain, such costs shall be paid by the Recipient.

Section 3.02. Computer-Based and Other Resources. From and after the date of this Agreement, each Party and its Affiliates shall cause all of their Representatives having access to the computer software, networks, hardware, technology or computer-based resources of the other Party and its Affiliates pursuant to the Separation and Distribution Agreement, or any Ancillary Agreement, or in connection with the performance, receipt or delivery of any Service, to comply with all security guidelines (including physical security, network access, internet security, confidentiality and personal data security guidelines) of such other Party and its Affiliates of which written notice is provided by such other Party. Each Party shall ensure that the access contemplated by this Section 3.02 shall be used by its Representatives only for the purposes contemplated by, and subject to the terms of, this Agreement.

Section 3.03. Access to Facilities. (a) SLM BankCo shall, and shall cause its Subsidiaries to, allow Navient and its Representatives reasonable access to the facilities of SLM BankCo necessary for Navient to fulfill its obligations under this Agreement.

(b) Navient shall, and shall cause its Subsidiaries to, allow SLM BankCo and its Representatives reasonable access to the facilities of Navient necessary for SLM BankCo to fulfill its obligations under this Agreement.

- (c) Notwithstanding the other rights of access of the Parties under this Agreement, each Party shall, and shall cause its Subsidiaries to, afford the other Party, its Subsidiaries and Representatives, following not less than five business days' prior written notice from the other Party, reasonable access during normal business hours to the facilities, information, systems, infrastructure and personnel of the relevant Providers as reasonably necessary for the other Party to verify the adequacy of internal controls over information technology, reporting of financial data and related processes employed in connection with the Services, including in connection with verifying compliance with Section 404 of the Sarbanes-Oxley Act of 2002; provided, however, such access shall not unreasonably interfere with any of the business or operations of such Party or its Subsidiaries.
- (d) Except as otherwise permitted by the other Party in writing, each Party shall permit only its authorized Representatives, contractors, invitees or licensees to access the other Party's facilities.

Section 3.04. <u>Cooperation</u>. It is understood that it will require the significant efforts of both Parties to implement this Agreement and to ensure performance of this Agreement by the Parties at the agreed-upon levels in accordance with all of the terms and conditions of this Agreement. The Parties will cooperate, acting in good faith and using commercially reasonable efforts, to effect a smooth and orderly transition and performance of the Services provided under this Agreement from the Provider to the Recipient (including the assignment or transfer of the rights and obligations under any third-party contracts relating to the Services); <u>provided</u>, <u>however</u>, that this <u>Section 3.04</u> shall not require either Party to incur any out-of-pocket costs or expenses unless and except as expressly provided in this Agreement or otherwise agreed to in writing by the Parties. Any Non-assignable Contract or Shared Contract shall be treated in the manner provided in <u>Section 2.5</u> of the Separation and Distribution Agreement.

Section 3.05. Security and Privacy.

- (a) Receiving Party; Disclosing Party. For purposes of this Section 3.05, "Receiving Party" means a party that obtains, maintains, processes, receives or otherwise is permitted access to Confidential Information of the other party ("Disclosing Party").
- (b) Disclosing Party Security Requirements; Security Questionnaire. Receiving Party shall comply, and shall cause its employees ("Receiving Party Personnel") to comply (to the extent applicable to individuals), with the provisions set forth in Exhibit II (the "Disclosing Party Security Requirements"). As periodically requested by Disclosing Party, but no more frequently than annually, Receiving Party shall promptly, fully and accurately complete Disclosing Party's Information Security Questionnaire and other documents or requests for information regarding Receiving Party's information security practices relating to Disclosing Party's Confidential Information (e.g., summaries of security audits, summaries of test results or other equivalent evaluations of Receiving Party's information security practices).
- (c) Customer Information Handling Requirements. Receiving Party hereby agrees that (i) it shall comply, (ii) it shall cause all Receiving Party Personnel to comply and (iii) it shall cause all agents, contractors, subcontractors, and vendors that perform services that are used by Receiving Party to provide the Services (collectively, "Subcontractors") to comply, to the extent

they have access to Disclosing Party Customer Information (as defined below), with all reuse, redisclosure and other customer information handling, processing, security and protection requirements that are specifically required of a non-affiliated third-party processor or servicer (or subcontractor) under the Federal Trade Commission's Privacy of Consumer Financial Information; Final Rule (16 CFR 313) (the "FTC Final Rule"), the Joint Banking Agencies' Privacy of Consumer Financial Information; Final Rule (12 CFR Parts 40, 216, 332 and 573) (the "Banking Agency Regulation P"), or the Bureau of Consumer Financial Protection's Privacy of Consumer Financial Information (Regulation P) (12 CFR 1016) (the "CFPB Regulation P"), as applicable, each implementing Title V of the Gramm-Leach-Bliley Act, Public Law 106-102 (the "GLB Requirements") and other applicable federal and state consumer privacy Laws. Without limiting the foregoing, Receiving Party agrees that:

- (i) it is prohibited from disclosing or using any "nonpublic personal information" (as defined in the GLB Requirements) that it may obtain, maintain, process or otherwise receive from, through or on behalf of Disclosing Party in connection with this Agreement or as may have been received pursuant to the Separation and Distribution Agreement (the "Disclosing Party Customer Information"), except solely to carry out the purposes for which it was disclosed, including use under an applicable exception contained in Section 313.14 or 313.15 of the FTC Final Rule, Section 332.14 or 332.15 of the Banking Agency Regulation P, or Section 1016.14 or 1016.15 of the CFPB Regulation P, as applicable, in the ordinary course of business to carry out those purposes; and
- (ii) it has implemented and will maintain a written information security program that complies with applicable state laws and regulations pertaining to the protection of personal information (e.g., MA 201 CMR 17.00) and that is designed to meet the following objectives as set forth in the Interagency Guidelines Establishing Information Security Standards; Final Rule (12 CFR Part 30, et al.) (the "Information Security Program Requirements"):
 - (1) Ensure the security and confidentiality of the Disclosing Party Customer Information;
 - (2) Protect against any anticipated threats or hazards to the security or integrity of such information;
 - (3) Protect against unauthorized access to or use of such information that could result in substantial harm or inconvenience to any customer; and
 - (4) Ensure the proper disposal of Disclosing Party Customer Information and "consumer information" (as that term is defined in the Interagency Guidelines Establishing Information Security Standards; Final Rule (12 CFR Part 30, et al.)).

(d) Security Incidents. Unless specifically requested by law enforcement not to communicate with Disclosing Party, Receiving Party shall report to Disclosing Party all known Security Incidents, For clarity, with respect to SLM BankCo Confidential Information accessed by Navient Representatives during the IT Transition, Navient shall be deemed to be the Receiving Party and SLM BankCo shall be deemed to be the Disclosing Party for purposes of SLM BankCo Confidential Information (including SLM BankCo's Disclosing Party Customer Information) hosted by Navient. With respect to Navient Confidential Information accessed by SLM BankCo Representatives during the IT Transition, SLM BankCo shall be deemed to be the Receiving Party and Navient shall be deemed to be the Disclosing Party for purposes of Navient Confidential Information (including Navient's Disclosing Party Customer Information) accessible by SLM BankCo. "Security Incident" means any unauthorized action by a known or unknown person or automated program (e.g., worm, virus, web crawler, malware, etc.) that leads to one of the following: unintended disclosure of confidential, customer, employee or other sensitive information, denial of service, misuse of system access, or unauthorized access or intrusion, all to the extent they affect the security, confidentiality or integrity of the Disclosing Party's Confidential Information received, stored, processed, or maintained by Receiving Party. "Security Incident" shall also include any contact by a law enforcement agency with Receiving Party regarding any of Disclosing Party's Confidential Information, unless specifically mandated by law enforcement not to communicate with Disclosing Party. To the extent Receiving Party becomes aware of any Security Incidents occurring with respect to its Subcontractors that have access (either authorized or unauthorized) to Disclosing Party's Confidential Information, Receiving Party shall be required to report such Security Incidents in accordance with the provisions of this Section. If a Security Incident occurs, Receiving Party shall notify Disclosing Party within 24 hours in accordance with the procedure and contact information set forth on Exhibit II and provide the following information, to the extent known at such time: nature and impact of the Security Incident; actions already taken by Receiving Party; Receiving Party's assessment of immediate risk; and corrective measures to be taken, evaluation of alternatives and next steps. Receiving Party shall continue providing (i) appropriate status reports to Disclosing Party regarding the resolution of the Security Incident and prevention of future such Security Incidents, and (ii) cooperation, as reasonably requested by Disclosing Party, in order to further investigate and resolve the Security Incident. Disclosing Party may require that the Services provided by Receiving Party to Disclosing Party be suspended, connectivity with Receiving Party be terminated or that other appropriate action be taken pending such resolution. Receiving Party shall preserve evidence of all Security Incidents and allow external forensic analysis either onsite or through shipment of components. Except as otherwise provided in Section 10.13 of the Separation and Distribution Agreement, Disputes arising in connection with a Security Incident, including without limitation, Disputes regarding the occurrence or non-occurrence of a Security Incident or the appropriate remediation measures, shall be addressed using the same escalation processes and dispute resolution procedures as applicable to Disputes arising in connection with the

(e) Security Audits. During the Term and thereafter for as long as Receiving Party retains Disclosing Party Confidential Information, Disclosing Party, its representatives and agents will be entitled to conduct audits of Receiving Party's relevant operations, facilities, and systems to confirm that Receiving Party has complied with the Disclosing Party Security Requirements and the Information Security Program Requirements (each, a "Security Audit"); provided that, following the termination of this Agreement, such Security Audits may only be

conducted by the Disclosing Party once per consecutive 12-month period commencing upon the effective date of termination of this Agreement. Any Security Audit shall be scheduled with reasonable prior notice and conducted during normal business hours and shall not unreasonably interfere with Receiving Party's business activities. Receiving Party may require any such auditor to sign a customary confidentiality agreement. In the event that any Security Audit results in the discovery of material security risks to Disclosing Party Confidential Information, Receiving Party shall respond to Disclosing Party in writing with Receiving Party's plan to promptly take reasonable measures and corrective actions necessary to effectively eliminate the risk, at no cost to Disclosing Party. Receiving Party shall have 15 business days to cure such security risk, unless the parties mutually agree in writing to a longer period of time for such cure. Disclosing Party's right, and the right of its representatives and agents, to conduct Security Audits, and any exercise of such right, shall not in any way diminish or affect Receiving Party's duties and liabilities under this Agreement. The Parties acknowledge and agree that the audit rights and limitations provided in this paragraph (e) are independent from the audit rights and limitations provided in Section 8.17.

(f) Subcontractors. Receiving Party shall be responsible for the acts and omissions of its Subcontractors as if they were the acts and omissions of its employees. Without limiting the foregoing, Receiving Party (a) shall oversee any such Subcontractors that obtain, maintain, process, receive, or otherwise are permitted access to Disclosing Party's Confidential Information (including, without limitation, all Disclosing Party Customer Information) by taking reasonable steps to select and retain Subcontractors that are capable of maintaining appropriate safeguards to protect the security and confidentiality of the Disclosing Party's Confidential Information, (b) shall require Subcontractors to comply with (i) confidentiality provisions substantially similar to those set forth in this Agreement, (ii) to the extent the Subcontractor has access to Disclosing Party Customer Information, privacy and security provisions substantially similar to those set forth in the Customer Information Handling Requirements paragraphs of this Agreement, and (iii) to the extent the Subcontractor has access to, stores, or processes the Disclosing Party's customers' cardholder information, the PCI DSS (defined immediately below) in effect from time to time, and (c) shall take commercially reasonable steps to require Subcontractors to adhere to the Disclosing Party Security Requirements set forth in Exhibit II. Receiving Party shall provide Disclosing Party with a list of Subcontractors from time-to-time as reasonably requested by Disclosing Party.

(g) PCI Requirements. If Receiving Party has access to, stores, or processes the Disclosing Party's customers' cardholder information, Receiving Party hereby confirms that it has on file a current Report on Compliance, evidencing that it is in compliance with the payment card industry ("PCI") data security standard ("DSS"). Receiving Party shall provide Disclosing Party with a copy of the PCI DSS Attestation of Compliance Letter upon request. In addition to the foregoing, to the extent applicable to the Services being provided to Disclosing Party hereunder, Receiving Party will comply with and adhere to the PCI DSS in effect from time to time. Each party shall be responsible for the implementation, testing, and compliance with respect to PCI data security controls within their respective PCI DSS boundaries. These requirements are applicable to all infrastructure and systems processing or storing any cardholder information as defined by the PCI DSS. Any change in Receiving Party's PCI compliance and/or certification status shall be promptly communicated to Disclosing Party.

(h) Limitations on Access. Notwithstanding anything contained herein to the contrary, SLM BankCo shall not, and shall cause its Representatives to not, access or attempt to access any customer/loan information except (i) the Navient customer information as necessary to perform the SLM BankCo Services, and (ii) the customer information pertaining to SLM BankCo owned loans (including loans owned by third party lenders who have contracted with SLM BankCo to provide certain services) (all customer information other than that described in subsections (i) and (ii) is collectively referred to as the "Restricted Access Navient Customer Information." If SLM BankCo inadvertently or otherwise accesses any Restricted Access Navient Customer Information, SLM BankCo agrees to not to use or disclose the Restricted Access Navient Customer Information.

Section 3.06. IT Transition Completion Requirements; IT Transition Governance.

- (a) Upon approval by a majority of the members of the SMI Board, which approval shall include the approval of the Special Preferred Director, that the "<u>Completion Criteria</u>" for each "<u>IT Transition Milestone</u>", as each is described in <u>Schedule 1</u>, have been satisfied, the migration of SLM BankCo's customer data, servicing and origination service functions to an independent SLM BankCo platform or alternative long-term platform (the "IT Transition") shall be deemed to be complete.
- (b) The Chief Information Officers of each of Navient and SLM BankCo (each, a "<u>CIO</u>") shall coordinate and manage the delivery of Services under <u>Schedule 1</u> so as to allow the IT Transition Milestones to be completed by the applicable "<u>Milestone Dates</u>" set forth on <u>Schedule 1</u>. The CIO's shall promptly escalate all Service delivery issues, material delays in completing the IT Transition Milestones, and material disputes arising under <u>Schedule 1</u> to the Separation and Oversight Committee.
 - (c) Any proposal by either Party to:
 - (i) modify the IT Transition Milestones (including the addition or removal of Separation Projects comprising any IT Transition Milestone);
 - (ii) modify any of the Completion Criteria or Milestone Dates;
 - (iii) amend the material terms of Schedule 1 or any Statement of Work thereunder; or
 - (iv) modify the cost allocation, budgets or spend commitments provided in Schedule 1 in any material respect;

in each case, shall be promptly submitted to the Separation and Oversight Committee for approval. If the Separation and Oversight Committee does not unanimously approve any such proposal, such proposal shall be escalated to the SMI Board for approval by a majority of the members of the SMI Board, which approval shall include the approval of the Special Preferred Director.

- (d) If any IT Transition Milestone is not completed by the applicable Milestone Date, the SMI Board will promptly meet to review (i) the impact to the remaining IT Transition Milestones and associated Milestone Dates, (ii) the remediation efforts undertaken by the Parties in respect of such uncompleted IT Transition Milestone, and (iii) the incremental costs, if any, to be incurred by the Parties in respect of such uncompleted IT Transition Milestone and remediation efforts. The SMI Board shall review and approve any additional budget or cost allocation arising out of the uncompleted IT Transition Milestone in accordance with paragraph (c) above (including the requirement of approval by the Special Preferred Director).
- (e) Navient shall cause any proposed changes to the operations of SMI that would reasonably be expected to adversely affect, in any material respect, the ability of the Navient Group to timely satisfy Navient's obligations in respect of the Completion Criteria for any IT Transition Milestone to be promptly submitted to the SMI Board for approval by a majority of the members of the SMI Board of Directors of SMI, which approval shall include the approval of the Special Preferred Director; provided, however that the foregoing consent requirement shall not apply to the SMI projects and operations that are set forth on Exhibit IV.
- (f) Prior to any termination (other than a termination for Cause or at the request of SLM BankCo) or replacement of a Key Employee of SMI during the Preferred Stock Period, Navient shall provide prompt written notice to the Special Preferred Director of Navient's intention to terminate or replace such Key Employee. Navient shall cause any such proposed termination or replacement to be submitted to the SMI Board for approval by a majority of the members of the SMI Board, which approval shall include the approval of the Special Preferred Director. Notwithstanding anything to the contrary herein, SMI may terminate a Key Employee for Cause (without the approval of, or provision of prior notice to, SLM BankCo or the Special Preferred Director) to the extent it reasonably determines, in consultation with counsel, that Cause for termination of such Key Employee exists.
- (g) SLM BankCo shall not make any changes to the operations of the SLM BankCo Group that would reasonably be expected to adversely affect, in any material respect, the ability of the Navient Group to timely satisfy Navient's obligations in respect of the Completion Criteria for any IT Transition Milestone without the prior written consent of Navient. Further, SLM BankCo shall not reassign any Key Employee, or any other employee of SLM BankCo assigned to perform Services pursuant to Statement of Work C (Separation Projects) under Schedule 1 for Projects that have a material impact to the fulfillment of the IT Transition Milestones, in each case without Navient's prior written consent. Notwithstanding anything to the contrary herein, SLM BankCo may terminate a Key Employee for Cause (without the approval of Navient SLM BankCo) to the extent it reasonably determines, in consultation with counsel, that Cause for termination of such Key Employee exists.
- (h) Navient shall cause SMI to redeem all of the issued and outstanding shares of Preferred Stock, out of funds legally available therefor, for the Redemption Price on the date (the "Redemption Date") that is the earlier to occur of (i) three business days

following the determination, and approval by the SMI Board pursuant to paragraph (a) that the IT Transition has been completed, and (ii) the 24 month anniversary of the initial issuance of the Preferred Stock.

Section 3.07. Additional Special Preferred Director Approval Rights.

- (a) During the Preferred Stock Period, Navient shall not permit SMI to enter into any merger, consolidation or combination of SMI into or with any other corporation or entity (regardless of whether SMI is the surviving entity) without the approval of the majority of the members of the SMI Board, which approval shall include the approval of the Special Preferred Director.
- (b) During the Preferred Stock Period, Navient shall cause the Chief Executive Officer of SMI to certify to the holder(s) of the Preferred Stock, within five business days after the end of each calendar quarter, that none of the actions described in Sections 3.06(e) or (h) or paragraph (a) above have been effected by SMI during such quarter (or, if any such action has been taken, that the actions taken by SMI comply with the provisions of Sections 3.06 and 3.07(a).

Section 3.08. Disaster Recovery; Business Continuity.

- (a) <u>Disaster Recovery and Business Continuity</u>. Receiving Party will be responsible for disaster recovery and business continuity planning, testing, implementation, and execution related to all locations from which services are provided under this Agreement (including locations of Receiving Party's Subcontractors, "<u>Service Locations</u>"). Receiving Party shall provide the services described in this <u>Section 3.08</u> without extra charge to the Disclosing Party:
 - (i) maintaining and testing the Disaster Recovery/Business Continuity Plan (as defined below);
- (ii) implementing the Disaster Recovery/Business Continuity Plan in the event of a disaster affecting Service Locations within the specified timeframes;
- (iii) carrying out disaster recovery, business continuity or redundancy procedures related to the services in accordance with such Disaster Recovery/Business Continuity Plan; and
 - (iv) designing and implementing a Disaster Recovery/Business Continuity plan to minimize disruption to the business of Disclosing Party.
- (b) <u>Disaster Recovery/Business Continuity Plan</u>. Receiving Party shall maintain a Disaster Recovery/Business Continuity plan (the "<u>Disaster Recovery/Business Continuity Plan</u>") that covers services that are provided to Disclosing Party and each of the proposed Service Locations. The Receiving Party will provide a target recovery time objective (RTO) and recovery point objective (RPO) of 48 hours and 2 hours, respectively, unless a lower minimum is provided in the Disaster Recovery/Business Continuity Plan in respect of any particular application.

- (c) <u>Testing</u>. Receiving Party will test the Disaster Recovery/Business Continuity Plan on at least an annual basis or more often as required by any regulatory authority having authority over Receiving Party.
- (d) <u>Priority</u>. If a disaster or major site impact occurs involving the need to recover operations at a Service Location or to operate from a fallback site and, as a result, it is necessary for Receiving Party or its Subcontractors to allocate limited resources between or among several organizations, Receiving Party shall not treat Disclosing Party less favorably than any other customer for purposes of allocation of such resources.

ARTICLE IV COSTS AND DISBURSEMENTS

- Section 4.01. Costs and Disbursements. (a) Except as otherwise provided in this Agreement, a Recipient of Services shall pay to the Provider of such Services a fee for the Services (or category of Services, as applicable) (each fee constituting a "Service Charge" and, collectively, "Service Charges") as listed on the Schedules. During the term of this Agreement, the amount of a Service Charge for any Services (or category of Services, as applicable) may increase or decrease to the extent of: (i) any increases or decreases mutually agreed to by the Parties, (ii) any Service Charges applicable to any Additional Services, Service Increases, Service Decreases or New Services, and (iii) any increase or decrease in the rates or charges imposed by any unaffiliated third-party provider that is providing Services.
- (b) Except as otherwise provided in this Agreement or the Schedules, the Recipient shall reimburse the Provider for reasonable out-of-pocket costs and expenses incurred by the Provider or its Affiliates in connection with providing the Services (including necessary travel-related expenses) to the extent that such costs and expenses are not reflected in the Service Charge for such Services (each such cost or expense, a "Reimbursement Charge" and, collectively, "Reimbursement Charges"); provided, however, that any such cost or expense that is materially inconsistent with historical practice between the Parties for any Service (including business travel-related expenses) shall require advance approval of the Recipient. Any authorized travel-related expenses incurred in performing the Services shall be incurred and charged to the Recipient in accordance with the Provider's then-applicable business travel policies.
- (c) Unless otherwise provided on a Schedule, each Provider shall prepare a monthly invoice that reflects all Service Charges and Reimbursement Charges with respect to the Services provided to each Recipient during the prior calendar month; provided, however, that to the extent any third-party costs and expenses are to be included in any Service Charges or Reimbursement Charges, each Provider shall have up to 60 days following receipt of the applicable invoice for such third-party costs or expenses before invoicing to each Recipient. Such invoice shall be submitted to the Recipient within 30 calendar days following the end of the prior calendar month and shall provide the Recipient with documentation reasonably necessary to support the calculation of such Service Charges and any Reimbursement Charges.

- (d) The Recipient shall pay the amount of each monthly invoice of Service Charges and Reimbursement Charges by wire transfer (or such other method of payment as may be agreed between the Parties) to the Provider within 30 calendar days of the receipt of each such invoice, including appropriate documentation as described herein, as instructed by the Provider. In the absence of a timely notice of a billing dispute in accordance with the provisions of Article VII of the Separation and Distribution Agreement, if the Recipient fails to pay such undisputed amount by the due date, the Recipient shall be obligated to pay to the Provider, in addition to the amount due, interest at an annual default interest rate of the Prime Rate plus two percent (2%) per annum, or the maximum legal rate, whichever is lower (the "Interest Payment"), accruing from the date the payment was due through the date of actual payment. In the event of any billing dispute, the Recipient shall promptly pay any undisputed amount. All amounts due and payable hereunder shall be invoiced and paid in U.S. dollars.
- (e) Subject to the confidentiality provisions set forth in Section 8.03, each Party shall, and shall cause their respective Affiliates to, provide, upon 10 days' prior written notice from the other Party, any information within such Party's or its Affiliates' possession that the requesting Party reasonably requests in connection with any Services being provided to such requesting Party by an unaffiliated third-party provider, including any applicable invoices, agreements documenting the arrangements between such third-party provider and the Provider and other supporting documentation; provided, however, that each Party shall make no more than one such request during any fiscal quarter.

Section 4.02. Tax Matters.

- (a) Without limiting any provisions of this Agreement, the Recipient shall be responsible for (i) all excise, sales, use, transfer, stamp, documentary, filing, recordation and other similar Taxes, (ii) all value added, goods and services or similar recoverable indirect Taxes, if any, and (iii) any related interest and penalties incurred solely as a result of the Recipient's failure to timely pay such Taxes to Provider (collectively, "Transfer Taxes"), in each case imposed or assessed as a result of the provision of Services by the Provider. The Party required to account for Transfer Tax shall provide to the other Party evidence of the remittance of the amount of such Transfer Tax to the relevant Governmental Authority. The Provider agrees that it shall take commercially reasonable actions to cooperate with the Recipient in obtaining any refund, return, rebate, exemption, or the like of any Transfer Tax, including by filing any necessary exemption or other similar forms, certificates, or other similar documents. The Recipient shall promptly reimburse the Provider for any costs incurred by the Provider or its Affiliates in connection with the Recipient obtaining a refund or overpayment of refund, return, rebate, exemption, or the like of any Transfer Tax. For the avoidance of doubt, any applicable gross receipts-based or net income-based Taxes shall be borne by the Provider.
- (b) The Recipient shall be entitled to deduct and withhold Taxes required by any Governmental Requirements to be withheld on payments made pursuant to this Agreement. To the extent any amounts are so withheld, the Recipient shall (i) pay, in addition to the amount otherwise due to the Provider under this Agreement, such additional amount as is necessary to ensure that the net amount actually received by the Provider will equal the full amount the Provider would have received had no such deduction or withholding been required, (ii) pay such deducted and withheld amount to the proper Governmental Authority, and (iii) promptly provide

to the Provider evidence of such payment to such Governmental Authority. The Provider shall, prior to the date of any payment to be made pursuant to this Agreement, at the request of the Recipient, make commercially reasonable efforts to provide the Recipient any certificate or other documentary evidence (x) required by Governmental Requirements or (y) which the Provider is entitled by Governmental Requirements to provide in order to reduce the amount of any Taxes that may be deducted or withheld from such payment and the Recipient agrees to accept and act in reliance on any such duly and properly executed certificate or other applicable documentary evidence.

(c) If the Provider (i) receives any refund (whether by payment, offset, credit or otherwise) or (ii) utilizes any overpayment of Taxes that are borne by Recipient pursuant to this Agreement, then the Provider shall promptly pay, or cause to be paid, to the Recipient an amount equal to the deficiency or excess, as the case may be, with respect to the amount that the Recipient has borne if the amount of such refund or overpayment (including, for the avoidance of doubt, any interest or other amounts received with respect to such refund or overpayment) had been included originally in the determination of the amounts to be borne by Recipient pursuant to this Agreement, net of any additional Taxes and costs the Provider incurs or will incur as a result of the receipt of or in obtaining such refund or such overpayment.

Section 4.03. No Right to Set-Off. The Recipient shall timely pay the full amount of Service Charges and Reimbursement Charges and, except in the case of amounts disputed in good faith, shall not set-off, counterclaim or otherwise withhold any amount owed to the Provider under this Agreement on account of any obligation owed by the Provider to the Recipient.

ARTICLE V STANDARD FOR SERVICE

Section 5.01. Standard for Service. (a) The Provider agrees (i) to perform the Services with substantially the same nature, quality, standard of care and service levels at which the same or similar services were performed by or on behalf of the Provider (or its predecessor) during the 12 months prior to the Distribution Date or as otherwise provided in the Schedules or; and (ii) upon receipt of written notice from the Recipient identifying any outage, interruption or other failure of any Service, except as otherwise provided in the Schedules, to respond to such outage, interruption or other failure of such Service in a manner that is substantially similar to the manner in which such Provider or its Affiliates responded to any outage, interruption or other failure of the same or similar services prior to the Distribution Date. Except as otherwise provided in the Schedules, the Parties acknowledge that an outage, interruption or other failure of any Service shall not be deemed to be a breach of the provisions of this Section 5.01 so long as the applicable Provider complies with the foregoing clause (ii).

(b) Nothing in this Agreement shall require the Provider to perform or cause to be performed any Service to the extent the manner of such performance would be prohibited by or constitute a violation of applicable Law or any existing contract or agreement with a third party. If the Provider is or becomes aware of any potential violation on the part of the Provider, the Provider shall promptly send a written notice to the Recipient of any such potential violation. The Parties each agree to cooperate and use commercially reasonable efforts to obtain any

necessary third-party consents required under any existing contract or agreement with a third party to allow the Provider to perform or cause to be performed any Service in accordance with the standards set forth in Section 5.01(a) and/or the applicable Schedule. Any costs and expenses incurred by either Party in connection with obtaining any such third-party consent that is required to allow the Provider to perform or cause to be performed any Service shall be solely the responsibility of the Recipient. If, with respect to a Service, the Parties, despite the use of such commercially reasonable efforts, are unable to obtain a required third-party consent or the performance of such Service by the Provider would continue to constitute a violation of applicable Laws, the Provider shall use commercially reasonable efforts in good faith to provide such Services in a manner as closely as possible to the standards described in Section 5.01(a) and in the applicable Schedules.

Section 5.02. <u>Disclaimer of Warranties.</u> EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT (INCLUDING THE SCHEDULES AND EXHIBITS HERETO), THE PARTIES ACKNOWLEDGE AND AGREE THAT THE SERVICES ARE PROVIDED AS-IS, THAT EACH RECIPIENT ASSUMES ALL RISKS AND LIABILITY ARISING FROM OR RELATING TO ITS USE OF AND RELIANCE UPON THE SERVICES AND EACH PROVIDER, TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, MAKES NO REPRESENTATION OR WARRANTY WITH RESPECT THERETO. EXCEPT AS EXPRESSLY SET FORTH IN THIS AGREEMENT (INCLUDING THE SCHEDULES AND EXHIBITS HERETO), TO THE MAXIMUM EXTENT PERMITTED BY APPLICABLE LAW, EACH PROVIDER HEREBY EXPRESSLY DISCLAIMS ALL REPRESENTATIONS AND WARRANTIES REGARDING THE SERVICES, WHETHER EXPRESS OR IMPLIED, EITHER IN FACT OR BY OPERATION OF LAW, BY STATUTE OR OTHERWISE, INCLUDING ANY REPRESENTATION OR WARRANTY IN REGARD TO QUALITY, PERFORMANCE, NONINFRINGEMENT, COMMERCIAL UTILITY, MERCHANTABILITY OR FITNESS OF ANY SERVICE FOR A PARTICULAR PURPOSE.

Section 5.03. <u>Compliance with Laws and Regulations</u>. Each Party shall be responsible for its own compliance and its Subcontractors' compliance with any and all Laws applicable to its performance under this Agreement, including, without limitation, Laws applicable to a Receiving Party's obligations under <u>Section 3.05</u>. No Party shall knowingly take any action in violation of any such applicable Law that results in liability being imposed on the other Party.

ARTICLE VI LIMITED LIABILITY AND INDEMNIFICATION

Section 6.01. Consequential and Other Damages. Other than with respect to servicing and collections activities that are Navient Liabilities pursuant to Section 2.3(a)(i)(B) of the Separation and Distribution Agreement, notwithstanding anything to the contrary contained in the Separation and Distribution Agreement or this Agreement, except for breaches of confidentiality obligations (including, without limitation, breach of confidentiality obligations relating to customer information or a Receiving Party's breach of Section 3.05), or in the case of bad faith, gross negligence or willful misconduct (including the willful refusal to provide Services), no Party shall be liable to the other Party or any of its Affiliates or Representatives, whether in contract, tort (including negligence and strict liability) or otherwise, at law or equity, for any special, indirect, incidental, punitive or consequential damages whatsoever (including

lost profits or damages calculated on multiples of earnings approaches), which in any way arise out of, relate to or are a consequence of, the performance or nonperformance by such Party (including any Affiliates and Representatives and any unaffiliated third-party providers, in each case, providing any applicable Services) under this Agreement or the provision of, or failure to provide, any Services under this Agreement, including with respect to loss of profits, business interruptions or claims of customers, even if such Party has been advised of the possibility of such damages.

Section 6.02. Limitation of Liability.

- (a) Other than with respect to servicing and collections activities that are Navient Liabilities pursuant to Section 2.3(a)(i)(B) of the Separation and Distribution Agreement, and except as otherwise provided on a Schedule, the Liabilities of each Provider and its Affiliates and Representatives, collectively, under this Agreement for any act or failure to act in connection herewith (including the performance or breach of this Agreement), or from the sale, delivery, provision or use of any Services provided under or contemplated by this Agreement, whether in contract, tort (including negligence and strict liability) or otherwise, at law or equity, except for breaches of confidentiality obligations (including, without limitation, breach of confidentiality obligations relating to customer information or a Receiving Party's breach of Section 3.05) or in the case of bad faith, gross negligence or willful misconduct (including the willful refusal to provide Services), shall not exceed the total aggregate Service Charges (excluding for the avoidance of doubt any Reimbursement Charges) actually paid to such Provider pursuant to this Agreement; provided, that until the Operational Servicing Date, Navient's Liabilities under Statement of Work A (IT Hosting) to Schedule 1 shall be limited as provided in paragraph (b) below.
- (b) Notwithstanding the limitation set forth in paragraph (a) above and except for breaches of confidentiality obligations (including, without limitation, breach of confidentiality obligations relating to customer information or a Navient's breach of Section 3.05) or in the case of bad faith, gross negligence or willful misconduct (including the willful refusal to provide Services), the aggregate Liabilities of Navient and its Affiliates and Representatives, collectively, under this Agreement for Navient's act or failure to act under Statement of Work A (IT Hosting) to Schedule 1 (including the performance or breach of Statement of Work A) occurring prior to the Operational Servicing Date shall be limited as follows: (1) Navient shall not be liable for the first \$10,000,000 of Liabilities incurred by SLM BankCo attributable to such breach, and (2) if SLM BankCo's Liabilities exceed \$10,000,000, then Navient shall be liable for the next \$35,000,000 of such SLM BankCo Liabilities over \$10,000,000.

Section 6.03. Obligation To Re-perform; Liabilities. In the event of any breach of this Agreement by any Provider with respect to the provision of any Services (with respect to which the Provider can reasonably be expected to re-perform in a commercially reasonable manner), the Provider shall (a) promptly correct in all material respects such services at the request of the Recipient and at the sole cost and expense of the Provider and (b) subject to the limitations set forth in Sections 6.01 and 6.02, reimburse the Recipient and its Affiliates and Representatives for Liabilities attributable to such breach by the Provider. Except as provided on a Schedule, the remedy set forth in this Section

6.03 shall be the sole and exclusive remedy of the Recipient for any such breach of this Agreement solely in the event that re-performance is possible in a commercially reasonable manner. Any request for re-performance in accordance with this Section 6.03 by the Recipient must be in writing and specify in reasonable detail the particular error, defect or breach, and such request must be made no more than one month from the date such error, defect or breach becomes apparent or should have reasonably become apparent to the Recipient.

Section 6.04. <u>Indemnification</u>. Subject to the limitations and other provisions of this Agreement, the provisions of <u>Article IV</u> of the Separation and Distribution Agreement shall govern indemnification matters and procedures arising under or in connection with this Agreement.

Section 6.05. <u>Liability for Payment Obligations</u>. Nothing in this <u>Article VI</u> shall be deemed to eliminate or limit, in any respect, Navient's or SLM BankCo's express obligation in this Agreement to pay Service Charges and Reimbursement Charges for Services rendered in accordance with this Agreement.

Section 6.06. Exclusion of Other Remedies. Except as otherwise provided in the Schedules hereto, the provisions of Sections 6.03 of this Agreement shall, to the maximum extent permitted by applicable Law, be the sole and exclusive remedies of the Provider Indemnified Parties and the Recipient Indemnified Parties, as applicable, for any claim, loss, damage, expense or liability, whether arising from statute, principle of common or civil law, principles of strict liability, tort, contract or otherwise under this Agreement.

ARTICLE VII TERM AND TERMINATION

Section 7.01. <u>Term and Termination</u>. (a) This Agreement shall commence immediately upon the Distribution Date and shall terminate upon the earlier to occur of: (i) the last date on which either Party is obligated to provide any Service to the other Party in accordance with the terms of this Agreement (not to extend beyond the two-year anniversary of this Agreement) or (ii) the mutual written agreement of the Parties to terminate this Agreement in its entirety.

(b) (i) Without prejudice to a Recipient's rights with respect to a Force Majeure set forth in <u>Section 7.03</u>, and except as otherwise set forth in a Schedule, a Recipient may from time to time terminate this Agreement with respect to the entirety of any individual Service but not a portion thereof:

(A) for any reason or no reason, upon providing at least 30 days' prior written notice to the Provider; provided, however, that the Recipient shall pay to the Provider the necessary and reasonable documented out-of-pocket costs incurred in connection with the wind down of such Service other than any employee severance and relocation expenses, but including unamortized license fees and costs for equipment used to provide such Service, contractual obligations under agreements used to provide such Service, any breakage or termination fees and any other termination costs payable by the Provider with respect to any resources or pursuant to any

other third-party agreements that were used by the Provider to provide such Service (or an equitably allocated portion thereof, in the case of any such equipment, resources or agreements that also were used for purposes other than providing Services) ("<u>Termination Charges</u>"); or

- (B) if the Provider of such Service has failed to perform any of its material obligations under this Agreement with respect to such Service, unless the Provider has (x) cured such failure within the 30-day period following receipt by the Provider of a written notice of such failure from the Recipient or, (y) the Provider has made substantial progress to cure such failure and implemented a plan that results in a cure of such failure within 60 days of receipt by the Provider of the written notice from the Recipient.
- (ii) A Provider may terminate this Agreement with respect to one or more Services, in whole but not in part, at any time upon prior written notice to the Recipient if the Recipient has failed to perform any of its material obligations under this Agreement relating to such Services, including making payment of Service Charges and Reimbursement Charges when due, unless the Recipient has (x) cured such failure within the 30-day period following receipt by the Recipient of a written notice of such failure from the Provider or, (y) other than the Recipient's failure to pay Service Charges or Reimbursement Charges when due (which is governed by clause (x)), the Recipient has made substantial progress to cure such breach and implemented a plan that results in a cure of such breach within 60 days of receipt by the Recipient of the written notice from the Provider. The relevant Schedule shall be updated to reflect any terminated Service.
- (iii) In the event that any Service is terminated other than at the end of a month, the Service Charge associated with such Service shall be prorated appropriately. The Parties acknowledge that there may be interdependencies among the Services being provided under this Agreement that may not be identified on the applicable Schedules and agree that, if the Provider's ability to provide a particular Service in accordance with this Agreement is materially and adversely affected by the termination of another Service in accordance with this Section 7.01(b), then the Parties shall negotiate in good faith to amend the Schedule relating to such affected continuing Service, which amendment shall be consistent with the terms of, and the pricing methodology used for, comparable Services.
- (c) In connection with the termination of any Service identified on the Schedules as being subject to the provisions of this Section 7.01(c), if the Recipient reasonably determines that it will require such Service to continue beyond the date on which such Service is scheduled to terminate in the applicable Schedule, the Recipient may request the Provider to extend such Service for up to six months (unless otherwise provided in the Schedule applicable to such Service) (each, a "Service Extension") by written notice to the Provider no less than 90 days prior to the date of such scheduled termination, and the Parties shall use commercially reasonable efforts to comply with such Service Extension; provided, however, that (i) there shall be no more than one Service Extension with respect to each Service, (ii) the Provider shall not be obligated to provide such Service Extension if a third-party consent is required and cannot be

obtained by the Provider, (iii) each Service Extension shall be permissible under applicable Law, including bank and bank holding company regulations, and (iv) no Service Extension may extend past the conclusion of the Preferred Stock Period. Unless otherwise agreed, Service Charges relating to any Service Extension shall be increased by 5% of the Service Charge reflected on the Schedules with respect to such Service. Within five days following either Party's receipt of a written notice requesting a Service Extension, the Navient Services Manager and the SLM BankCo Services Manager shall in good faith (x) negotiate the terms of an amendment to the applicable Schedule, which amendment shall be consistent with the terms of, and the pricing methodology used for, the applicable Service (subject to the 5% increase of the applicable Service Charges, as described above); and (y) determine the costs and expenses (which shall not include any Service Charges payable under this Agreement), if any, that would be incurred by the Provider or the Recipient, as the case may be, in connection with the provision of such Service Extension, which costs and expenses shall be borne solely by the Recipient. Each such amended Schedule, as agreed to in writing by the Parties, shall be deemed part of this Agreement as of the date of such agreement and any Services provided pursuant to such Service Extensions shall be deemed "Services" provided under this Agreement, in each case subject to the terms and conditions of this Agreement. If a Recipient requests any subsequent extension of Services, such extension shall be subject to mutual agreement of the Parties and shall be provided at a mutually agreed market rate.

Section 7.02. Effect of Termination. Upon termination of any Service pursuant to this Agreement, the Provider of the terminated Service will have no further obligation to provide the terminated Service, and the relevant Recipient will have no obligation to pay any future Service Charges relating to any such Service; provided, however, that the Recipient shall remain obligated to the relevant Provider for (a) the Service Charges, Reimbursement Charges, and Transfer Taxes owed and payable in respect of Services provided prior to the effective date of termination and (b) any applicable Termination Charges payable in the event that the Recipient terminates such Service pursuant to Section 7.01(b)(i)(A). In connection with the termination of any Service, the provisions of this Agreement not relating solely to such terminated Service shall survive any such termination, and in connection with a termination of this Agreement, Article I, Article VI (including liability in respect of any indemnifiable Liabilities under this Agreement arising or occurring on or prior to the date of termination), Article VII, Article VIII, Section 3.05 (so long as the Receiving Party continues to hold or store Confidential Information) and liability for all due and unpaid Service Charges, Reimbursement Charges, Termination Charges and Transfer Taxes, shall continue to survive indefinitely.

Section 7.03. Force Majeure. Notwithstanding anything to the contrary herein, if either party is rendered unable, in whole or in part, by a Force Majeure to satisfy its obligations under this Agreement, such party shall not be deemed to have breached any such obligation upon delivery of written notice of such event to the other Party hereto, for so long as such party remains unable to perform such obligation as a result of such event. The following events shall not constitute force majeure events: (a) acts or omissions of subcontractors; and (b) delays or failures to the extent they could have been avoided or their impact mitigated through the use of commercially reasonable business continuity measures. If either Party is unable to materially perform its obligations under this Agreement and its performance is excused pursuant to this Section 7.03 for a period of 30 consecutive days, the other Party may terminate this Agreement upon at least three business days' written notice and shall not be required to pay any Termination Charges pursuant to Section 7.01(b).

ARTICLE VIII GENERAL PROVISIONS

Section 8.01. No Agency. Nothing in this Agreement shall be deemed in any way or for any purpose to constitute any Party an agent of an unaffiliated party in the conduct of such other Party's business. A Provider of any Service under this Agreement shall act as an independent contractor and not as the agent of the Recipient in performing such Service, maintaining control over its employees, its subcontractors and their employees and complying with all withholding of income at source requirements, whether federal, national, state, local or foreign.

Section 8.02. <u>Subcontractors</u>. A Provider may hire or engage one or more subcontractors to perform any or all of its obligations under this Agreement; <u>provided, however,</u> that (a) such Provider shall use the same degree of care in selecting any such subcontractor as it would if such contractor was being retained to provide similar services to the Provider and (b) such Provider shall in all cases remain primarily responsible for all of its obligations under this Agreement with respect to the scope of the Services, the standard for services as set forth in <u>Article V</u> and the content of the Services provided to the Recipient.

Section 8.03. Treatment of Confidential Information. (a) The Parties shall not, and shall cause all other persons providing Services or having access to information of the other Party that is confidential or proprietary (including, without limitation, Disclosing Party Customer Information, "Confidential Information") not to, disclose to any other person or use, except for purposes of this Agreement, any Confidential Information of the other Party; provided, however, that the Confidential Information may be used by such Party to the extent that such Confidential Information has been (i) in the public domain through no fault of such Party or any member of such Group or any of their respective Representatives, (ii) later lawfully acquired from other sources by such Party (or any member of such Party's Group) which sources are not themselves bound by a confidentiality obligation or (iii) independently generated without reference to any Confidential Information of the other Party; provided, further, that each Party may disclose Confidential Information of the other Party, to the extent not prohibited by applicable Law: (i) to its Representatives on a need-to-know basis in connection with the performance of such Party's obligations under this Agreement; (ii) in any report, statement, testimony or other submission required to be made to any Governmental Authority having jurisdiction over the disclosing Party; or (iii) in order to comply with applicable Law, or in response to any summons, subpoena or other legal process or formal or informal investigative demand issued to the disclosing Party in the course of any litigation, investigation or administrative proceeding. In the event that a Party becomes legally compelled (based on advice of counsel) by deposition, interrogatory, request for documents subpoena, civil investigative demand or similar judicial or administrative process to disclose any Confidential Information of the other Party, such disclosing Party shall provide the other Party with prompt prior written notice of such requirement, and, to the extent reasonably practicable, cooperate with the other Party (at such other Party's expense) to obtain a protective order or similar remedy to cause such Confidential Information not to be disclosed, including interposing all available objections thereto, such as objections based on settlement privilege. In the event that such protective order or other similar remedy is not obtained, the disclosing Party

shall furnish only that portion of the Confidential Information that has been legally compelled, and shall exercise its commercially reasonable efforts (at such other Party's expense) to obtain assurance that confidential treatment will be accorded such Confidential Information.

- (b) Upon the termination of this Agreement, either Party may request, in such Party's sole discretion, that all Confidential Information belonging to such Party either be promptly returned to such Party or promptly destroyed by the other Party, and in either case not retained by such other Party or its Affiliates or their respective Subcontractors in any form. Notwithstanding anything to the contrary contained herein, each Party's legal department may retain an archival copy of all or any portion of such Confidential Information to the extent required by applicable Law. The rights and obligations of the Parties regarding the non-disclosure and use of Confidential Information exchanged under this Agreement shall survive any return, retention or destruction of any Confidential Information.
 - (c) All Confidential Information shall remain the property of the Disclosing Party.
- (d) Each Party shall, and shall cause its Representatives to, protect the Confidential Information of the other Party by using the same degree of care to prevent the unauthorized disclosure of such as the Party uses to protect its own confidential information of a like nature, but in any event no less than a reasonable degree of care.
- (e) Each Party shall be liable for any failure by its respective Representatives to comply with the restrictions on use and disclosure of Confidential Information contained in this Agreement.
- (f) Each Party shall comply with all applicable local, state, national, federal and foreign privacy and data protection Laws that are or that may in the future be applicable to the provision of Services under this Agreement.
- Section 8.04. <u>Further Assurances</u>. Each Party covenants and agrees that, without any additional consideration, it shall execute and deliver any further legal instruments and perform any acts that are or may become necessary to effectuate this Agreement.

Section 8.05. <u>Dispute Resolution</u>. Any Dispute shall be resolved in accordance with the procedures set forth in <u>Article VII</u> and <u>Section 10.13</u> of the Separation and Distribution Agreement, which shall be the sole and exclusive procedures for the resolution of any such Dispute unless otherwise specified herein or in <u>Article VII</u> or <u>Section 10.13</u> of the Separation and Distribution Agreement.

Section 8.06. Notices. Except with respect to routine communications by the Navient Services Manager and the SLM BankCo Services Manager under Section 2.07, all notices, requests, claims, demands and other communications under this Agreement shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by facsimile or electronic transmission with receipt confirmed (followed by delivery of an original via overnight courier service) or by registered or certified mail (postage prepaid, return receipt requested) to the respective Parties at the following addresses (or at such other address for a Party as shall be specified in a notice given in accordance with this Section 8.06):

(i) if to Navient:

Navient Corporation Attention: Paul Mayer 2001 Edmund Halley Drive Reston, Virginia 20191

with copies to:

Navient Corporation Attention: General Counsel 2001 Edmund Halley Drive Reston, Virginia 20191

(ii) if to SLM BankCo:

SLM Corporation Attention: Paul Thome 175 South West Temple, 6th Floor Salt Lake City, UT 84101

with copies to:

SLM Corporation Attention: General Counsel 300 Continental Drive Newark, Delaware 19713:

Section 8.07. Severability. If any term or other provision of this Agreement is invalid, illegal or incapable of being enforced under any Law or as a matter of public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated by this Agreement is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in a mutually acceptable manner in order that the transactions contemplated by this Agreement be consummated as originally contemplated to the greatest extent possible.

Section 8.08. Entire Agreement. This Agreement, together with the documents referenced herein (including the Schedules, the Separation and Distribution Agreement and any other Ancillary Agreements) constitutes the entire agreement between the Parties and their respective Affiliates with respect to the subject matter hereof and supersedes all prior written and

oral and all contemporaneous oral agreements, negotiations, discussions, writings, understandings, commitments and conversations with respect to such subject matter, and there are no agreements or understandings between the Parties other than those set forth or referred to herein or therein.

Section 8.09. No Third-Party Beneficiaries. Except as provided in Article VI with respect to Provider Indemnified Parties and Recipient Indemnified Parties, this Agreement is for the sole benefit of the Parties and their permitted successors and assigns and nothing in this Agreement, express or implied, is intended to or shall confer upon any other Person, including any union or any employee or former employee of Navient or SLM BankCo, any legal or equitable right, benefit or remedy of any nature whatsoever, including any rights of employment for any specified period, under or by reason of this Agreement.

Section 8.10. Governing Law. This Agreement (and any claims or disputes arising out of or related to this Agreement or to the transactions contemplated by this Agreement or to the inducement of any Party to enter into this Agreement or the transactions contemplated by this Agreement, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall in all respects be governed by, and construed in accordance with, the Laws of the State of Delaware, including all matters of construction, validity and performance, in each case without reference to any conflict of Law rules that might lead to the application of the Laws of any other jurisdiction.

Section 8.11. Amendment; Waiver. No provision of this Agreement, including any Schedules to this Agreement, may be amended, supplemented or modified except by a written instrument making specific reference to this Agreement or any such Schedules to this Agreement, as applicable, signed by each of the Parties. Waiver by a Party of any default by the other Party of any provision of this Agreement shall not be deemed a waiver by the waiving Party of any subsequent or other default, nor shall it prejudice the rights of the waiving Party.

Section 8.12. Rules of Construction. Interpretation of this Agreement shall be governed by the following rules of construction: (a) words in the singular shall be held to include the plural and vice versa, and words of one gender shall be held to include the other gender as the context requires; (b) references to the terms Article, Section, paragraph, Exhibit and Schedule are references to the Articles, Sections, paragraphs, Exhibits and Schedules of this Agreement unless otherwise specified; (c) references to "\$" shall mean U.S. dollars; (d) the word "including" and words of similar import when used in this Agreement shall mean "including without limitation," unless otherwise specified; (e) the word "or" shall not be exclusive; (f) references to "written" or "in writing" include in electronic form; (g) provisions shall apply, when appropriate, to successive events and transactions; (h) the headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement; (i) Navient and SLM BankCo have each participated in the negotiation and drafting of this Agreement and if an ambiguity or question of interpretation should arise, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or burdening either Party by virtue of the authorship of any of the provisions in this Agreement or any interim drafts of this Agreement; (j) a reference to any Person includes such Person's successors and permitted assigns; (k) any reference to "days" means calendar days unless Business Days are expressly specified; and (l) when calculating the period of time before

which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded, and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day.

Section 8.13. <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, and by each Party in separate counterparts, each of which when executed shall be deemed to be an original but all of which taken together shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement by facsimile or portable document format (PDF) shall be as effective as delivery of a manually executed counterpart of this Agreement.

Section 8.14. <u>Assignability</u>. This Agreement shall not be assigned or transferred by operation of Law or otherwise without the prior written consent of Navient and SLM BankCo, except that each Party may:

- (a) assign all of its rights and obligations under this Agreement to any of its Subsidiaries; <u>provided</u>, that no such assignment shall release Navient or SLM BankCo, as the case may be, from any liability or obligation under this Agreement;
- (b) in connection with the divestiture of any Subsidiary or business of such Party that is a Recipient to an acquiror that is not a competitor of the Provider, assign to the acquiror of such Subsidiary or business its rights and obligations as a Recipient with respect to the Services provided to such divested Subsidiary or business under this Agreement; provided, however, that (i) no such assignment shall release Navient or SLM BankCo, as the case may be, from any liability or obligation under this Agreement; (ii) any and all costs and expenses incurred by either Party in connection with such assignment (including in connection with clause (iii) of this proviso) shall be borne solely by the assigning Party; and (iii) the Parties shall in good faith negotiate any amendments to this Agreement, including the Schedules and Exhibits hereto, that may be necessary or appropriate in order to assign such Services; and
- (c) in connection with the divestiture of any Subsidiary or business of such Party that is a Recipient to an acquiror that is a competitor of the Provider, assign to the acquiror of such Subsidiary or business its rights and obligations as a Recipient with respect to the Services provided to such divested Subsidiary or business under this Agreement; provided, however, that (i) no such assignment shall release Navient or SLM BankCo, as the case may be, from any liability or obligation under this Agreement; (ii) any and all costs and expenses incurred by either Party in connection with such assignment (including in connection with clause (iii) of this proviso) shall be borne solely by the assigning Party; (iii) the Parties shall in good faith negotiate any amendments to this Agreement, including the Schedules and Exhibits hereto, that may be necessary or appropriate in order to ensure that such assignment will not (x) materially and adversely affect the businesses and operations of each of the Parties and their respective Affiliates or (y) create a competitive disadvantage for the Provider with respect to an acquiror that is a competitor; and (iv) no Party shall be obligated to provide any such assigned Services to an acquiror that is a competitive disadvantage for such Party with respect to such acquiror.

Section 8.15. <u>Public Announcements</u>. From and after the Distribution Date, the Parties shall consult with each other before issuing, and give each other the opportunity to review and comment upon, that portion of any press release or other public statement that relates to the transactions contemplated by this Agreement, and shall not issue any such press release or make any such public statement prior to such consultation, except (a) as may be required by applicable Law, court process or by obligations pursuant to any listing agreement with any national securities exchange or national securities quotation system; or (b) as otherwise set forth in the Separation and Distribution Agreement.

Section 8.16. Non-Recourse. No past, present or future director, officer, employee, incorporator, member, partner, shareholder, Affiliate, agent, attorney or representative of either Navient or SLM BankCo or their Affiliates shall have any liability for any obligations or liabilities of Navient or SLM BankCo, respectively, under this Agreement or for any claims based on, in respect of, or by reason of, the transactions contemplated by this Agreement.

Section 8.17. <u>Audit Rights</u>. Each Party may, at such Party's sole cost and expense, from time to time (but no more frequently than once per year) audit the books and records of the other Party reasonably relating to the Services that the other Party is obligated to provide under this Agreement and to otherwise verify such Party's compliance with the terms and conditions of this Agreement. Each Party shall reasonably cooperate with any audit conducted by the other Party pursuant to this <u>Section 8.17</u>; <u>provided</u> that the Party conducting the audit shall give the other Party reasonable written notice of any audit, but in no event less than 30 days' notice, prior to such audit. Any audit pursuant to this <u>Section 8.17</u> shall be conducted during normal business hours for the applicable location in a manner that does not materially disrupt the operations of the other Party. The Parties acknowledge and agree that the audit rights and limitations provided in this <u>Section 8.17</u> are independent from the audit rights and limitations provided in <u>Section 3.05(e)</u>.

Section 8.18. <u>Title to Intellectual Property</u>. Except as expressly provided for under the terms of this Agreement (including the Schedules and Exhibits hereto), each Party acknowledges that it shall acquire no right, title or interest (including any license rights or rights of use) in any Intellectual Property which is owned or licensed by the other Party, by reason of the provision or receipt of the Services provided hereunder. Neither Party shall remove or alter any copyright, trademark, confidentiality or other proprietary notices that appear on any Intellectual Property owned or licensed by the other Party. Neither Partyshall attempt to decompile, translate, reverse engineer or make excessive copies of any Intellectual Property owned or licensed by the other Party, and each Party shall promptly notify the other Party of any such attempt, regardless of whether by the notifying Party or any third party, of which the notifying Partybecomes aware.

Section 8.19. Order of Precedence. The following shall apply to the extent of any conflict among the terms in the various documents within this Agreement (including Schedules and Exhibits):

(a) to the extent the conflicting provisions can reasonably be interpreted so that such provisions are consistent with each other, such consistent interpretations will prevail; and

(b) to the extent paragraph (a) above does not resolve such conflict, then the Schedules and Exhibits will prevail over a conflicting term in this Agreement.

[The remainder of this page is intentionally left blank.]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed on the date first written above by their respective duly authorized officers.

NAVIENT CORPORATION

By: /s/ John F. Remondi

Name: John F. Remondi

Title: Chief Executive Officer

SLM CORPORATION

By: /s/ Raymond Quinlan

Name: Raymond Quinlan
Title: Chief Executive Officer

Exhibit I Services Managers

Overall Services Managers

 Initial SLM BankCo Services Manager:
 Paul Thome

 Initial Navient Services Manager:
 Paul Mayer

Service Schedule Services Managers

Schedule 1 (Comprehensive IT):

 Initial SLM BankCo Services Manager:
 Mike Brannon

 Initial Navient Services Manager:
 Mary Jo Adams

Schedule 2 (Short-Term Shared Loan Servicing):

 Initial SLM BankCo Services Manager:
 Tim Hanrahan

 Initial Navient Services Manager:
 Mike Maier

Schedule 3 (Customer Experience):

 Initial SLM BankCo Services Manager:
 Tim Hanrahan

 Initial Navient Services Manager:
 Lisa Stashik

Schedule 4 (Trust Administration):

 Initial SLM BankCo Services Manager:
 Lance Welch

 Initial Navient Services Manager:
 Scott Booher

Schedule 5 (Facilities Services):

 Initial SLM BankCo Services Manager:
 Larry Zepp

 Initial Navient Services Manager:
 Joe Muffler

Schedule 6 (Third Party TSA Support):

 Initial SLM BankCo Services Manager:
 David O'Connell

 Initial Navient Services Manager:
 Mary Jo Adams

Schedule 7 (Government Relations):

 Initial SLM BankCo Services Manager:
 Rick Nelson

 Initial Navient Services Manager:
 Tim Morrison

Schedule 8 (Services Business Development Support and IT Services Support):

<u>Initial SLM BankCo Services Manager:</u> Kirk Etten

<u>Initial Navient Services Manager:</u>
Wendy Kincaid

Schedule 9 (Financial and HR Systems Support Services)

 Initial SLM BankCo Services Manager:
 Brian Carp

 Initial Navient Services Manager:
 Peter Strang

Exhibit II DISCLOSING PARTY SECURITY REQUIREMENTS

Access to Disclosing Party Confidential Information (including, without limitation, any non-public personal information or personally identifiable information of customers or consumers for which the Disclosing Party is the custodian) shall, subject to the other applicable provisions of the Agreement, be restricted to the Receiving Party with whom the Disclosing Party is contracting under the Agreement and the employees and, to the extent permitted by the Agreement, the permitted Subcontractors of the Disclosing Party (and their respective personnel) (each, including the Receiving Party, an "Accessing Party") who (1) shall provide the Services under the Agreement and (2) have been properly trained and instructed as to all obligations set forth in the Agreement and this Exhibit H with respect to the access and use of such Disclosing Party Confidential Information. Any Accessing Party hardware or software used to accesses, process, store or transmit any Disclosing Party Confidential Information shall be referred to herein as an "Affected Computing Device." In order to prevent unauthorized use, access, modification, or disruption to Disclosing Party Confidential Information, information security measures shall be implemented with respect to any Affected Computing Device.

In addition to the security of Affected Computing Devices, this Exhibit must take a broader view of security. The security of these devices is dependent upon the Accessing Party maintaining a secure infrastructure that implements an array of industry standard controls. Therefore, this Exhibit covers the infrastructure necessary to protect the Affected Computing Device. This does not cover the Accessing Party's entire infrastructure and controls regime, but only the slice which assures the security of the Affected Computing Devices.

Access Control

- 1. Access to Affected Computing Devices shall be configured with the concept of "Least Privilege" enforced. Access to Affected Computing Devices shall be granted only to those individuals who shall have it in order to provide the Services.
- 2. Access to Affected Computing Devices shall be granted in line with formally documented user administration processes that are designed to ensure:
 - a. All users granted access to Affected Computing Devices shall be explicitly approved by appropriate Accessing Party management;
 - b. Provisioning and de-provisioning of access to critical Affected Computing Devices is reviewed at regularly scheduled intervals by appropriate Accessing Party management.
- 3. All Affected Computing Devices shall be configured to require users to periodically change their passwords. Password length, complexity and reuse shall adhere to industry standard practices.

- 4. Accessing Party shall limit access by means of wireless networking to Disclosing Party Confidential Information to the minimum required situations. Where such wireless access is necessary, such Accessing Party shall require that authentication, encryption, and security for the wireless network meets generally accepted industry standard practices.
- 5. Accessing Party authentication credentials that are no longer required or that are no longer authorized to obtain access to Disclosing Party Confidential Information from Affected Computing Devices shall be (promptly disabled.
- 6. In the event that Accessing Party is provided with Disclosing Party administered credentials or devices to permit access to Disclosing Party Confidential Information, Accessing Party shall promptly notify the Disclosing Party, within 24 hours in a manner that is mutually agreed upon by the parties, of a personnel change whereby an individual no longer requires Disclosing Party administered authentication credentials to Disclosing Party Confidential Information.
- 7. Network protection mechanisms meeting industry standard practices shall be implemented for all Accessing Party company networks accessing Disclosing Party Confidential Information.
- 8. Accessing Party's personnel shall exercise restraint with their access privileges to Disclosing Party Confidential Information, shall not attempt to circumvent or subvert any security measures and not use their access for anything other than the provision of Services.
- 9. Access to Disclosing Party Confidential Information over the public Internet shall be restricted.

Logging & Monitoring

- 10. All Affected Computing Devices shall be configured to audit relevant security events using industry standard practices.
- 11. Audit and event logs for Affected Computing Devices shall be adequately protected to prevent contents from being modified or deleted in an unauthorized fashion.
- 12. System logs are to be analyzed for suspicious activity.

Patching & Configuration Management

- 13. Accessing Party shall maintain a security patching/update validation process that is designed to ensure that all Affected Computing Devices are patched in a timely manner consistent with industry standard practices.
- 14. Accessing Party shall follow industry standard practices to implement secure system configurations.
- 15. Accessing Party change control management procedures shall be documented and followed.

Storage & Media Protection

- 16. Servers, enterprise data storage devices, backup tapes and media, and other Affected Computing Devices used to support network communications shall be located in a secure and restricted access location within Accessing Party facilities, or approved Subcontractor facilities,
- 17. All backup media, including tapes, sent off site that contains Disclosing Party Confidential Information shall be encrypted using industry standard practices.
- 18. When Disclosing Party Confidential Information is required to be destroyed or disposed per the terms of the Agreement, such Disclosing Party Confidential Information, whether in paper, electronic or other form requires secure disposal or destruction. These measures shall be in line with industry standard practices.

Physical & Environmental Protection

- 19. Accessing Party Facilities at which Affected Computing Devices are stored or maintained shall have appropriate controls implemented that restrict physical access to only authorized personnel. Visitor access to such facilities shall be based on authentication of visitor identity and visitors shall be escorted.
- 20. Data centers housing Disclosing Party Confidential Information are required to have controlled access with working security cameras. Data center access control lists shall be kept current.
- 21. Accessing Party shall provide at least 90 days' advance notification to Disclosing Party prior to relocation or transfer of Accessing Party's datacenter housing Disclosing Party Confidential Information.

Vulnerability Assessments

- 22. Accessing Party shall maintain a vulnerability management and assessment program based on industry standard practices that frequently assesses Affected Computing Devices and mitigates or eliminates vulnerabilities.
- 23. Routine network and database scans shall be scheduled. The scan results shall be analyzed and vulnerabilities identified and remediated within a timeframe commensurate with the relative risk.
- 24. Accessing Party shall engage, at its expense, an unrelated security firm to perform an annual penetration test of Affected Computing devices. Accessing Party shall implement security controls and practices to mitigate risks identified during such an assessment.

Encryption

25. When Disclosing Party Confidential Information is being accessed, or transmitted, over the Internet or via a public switched network, the communications session shall utilize a secure transport mechanism meeting industry standard for encryption. All laptops, hand-held devices and removable storage devices, shall utilize full disk encryption meeting industry standards for encryption.

Malware Protection

26. Affected Computing Devices shall be appropriately and reasonably protected against malicious software. All such Affected Computing Devices shall be configured with up-to-date anti-virus.

Security Awareness & Training

27. All Accessing Party personnel who have access to Disclosing Party Confidential Information are required to successfully complete initial security awareness training, including affirmative acknowledgement of their security and privacy responsibilities, and annual refresher training thereafter

Host-Based Security - End Point Computing

28. Only devices provided by or under the management of Accessing Party personnel or by Disclosing Party may be used to access Disclosing Party Confidential Information. Public resources such as hotel PC kiosks, or other public-access terminals such as those available in malls and airports, may not be used for this purpose.

Policies and Procedures

- 29. Upon request, but no more frequently than annually, Accessing Party shall permit Disclosing Party to review its information security, physical security, and privacy policy at the Accessing Party's facility.
- 30. Accessing Party shall develop, implement, and maintain logical network diagrams and security documentation for the Affected Computing Devices to provide an overview of the security requirements and a description of the security controls in place.

Change Management

31. Accessing Party shall utilize a change management process based on industry standards that is designed to ensure that Accessing Party information security personnel have insight and approval over changes affecting security devices (e.g., firewalls, VPNs, IPS) and systems handling authentication, authorization, and auditing.

Data Loss Prevention

32. Accessing Party shall maintain an approach to data loss prevention in-line with industry standards to mitigate the risk of unauthorized data disclosure from applications and network infrastructure.

Asset Inventory

33. Accessing Party shall maintain an asset inventory of all critical Affected Computing Devices that access, store, process or transmit Disclosing Party Confidential Information.

Remote Access

- 34. In the event Accessing Party requires remote access to Disclosing Party's network:
 - a. Accessing Party's use of individually assigned Disclosing Party remote access credentials and authentication devices are for the sole use of that individual and not to be shared with anyone else for any reason.
 - b. Accessing Party remote access to Disclosing Party's network will require a Disclosing Party approved or Disclosing Party-provided multi-factor authentication.

Background Investigations

- 35. Accessing Party on-boarding process for personnel that will have access to Disclosing Party Confidential Information shall include, at its expense, an industry standard background investigation.
- 36. Accessing Party shall not assign any individual to Disclosing Party who (1) has been convicted of a felony of any nature or a misdemeanor of violence, theft or fraud or a crime involving dishonesty or breach of trust; (2) has an unverifiable SSN, address, employment, or education; or (3) appears on the OFAC list.
- 37. Accessing Party shall promptly notify Disclosing Party in the event Accessing Party becomes aware that any individual assigned to Disclosing Party is under investigation or arrested for, or convicted of, a felony of any nature or a misdemeanor of violence, theft or fraud or a crime involving dishonesty or breach of trust.

Systems Development and Maintenance

38. Accessing Party will annually conduct appropriate application security reviews for critical applications and prior to the promotion of production changes to critical application.

Segregation of Duties

39. Accessing Party shall safeguard Disclosing Party Confidential Information by employing commercially reasonable practices, techniques and/or technologies, to implement segregation of duties.

Cloud Service Delivery1

40. Accessing Party shall document controls used to maintain logical separation of data (prevent inadvertent release of data) in multi-tenant environments.

¹ Adapted from the Cloud Security Alliance Cloud Controls Matrix Version 1.1 (2010), http://www.cloudsecurityalliance.org/cm.html

Exhibit III Key Employees

A. Navient Key Employees

- 1. Pat Lawicki
- 2. Cheri Dayton
- 3. Jon Jones
- 4. Matt Anderson
- 5. Michael Castagna
- 6. Jeff Dossman
- 7. Rich Jackson
- 8. Brenda Nethery
- 9. Mark Perrault
- 10. Carol Swartz

B. SLM BankCo Key Employees

- 1. Dan Kennedy
- 2. Jerry Archer
- 3. Mike Brannon
- 4. Mike Migliore
- 5. Mike Bandy
- 6. John Sullivan

Exhibit IV Excluded SMI Projects

- 1. Regulatory, legislative, and production ticket support
- 2. FDR (by FDR), CLASS, CLASS ED, Collection systems development
- 3. Collection Platform Replacement
- 4. Telephony Replacement Projects, including I3/IVR
- 5. xPressions Deployment
- 6. IT Cost Reduction Initiatives
- 7. IT On-going Upgrades/Deployments/Maintenance
- 8. Business activities included in the Navient business plan presented to the SLM Board in October 2013.
- 9. Initiatives to support ED Servicing including default prevention and customer experience initiatives to support ED Scorecard improvement

Schedule 1
Comprehensive Information Technology Services

[See attached.]

Schedule 2
Short-Term Shared Loan Servicing Services

[See attached.]

-44-

<u>Schedule 3</u> Customer Experience Services

[See attached.]

<u>Schedule 4</u> Trust Administration Services

[See attached.]

-46-

<u>Schedule 5</u> Facilities Services

[See attached.]

-47-

<u>Schedule 6</u> Third Party TSA Support Services

[See attached.]

-48-

<u>Schedule 7</u> Government Relations Services

[See attached.]

-49-

<u>Schedule 8</u> Services Business Development Support and IT Services

[See attached.]

Schedule 9
Financial and HR Systems Support Services

[See attached.]

EMPLOYEE MATTERS AGREEMENT

between

NEW BLC CORPORATION

and

NAVIENT CORPORATION

dated as of

April 28, 2014

TABLE OF CONTENTS

ARTICLE I DEFINITIONS		PAGE 2
Section 1.1 Section 1.2	Definitions Interpretation	2 8
ARTICLE II ASSIGNMENT OF EMPLOYEES		10
Section 2.1 Section 2.2 Section 2.3 Section 2.4	Active Employees Former Employees Employment Law Obligations Employee Records	10 11 11
ARTICLE III EQU	ITY AND INCENTIVE COMPENSATION PLANS	13
Section 3.1 Section 3.2 Section 3.3 Section 3.4 Section 3.5 Section 3.6 Section 3.7 Section 3.8 Section 3.9 ARTICLE IV GEN Section 4.1 Section 4.2	General Principles Restricted Stock Restricted Stock Units Stock Options Performance Stock Units Section 16(b) of the Exchange Act; Code Sections 162(m) and 409A Certain Bonus Payments ESPP Registration Statements; Blackouts JERAL PRINCIPLES FOR ALLOCATION OF LIABILITIES General Principles Sponsorship and/or Establishment of NewCo Plans	13 15 15 16 17 18 19 20 20
Section 4.2 Section 4.3 Section 4.4	Sponsorship and/or Establishment of NewCo Plans Service Credit Plan Administration	21 22 23
ARTICLE V COLI POLICIES		23
Section 5.1	COLI	23
ARTICLE VI THRIFT PLANS		24
Section 6.1 Section 6.2 Section 6.3 Section 6.4 Section 6.5	General Principles Treatment of SLM BankCo Common Stock and NewCo Common Stock Transfer of Accounts Supplemental 401(k) Plan Deferred Compensation Plans	24 24 25 25 26

i

ARTICLE VII WEL	EADE DI ANG	26
ARTICLE VII WEL		26
Section 7.1 Section 7.2 Section 7.3 Section 7.4 Section 7.5 Section 7.6	Establishment of NewCo Welfare Plans Transitional Matters Under NewCo Welfare Plans Continuity of Benefits, Benefit Elections and Beneficiary Designations Insurance Contracts Third-Party Vendors Claims Experience	26 27 28 29 29 29
	IEFIT ARRANGEMENTS	29
ARTICLE IX WORI	KERS' COMPENSATION AND UNEMPLOYMENT COMPENSATION	29
Section 9.1 Section 9.2	Workers' Compensation Unemployment Compensation	29 30
ARTICLE X RETE	NTION, SEVERANCE AND OTHER MATTERS	30
Section 10.1 Section 10.2 Section 10.3 Section 10.4 Section 10.5 Section 10.6	Retention Agreements Severance Accrued Time Off Leaves of Absence Director Programs Restrictive Covenants in Employment and Other Agreements	30 30 31 31 31
ARTICLE XI LEGA	ACY LIABILITY	32
Section 11.1 Section 11.2	Legacy Pension Plan Scholarships	32 32
ARTICLE XII GENI	ERAL PROVISIONS	32
Section 12.1 Section 12.2 Section 12.3 Section 12.4 Section 12.5 Section 12.6	Preservation of Rights to Amend Confidentiality Administrative Complaints/Litigation Reimbursement and Indemnification Costs of Compliance with Agreement Fiduciary Matters	32 32 32 33 33 33
Section 12.7 Section 12.8 Section 12.9 Section 12.10 Section 12.11 Section 12.12	Entire Agreement Binding Effect; No Third-Party Beneficiaries; Assignment Amendment; Waivers Remedies Cumulative Notices Counterparts	33 34 34 34 34 34
Section 12.13 Section 12.14 Section 12.15 Section 12.16 Section 12.17	Severability Governing Law Performance Construction Effect if Distribution Does Not Occur	35 35 35 35 35 35

EMPLOYEE MATTERS AGREEMENT

THIS EMPLOYEE MATTERS AGREEMENT dated as of April 28, 2014, between New BLC Corporation, a Delaware corporation ("SLM BankCo"), and Navient Corporation, a Delaware corporation ("NewCo"). SLM BankCo and NewCo are sometimes referred to herein, individually, as a "Party," and, collectively, as the "Parties." Capitalized terms used herein and not otherwise defined shall have the respective meanings ascribed to such terms in Article I hereof

RECITALS

WHEREAS, the Board of Directors of SLM Corporation, a Delaware Corporation ("Existing SLM"), has determined that it would be in the best interests of its stockholders that its education loan management business be separated from its consumer banking business (the "Separation");

WHEREAS, the Separation will involve an internal corporate reorganization of Existing SLM and its subsidiaries, pursuant to which all of the assets and liabilities associated with Existing SLM's loan management, services and asset recovery business will be transferred to NewCo and its subsidiaries, and those assets and liabilities (to the extent not assumed by NewCo) associated with its consumer banking business will remain with or be transferred to SLM BankCo and its subsidiaries;

WHEREAS, as part of the internal corporation reorganization, (x) SLM BankCo, which is a newly formed holding company, will become the publicly traded successor to Existing SLM by means of a holding company merger effected pursuant to Section 251(g) of the Delaware General Corporation Law (the "DGCL"), and (y) Existing SLM will become a subsidiary of NewCo and retain directly or indirectly the assets and liabilities associated with Existing SLM's businesses other than the consumer banking business to be retained by or transferred to SLM BankCo and its subsidiaries;

WHEREAS, immediately following the internal corporate reorganization, SLM BankCo will own all of the issued and outstanding shares of NewCo common stock, which it will distribute to SLM BankCo stockholders, on a *pro rata* basis, as of a record date previously determined by the Existing SLM Board of Directors (the "**Distribution**"), thereby completing the Separation;

WHEREAS, the Parties and Existing SLM are entering into the Separation Agreement, which sets forth the agreement of the Parties and Existing SLM regarding the internal corporate reorganization and other transactions necessary to effect the Separation, including the Distribution;

WHEREAS, the Separation Agreement provides, among other things, subject to the terms and conditions thereof, for the execution and delivery of various agreements, including this Agreement, in order to facilitate and provide for the Separation, provide a framework for the relationship of SLM BankCo and NewCo after the Separation and provide for the allocation between SLM BankCo and NewCo of all of the assets, liabilities, and obligations of Existing SLM and its subsidiaries attributable to periods prior to and after the Separation; and

WHEREAS, this Agreement is being entered into in order to allocate between SLM BankCo and NewCo certain assets, liabilities and obligations, and related responsibilities, with respect to employees, employee compensation, benefit plans and programs, and to set forth the agreement of SLM BankCo and NewCo with respect to the matters set forth herein.

NOW, THEREFORE, in consideration of the premises and the mutual covenants and agreements herein contained, the Parties, intending to be legally bound, agree as follows:

ARTICLE I

DEFINITIONS

Definitions. As used in this Agreement, the following terms shall have the meanings set forth in this Section 1.1:

- (a) "2014 Pre-Spin Bonus" has the meaning set forth in Section 3.7.
- (b) "Adjusted SLM BankCo RSA" has the meaning set forth in Section 3.2(a).
- (c) "Adjusted SLM BankCo RSU" has the meaning set forth in Section 3.3(b).
- (d) "Adjusted NewCo RSU" has the meaning set forth in Section 3.3(b).
- (e) "Affiliate" shall mean, when used with respect to a specified Person, a Person that, directly or indirectly, through one or more intermediaries, controls, is controlled by or is under common control with such specified Person. For the purpose of this definition, "control" (including with correlative meanings, "controlled by" and "under common control with"), when used with respect to any specified Person shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or other interests, by contract, agreement, obligation, indenture, instrument, lease, promise, arrangement, release, warranty, commitment, undertaking or otherwise. It is expressly agreed that, from and after the Effective Time, for purposes of this Agreement and the Ancillary Agreements, (a) no member of the NewCo Group shall be deemed an Affiliate of any member of the SLM BankCo Group and (b) no member of the SLM BankCo and SMI as "affiliates" for purposes of federal customer data privacy laws, including the Gramm-Leach-Bliley Act during the Preferred Stock Period (as defined in the Separation Agreement).
- (f) "Agreement" means this Employee Matters Agreement together with all Schedules hereto and all amendments, changes and supplements hereto and thereto entered into in accordance with Section 12.9.
 - (g) "Ancillary Agreements" has the meaning set forth in the Separation Agreement.
- (h) "Benefit Arrangement" means any contract, agreement, policy, practice, program, plan, trust or arrangement (other than any deferred compensation, profit sharing, bonus, stock-based compensation or other form of incentive compensation) providing for benefits,

perquisites or compensation of any nature to any Employee, or to any family member, dependent or beneficiary of any such Employee, including travel and accident insurance, tuition reimbursement, vacation, sick, personal or bereavement days, and holidays.

- (i) "COBRA" means the U.S. Consolidated Omnibus Budget Reconciliation Act of 1985, as codified at Part 6 of Subtitle B of Title I of ERISA and at Code Section 4980B.
 - (j) "Code" means the Internal Revenue Code of 1986.
 - (k) "Code Sections" means a section of the Code.
 - (1) "Confidential Information" has the meaning set forth in the Separation Agreement.
 - (m) "Deferred Compensation Plan" has the meaning set forth in Section 6.5.
 - (n) "DGCL" has the meaning set forth in the recitals to this Agreement.
 - (o) "Distribution" has the meaning set forth in the recitals to this Agreement.
 - (p) "Distribution Date" has the meaning set forth in the Separation Agreement.
 - (q) "Effective Time" means 4 p.m., Eastern Standard Time, on the Distribution Date.
 - (r) "Employee" means any SLM BankCo Employee, NewCo Employee or Former NewCo Employee.
 - (s) "Employee Records" has the meaning set forth in Section 2.4(a).
 - (t) "ERISA" means the U.S. Employee Retirement Income Security Act of 1974.
 - (u) "ESPP" means the Amended and Restated Sallie Mae Employee Stock Purchase Plan.
 - (v) "Exchange Act" means the Securities Exchange Act of 1934.
 - (w) "Existing SLM" has the meaning set forth in the preamble to this Agreement.
 - (x) "FMLA" means the U.S. Family and Medical Leave Act.
 - (y) "Former NewCo Employee" has the meaning set forth in Section 2.2(a).
 - (z) "Group" means collectively the Sallie Mae Group, NewCo Group and the SLM BankCo Group.
 - (aa) "Initial Employment Date" means, for any person, the first date such person was employed by any member of the Sallie Mae Group.
 - (bb) "IRS" means the U.S. Internal Revenue Service.

- (cc) "Legacy Pension Liabilities" has the meaning set forth in Section 11.1.
- (dd) "Legacy Performance Stock Units" has the meaning set forth in Section 3.5(a).
- (ee) "Legacy Sallie Mae Employee" has the meaning set forth in Section 2.2(a).
- (ff) "Legacy SLM BankCo Option" has the meaning set forth in Section 3.4(a).
- (gg) "Measurement Date" has the meaning set forth in Section 3.5(a).
- (hh) "Merger" has the meaning set forth in the Separation Agreement.
- (ii) "Merger Date" means the date on which the Merger becomes effective in accordance with Section 251 and Section 103 under the DGCL.
- (jj) "NASDAQ" means the NASDAQ Global Select Market.
- (kk) "NewCo" has the meaning set forth in the preamble to this Agreement.
- (II) "NewCo Benefit Arrangement" means any Benefit Arrangement sponsored by a member of the NewCo Group.
- (mm) "NewCo Common Stock" means the common stock of NewCo, par value \$0.01 per share.
- (nn) "NewCo Director" means an individual who is a non-employee director of NewCo following the Distribution.
- (00) "NewCo Employee" means any individual who is employed by a member of the NewCo Group on the Distribution Date.
- (pp) "NewCo Entity" means any (x) member of the NewCo Group (together with each current and former, direct or indirect, Subsidiary of any such member (and of any such former Subsidiary)) and (y) any former Subsidiary of Existing SLM, or line of business of a Subsidiary of Existing SLM, that was sold or otherwise disposed of prior to the Distribution Date. For purposes of this definition, all lines of business that are treated as "discontinued operations" in the consolidated financial statements of NewCo shall be deemed to have been part of a NewCo Entity. For purposes of this definition, no SLM BankCo Entity shall be considered a NewCo Entity. The Parties acknowledge that this term is defined differently in the Separation Agreement.
 - (qq) "NewCo Equity Awards" means NewCo RSAs, NewCo RSUs or Post-Distribution NewCo Options.
- (rr) "NewCo ESPP" means an employee stock purchase plan sponsored by NewCo with terms substantially similar to those of the ESPP as of the Distribution Date.
 - (ss) "NewCo Group" means, collectively, NewCo and each NewCo Subsidiary.

- (tt) "NewCo New Equity Plan" means the Navient Corporation 2014 Omnibus Incentive Plan, under which the NewCo equity-based awards described in Article III shall be issued.
 - (uu) "NewCo Reimbursement Account" has the meaning set forth in Section 7.3(b).
 - (vv) "NewCo RSU" has the meaning set forth in Section 3.3(a).
 - (ww) "NewCo Subsidiary" means any direct or indirect Subsidiary of NewCo, as of the Distribution Date.
 - (xx) "NewCo Supplemental 401(k) Plan" has the meaning set forth in Section 6.4.
 - (yy) "NewCo Thrift Plan" has the meaning set forth in Section 6.1.
 - (zz) "NewCo Thrift Plan Beneficiaries" has the meaning set forth in Section 6.1.
- (aaa) "NewCo Welfare Plan" means any Welfare Plan sponsored or maintained by any one or more members of the NewCo Group on the Distribution Date.
 - (bbb) "NewCo Welfare Plan Participants" has the meaning set forth in Section 7.1.
 - (ccc) "Participating NewCo Employers" has the meaning set forth in Section 7.1.
 - (ddd) "Participation Period" has the meaning set forth in Section 7.3(b).
 - (eee) "Party" or "Parties" has the meaning set forth in the preamble to this Agreement.
- (fff) "Person" means an individual, a partnership, a corporation, a limited liability company, an association, a joint stock company, a trust, a joint venture, a union, an unincorporated organization or a governmental entity or any department, agency or political subdivision thereof.
- (ggg) "Post-Distribution NewCo Share Price" means the volume-weighted average of the "when issued" trading price on NASDAQ of a share of NewCo Common Stock on the five trading days ending on the Distribution Date.
 - (hhh) "Post-Distribution NewCo Option" has the meaning set forth in Section 3.4(a).
 - (iii) "Post-Distribution SLM BankCo Option" has the meaning set forth in Section 3.4(a).
- (jjj) "Post-Distribution SLM BankCo Option Price Target" means, as to a particular Post-Distribution SLM BankCo Option that is to include a Price Target, the number determined by multiplying the Price Target of the Legacy SLM BankCo Option in respect of which such Post-Distribution SLM BankCo Option is to be granted by a fraction, the numerator of which is the Post-Distribution SLM BankCo Share Price and the denominator of which is the Pre-Distribution SLM BankCo Share Price.

- (kkk) "Post-Distribution SLM BankCo Share Price" means the volume-weighted average of the "ex-dividend" trading price of a share of SLM BankCo Common Stock on the five trading days ending on the Distribution Date.
- (Ill) "Pre-Distribution SLM BankCo Share Price" means the sum of the Post-Distribution SLM BankCo Share Price and the Post-Distribution NewCo Share Price.
 - (mmm) "Price Target" means any SLM BankCo Stock Price (or Prices) that is (or are) specified as an exercisability trigger in a SLM BankCo Option.
 - (nnn) "Privacy Contract" means any contract entered into in connection with applicable privacy protection laws or regulations.
- (000) "Registration Statement Effectiveness Date" means the first date on which the applicable registration statement on Form S-8 (or other appropriate form) contemplated by Section 3.9 shall be effective under the Securities Act.
 - (ppp) "Post-Distribution NewCo Option" has the meaning set forth in Section 3.4(a).
- (qqq) "Post-Distribution NewCo Option Price Target" means, as to a particular Post-Distribution NewCo Option that is to include a Price Target, the number determined by multiplying the Price Target of the Legacy SLM BankCo Option in respect of which such Post-Distribution NewCo Option is to be granted multiplied by (ii) a fraction, the numerator of which is the Post-Distribution NewCo Share Price and the denominator of which is the Pre-Distribution SLM BankCo Share Price.
 - (rrr) "Post-Distribution SLM BankCo Option" has the meaning set forth in Section 3.4(a).
- (sss) "Sallie Mae Group" means Existing SLM and its predecessors and every Subsidiary of Existing SLM and such predecessors in existence at any time prior to the Merger Date.
 - (ttt) "SEC" means the Securities and Exchange Commission.
 - (uuu) "Securities Act" means the Securities Act of 1933.
 - (vvv) "Separation" has the meaning set forth in the recitals to this Agreement.
- (www) "Separation Agreement" means the Separation and Distribution Agreement, dated as of April 28, 2014, among the Parties and Existing SLM, together with all Schedules and all amendments and supplements thereto.
 - (xxx) "SLM BankCo" has the meaning set forth in the preamble to this Agreement.
- (yyy) "SLM BankCo Benefit Arrangement" means any Benefit Arrangement sponsored or maintained by a member of the SLM BankCo Group on the Distribution Date. For purposes of this definition, a Benefit Arrangement sponsored or maintained by any one or more members of the SLM BankCo Group includes any former Benefit Arrangement of Existing SLM

that has been assumed by one or more members of the SLM BankCo Group (expressly or by the terms of the applicable Benefit Arrangement). Effective on the Merger Date, SLM BankCo shall assume each Benefit Arrangement then sponsored or maintained by Existing SLM, unless otherwise provided in the Agreement.

- (zzz) "SLM BankCo Common Stock" means (i) for any period prior to the Merger Date, the common stock of Existing SLM, par value \$0.20 per share, and (ii) for any period from and following the Merger Date, the common stock of SLM BankCo, par value \$0.20 per share.
 - (aaaa) "SLM BankCo Deferred Compensation Plan" has the meaning set forth in Section 6.5.
 - (bbbb) "SLM BankCo Director" means an individual who is a non-employee director of SLM BankCo following the Distribution Date.
 - (cccc) "SLM BankCo Employee" means any individual who is employed by a member of the SLM BankCo Group on the Distribution Date.
- (dddd) "SLM BankCo Entity" means any member of the SLM BankCo Group. For purposes of this definition, no NewCo Entity shall be considered a SLM BankCo Entity. The Parties acknowledge that this term is defined differently in the Separation Agreement.
- (eeee) "SLM BankCo Equity Awards" means Legacy SLM BankCo Options, SLM BankCo Options, SLM BankCo RSAs, SLM BankCo RSUs or Post-Distribution SLM BankCo Options.
- (ffff) "SLM BankCo Group" means, collectively, SLM BankCo and each SLM BankCo Subsidiary. For purposes of this definition, no member of the NewCo Group shall be deemed a member of the SLM BankCo Group.
- (gggg) "SLM BankCo Legacy Equity Plan" means any equity plan sponsored or maintained by the SLM BankCo Group immediately prior to the Distribution Date. For purposes of this definition, an equity plan sponsored or maintained by SLM BankCo includes a former equity plan of Existing SLM that has been assumed by SLM BankCo (expressly or by the terms of the applicable equity plan). Effective on the Merger Date, SLM BankCo shall assume each equity plan then sponsored or maintained by Existing SLM, unless otherwise provided in the Agreement.
- (hhhh) "SLM BankCo Options" means options to purchase shares of SLM BankCo Common Stock granted pursuant to any of the SLM BankCo Legacy Equity Plans, including options granted prior to the Merger Date by Existing SLM and assumed by SLM BankCo.
 - (iiii) "SLM BankCo Rabbi Trust" has the meaning set forth in Section 5.1.
- (jjjj) "SLM BankCo RSAs" means unvested restricted stock awards issued under any of the SLM BankCo Legacy Equity Plans, including awards issued by Existing SLM prior to the Merger Date and assumed by SLM BankCo.

- (kkkk) "SLM BankCo RSUs" means restricted stock units or deferred stock units issued under any of the SLM BankCo Legacy Equity Plans that are not subject to performance conditions, including units issued by Existing SLM prior to the Merger Date and assumed by SLM BankCo.
- (IllI) "SLM BankCo Subsidiary" means any Subsidiary of SLM BankCo as of the Distribution Date. For purposes of this definition, no NewCo Subsidiary shall be considered a SLM BankCo Subsidiary.
 - (mmmm) "SLM BankCo Thrift Plan" means the Sallie Mae 401(k) Savings Plan.
 - (nnnn) "SLM BankCo Thrift Plan Beneficiaries" has the meaning set forth in Section 6.1.
- (0000) "SLM BankCo Welfare Plan" means any Welfare Plan sponsored or maintained by any one or more members of the SLM BankCo Group on the Distribution Date. For purposes of this definition, a Welfare Plan sponsored or maintained by any one or more members of the SLM BankCo Group includes former Welfare Plans of Existing SLM that have been assumed by a member of the SLM BankCo Group (expressly or by the terms of the applicable Welfare Plan). Effective on the Merger Date, SLM BankCo shall assume each Welfare Plan then sponsored or maintained by Existing SLM, unless otherwise provided in the Agreement.
- (pppp) "Subsidiary" means, with respect to any specified Person, any corporation, partnership, limited liability company, joint venture or other organization, whether incorporated or unincorporated, of which at least a majority of the securities or interests having by the terms thereof ordinary voting power to elect at least a majority of the board of directors or others performing similar functions with respect to such corporation or other organization is directly or indirectly owned or controlled by such specified Person or by any one or more of its Subsidiaries, or by such specified Person and one or more of its Subsidiaries.
 - (qqqq) "Supplemental 401(k) Plan" has the meaning set forth in Section 6.4.
 - (rrrr) "U.S." means the United States of America.
 - (ssss) "WARN" means the U.S. Worker Adjustment and Retraining Notification Act, and any applicable state or local law equivalent.
- (tttt) "Welfare Plan" means a "welfare plan" as defined in ERISA Section 3(1) and also means a cafeteria plan under Code Section 125 and any benefits offered thereunder, including pre-tax premium conversion benefits, a dependent care assistance program, contribution funding toward a health savings account and flex or cashable credits.

Interpretation. In this Agreement, unless the context clearly indicates otherwise:

- (uuuu) words used in the singular include the plural and words used in the plural include the singular;
- (vvvv) if a word or phrase is defined in this Agreement, its other grammatical forms, as used in this Agreement, shall have a corresponding meaning;

(wwww) reference to any gender includes the other gender and the neuter;

(xxxx) the words "include," "includes" and "including" shall be deemed to be followed by the words "without limitation";

(yyyy) the words "shall" and "will" are used interchangeably and have the same meaning;

(zzzz) the word "or" shall have the inclusive meaning represented by the phrase "and/or";

(aaaaa) relative to the determination of any period of time, "from" means "from and including," "to" means "to but excluding" and "through" means "through and including";

(bbbbb) all references to a specific time of day in this Agreement shall be based upon Eastern Standard Time or Eastern Daylight Savings Time, as applicable, on the date in question;

(cccc) whenever this Agreement refers to a number of days, such number shall refer to calendar days unless business days are specified;

(ddddd) accounting terms used herein shall have the meanings historically ascribed to them by Existing SLM and its Subsidiaries for periods prior to the Merger Date, including NewCo for this purpose, in its and their internal accounting and financial policies and procedures in effect immediately prior to the date of this Agreement;

(eeeee) reference to any Article, Section or Schedule means such Article or Section of, or such Schedule to, this Agreement, as the case may be, and references in any Section or definition to any clause means such clause of such Section or definition;

(fffff) the words "this Agreement," "herein," "herein," "hereof," "hereto" and words of similar import shall be deemed references to this Agreement as a whole and not to any particular Section or other provision of this Agreement;

(ggggg) reference to any agreement, instrument or other document means such agreement, instrument or other document as amended, supplemented and modified from time to time to time to the extent permitted by the provisions thereof and not prohibited by this Agreement;

(hhhhh) reference to any Law (including statutes and ordinances) means such Law (including any and all rules and regulations promulgated thereunder) as amended, modified, codified or reenacted, in whole or in part, and in effect at the time of determining compliance or applicability;

(iiiii) references to any Person include such Person's successors and assigns but, if applicable, only if such successors and assigns are permitted by this Agreement; and any reference to a third party shall be deemed to mean a Person who is not a Party or an Affiliate of a Party;

- (jjjjj) if there is any conflict between the provisions of the main body of this Agreement and the Schedules hereto, the provisions of the main body of this Agreement shall control unless explicitly stated otherwise in such Schedule;
 - (kkkkk) unless otherwise specified in this Agreement, all references to dollar amounts herein shall be in respect of lawful currency of the U.S.;
- (IIIII) the titles to Articles and headings of Sections contained in this Agreement, in any Schedule and in the table of contents to this Agreement have been inserted for convenience of reference only and shall not be deemed to be a part of or to affect the meaning or interpretation of this Agreement; and

(mmmmm) any portion of this Agreement obligating a Party to take any action or refrain from taking any action, as the case may be, shall mean that such Party shall also be obligated to cause its relevant Subsidiaries to take such action or refrain from taking such action, as the case may be.

ARTICLE II

ASSIGNMENT OF EMPLOYEES

Active Employees.

- (a) <u>NewCo Employees</u>. Except as otherwise set forth in this Agreement, effective as of the Distribution Date, the employment of the NewCo Employees will be continued by a member of the NewCo Group. Prior to such date, NewCo Employees will be transferred and assigned to the applicable member of the NewCo Group. Each of the Parties agrees to execute, and to seek to have the applicable employees execute, such documentation as may be necessary to reflect such assignments and transfers.
- (b) <u>SLM BankCo Employees</u>. Except as otherwise set forth in this Agreement, effective as of the Distribution Date, the employment of the SLM BankCo Employees will be continued by a member of the SLM BankCo Group. Prior to such date, SLM BankCo Employees will be transferred and assigned to the applicable member of the SLM BankCo Group. Each of the Parties agrees to execute, and to seek to have the applicable employees execute, such documentation as may be necessary to reflect such assignments and transfers.
- (c) At-Will Status. Notwithstanding the above or any other provision of this Agreement, nothing in this Agreement shall create any obligation on the part of any member of the SLM BankCo Group or any member of the NewCo Group to continue the employment of any employee for any period following the date of this Agreement or the Distribution or to change the employment status of any employee from "at will," to the extent such employee is an "at will" employee under applicable law.
- (d) <u>Severance</u>. The Distribution and the assignment, transfer or continuation of the employment of employees as contemplated by this Section 2.1 shall not be deemed a severance of employment of any employee for purposes of this Agreement or any plan, policy, practice or arrangement of any member of the Group.

(e) Change of Control/Change in Control. Neither the completion of the Distribution nor any transaction in connection with the Distribution (including the Merger) shall be deemed a "change of control" or "change in control" for purposes of any plan, policy, practice or arrangement relating to directors, former directors, employees, former employees or consultants of any member of the Group.

Former Employees.

(f) Any individual who has previously worked as an employee for a member (or former member) of the Sallie Mae Group but who is not employed by the Sallie Mae Group as of the time immediately prior to the Distribution Date shall be referred to as a "Legacy Sallie Mae Employee". For purposes of this Agreement, any Legacy Sallie Mae Employee shall be deemed to be a "Former NewCo Employee."

Employment Law Obligations.

- (g) WARN Act. After the Distribution Date, (i) SLM BankCo shall be responsible for providing any necessary WARN notice (and meeting any similar state law notice requirements) with respect to any termination of any SLM BankCo Employee and (ii) NewCo shall be responsible for providing any necessary WARN notice (and meeting any similar state law notice requirements) with respect to any termination of any NewCo Employee.
- (h) Compliance With Employment Laws. On and after the Distribution Date, (i) each member of the SLM BankCo Group shall be responsible for adopting and maintaining any policies or practices, and for all other actions and inactions, necessary to comply with employment-related laws and requirements relating to the employment of SLM BankCo Employees, and (ii) each member of the NewCo Group shall be responsible for adopting and maintaining any policies or practices, and for all other actions and inactions, necessary to comply with employment-related laws and requirements relating to the employment of NewCo Employees and the treatment of any applicable Former NewCo Employees in respect of their former employment. Any liabilities with respect to Legacy Sallie Mae Employees that are not specifically allocated within this Agreement shall be assumed and satisfied by NewCo.

Employee Records.

- (i) <u>Records Relating to SLM BankCo Employees</u>. All records, data and other employee-related information in any form (collectively, "**Employee Records**") relating to SLM BankCo Employees shall be the property of the SLM BankCo Group, except that Employee Records pertaining to such an employee and relating to any period that such employee was (i) employed by NewCo or (ii) covered under any employee benefit plan sponsored by any member of the NewCo Group (to the extent that such records or data relate to such coverage) prior to the Distribution Date shall be jointly owned by those members of the NewCo Group and the SLM BankCo Group.
- (j) Records Relating to NewCo Employees and Former NewCo Employees. All Employee Records relating to NewCo Employees or Former NewCo Employees shall be the property of the NewCo Group, except that Employee Records pertaining to such an employee and relating to any period that such employee was (i) employed by any member of the SLM BankCo Group or (ii) covered under any employee benefit plan sponsored by any member of the SLM BankCo Group (to the extent that such records or data relate to such coverage) prior to the Distribution Date shall be jointly owned by those members of the SLM BankCo Group and the NewCo Group.

- (k) Sharing of Records. Notwithstanding Sections 2.4(a) and (b), each of NewCo and SLM BankCo shall allow the other to retain a copy of the Employee Records owned by each of them, and use their respective commercially reasonable efforts to provide the other such Employee Records as may be necessary or appropriate to carry out their respective obligations under applicable law (subject to any relevant privacy protection laws or regulations in any applicable jurisdictions or Privacy Contract), this Agreement, any other Ancillary Agreement or the Separation Agreement, and for the purposes of administering their respective employee benefit plans and policies. All information and records regarding employment, personnel and employee benefit matters, including Employee Records, of SLM BankCo Employees shall be accessed, retained, held, used, copied and transmitted after the Distribution Date by members of the SLM BankCo Group in accordance with all applicable laws, policies and Privacy Contracts relating to the collection, storage, retention, use, transmittal, disclosure and destruction of such records. All information and records regarding employment, personnel and employee benefit matters, including Employee Records, of NewCo Employees and Former NewCo Employees shall be accessed, retained, held, used, copied and transmitted after the Distribution Date by members of the NewCo Group in accordance with all applicable laws, policies and Privacy Contracts relating to the collection, storage, retention, use, transmittal, disclosure and destruction of such records.
- (l) Access to Records. To the extent not inconsistent with this Agreement and any applicable privacy protection laws or regulations or Privacy Contracts, access to all employment, personnel and employee benefit matters, including Employee Records, after the Distribution Date will be provided to members of the SLM BankCo Group and members of the NewCo Group in accordance with the Separation Agreement. In addition, notwithstanding anything to the contrary, SLM BankCo shall be provided reasonable access to those records that are the property of the NewCo Group necessary for the administration of any plans or programs on behalf of SLM BankCo Employees after the Distribution Date to the extent permissible under applicable privacy protection laws or regulations or Privacy Contracts. SLM BankCo shall also be permitted to retain copies of all restrictive covenant agreements with any NewCo Employee or Former NewCo Employee in which any member of the SLM BankCo Group has a valid business interest. In addition, notwithstanding anything to the contrary, NewCo shall be provided reasonable access to those records that are the property of the SLM BankCo Group necessary for the administration of any plans or programs on behalf of NewCo Employees and Former NewCo Employees after the Distribution Date to the extent permissible under applicable privacy protection laws or regulations or Privacy Contracts. NewCo shall also be permitted to retain copies of all restrictive covenant agreements with any SLM BankCo Employee in which any member of the NewCo Group has a valid business interest.
- (m) <u>Maintenance of Records</u>. With respect to retaining, destroying, transferring, sharing, copying and permitting access to information and records regarding employment, personnel and employee benefit matters, including Employee Records, SLM BankCo and NewCo shall (and shall cause their respective Subsidiaries to) comply with all applicable laws,

regulations, Privacy Contracts and internal policies, and shall indemnify and hold harmless each other from and against any and all liability, claims, actions, and damages that arise from a failure (by the indemnifying party or its affiliates or their respective agents) to so comply with all applicable laws, regulations, Privacy Contracts and internal policies applicable to such information and records.

- (n) No Access to Computer Systems or Files. No provision of this Agreement shall give (i) any member of the SLM BankCo Group direct access to the computer systems or other files, records or databases of any member of the NewCo Group to access Employee Records or (ii) any member of the NewCo Group direct access to the computer systems or other files, records or databases of any member of the SLM BankCo Group to access Employee Records, unless expressly and specifically permitted by the owner of such systems, files, records or databases.
- (o) <u>Relation to Separation Agreement</u>. The provisions of this Section 2.4 shall be in addition to, and not in derogation of, the provisions of the Separation Agreement governing Confidential Information, including Section 6.9 of the Separation Agreement. To the extent of any inconsistency between the provisions of this Section 2.4 of this Agreement and Section 6.9 of the Separation Agreement, the provisions of this Section 2.4 shall control.
- (p) <u>Confidentiality</u>. Except to the extent otherwise set forth in this Agreement, all Employee Records shall, in each case, be subject to the confidentiality provisions of the Separation Agreement and any Privacy Contract or other applicable agreement and applicable law.
- (q) <u>Cooperation</u>. Each Party shall use commercially reasonable efforts to cooperate to share, retain and maintain data and records that are necessary or appropriate to further the purposes of this Section 2.4 and for each Party to administer its respective benefit plans to the extent consistent with this Agreement and applicable law, and each Party agrees to cooperate as long as is reasonably necessary to further the purposes of this Section 2.4. Except as provided under any Ancillary Agreement, no Party shall charge another Party a fee for such cooperation.
- (r) <u>HIPAA Business Associate Agreement</u>. NewCo and SLM BankCo shall enter into a Business Associate agreement under the Health Insurance Portability and Accountability Act of 1996, as amended, with respect to the group health plan of the other Party, in such form as NewCo and SLM BankCo shall mutually agree.

ARTICLE III

EQUITY AND INCENTIVE COMPENSATION PLANS

General Principles.

(a) For the avoidance of doubt, the provisions of this Article III shall not apply unless the Distribution takes place. The adjustments and replacements of awards contemplated by this Article III may be made in advance of the Distribution, but no adjusted or replacement award may be settled or exercised prior to the Distribution, and if the Distribution does not occur by December 31, 2014, such adjustments and/or replacements shall be null and void. Each of SLM BankCo and NewCo shall take any and all reasonable action as shall be necessary and appropriate to further the provisions of this Article III.

- (b) Where an award granted under one of the SLM BankCo Legacy Equity Plans is adjusted or replaced in whole or in part by an award under either a SLM BankCo Legacy Equity Plan or the NewCo New Equity Plan in accordance with the provisions of this Article III, such award generally shall be on terms which are in all material respects identical to the terms of the award which it replaces (including any requirements of continued employment) but subject to any necessary changes to take into account that (i) an award may relate to NewCo Common Stock, (ii) the NewCo New Equity Plan is administered by NewCo and the compensation committee of its board of directors, (iii) if applicable, the grantee under the award is employed or affiliated with a new employer or plan sponsor, and (iv) except for Price Targets under a Post-Distribution NewCo Option or a Post-Distribution SLM BankCo Option, the award is not subject to any performance conditions. Where an award granted under one of the SLM BankCo Legacy Equity Plans is adjusted in accordance with the provisions of this Article III, such award shall otherwise continue to retain the same terms and conditions of the original award, subject to any necessary changes to take into account the adjustments required by this Article III.
- (c) Following the Distribution Date, a grantee who has outstanding awards under one or more of the SLM BankCo Legacy Equity Plans and/or replacement awards under the NewCo New Equity Plan shall be considered to have been employed continuously by the applicable plan sponsor before the Distribution for purposes of (1) vesting and (2) determining the date of termination of employment as it applies to any such award. SLM BankCo will take such action as is necessary such that NewCo Employees that hold SLM BankCo Equity Awards as of or following the Distribution Date will not incur a termination of employment as a result of the Distribution for purposes of such awards. NewCo will take such action as is necessary such that SLM BankCo Employees that hold NewCo Equity Awards as of or following the Distribution Date will not incur a termination of employment as a result of the Distribution for purposes of such awards. Following the Distribution Date, a grantee who has outstanding awards under one or more of the SLM BankCo Legacy Equity Plans and/or replacement awards under the NewCo New Equity Plan shall be considered to be employed by the applicable plan sponsor only for service with the applicable of SLM BankCo or NewCo whichever is the employer of the individual as of a time immediately after the Distribution Date. For the avoidance of doubt, (1) a NewCo Employee or NewCo Director who has a SLM BankCo Equity Award will continue to vest in such award by reason of service with NewCo but upon termination of service with NewCo will be treated for purposes of the SLM BankCo Equity Award as having then terminated service, and (2) a SLM BankCo Employee or SLM BankCo Director who has a NewCo Equity Award will continue to vest in such award by reason of service with SLM BankCo but upon termination of service with SLM BankCo will be treated for purposes of the NewCo Equity Award as having then terminated service.
- (d) No award described in this Article III, whether outstanding or to be issued, adjusted, substituted or cancelled by reason of or in connection with the Distribution, shall be adjusted, settled or cancelled or become exercisable if in the judgment of the administrator of the applicable plan or program such action would be inconsistent with applicable law, including federal securities laws. Any period of exercisability will not be extended on account of a period during which such an award is not exercisable in accordance with the preceding sentence.

Restricted Stock.

(e) Each grantee under the SLM BankCo Legacy Equity Plans who holds one or more SLM BankCo RSAs shall receive in lieu of such award an adjusted award of SLM BankCo Restricted Stock (an "Adjusted SLM BankCo RSA") with respect to a number of shares of SLM BankCo Common Stock equal to a fraction, the numerator of which is the product of the Pre-Distribution SLM BankCo Share Price and the number of shares subject to the SLM BankCo RSA and the denominator of which is the Post-Distribution SLM BankCo Share Price, rounded up to the nearest whole share. SLM BankCo (or one or more of the SLM BankCo Subsidiaries, as designated by SLM BankCo) shall be responsible for (i) the satisfaction of all tax reporting and withholding requirements in respect of the Adjusted SLM BankCo RSAs and (ii) remitting the appropriate tax or withholding amounts to the appropriate taxing authorities. Except as provided in the foregoing provisions of this Section 3.2(a), the Adjusted SLM BankCo RSAs shall be granted on terms which are in all material respects identical (including with respect to vesting) to the terms of the SLM BankCo RSAs with respect to which they are granted.

Restricted Stock Units.

- (f) Each grantee under the SLM BankCo Legacy Equity Plans who holds one or more SLM BankCo RSUs granted prior to February 4, 2014, or granted in connection with awards under the 2013 Management Incentive Plan, except each grantee of awards set forth on Schedule 3.3(b), shall retain each such SLM BankCo RSU and shall receive a number of additional restricted stock units with respect to NewCo Common Stock (the "NewCo RSUs") equal to the number of shares received by a stockholder of SLM BankCo Common Stock in connection with the Distribution with respect to the number of shares of SLM BankCo Common Stock subject to such grantee's SLM BankCo RSUs. Except as provided in this Agreement, NewCo RSUs shall be granted on terms which are in all material respects identical (including with respect to vesting) to the terms of the SLM BankCo RSUs with respect to which they are granted.
- (g) Each grantee under the SLM BankCo Legacy Equity Plans who holds one or more SLM BankCo RSUs granted on or after February 4, 2014, (other than SLM BankCo RSUs granted in connection with awards under the 2013 Management Incentive Plan) and who will be an SLM BankCo Employee, as well as each grantee of awards set forth on Schedule 3.3(b), shall receive in substitution for such SLM BankCo RSU (which shall be cancelled) an adjusted award of restricted stock units with respect to solely SLM BankCo Common Stock (an "Adjusted SLM BankCo RSU"), with the number of shares of SLM BankCo Common Stock subject to such Adjusted SLM BankCo RSU equal to a fraction, the numerator of which is the product of the Pre-Distribution SLM BankCo Share Price and the number of shares subject to the SLM BankCo RSU and the denominator of which is the Post-Distribution SLM BankCo RSU sgranted on or after February 4, 2014, (other than SLM BankCo RSUs granted in connection with awards under the 2013 Management Incentive Plan) and who will not be an SLM BankCo Employee shall receive in substitution for each such SLM BankCo RSU (which shall be cancelled) an adjusted award of restricted stock units with respect to solely NewCo Common Stock (an "Adjusted NewCo RSU"), with the number of shares of NewCo Common Stock subject to such Adjusted NewCo RSU equal to a fraction, the numerator of which is the Product of the Pre-Distribution SLM BankCo Share Price and the number of shares subject to the SLM BankCo RSU and the denominator of which is the Post-Distribution NewCo Share Price, rounded up to the nearest whole share.

(h) SLM BankCo (or one or more of the SLM BankCo Subsidiaries, as designated by SLM BankCo) shall be responsible for (i) the satisfaction of all tax reporting and withholding requirements in respect of the SLM BankCo RSUs and NewCo RSUs issued to SLM BankCo Employees and (ii) remitting the appropriate tax or withholding amounts to the appropriate taxing authorities in the case of SLM BankCo Employees or ensuring the remittance of the appropriate tax or withholding amounts to NewCo in the case of NewCo Employees and Former NewCo Employees that hold SLM BankCo RSUs. NewCo shall be responsible for (i) the satisfaction of all tax reporting and withholding requirements in respect of the NewCo RSUs and SLM BankCo RSUs issued to NewCo Employees and Former NewCo Employees and (ii) remitting the appropriate tax or withholding amounts to the appropriate taxing authorities in the case of NewCo Employees and Former NewCo Employees or ensuring the remittance of the appropriate tax or withholding amounts to SLM BankCo in the case of SLM BankCo Employeeswho hold NewCo RSUs.

Stock Options.

- (i) Each grantee under any of the SLM BankCo Legacy Equity Plans who, as of the Distribution Date, holds one or more SLM BankCo Options (each, a "Legacy SLM BankCo Option"), shall receive, in substitution for each such Legacy SLM BankCo Option (which shall be cancelled), both a NewCo Option (a "Post-Distribution NewCo Option") with respect to shares of NewCo Common Stock and a SLM BankCo Option (a "Post-Distribution SLM BankCo Option") with respect to shares of SLM BankCo Common Stock. The shares of NewCo Common Stock and SLM BankCo Common Stock subject to the Post-Distribution NewCo Option and Post-Distribution SLM BankCo Option, respectively, shall be equal to the number of shares of SLM BankCo Common Stock subject to the Legacy SLM BankCo Option. The exercise price of each Post-Distribution SLM BankCo Option and Post-Distribution NewCo Option shall bear the same ratio to the Post-Distribution SLM BankCo Share Price and the Post-Distribution NewCo Share Price, respectively, as the exercise price of the Legacy SLM BankCo Option being replaced bears to the Pre-Distribution SLM BankCo Share Price. In the case of a Legacy SLM BankCo Option that includes a Price Target, the Price Targets for the Post-Distribution SLM BankCo Option and Post-Distribution NewCo Option that are substituted therefor shall be, respectively, the applicable Post-Distribution SLM BankCo Option Price Target and Post-Distribution NewCo Option Price Target, with appropriate provisions to account for periods that overlap the Distribution Date.
- (j) SLM BankCo (or one or more of the SLM BankCo Subsidiaries, as designated by SLM BankCo) shall be responsible for (i) the satisfaction of all tax reporting and withholding requirements in respect of the exercise of Post-Distribution SLM BankCo Options and Post-Distribution NewCo Options held by SLM BankCo Employees and (ii) remitting the appropriate tax or withholding amounts to the appropriate taxing authorities in the case of SLM BankCo Employees or ensuring the remittance of the appropriate tax or withholding amounts to NewCo in the case of NewCo Employees and Former NewCo Employees that receive Post-Distribution SLM BankCo Options. NewCo shall be responsible for (i) the satisfaction of all tax reporting and withholding requirements in respect of the exercise of Post-Distribution NewCo Options and Post-Distribution BankCo Options held by NewCo Employees and Former NewCo Employees

and (ii) remitting the appropriate tax or withholding amounts to the appropriate taxing authorities in the case of NewCo Employees and Former NewCo Employees or ensuring the remittance of the appropriate tax or withholding amounts to SLM BankCo in the case of SLM BankCo Employees who receive Post-Distribution NewCo Options.

(k) Post-Distribution NewCo Options and Post-Distribution SLM BankCo Options shall not be exercisable until the applicable Registration Statement Effectiveness Date and may be further restricted as described in Section 3.9. Except as provided in this Section 3.4, Post-Distribution NewCo Options and Post-Distribution SLM BankCo Options shall be granted on terms which are in all material respects identical (including with respect to vesting) to the terms of the Legacy SLM BankCo Options which they replace.

Performance Stock Units.

(l) Each grantee under any of the SLM BankCo Legacy Equity Plans who, but for this Section 3.5, would hold, as of the Distribution Date, one or more performance stock unit awards (each, a "Legacy Performance Stock Unit") shall receive on a date prior to the Distribution Date (but subject to the occurrence of the Distribution Date) and as a replacement award in substitution for each Legacy Performance Stock Unit (which shall be cancelled, subject to the occurrence of the Distribution Date), a number of SLM BankCo RSUs equal to the number of shares of SLM BankCo Common Stock that would vest under the legacy performance-based award based on the actual performance to the last day of the most recent calendar quarter ending prior to or coincident with the Distribution Date (the "Measurement Date") and the lesser of (i) the target performance specified for such Legacy Performance Stock Unit, prorated for that portion of the performance period which will occur after the Measurement Date and (ii) projected performance for that portion of the performance period which will occur after the Measurement Date (as determined by the compensation committee of the Board of Directors of SLM BankCo). SLM BankCo RSUs issued in substitution for Legacy Performance Stock Unit pursuant to this Section 3.5 shall then be adjusted pursuant to Section 3.3(a). In the event the Distribution Date does not occur, the transactions contemplated by this Section 3.5 shall be null and void and of no effect, and the Legacy Performance Stock Units shall remain outstanding in accordance with their terms.

Section 16(b) of the Exchange Act; Code Sections 162(m) and 409A.

(m) By approving the adoption of this Agreement and in accordance with the provisions of Rule 16b-3 under the Exchange Act, the board of directors of SLM BankCo intends to exempt from the short-swing profit recovery provisions of Section 16(b) of the Exchange Act all acquisitions from and dispositions to SLM BankCo of SLM BankCo Equity Awards by directors and executive officers of SLM BankCo contemplated by this Article III. The board of directors of SLM BankCo also intends to expressly approve, in respect of all SLM BankCo Equity Awards granted or issued to directors and executive officers of SLM BankCo in accordance with this Article III, the use of any method for the payment of an exercise price and the satisfaction of any applicable tax withholding (specifically including the actual or constructive tendering of shares of SLM BankCo Common Stock in payment of an exercise price and the withholding of such shares from delivery under options in satisfaction of applicable tax withholding requirements) to the extent such method is permitted under the applicable equity incentive plan and award agreement.

- (n) By approving the adoption of this Agreement and in accordance with the provisions of Rule 16b-3 under the Exchange Act, the board of directors of NewCo intends to exempt from the short-swing profit recovery provisions of Section 16(b) of the Exchange Act all acquisitions from and dispositions to NewCo of NewCo Equity Awards by directors and executive officers of NewCo contemplated by this Article III. The board of directors of NewCo also intends to expressly approve, in respect of all NewCo Equity Awards granted or issued to directors and executive officers of NewCo in accordance with this Article III, the use of any method for the payment of an exercise price and the satisfaction of any applicable tax withholding (specifically including the actual or constructive tendering of shares of NewCo Common Stock in payment of an exercise price and the withholding of such shares from delivery under options in satisfaction of applicable tax withholding requirements) to the extent such method is permitted under the applicable equity incentive plan and award agreement.
- (o) Notwithstanding anything in this Agreement to the contrary (including the treatment of supplemental and deferred compensation plans, outstanding long-term incentive awards and annual incentive awards as described herein), SLM BankCo and NewCo agree to negotiate in good faith regarding the need for any treatment different from that otherwise provided herein to ensure that (i) a federal income tax deduction for the payment of such supplemental or deferred compensation or long-term incentive award, annual incentive award or other compensation is not limited by reason of Code Section 162(m), and (ii) the treatment of such supplemental or deferred compensation or long-term incentive award, annual incentive award or other compensation does not cause the imposition of a tax under Code Section 409A.

Certain Bonus Payments.

- (p) Annual incentive bonuses in respect of calendar year 2014, prorated for the period commencing on January 1, 2014 and ending on the Distribution Date, may be paid to SLM BankCo Employees and NewCo Employees by SLM BankCo and NewCo, respectively, in the discretion of the compensation committee of each of SLM BankCo and NewCo, at the time such bonuses are normally paid in accordance with the bonus metrics determined by the compensation committee of the board of directors of Existing SLM (the "2014 Pre-Spin Bonus"). Each employee's 2014 Pre-Spin Bonus shall be based on actual performance results based on the bonus metrics set by the compensation committee of the board of directors of Existing SLM, as such performance is certified by the board of directors of Existing SLM, for the period commencing on January 1, 2014 and ending on the last day of the full calendar quarter preceding the Distribution Date.
- (q) SLM BankCo shall assume responsibility and liability for payment of 2014 Pre-Spin Bonuses to any SLM BankCo Employee. NewCo shall assume responsibility and liability for payment of 2014 Pre-Spin Bonuses to all individuals other than SLM BankCo Employees.
- (r) Any other cash bonuses accrued or earned as of the Distribution Date but not yet paid shall be paid following the Distribution Date by SLM BankCo in the case of SLM BankCo Employees or by NewCo in the case of NewCo Employees and Former NewCo Employees at such time and in such amount as is prescribed by the terms of the applicable bonus arrangement.

ESPP.

- (s) The administrator of the ESPP has taken all actions necessary and appropriate to suspend payroll deductions for and operation of the ESPP prior to the Distribution Date. SLM BankCo shall assume the ESPP effective as of the Distribution Date.
- (t) Effective no later than immediately prior to the Distribution Date, NewCo shall establish the NewCo ESPP, which plan shall be approved by SLM BankCo in its capacity as the sole stockholder of NewCo prior to the Distribution Date.

Section 3.2 Registration Statements; Blackouts

- (a) Before the Distribution Date or as soon as reasonably practicable thereafter and subject to compliance with applicable law, NewCo shall prepare and file with the SEC a registration statement on Form S–8 (or, if NewCo is not eligible to use Form S-8, another appropriate form for which NewCo is eligible) registering under the Securities Act the offer and sale of a number of shares of NewCo Common Stock equal to not less than the aggregate number of shares subject to all NewCo Equity Awards to be issued to the NewCo Employees and Former NewCo Employees in accordance with this Article III. NewCo shall use commercially reasonable efforts to cause such registration statement to be kept effective (and the prospectus or prospectuses forming a part thereof to be kept current to the extent required by the Securities Act) for so long as any of such NewCo Equity Awards remain outstanding.
- (b) Before the Distribution Date or as soon as reasonably practicable thereafter and subject to compliance with applicable law, SLM BankCo shall prepare and file with the SEC a registration statement on Form S–8 (or, if SLM BankCo is not eligible to use Form S-8, another appropriate form for which SLM BankCo is eligible) registering under the Securities Act the offer and sale of a number of shares of SLM BankCo Common Stock equal to not less than the aggregate number of shares subject to all SLM BankCo Equity Awards to be issued to SLM BankCo Employees in accordance with this Article III. NewCo shall use commercially reasonable efforts to cause such registration statement to be kept effective (and the prospectus or prospectuses forming a part thereof to be kept current to the extent required by the Securities Act) for so long as any of such SLM BankCo Equity Awards remain outstanding.
- (c) Each of NewCo and SLM BankCo covenants and agrees to use its commercially reasonable efforts to timely effect the adjustments and issuances of equity awards contemplated by this Article III, but acknowledges that such adjustments and issuances are subject to possible delays due to administrative issues and in order to comply with the requirements of applicable law and NASDAQ listing rules.
- (d) The Parties acknowledge that the exercise and settlement of equity-based awards may be subject to blackout periods as necessary to comply with applicable law, and the Parties agree to use commercially reasonable efforts to cooperate in complying with such requirements.

ARTICLE IV

GENERAL PRINCIPLES FOR ALLOCATION OF LIABILITIES

General Principles.

- (a) Each member of the SLM BankCo Group and each member of the NewCo Group shall take any and all reasonable action as shall be necessary or appropriate so that active participation in the SLM BankCo Thrift Plan, SLM BankCo Welfare Plans and SLM BankCo Benefit Arrangements by all NewCo Employees and Former NewCo Employees shall terminate in connection with the Distribution as and when provided under this Agreement (or if not specifically provided under this Agreement, as of the Effective Time), and each member of the NewCo Group shall cease to be a participating employer under the terms of the SLM BankCo Thrift Plan, SLM BankCo Welfare Plans and SLM BankCo Benefit Arrangements as of such time.
- (b) Each member of the NewCo Group and each member of the SLM BankCo Group shall take any and all reasonable action as shall be necessary or appropriate so that active participation in the NewCo Thrift Plan, NewCo Welfare Plans and NewCo Benefit Arrangements by all SLM BankCo Employees shall terminate in connection with the Distribution as and when provided under this Agreement (or if not specifically provided under this Agreement, as of the Effective Time), and each member of the SLM BankCo Group shall cease to be a participating employer under the terms of the NewCo Thrift Plan, NewCo Welfare Plans and NewCo Benefit Arrangements as of such time.
- (c) Except as otherwise provided in this Agreement, (i) one or more members of the NewCo Group (as designated by NewCo) shall continue to be responsible for or assume, effective as of the Distribution Date, all employee benefits liabilities for NewCo Employees and Former NewCo Employees, and the assets relating to such employee benefits for NewCo Employees and Former NewCo Employees shall be transferred to or continue to be held by one or more members of the NewCo Group (as designated by NewCo) and (ii) and one or more members of the SLM BankCo Group (as designated by SLM BankCo) shall continue to be responsible for or assume all employee benefits liabilities for SLM BankCo Employees and the assets relating to such employee benefits for SLM BankCo Employees shall be transferred to or continue to be held by one or more members of the SLM BankCo Group (as designated by SLM BankCo).
- (d) Except as otherwise provided in this Agreement, effective as of the day after the Distribution Date, one or more members of the NewCo Group (as determined by NewCo) shall assume or continue the sponsorship of, and no member of the SLM BankCo Group shall have any further liability for or under, the following agreements, obligations and liabilities, and NewCo shall indemnify each member of the SLM BankCo Group, and the officers, directors, and employees of each member of the SLM BankCo Group, and hold them harmless with respect to such agreements, obligations or liabilities:
 - (i) any and all individual employment agreements entered into between any member of the SLM BankCo Group and any NewCo Employee or Former NewCo Employee;

- (ii) any and all service agreements entered into between any member of the SLM BankCo Group and any individual who is an independent contractor providing services primarily for the business activities of the NewCo Group;
- (iii) any and all wages, salaries, incentive compensation (as the same may be modified by this Agreement), commissions and bonuses payable to any NewCo Employees or Former NewCo Employees after the Distribution Date, without regard to when such wages, salaries, incentive compensation, commissions and bonuses are or may have been earned;
- (iv) any and all moving expenses and obligations related to relocation, repatriation, transfers or similar items incurred by or owed to any NewCo Employees or Former NewCo Employees, whether or not accrued as of the Distribution Date (other than such expenses and obligations incurred by SLM BankCo on or prior to the Distribution Date as a result of which there is an existing liability as of the Distribution Date, all of which shall remain SLM BankCo's obligation);
- (v) any and all immigration-related, visa, work application or similar rights, obligations and liabilities related to any NewCo Employees or Former NewCo Employees; and
- (vi) any and all liabilities and obligations whatsoever with respect to claims made by or with respect to any NewCo Employees or Former NewCo Employees in connection with any employee benefit plan, program or policy not otherwise retained or assumed by any member of the SLM BankCo Group pursuant to this Agreement, including such liabilities relating to actions or omissions of or by any member of the NewCo Group or any officer, director, employee or agent thereof on or prior to the Distribution Date.
- (e) Except as otherwise provided in this Agreement, effective as of the Effective Time, no member of the NewCo Group shall have any further liability for, and SLM BankCo shall indemnify each member of the NewCo Group, and the officers, directors, and employees of each member of the NewCo Group, and hold them harmless with respect to any and all liabilities and obligations whatsoever with respect to, claims made by or with respect to any SLM BankCo Employees in connection with any employee benefit plan, program or policy not otherwise retained or assumed by any member of the NewCo Group pursuant to this Agreement, including such liabilities relating to actions or omissions of or by any member of the SLM BankCo Group or any officer, director, employee or agent thereof on or prior to the Distribution Date.

Sponsorship and/or Establishment of NewCo Plans. Except as otherwise provided in this Agreement, sponsorship of benefit plans that cover solely NewCo Employees and Former NewCo Employees shall become effective not later than as of the Effective Time by a member of the NewCo Group, and to the extent necessary to achieve such sponsorship, each member of the SLM BankCo Group and each member of the NewCo Group shall take appropriate action, including transfer of sponsorship of each such plan. SLM BankCo Welfare Plans in which both (i) SLM BankCo Employees and (ii) NewCo Employees or Former NewCo Employees participate shall be divided into two separate plans, with one covering SLM BankCo Employees sponsored by a member of the SLM BankCo Group, and the other covering NewCo Employees and Former NewCo Employees sponsored by a member of the NewCo Group.

Service Credit.

- (f) Service for Eligibility and Vesting Purposes. Except as otherwise provided in this Agreement, for purposes of eligibility and vesting under the NewCo Thrift Plan, NewCo Welfare Plans and NewCo Benefit Arrangements, NewCo shall, and shall cause each member of the NewCo Group to, credit each NewCo Employee and Former NewCo Employee with service for any period of employment with any member of the SLM BankCo Group on or prior to the Distribution Date to the same extent such service would be credited if it had been performed for a member of the NewCo Group. Except as otherwise provided in this Agreement, for purposes of eligibility and vesting under the SLM BankCo Thrift Plan, SLM BankCo Welfare Plans and SLM BankCo Benefit Arrangements, SLM BankCo shall, and shall cause each member of the SLM BankCo Group to, credit each SLM BankCo Employee and Former SLM BankCo Employee with service for any period of employment with any member of the NewCo Group on or prior to the Distribution Date to the same extent such service would be credited if it had been performed for a member of the SLM BankCo Group.
- (g) Service for Benefit Purposes. Except as otherwise provided in this Agreement, (i) for purposes of benefit levels and accruals and benefit commencement entitlements under the NewCo Thrift Plan, NewCo Welfare Plans and NewCo Benefit Arrangements, NewCo shall, and shall cause each member of the NewCo Group to, credit each NewCo Employee and Former NewCo Employee with service for any period of employment with any member of the SLM BankCo Group on or prior to the Distribution Date to the same extent that such service is taken into account pursuant to the terms of the SLM BankCo Thrift Plan and SLM BankCo Welfare Plans, and (ii) for purposes of benefit commencement entitlements under the SLM BankCo Thrift Plan, SLM BankCo Welfare Plans and SLM BankCo Benefit Arrangements, SLM BankCo shall, and shall cause each member of the SLM BankCo Group to, credit each SLM BankCo Employee with service for any period of employment with any member of the NewCo Group on or prior to the Distribution Date to the same extent such service would be credited if it had been performed for a member of the SLM BankCo Group.
- (h) Evidence of Prior Service. Notwithstanding anything to the contrary, but subject to applicable law, upon reasonable request by SLM BankCo or NewCo to the other, the first Party will provide to the other Party copies of any records available to the first Party to document the service, plan participation and membership of an Employee and cooperate with the first Party to resolve any discrepancies or obtain any missing data for purposes of determining benefit eligibility, participation, vesting and calculation of benefits with respect to such Employee.
- (i) <u>Post-Distribution Transfers</u>. In the event, following the Distribution Date, a NewCo Employee leaves the employ of the NewCo Group to become an employee of the SLM BankCo Group, such NewCo Employee will experience a separation from service, severance from employment and termination of employment, as applicable, for purposes of the NewCo Thrift Plan, NewCo Welfare Plans and any other compensatory plan or arrangement maintained by the NewCo Group. In the event, following the Distribution Date, a SLM BankCo Employee leaves the employ of the SLM BankCo Group to become an employee of the NewCo Group, such BankCo Employee will experience a separation from service, severance from employment and termination of employment, as applicable, for purposes of the SLM BankCo Thrift Plan, SLM BankCo Welfare Plans and any other compensatory plan or arrangement maintained by the SLM BankCo Group.

Plan Administration.

- (j) <u>Transition Services</u>. The Parties acknowledge that the SLM BankCo Group or the NewCo Group may provide administrative services for certain of the other Party's benefit programs for a transitional period under the terms of an applicable transition services agreement. The Parties agree to enter into a business associate agreement (if required by applicable health information privacy laws) in connection with such transition services agreement.
- (k) Administration. NewCo shall use its reasonable best efforts to, and shall cause each member of the NewCo Group to use its reasonable best efforts to, administer its benefit plans in a manner that does not jeopardize the tax-favored status of the tax-favored benefit plans maintained by any member of the SLM BankCo Group. SLM BankCo shall use its reasonable best efforts to, and shall cause each member of the SLM BankCo Group to use its reasonable best efforts to, administer its benefit plans in a manner that does not jeopardize the tax-favored status of the tax-favored benefit plans maintained by any member of the NewCo Group.
- (1) <u>Participant Elections and Beneficiary Designations</u>. All participant elections and beneficiary designations made under any plan sponsored by a member of the SLM BankCo Group prior to the effective date as of which assets or liabilities relating to that plan are transferred or allocated to a member of the NewCo Group shall continue in effect under any plan maintained by any member of the NewCo Group to which liabilities are transferred or allocated pursuant to this Agreement until such time as any applicable participant changes his elections or beneficiary designations in accordance with the procedures of the relevant plan, as the case may be, including deferral, investment, and payment form elections, dividend elections, coverage options and levels, beneficiary designations and the rights of alternate payees under qualified domestic relations orders.

ARTICLE V

COLI POLICIES

COLI. Notwithstanding any other provision of the Agreement, on or prior to the Distribution Date, a member of the NewCo Group shall assume (or retain, as applicable) sponsorship and responsibility for that certain trust established in connection with the Sallie Mae Deferred Compensation Plan for Key Employees (the "SLM BankCo Rabbi Trust"), and the company-owned life insurance policies, which were adopted in connection with a deferred compensation arrangement but not held in the SLM BankCo Rabbi Trust shall be transferred to (or, as applicable, retained by) a member of the NewCo Group.

ARTICLE VI

THRIFT PLANS

General Principles. Effective as of a date on or prior to the Distribution Date, NewCo will establish and adopt a qualified employee cash or deferred arrangement under Code Section 401(k) (the "NewCo Thrift Plan") intended to be qualified under Code Section 401(a) and containing provisions that provide, among other things, (i) benefits for each NewCo Employee and Former NewCo Employee who was a participant (or former participant with a remaining account balance) in the SLM BankCo Thrift Plan as of the date immediately prior to the establishment of the NewCo Thrift Plan (and each beneficiary and alternate payee of such person) (the "NewCo Thrift Plan Beneficiaries") identical (except as provided in this Article VI) to those in effect for the NewCo Thrift Plan Beneficiaries under the SLM BankCo Thrift Plan as of the date of transfer of assets and liabilities with respect to such plan (as described below), and (ii) SLM BankCo Employees (and each beneficiary or alternate payee of such person) (the "SLM BankCo Thrift Plan Beneficiaries") with participant account balances reflecting shares of NewCo Common Stock received in the Distribution. Each NewCo Employee who was an active participant in the SLM BankCo Thrift Plan as of the date immediately prior to the establishment of the NewCo Thrift Plan shall participate in the NewCo Thrift Plan effective from and after its establishment. NewCo Employees and Former NewCo Employees shall not make or receive additional contributions under the SLM BankCo Thrift Plan after the effective date of the NewCo Thrift Plan, unless any such NewCo Employee or Former NewCo Employee shall become employed by any member of the SLM BankCo Group after such date and such member participates in the SLM BankCo Thrift Plan. A SLM BankCo Employee shall not participate in the NewCo Thrift Plan unless any such SLM BankCo Employee shall become employed by any member of the NewCo Group after the effective date of the NewCo Thrift Plan and such member participates in the NewCo Thrift Plan. The interest of each NewCo Thrift Plan Beneficiary in the SLM BankCo Thrift Plan attributable to employer matching contributions and employer core contributions as of the Distribution Date (which shall be limited to the amounts invested in the SLM BankCo Common Stock fund) shall be 100% vested on the Distribution Date. The interest of each SLM BankCo Thrift Plan Beneficiary in the NewCo Thrift Plan attributable to employer matching contributions and employer core contributions as of the Distribution Date (which shall be limited to the amounts invested in the NewCo Common Stock fund) shall be 100% vested on the Distribution Date. The participating employers in each of the NewCo Thrift Plan and SLM BankCo Thrift Plan as of the Effective Time are described in Schedule 6.1.

Treatment of SLM BankCo Common Stock and NewCo Common Stock.

(a) NewCo Common Stock Fund. The NewCo Thrift Plan will provide as of the Distribution Date: (i) for the establishment of a NewCo Common Stock fund; (ii) that such NewCo Common Stock fund shall receive and hold all shares of NewCo Common Stock to be distributed in the Distribution on behalf of NewCo Thrift Plan Beneficiaries and SLM BankCo Thrift Plan Beneficiaries; (iii) that, following the Distribution Date, contributions made by or on behalf of NewCo Thrift Plan Beneficiaries may be allocated to the NewCo Common Stock fund; (iv) that the SLM BankCo Thrift Plan Beneficiaries will be prohibited from increasing their holdings in the NewCo Common Stock fund; (v) that the SLM BankCo Thrift Plan Beneficiaries

may elect to liquidate their holdings in the NewCo Common Stock fund and invest those monies in any other investment fund offered under the NewCo Thrift Plan; and (vi) that the SLM BankCo Thrift Plan Beneficiaries may elect to receive their holdings in the NewCo Thrift Plan in accordance with the distribution options provided under such plan to terminated employees. Additionally, NewCo shall cause the NewCo Thrift Plan to provide that the SLM BankCo Thrift Plan Beneficiaries shall participate in the NewCo Thrift Plan in respect of their accounts thereunder; provided, however, NewCo may in its discretion provide that the NewCo Common Stock fund shall no longer be offered as an investment alternative under the NewCo Thrift Plan.

- (b) <u>SLM BankCo Common Stock Fund</u>. SLM BankCo shall amend the SLM BankCo Thrift Plan, on or prior to the Distribution Date, to provide that, following the Distribution: (i) the SLM BankCo Common Stock fund will hold the assets of the accounts of the NewCo Thrift Plan Beneficiaries invested in the SLM BankCo Common Stock fund; (ii) the NewCo Thrift Plan Beneficiaries will be prohibited from increasing their holdings in the SLM BankCo Common Stock fund; (iii) the NewCo Thrift Plan Beneficiaries may elect to liquidate their holdings in the SLM BankCo Common Stock fund and invest those monies in any other investment fund offered under the SLM BankCo Thrift Plan; and (iv) the NewCo Thrift Plan Beneficiaries may elect to receive their holdings in the SLM BankCo Thrift Plan in accordance with the distribution options available under such plan to terminated employees. SLM BankCo shall cause the SLM BankCo Thrift Plan to provide that NewCo Thrift Plan Beneficiaries shall participate in the SLM BankCo Thrift Plan in respect of their accounts thereunder; provided, however, SLM BankCo may in its discretion provide that the SLM BankCo Common Stock fund shall no longer be offered as an investment alternative under the SLM BankCo Thrift Plan.
- (c) Transfer of Accounts. Effective as soon as practicable following the Distribution Date, SLM BankCo shall cause to be transferred from the trust under the SLM BankCo Thrift Plan to the trust under the NewCo Thrift Plan the aggregate amount that is credited to the accounts of the NewCo Thrift Plan Beneficiaries as of such date. The transfer shall, to the extent reasonably possible, be an in-kind transfer, subject to the reasonable consent of the trustee of the NewCo Thrift Plan trust and shall include the transfer of the aggregate assets held in the accounts relating to each NewCo Thrift Plan Beneficiary under the SLM BankCo Thrift Plan and any participant loan notes held under such plans. SLM BankCo shall cause the SLM BankCo Thrift Plan to allocate to the NewCo Thrift Plan a proportionate share of any forfeiture account under the SLM BankCo Thrift Plan.

Supplemental 401(k) Plan. Effective prior to the Distribution Date, a member of the SLM BankCo Group shall assume responsibility for the Sallie Mae Supplemental 401(k) Savings Plan (the "Supplemental 401(k) Plan"), and a mirror plan (the "NewCo Supplemental 401(k) Plan") will be established by a member of the NewCo Group. As of the Effective Time, the NewCo Supplemental 401(k) Plan will assume all liability under the Supplemental 401(k) Plan for benefits to NewCo Employees and Former NewCo Employees, including any notional investment accounts deemed invested in NewCo Common Stock or SLM BankCo Common Stock. The NewCo Supplemental 401(k) Plan will continue to honor any deferral elections and payment timing provisions in effect under the Supplemental 401(k) Plan for NewCo Employees and Former NewCo Employees.

Deferred Compensation Plans. Effective prior to the Distribution Date, a member of the NewCo Group shall assume responsibility for the Sallie Mae Deferred Compensation Plan for Key Employees and the SLM Corporation Deferred Compensation Plan for Directors (collectively, the "Deferred Compensation Plans"), and mirror plans (the "SLM BankCo Deferred Compensation Plans") will be established by a member of the SLM BankCo Group. As of the Effective Time, the SLM BankCo Deferred Compensation Plans will assume all liability under the Deferred Compensation Plans for benefits to SLM BankCo Employees and SLM BankCo Directors, including any notional investment accounts deemed invested in SLM BankCo Common Stock or NewCo Common Stock. The SLM BankCo Deferred Compensation Plan will continue to honor any deferral elections and payment timing provisions in effect under the Deferred Compensation Plan for SLM BankCo Employees and SLM BankCo Directors.

ARTICLE VII

WELFARE PLANS

Establishment of NewCo Welfare Plans. Except as provided below, the members of the NewCo Group who had previously adopted a SLM BankCo Welfare Plan and were participating employers therein ("Participating NewCo Employers") will, as of not later than the day following the Distribution Date, withdraw from such participation, and, effective as of the day following the Distribution Date, one or more of the Participating NewCo Employers will assume sponsorship, under newly established welfare plans, of the coverage and benefits which were offered under such plans to the NewCo Employees and the Former NewCo Employees (and their eligible spouses and dependents as the case may be) of the Participating NewCo Employers (collectively, the "NewCo Welfare Plan Participants"). Such coverage and benefits shall then be provided to the NewCo Welfare Plan Participants on an uninterrupted basis under the newly established NewCo Welfare Plans which shall contain substantially the same benefit provisions as in effect under the corresponding SLM BankCo Welfare Plan immediately prior to the Effective Time. Except as provided below, effective as of the Effective Time, liabilities relating to the NewCo Welfare Plan Participants shall be spun off from each SLM BankCo Welfare Plan and allocated to the corresponding new NewCo Welfare Plan. The participating employers in each of the NewCo Welfare Plan and SLM BankCo Welfare Plan as of the Effective Time are described in Schedule 6.1.

As a result of withdrawal from participation in the SLM BankCo Welfare Plans by the Participating NewCo Employers, the NewCo Welfare Plan Participants ceased to be eligible for coverage under the SLM BankCo Welfare Plans as of the Effective Time. NewCo Welfare Plan Participants shall not participate in any SLM BankCo Welfare Plans after the Distribution Date, unless they shall become employed after such date by any member of the SLM BankCo Group that participates in such plans and meet the terms and conditions of participation thereunder. SLM BankCo Employees shall not participate in any NewCo Welfare Plans, unless they shall become employed after the Distribution Date by any member of the NewCo Group that participates in such plans and meet the terms and conditions of participation thereunder.

Transitional Matters Under NewCo Welfare Plans.

(a) Treatment of Claims Incurred.

- (i) Self-Insured Benefits. NewCo will assume and be responsible for the funding of payment for any unpaid covered claim and eligible expense:
- (A) incurred by any NewCo Welfare Plan Participant prior to the Effective Time under a SLM BankCo Welfare Plan that is not described in section 7.2(a)(ii) below, to the extent such participant has coverage under such plan as, or through, an employee or former employee of a Participating NewCo Employer on the date such claim or expense is incurred; or
- (B) incurred by any NewCo Employee or Former NewCo Employee prior to the Effective Time under a SLM BankCo Benefit Arrangement that is not described in section 7.2(a)(ii) below.

No member of the SLM BankCo Group shall be responsible for any liability with respect to any such claims or expenses.

- (ii) <u>Insured Benefits</u>. With respect to benefits that, on or prior to the Distribution Date, were provided for under the SLM BankCo Welfare Plans through the purchase of insurance, SLM BankCo shall cause the SLM BankCo Welfare Plans to fully perform, pay and discharge all claims of NewCo Welfare Plan Participants that were incurred prior to the Distribution Date.
- (iii) <u>Claims Incurred</u>. For purposes of this Section 7.2(a), a claim or liability is deemed to be incurred (A) with respect to medical, dental, vision and/or prescription drug benefits, upon the rendering of health services giving rise to such claim or liability; (B) with respect to life insurance, accidental death and dismemberment and business travel accident insurance, upon the occurrence of the event giving rise to such claim or liability; (C) with respect to long-term disability benefits, upon the date of an individual's disability, as determined by the disability benefit insurance carrier or claim administrator, giving rise to such claim or liability; and (D) with respect to a period of continuous hospitalization, upon the date of admission to the hospital, unless otherwise provided under the terms of the applicable SLM BankCo Welfare Plan or SLM BankCo Benefit Arrangement.
- (b) <u>Credit for Deductibles and Other Limits</u>. With respect to each NewCo Welfare Plan Participant, the NewCo Welfare Plans will give credit in plan year 2014 for any amount paid, number of services obtained or visits provided under the comparable type SLM BankCo Welfare Plan by such NewCo Welfare Plan Participant in plan year 2014 toward deductibles, out-of-pocket maximums, limits on number of services or visits, or other similar limitations to the extent such amounts are taken into account under the comparable type SLM BankCo Welfare Plan. For purposes of any life-time maximum benefit limit payable to a NewCo Welfare Plan Participant under any NewCo Welfare Plan, the NewCo Welfare Plans will recognize any expenses paid or reimbursed by a SLM BankCo Welfare Plan with respect to such participant on or prior to the Distribution Date to the same extent such expense payments or reimbursements would be recognized in respect of an active plan participant under that SLM BankCo Welfare Plan.

- (c) <u>COBRA</u>. Effective as of the Effective Time, NewCo has assumed and will satisfy all requirements under COBRA with respect to all NewCo Employees and Former NewCo Employees and their qualified beneficiaries, including for individuals who are already receiving benefits as of such date under COBRA.
- (d) <u>Long-Term Care Insurance</u>. Effective as of the Effective Time, any long-term care arrangements provided under a SLM BankCo Benefits Arrangement will be terminated, and NewCo shall assume responsibility for any employer obligations associated with long-term care coverage provided under a SLM BankCo Benefits Arrangement.

Continuity of Benefits, Benefit Elections and Beneficiary Designations.

- (e) <u>Benefit Elections and Designations</u>. As of the Distribution Date (or such other date provided for under subsection 7.3(b)), NewCo has caused the NewCo Welfare Plans to recognize and give effect to all elections and designations (including all coverage and contribution elections and beneficiary designations) made by each NewCo Welfare Plan Participant under, or with respect to, the corresponding SLM BankCo Welfare Plan for plan year 2014.
- (f) Additional Details Regarding Reimbursement Accounts. To the extent any NewCo Welfare Plan provides or constitutes a health care flexible spending account, dependent care flexible spending account, or health reimbursement arrangement (each a "NewCo Reimbursement Account"), such NewCo Welfare Plan shall be deemed effective as of January 1, 2014 rather than the Effective Time. It is the intention of the Parties that all activity under a NewCo Welfare Plan Participant's flexible spending account or health reimbursement arrangement with SLM BankCo for plan year 2014 be treated instead as activity under the corresponding NewCo Reimbursement Account. Accordingly, (i) any period of participation by a NewCo Welfare Plan Participant in a SLM BankCo flexible spending account or health reimbursement arrangement during plan year 2014 (the "Participation Period") will be deemed a period when the NewCo Welfare Plan Participated in the corresponding NewCo Reimbursement Account; (ii) all expenses incurred during a Participation Period will be deemed incurred while the participant's coverage was in effect under the corresponding NewCo Reimbursement Account; and (iii) all elections and reimbursements made with respect to a Participation Period under a SLM BankCo flexible spending account or health reimbursement arrangement will be deemed to have been made with respect to the corresponding NewCo Reimbursement Account. As soon as practicable following the Effective Time, BankCo will transfer to NewCo the amount of employee contributions to healthcare flexible spending accounts, dependent care flexible spending accounts, and health reimbursement arrangements collected on behalf of NewCo Welfare Plan Participants as of the Effective Time, less the amount of account benefits paid on behalf of NewCo Welfare Plan Participants as of the Effective Time exceeds the amount of contributions collected from NewCo Welfare Plan Participants, NewCo shall make a payment to BankCo in amount equal to such deficit).

(g) Employer Non-elective Contributions. As of the Distribution Date, NewCo has caused any NewCo Welfare Plan that constitutes a cafeteria plan under Section 125 of the Code to recognize and give effect to all non-elective employer contributions payable and paid toward coverage of a NewCo Welfare Plan Participant under the corresponding SLM BankCo Welfare Plan that is a cafeteria plan under Section 125 of the Code for the applicable cafeteria plan year.

Insurance Contracts. To the extent any SLM BankCo Welfare Plan is funded through the purchase of an insurance contract or is subject to any stop loss contract, SLM BankCo and NewCo will cooperate and use their commercially reasonable efforts to replicate such insurance contracts (except to the extent changes are required under applicable state insurance laws) and to maintain any pricing discounts or other preferential terms for both SLM BankCo and NewCo for a reasonable term. Neither Party shall be liable for failure to obtain such pricing discounts or other preferential terms for the other Party. Each Party shall be responsible for any additional premiums, charges or administrative fees that such Party may incur pursuant to this Section 7.4.

Third-Party Vendors. Except as provided below, to the extent any SLM BankCo Welfare Plan is administered by a third-party vendor, SLM BankCo and NewCo will cooperate and use their commercially reasonable efforts to replicate any contract with such third-party vendor and to maintain any pricing discounts or other preferential terms for both SLM BankCo and NewCo for a reasonable term. Neither Party shall be liable for failure to obtain such pricing discounts or other preferential terms for the other Party. Each Party shall be responsible for any additional premiums, charges or administrative fees that such Party may incur pursuant to this Section 7.5.

Claims Experience. Notwithstanding the foregoing, SLM BankCo and NewCo shall use commercially reasonable efforts to ensure that any claims experience under the SLM BankCo Welfare Plans attributable to NewCo Welfare Beneficiaries shall be available to the NewCo Welfare Plans, to the extent permitted by any applicable privacy protection laws or regulations or Privacy Contracts.

ARTICLE VIII

BENEFIT ARRANGEMENTS

Except as otherwise provided under this Agreement, effective as of the Distribution Date, NewCo Employees and Former NewCo Employees are no longer eligible to participate in any SLM BankCo Benefit Arrangement.

ARTICLE IX

WORKERS' COMPENSATION AND UNEMPLOYMENT COMPENSATION

Workers' Compensation. Effective as of the Distribution Date, NewCo shall have (and, to the extent it has not previously had such obligations, assume) the obligations for all claims and liabilities relating to workers' compensation benefits for all NewCo Employees and Former NewCo Employees. Effective as of the Distribution Date, SLM BankCo shall have (and, to the extent it has not previously had such obligations, assume) the obligations for all claims and liabilities relating to workers' compensation benefits for all SLM BankCo Employees. NewCo and SLM BankCo shall use commercially reasonable efforts to provide that workers' compensation insurance costs are not adversely affected for either of them by reason of the Distribution.

Unemployment Compensation. Effective as of the Distribution Date, NewCo shall have (and, to the extent it has not previously had such obligations, assume) the obligations for all claims and liabilities relating to unemployment compensation benefits for all NewCo Employees and Former NewCo Employees. NewCo shall use commercially reasonable efforts to provide that unemployment insurance costs are not adversely affected by reason of the Distribution.

ARTICLE X

RETENTION, SEVERANCE AND OTHER MATTERS

Retention Agreements.

- (a) <u>NewCo Obligations</u>. Effective as of the Distribution Date, NewCo hereby assumes Existing SLM's rights and obligations arising under any retention agreements other than those described in Schedule 10.1(b) and agrees to honor the terms and conditions of those agreements applicable to NewCo as a successor under the terms of such agreements. Except for NewCo's assumption of the retention agreements as described above, the terms of the retention agreements shall in all other respects be unaffected.
- (b) <u>SLM BankCo Obligations</u>. SLM BankCo shall continue to be responsible for and remain obligated under the retention agreements described in Schedule 10.1(b) and agrees to honor the terms and conditions of those agreements.
- (c) <u>Additional Obligations</u>. NewCo and SLM BankCo shall each be solely responsible for any other retention arrangements entered into by any member of the NewCo Group or any member of the SLM BankCo Group, respectively, and that are not otherwise allocated by this Agreement to a member of either the SLM BankCo Group or the NewCo Group.

Severance.

- (d) Except as otherwise provided in this Agreement, immediately following the Distribution, SLM BankCo shall have no liability or obligation under any SLM BankCo severance plan or policy with respect to NewCo Employees or Former NewCo Employees. NewCo shall be responsible for paying severance benefits, in accordance with a severance plan or policy that it may establish in its discretion, to individuals who transfer from employment with a member of the SLM BankCo Group to employment with a member of the NewCo Group in connection with the Distribution and who are involuntarily terminated following the Distribution Date.
- (e) Except as otherwise provided in this Agreement, effective after the Distribution Date, NewCo shall assume and shall be responsible for administering all payments and benefits under the applicable SLM BankCo severance policies or any termination agreements with Former NewCo Employees whose employment terminated prior to the Distribution Date for an eligible reason under such policies or in accordance with such agreements.

Accrued Time Off. NewCo shall recognize and assume all liability for all vacation, holiday, sick leave, flex days, personal days and paid-time off with respect to NewCo Employees accrued prior to the Distribution Date, and NewCo shall credit each NewCo Employee with such accrual.

Leaves of Absence. Employees of the Group on approved leave of absence as of the time immediately prior to the Distribution Date will be allocated to and assigned to be an employee of either the NewCo Group or the BankCo Group as of the Effective Time. NewCo will continue to apply the appropriate leave of absence policies applicable to inactive NewCo Employees who are on an approved leave of absence as of the Distribution Date and the terms of any such leave policy for such employees shall be honored by NewCo. Leaves of absence taken by NewCo Employees prior to the Distribution Date shall be deemed to have been taken as employees of a member of the NewCo Group.

Director Programs. SLM BankCo shall retain responsibility for the payment of any fees payable in respect of service on the SLM BankCo board of directors that are payable but not yet paid as of the Distribution Date, and NewCo shall not have any responsibility for any such payments.

Restrictive Covenants in Employment and Other Agreements.

(f) To the fullest extent permitted by the agreements described in this Section 10.6(a) and applicable law, SLM BankCo shall assign, or cause any member of the SLM BankCo Group to assign, to NewCo or a member of the NewCo Group, as designated by NewCo, all agreements containing restrictive covenants (including confidentiality and non-competition provisions) between a member of the SLM BankCo Group and a NewCo Employee or Former NewCo Employee, with such assignment effective as of the Effective Time. To the extent that assignment of such agreements is not permitted, effective as of the Effective Time, each member of the NewCo Group shall be considered to be a successor to each member of the SLM BankCo Group for purposes of, and a third-party beneficiary with respect to, all agreements containing restrictive covenants (including confidentiality and non-competition provisions) between a member of the SLM BankCo Group and a NewCo Employee or Former NewCo Employee whom NewCo reasonably determines has substantial knowledge of the business activities of the NewCo Group, such that each member of the NewCo Group shall enjoy all the rights and benefits under such agreements (including rights and benefits as a third-party beneficiary), with respect to the business operations of the NewCo Group; provided, however, that in no event shall SLM BankCo be permitted to enforce such restrictive covenant agreements against NewCo Employees or Former NewCo Employees for action taken in their capacity as employees of a member of the NewCo Group.

(g) To the fullest extent permitted by the agreements described in this Section 10.6(b) and applicable law, NewCo shall assign, or cause any member of the NewCo Group to assign, to SLM BankCo or a member of the SLM BankCo Group, as designated by SLM BankCo, all agreements containing restrictive covenants (including confidentiality and non-competition provisions) between a member of the NewCo Group and a SLM BankCo Employee, with such assignment effective as of the Effective Time. To the extent that assignment of such agreements is not permitted, effective as of the Effective Time, each member of the SLM BankCo Group shall be considered to be a successor to each member of the NewCo Group for purposes of, and a third-party beneficiary with respect to, all agreements containing restrictive covenants (including confidentiality and non-competition provisions) between a member of the NewCo Group and a

SLM BankCo Employee whom SLM BankCo reasonably determines has substantial knowledge of the business activities of the SLM BankCo Group, such that SLM BankCo and each member of the SLM BankCo Group shall enjoy all the rights and benefits under such agreements (including rights and benefits as a third-party beneficiary), with respect to the business operations of the SLM BankCo Group; provided, however, that in no event shall NewCo be permitted to enforce such restrictive covenant agreements against SLM BankCo Employees for action taken in their capacity as employees of a member of the SLM BankCo Group.

ARTICLE XI

LEGACY LIABILITY

Legacy Pension Plan. NewCo shall assume, and no member of the SLM BankCo Group shall have any further liability for or under, the Sallie Mae Cash Account Retirement Pension Plan liabilities ("Legacy Pension Liabilities"), and NewCo shall indemnify each member of the SLM BankCo Group, and the officers, directors, and employees of each member of the SLM BankCo Group, and hold them harmless as against the Legacy Pension Liabilities.

Scholarships. SLM BankCo shall assume liability for providing previously awarded company-sponsored scholarship benefits to SLM BankCo Employees, Former BankCo Employees and their children, as applicable, and NewCo shall assume liability for providing previously awarded company-sponsored scholarship benefits to NewCo Employees, Former NewCo Employees and their children, as applicable.

ARTICLE XII

GENERAL PROVISIONS

Preservation of Rights to Amend. The rights of each member of the SLM BankCo Group and each member of the NewCo Group to amend, waive, or terminate any plan, arrangement, agreement, program, or policy referred to herein shall not be limited in any way by this Agreement.

Confidentiality. Each Party agrees that any information conveyed or otherwise received by or on behalf of a Party in conjunction herewith that is not otherwise public through no fault of such Party is confidential and is subject to the terms of the confidentiality provisions set forth in the Separation Agreement.

Administrative Complaints/Litigation. Except as otherwise provided in this Agreement, on and after the Distribution Date, NewCo shall assume, and be solely liable for, the handling, administration, investigation and defense of actions, including ERISA, occupational safety and health, employment standards, union grievances, wrongful dismissal, discrimination or human rights and unemployment compensation claims asserted at any time against SLM BankCo or any member of the SLM BankCo Group by any NewCo Employee or Former NewCo Employee (including any dependent or beneficiary of any such Employee) or any other person, to the extent such actions or claims arise out of or relate to employment or the provision of services (whether as an employee, contractor, consultant or otherwise) to or with respect to the business activities of any member of the NewCo Group, whether or not such employment or services were

performed before or after the Distribution. To the extent that any legal action relates to a putative or certified class of plaintiffs, which includes both SLM BankCo Employees and NewCo Employees (or Former NewCo Employees) and such action involves employment or benefit plan related claims, reasonable costs and expenses incurred by the Parties in responding to such legal action shall be allocated among the Parties equitably in proportion to a reasonable assessment of the relative proportion of employees included in or represented by the putative or certified plaintiff class. The procedures contained in the indemnification and related litigation cooperation provisions of the Separation Agreement shall apply with respect to each Party's indemnification obligations under this Section 12.3.

Reimbursement and Indemnification. SLM BankCo and NewCo hereto agrees to reimburse the other Party, within 60 days of receipt from the other Party of reasonable verification, for all costs and expenses which the other Party may incur on its behalf as a result of any of the respective SLM BankCo and NewCo Welfare Plans, Thrift Plan and Benefit Arrangements and, as contemplated by Section 10.2, any termination or severance payments or benefits. All liabilities retained, assumed or indemnified against by NewCo pursuant to this Agreement, and all liabilities retained, assumed or indemnified against by SLM BankCo pursuant to this Agreement, shall in each case be subject to the indemnification provisions of the Separation Agreement. Notwithstanding anything to the contrary, (i) no provision of this Agreement shall require any member of the NewCo Group to pay or reimburse to any member of the SLM BankCo Group any benefit-related cost item that a member of the NewCo Group has previously paid or reimbursed to any member of the NewCo Group any benefit-related cost item that a member of the SLM BankCo Group has previously paid or reimbursed to any member of the NewCo Group any benefit-related cost item that a member of the SLM BankCo Group has previously paid or reimbursed to any member of the NewCo Group any benefit-related cost item that a member of the SLM BankCo Group has previously paid or reimbursed to any member of the NewCo Group.

Costs of Compliance with Agreement. Except as otherwise provided in this Agreement or any other Ancillary Agreement, each Party shall pay its own expenses in fulfilling its obligations under this Agreement.

Fiduciary Matters. SLM BankCo and NewCo each acknowledge that actions required to be taken pursuant to this Agreement may be subject to fiduciary duties or standards of conduct under ERISA or other applicable law, and no Party shall be deemed to be in violation of this Agreement if it fails to comply with any provisions hereof based upon its good faith determination (as supported by advice from counsel experienced in such matters) that to do so would violate such a fiduciary duty or standard. Each Party shall be responsible for taking such actions as are deemed necessary and appropriate to comply with its own fiduciary responsibilities and shall fully release and indemnify the other Party for any liabilities caused by the failure to satisfy any such responsibility.

Entire Agreement. This Agreement, together with the documents referenced herein (including the Separation Agreement, the Ancillary Agreements and the plans and agreements referenced herein), constitutes the entire agreement and understanding among the Parties with respect to the subject matter hereof and supersedes all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof. To the extent any provision of this Agreement conflicts with the provisions of the Separation Agreement, the provisions of this Agreement shall be deemed to control with respect to the subject matter hereof.

Binding Effect; No Third-Party Beneficiaries; Assignment. This Agreement shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. This Agreement is solely for the benefit of the Parties and should not be deemed to confer upon any third parties any remedy, claim, liability, reimbursement, cause of action or other right in excess of those existing without reference to this Agreement. Nothing in this Agreement is intended to amend any employee benefit plan or affect the applicable plan sponsor's right to amend or terminate any employee benefit plan pursuant to the terms of such plan. Except as otherwise provided in Section 10.1(a), the provisions of this Agreement are solely for the benefit of the Parties, and no current or former Employee, officer, director or independent contractor or any other individual associated therewith shall be regarded for any purpose as a third-party beneficiary of this Agreement. This Agreement may not be assigned by any Party, except with the prior written consent of the other Parties.

Amendment; Waivers. No change or amendment may be made to this Agreement except by an instrument in writing signed on behalf of each of the Parties. Any Party may, at any time, (i) extend the time for the performance of any of the obligations or other acts of another Party, (ii) waive any inaccuracies in the representations and warranties of another Party contained herein or in any document delivered pursuant hereto, and (iii) waive compliance by another Party with any of the agreements, covenants or conditions contained herein. Any such extension or waiver shall be valid only if set forth in an instrument in writing signed by the Party to be bound thereby. No failure or delay on the part of any Party in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty, covenant or agreement contained herein, nor shall any single or partial exercise of any such right preclude other or further exercises thereof or of any other right.

Remedies Cumulative. All rights and remedies existing under this Agreement or the Schedules attached hereto are cumulative to, and not exclusive of, any rights or remedies otherwise available.

Notices. Unless otherwise expressly provided herein, all notices, claims, certificates, requests, demands and other communications hereunder shall be in writing and shall be deemed to be duly given: (i) when personally delivered, (ii) if mailed by registered or certified mail, postage prepaid, return receipt requested, on the date the return receipt is executed or the letter is refused by the addressee or its agent, (iii) if sent by overnight courier which delivers only upon the executed receipt of the addressee, on the date the receipt acknowledgment is executed or refused by the addressee or its agent, or (iv) if sent by facsimile or electronic mail, on the date confirmation of transmission is received (provided that a copy of any notice delivered pursuant to this clause (iv) shall also be sent pursuant to clause (i), (ii) or (iii)), addressed to the attention of the addressee's General Counsel at the address of its principal executive office or to such other address or facsimile number for a Party as it shall have specified by like notice.

Counterparts. This Agreement may be executed in multiple counterparts, each of which when executed shall be deemed to be an original but all of which together shall constitute one and the same agreement.

Severability. If any term or other provision of this Agreement or the Schedules attached hereto is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the court, administrative agency or arbitrator shall interpret this Agreement so as to affect the original intent of the Parties as closely as possible in an acceptable manner to the end that transactions contemplated hereby are fulfilled to the fullest extent possible. If any sentence in this Agreement is so broad as to be unenforceable, the provision shall be interpreted to be only so broad as is enforceable.

Governing Law. To the extent not preempted by applicable federal law, this Agreement shall be governed by, and construed and enforced in accordance with, the substantive laws of the State of Delaware, without regard to any conflicts of law provisions thereof that would result in the application of the laws of any other jurisdiction.

Performance. Each of SLM BankCo and NewCo shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any member of the SLM BankCo Group and any member of the NewCo Group, respectively. The Parties each agree to take such further actions and to execute, acknowledge and deliver, or to cause to be executed, acknowledged and delivered, all such further documents as are reasonably requested by the other for carrying out the purposes of this Agreement or of any document delivered pursuant to this Agreement.

Construction. This Agreement shall be construed as if jointly drafted by the Parties and no rule of construction or strict interpretation shall be applied against any Party.

Effect if Distribution Does Not Occur. Notwithstanding anything in this Agreement to the contrary, if the Separation Agreement is terminated prior to the Distribution Date, this Agreement shall be of no further force and effect.

[Signature Page Follows]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed in their names by a duly authorized officer as of the date first written above.

NEW BLC CORPORATION

By: /s/ John F. Remondi

Name: John F. Remondi Title: Chief Executive Officer

NAVIENT CORPORATION

By: /s/ Raymond Quinlan

Name: Raymond Quinlan Title: Chief Executive Officer

Schedules to the Employee Matters Agreement

Schedule 3.3(b) RSU Awards to be Converted via Concentration Method

One-time equity grant of restricted stock units issued to Raymond J. Quinlan on January 21, 2014

Schedule 6.1 Participating Employers as of the Effective Time

Participating employers in the NewCo Thrift Plan and NewCo Welfare Plan:

Asset Performance Group, LLC

Navient Solutions, Inc. (successor to Sallie Mae, Inc.)

Navient Investment Corporation

Student Assistance Corporation

Navient Education Loan Corp.

Navient DE Corporation

Navient Credit Finance Corporation

RKL Financial Corporation

Pioneer Credit Recovery, Inc.

Student Outreach Solutions, Inc.

General Revenue Corporation

Participating employers in the SLM BankCo Thrift Plan and SLM BankCo Welfare Plan:

Upromise, Inc.

Sallie Mae Bank

SMB Servicing Company, Inc.

Schedule 10.1(b) Retention Arrangements Assumed by SLM BankCo

Employment Agreement, dated as of April 21, 2014, by and between Laurent C. Lutz, SLM Corporation, and New BLC Corporation

Letter Agreement, dated April 15, 2014, by and between Joeseph A. DePaulo, SLM Corporation, and New BLC Corporation

Letter Agreement, dated January 15, 2014, by and between Raymond J. Quinlan and SLM Corporation

Exhibit 10.3

Execution Copy

TAX SHARING AGREEMENT

BETWEEN

NAVIENT CORPORATION

AND

NEW BLC CORPORATION

DATED AS OF APRIL 29, 2014

TABLE OF CONTENTS

Article I Definition of Terms		1
Article II Allocati	ion of Tax Liabilities and Tax Benefits	10
Section 2.1	Liability for and Payment of Taxes	10
(a)	Navient Liabilities and Payments	10
(b)	SLM BankCo Liabilities and Payments	11
Section 2.2	Special Rules for Certain Tax Items	11
(a)	Spin-Off Taxes and Spin-Off Tax Items	11
(b)	Bank Loan Sales	11
(c)	Section 108(i) Income	11
	ation and Filing of Tax Returns	12
Section 3.1	Joint Returns	12
(a)	Preparation and Filing	12
(b)	Draft Joint Returns	12
(c)	Provision of Information	13
(d)	Information with Respect to Estimated Payments and Extension Payments	13
(e)	Provision of Assistance with Respect to Joint Returns	13
(f)	Engagement Letter for Certain 2014 Tax Returns	14
(g)	Allocation of Third-Party Preparer Expenses	14
Section 3.2	Separate Returns	15
(a)	Separate Returns to be Prepared by SLM BankCo	15
(b)	Separate Returns to be Prepared by Navient	15
(c)	Provision of Information and Assistance	15
	Additional Rules Relating to the Preparation of Tax Returns	15
(a)	General Rule	15
(b)	Navient Separate Returns	16
(c)	SLM BankCo Separate Returns	16
(d)	Election to File Consolidated, Combined or Unitary Tax Returns	16
(e)	Standard of Performance	16
Section 3.4	Reliance on Exchanged Information	16
Section 3.5	Allocation of Tax Items	17
Article IV Tax Pa	•	17
Section 4.1	Payment of Taxes to Tax Authority	17
Section 4.2	Indemnification Payments	17
(a)	Tax Payments	17
(b)	Credit for Prior Tax Payments	18
Section 4.3	Initial Determinations and Subsequent Adjustments	18
Section 4.4	Interest on Late Payments	19
Section 4.5	Payments by or to Other Group Members	19
Section 4.6	Procedural Matters	19
Section 4.7	Tax Consequences of Payments; Tax Gross-Up	19
	i	

Article V Assistan	ce and Cooperation	20
Section 5.1	Cooperation	20
Section 5.1 Section 5.2	Supplemental Rulings and Supplemental Tax Opinions	20
Article VI Tax Rec	21	
Section 6.1	Retention of Tax Records	21
Section 6.2	Access to Tax Records	21
Section 6.3	Confidentiality	21
Article VII Tax Co	ontests	21
Section 7.1	Notices	21
Section 7.2	Control of Tax Contests	22
(a)	General Rule	22
(b)	Participation and Settlement Rights	22
Section 7.3	Cooperation	23
Article VIII Restric	ction on Certain Actions of SLM BankCo and Navient	23
Section 8.1	General Restrictions	23
Section 8.2	Certain Navient Actions Following the Effective Time	23
Section 8.3	Certain SLM BankCo Actions Following the Effective Time	24
	C	
Article IX General		25
Section 9.1	Counterparts; Corporate Power	25
Section 9.2	Governing Law	26
Section 9.3	Assignability	26
Section 9.4	Third-Party Beneficiaries	26
Section 9.5	Notices	26
Section 9.6	Severability	26
Section 9.7	Force Majeure	27
Section 9.8	Expenses	27
Section 9.9	Headings	27
Section 9.10		27
Section 9.11	Waivers of Default	27
Section 9.12		27
Section 9.13	1	27
Section 9.14	•	28
Section 9.15		28
Section 9.16		28
Section 9.17		28
Section 9.18	ϵ	28
Section 9.19	Disputes	28

TAX SHARING AGREEMENT

This TAX SHARING AGREEMENT, dated as of April 29, 2014 (this "Agreement"), is by and between Navient Corporation, a Delaware corporation ("Navient") and New BLC Corporation, a Delaware corporation ("SLM BankCo"). Unless otherwise defined in this Agreement, all capitalized terms used in this Agreement have the meanings set forth in the Separation and Distribution Agreement, dated as of April 28, 2014, by and among SLM Corporation, a Delaware corporation ("SLM"), SLM BankCo and Navient (as amended, modified or supplemented from time to time in accordance with its terms, the "Separation Agreement").

RECITALS

WHEREAS, the Existing SLM Board has determined that it is in the best interests of Existing SLM and its shareholders to separate into two separate publicly-traded companies;

WHEREAS, in furtherance of the foregoing, the Existing SLM Board has determined that it is appropriate and desirable to undergo the Separation and Distribution;

WHEREAS, SLM, SLM BankCo and Navient have entered into the Separation Agreement; and

WHEREAS, the parties desire to provide for and agree upon the allocation between the parties of Taxes and Tax Benefits arising prior to, as a result of, and subsequent to the External Spin-Off, and provide for and agree upon other matters relating to Taxes.

NOW, THEREFORE, in consideration of the foregoing and the covenants and agreements set forth below, the parties hereto agree as follows:

Article I

Definition of Terms

For purposes of this Agreement (including the recitals hereof), the following terms have the following meanings:

"2013 Engagement Letter" means the engagement letter between KPMG LLP and SLM (to be assigned by SLM to SLM BankCo in connection with the External Spin-Off), executed on March 12, 2014, pursuant to which SLM engages KPMG LLP to assist in preparing certain U.S. federal, state and local income, franchise and similar Tax Returns for taxable periods ending during 2013.

"2014 Accountant" has the meaning set forth in Section 3.1(f).

"2014 Engagement Letter" has the meaning set forth in Section 3.1(f).

"Affiliate" has the meaning set forth in the Separation Agreement.

- "Agreement" has the meaning set forth in the preamble hereof.
- "Ancillary Agreements" has the meaning set forth in the Separation Agreement.
- "Applicable Date" has the meaning set forth in Section 4.2(a).
- "Bank Loan Sales" means any sales or transfers on or prior to the Spin-Off Date of student loans by Sallie Mae Bank to (or treated as being made to, for U.S. federal income tax purposes) a Person that is a member of the Navient Group as of the Spin-Off Date (or any predecessor of such entity).
- "Business Day" means any day other than a Saturday, a Sunday or a day on which commercial banking institutions in Delaware are authorized or obligated by law or executive order to be closed.
 - "Change of Control" has the meaning set forth in the Separation Agreement.
 - "Code" means the U.S. Internal Revenue Code of 1986, as amended from time to time, or any successor law.
 - "Consolidated Return" means (i) any Joint Return and (ii) any other Tax Return that reflects Consolidated Taxes.
- "Consolidated Tax" means (i) any Tax, for any Tax Year, which is based on or determined by reference to Tax Items, operations, activities or assets of both the Navient Operations and the SLM BankCo Operations, and (ii) any Spin-Off Taxes.
 - "Consumer Banking Business" has the meaning set forth in the Ruling Request.
- "Controlling Party" means, with respect to a Tax Contest, the party that controls such Tax Contest (or whose Group member controls such Tax Contest) pursuant to Section 7.2(a).
- "Control" means, with respect to any Person, the possession, directly or indirectly, of the power to direct or cause the direction of the management or policies of such Person, whether through ownership of securities or partnership, membership, limited liability company, or other ownership interests, by contract or otherwise, and the terms "Controlling" and "Controlled" have meanings correlative to the foregoing.
 - "Dispute" has the meaning set forth in the Separation Agreement.
 - "Due Date" has the meaning set forth in Section 4.4.
 - "Education Loan Management Business" has the meaning set forth in the Ruling Request.
 - "Effective Time" means the time at which the External Spin-Off is effective on the Spin-Off Date.
 - "External Contribution" has the meaning set forth in the Ruling Request.

- "External Spin-Off" has the meaning set forth in the Ruling Request.
- "Governmental Entity" means any national, supranational, federal, state, provincial, local or similar government; any instrumentality or subdivision thereof; any court, administrative agency, department, board, bureau or commission thereof or other governmental authority or instrumentality; or any quasi-governmental or private body authorized to exercise any tax, governmental or quasi-governmental authority.
 - "Group" means the SLM BankCo Group or the Navient Group, as the context requires.
 - "Information" has the meaning set forth in the Separation Agreement.
 - "Internal Contribution" has the meaning set forth in the Ruling Request.
 - "Internal Controlled" has the meaning set forth in the Ruling Request.
 - "Internal Controlled Business" means the "Bank Private Education Loan Servicing Business" as defined in the Ruling Request.
 - "Internal Distributing" has the meaning set forth in the Ruling Request.
 - "Internal Distributing Business" means the "Loan Servicing Business" as defined in the Ruling Request.
 - "Internal Spin-Off" has the meaning set forth in the Ruling Request.
 - "IRS" means the Internal Revenue Service.
- "Joint Return" means any Tax Return, for any Tax Year, that includes Tax Items, operations, activities or assets of both the SLM BankCo Operations and the Navient Operations, determined without regard to Tax Items carried forward to such Tax Year.
- "Law" means any national, supranational, federal, state, provincial, local or similar law (including common law), statute, code, order, ordinance, rule, regulation, treaty (including any income tax treaty), license, permit, authorization, approval, consent, decree, injunction, binding judicial or administrative interpretation or other requirement, in each case, enacted, promulgated, issued or entered by a Governmental Entity.
 - "Navient" has the meaning set forth in the preamble hereof.
 - "Navient Business" means the "Pre-Separation Education Loan Management Business" as defined in the Separation Agreement.
 - "Navient Group" has the meaning set forth in the Separation Agreement.
 - "Navient Joint Return" has the meaning set forth in Section 3.1(a).

"Navient Net Obligation" with respect to any Consolidated Tax for any Tax Year (or portion thereof) as of any date, means the amount equal to (i) the Navient Tax Amount for such Consolidated Tax for such Tax Year (or portion thereof), plus (ii) the gross amount (without duplication of any amounts) remitted pursuant to this Agreement (or deemed to be remitted pursuant to Section 4.2(b) of this Agreement) to Navient by SLM BankCo on or prior to such date, or refunded to the Navient Group by the applicable Tax Authority after the Effective Time but on or prior to such date or refunded to the SLM group at or prior to the Effective Time (and not treated as refunded to the SLM BankCo Group under the Separate Return Method applied as of the Effective Time), for such Consolidated Tax for such Tax Year (or portion thereof), minus (iii) the gross amount (without duplication of any amounts) remitted pursuant to this Agreement (or deemed to be remitted pursuant to Section 4.2(b) of this Agreement) by Navient to SLM BankCo or the applicable Tax Authority for such Consolidated Tax for such Tax Year (or portion thereof) on or prior to such date.

"Navient Operations" means (i) with respect to any Tax Year (or portion thereof) that ends on or before the Spin-Off Date, the assets (other than any SLM BankCo Assets), business, operations and activities of the Navient Business, and (ii) with respect to any Tax Year (or portion thereof) that begins after the Spin-Off Date, the assets, business, operations and activities of the Navient Group.

"Navient Separate Return" has the meaning set forth in Section 3.2(b).

"Navient Separate Tax" means any Tax (other than any Spin-Off Tax), for any Tax Year, which (i) is based solely on or determined solely by reference to Tax Items, operations, activities or assets of the Navient Operations, and (ii) is not based on or determined by reference to any Tax Items, operations, activities or assets of the SLM BankCo Operations.

"Navient Tax Amount" with respect to any Consolidated Tax (whether or not the actual amount of such Tax is positive) for any Tax Year (or portion thereof), means the amount equal to (i) the actual amount of such Consolidated Tax for such Tax Year (or portion thereof) minus (ii) the SLM BankCo Tax Amount for such Tax Year (or portion thereof). For the avoidance of doubt, the Navient Tax Amount may be a negative number.

"Navient Tax Obligation" with respect to any Consolidated Tax (whether or not the actual amount of such Tax is positive) for any Tax Year (or portion thereof), means the greater of (i) zero and (ii) the Navient Tax Amount for such Tax for such Tax Year (or portion thereof).

"NOL" means net operating loss.

"Non-Controlling Party" means, with respect to a Tax Contest, the party that is not the Controlling Party with respect to such Tax Contest.

"Non-Preparer" means, with respect to a Tax Return, the party that is not responsible for the preparation of such Tax Return under Section 3.1 or Section 3.2.

"Non-Preparer Item" has the meaning set forth in Section 7.2(b).

"Payment Date" means (i) with respect to any U.S. federal income tax return, the due date for any required installment of estimated taxes determined under Code Section 6655, the due date (determined without regard to extensions) for filing the return determined under Code Section 6072, and the date the return is filed, and (ii) with respect to any other Tax Return, the corresponding dates determined under applicable Tax Law.

"Permitted Supplemental Ruling" means, with respect to a specified action, a Supplemental Ruling to the effect that such action will not preclude (i) the Reorganization from qualifying as a reorganization within the meaning of Section 368(a)(1)(F) of the Code, (ii) the Internal Contribution and the Internal Spin-Off from qualifying as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code and (iii) the External Contribution and the External Spin-Off from qualifying as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code (except, in the case of holders of SLM stock, with respect to cash received in lieu of fractional shares).

"Person" means any individual, corporation, company, partnership, trust, incorporated or unincorporated association, joint venture or other entity of any kind.

"Preparer" means, with respect to a Tax Return, the party that is responsible for the preparation of such Tax Return pursuant to Section 3.1 or Section 3.2.

"Prime Rate" has the meaning set forth in the Separation Agreement.

"Reorganization" has the meaning set forth in the Ruling Request.

"Requesting Party" has the meaning set forth in Section 5.2.

"Return Signer" has the meaning set forth in Section 3.1(a).

"Ruling" means the private letter ruling (if any) in connection with the Separation and Distribution issued by the IRS in response to the Ruling Request.

"Ruling Request" means (1) the request for ruling, dated August 21, 2013, filed on behalf of SLM with the IRS in connection with the Separation and Distribution, (2) the supplements to ruling request, dated November 27, 2013, March 25, 2014, and March 27, 2014, filed on behalf of SLM with the IRS in connection with the Separation and Distribution, (3) the document titled "Description of GLBA Issues and the Bank Regulatory Concept of Control," dated March 17, 2014, and the accompanying exhibits, filed on behalf of SLM with the IRS in connection with the Separation and Distribution and (4) any other correspondence or supplemental materials submitted to the IRS in connection with the request to obtain a private letter ruling pursuant to the foregoing submissions.

"Sallie Mae Bank" means Sallie Mae Bank, a Utah industrial bank and insured depository institution.

"Section 108(i) Income" means any income that both (i) arose in connection with the reacquisition after December 31, 2008, and before January 1, 2011, of a debt instrument or a swap unwind related to such a reacquisition, and (ii) was deferred by the SLM consolidated group or any member thereof under Section 108(i) of the Code.

"Section 108(i) Installment Date" for the first, second, third or fourth quarter of a calendar year, means the due date by which SLM BankCo's first, second, third or fourth installment, respectively, of estimated U.S. federal income taxes for such year is required to be paid; provided that if SLM BankCo is not a calendar-year taxpayer or is not required to pay estimated U.S. federal income taxes for any year, the Section 108(i) Installment Dates shall be determined as though SLM BankCo were a calendar-year taxpayer required to make estimated U.S. federal income tax payments for such year. For the avoidance of doubt, as of the date of this Agreement, the Section 108(i) Installment Date (i) for the first quarter of a calendar year is generally April 15 of such year, (iii) for the second quarter of a calendar year is generally June 15 of such year, (iii) for the third quarter of a calendar year is generally September 15 of such year, and (iv) for the fourth quarter of a calendar year is generally December 15 of such year, except in each case to the extent the applicable date falls on a Saturday, Sunday or legal holiday, in which case the Section 108(i) Installment Date is generally the next business day after such date.

"Section 108(i) Quarterly Amount" for any calendar quarter from and including second quarter 2014 through and including fourth quarter 2018, means the amount equal to the sum of (i) the Section 108(i) Quarterly Federal Amount for such quarter and (ii) one hundred thirty thousand five hundred dollars (\$130,500).

"Section 108(i) Quarterly Federal Amount" for any calendar quarter from and including second quarter 2014 through and including fourth quarter 2018, means the amount equal to (i) fourteen million seven hundred seventy-seven thousand seven hundred dollars (\$14,777,700), multiplied by (ii) the Section 108(i) Tax Factor for such quarter.

"Section 108(i) Tax Factor" for any calendar quarter means one (1); provided that, if the highest statutory marginal U.S. federal income tax rate applicable to ordinary income of a U.S. corporation is changed under applicable Law to a new rate after the date of this Agreement, and such new rate is applicable for the calendar year that includes such quarter, the Section 108(i) Tax Factor for such quarter shall be equal to (i) the highest statutory marginal U.S. federal income tax rate applicable to ordinary income of a U.S. corporation for the calendar year that includes such quarter (expressed as a percentage) divided by (ii) thirty-five percent (35%). For the avoidance of doubt, as of the date of this Agreement, the highest statutory marginal U.S. federal income tax rate applicable to ordinary income of a U.S. corporation (within the meaning of this definition) is thirty-five percent (35%).

"Separate Return" means any Tax Return that is not a Joint Return.

"Separate Return Method" means the principles and methodology used to calculate net taxes payable/receivable by the SLM BankCo Group for purposes of the audited carve-out financial statements of SLM BankCo and its subsidiaries for 2011 through 2014, as adjusted pursuant to Section 2.2 and the indemnification provisions of Article III. For purposes of applying the Separate Return Method under this Agreement, the Section 108(i) Quarterly Amounts shall not be treated as tax receivables or otherwise taken into account.

- "Separation Agreement" has the meaning set forth in the preamble hereof.
- "Separation Effective Time" means the "Effective Time" as defined in the Separation Agreement.
- "SLM" has the meaning set forth in the preamble hereof.
- "SLM BankCo" has the meaning set forth in the preamble hereof.
- "SLM BankCo Business" means the "Pre-Separation Consumer Banking Business" as defined in the Separation Agreement.
- "SLM BankCo Group" has the meaning set forth in the Separation Agreement.
- "SLM Bankco Joint Return" has the meaning set forth in Section 3.1(a).
- "SLM BankCo Net Obligation" with respect to any Consolidated Tax for any Tax Year (or portion thereof) as of any date, means the amount equal to (i) the SLM BankCo Tax Amount for such Consolidated Tax for such Tax Year (or portion thereof), plus (ii) the gross amount (without duplication of any amounts) remitted pursuant to this Agreement (or deemed to be remitted pursuant to Section 4.2(b) of this Agreement) to SLM BankCo by Navient on or prior to such date, or refunded to the SLM BankCo Group by the applicable Tax Authority after the Effective Time but on or prior to such date or treated as refunded to the SLM BankCo Group under the Separate Return Method applied as of the Effective Time, for such Consolidated Tax for such Tax Year (or portion thereof), minus (iii) the gross amount (without duplication of any amounts) remitted pursuant to this Agreement (or deemed to be remitted pursuant to Section 4.2(b) of this Agreement) by SLM BankCo to Navient or the applicable Tax Authority for such Consolidated Tax for such Tax Year (or portion thereof) on or prior to such date.
- "SLM BankCo Operations" means (i) with respect to any Tax Year (or portion thereof) that ends on or before the Spin-Off Date, the assets (other than any Navient Assets), business, operations and activities of the SLM BankCo Business, and (ii) with respect to any Tax Year (or portion thereof) that begins after the Spin-Off Date, the assets, business, operations and activities of the SLM BankCo Group.
 - "SLM BankCo Separate Return" has the meaning set forth in Section 3.2(a).
- "SLM BankCo Separate Tax" means any Tax (other than any Spin-Off Tax), for any Tax Year, which (i) is based solely on or determined solely by reference to Tax Items, operations, activities or assets of the SLM BankCo Operations, and (ii) is not based on or determined by reference to any Tax Items, operations, activities or assets of the Navient Operations.
- "SLM BankCo Tax Amount" with respect to any Consolidated Tax (whether or not the actual amount of such Tax is positive) for any Tax Year (or portion thereof), means the amount of the net taxes payable/receivable of the SLM BankCo Group with respect to such Tax and Tax Year (or portion thereof) determined under the Separate Return Method (not taking into account any payments actually made or deemed to be made by the SLM group, Navient Group or

SLM BankCo Group with respect to such Tax). For the avoidance of doubt, the amount of the SLM BankCo Tax Amount with respect to a Consolidated Tax for a Tax Year may be negative if, for example, Tax Benefits allocable to SLM BankCo are used to reduce the amount of such Consolidated Tax for such Tax Year.

"SLM BankCo Tax Obligation" with respect to any Consolidated Tax (whether or not the actual amount of such Tax is positive) for any Tax Year (or portion thereof), means the greater of (i) zero and (ii) the SLM BankCo Tax Amount for such Tax for such Tax Year (or portion thereof).

"Spin-Off Date" means the date on which the External Spin-Off occurs.

"Spin-Off Tax Payment" means any payment (or portion thereof) required to be made by one party to the other party under this Agreement (including any indirect payment made under this Agreement by means of the first party satisfying a Tax liability imposed by applicable Tax Law on such other party without reimbursement by such other party) to the extent such payment arises as a result of an allocation of Spin-Off Taxes or Spin-Off Tax Items pursuant to Section 2.2(a)(i) or Section 2.2(a)(ii); provided that any additional amounts payable pursuant to Section 4.7 shall not be treated as Spin-Off Tax Payments.

"Spin-Off Taxes" means any Taxes of the SLM group, SLM BankCo Group or Navient Group or any of their respective members resulting from the Reorganization, Internal Contribution, External Contribution, Internal Spin-Off or External Spin-Off, or the contribution of all of the outstanding capital stock of Private ServiceCo to Sallie Mae Bank (pursuant to Section 2.1(q) of the Separation Agreement), excluding any Taxes resulting from any Bank Loan Sale.

"Spin-Off Tax Items" means any Tax Items of the SLM group, SLM BankCo Group or Navient Group or any of their respective members resulting from the Reorganization, Internal Contribution, External Contribution, Internal Spin-Off or External Spin-Off, or the contribution of all of the outstanding capital stock of Private ServiceCo to Sallie Mae Bank (pursuant to Section 2.1(q) of the Separation Agreement), excluding any gain, income or other Tax Item from any Bank Loan Sale.

"Subsidiary" has the meaning set forth in the Separation Agreement.

"Supplemental IRS Submissions" means any request for a Supplemental Ruling and each supplemental submission and other correspondence and supplemental materials submitted to the IRS in connection with obtaining any Supplemental Ruling.

"Supplemental Ruling" means any private letter ruling obtained by SLM, SLM BankCo or Navient from the IRS which supplements or otherwise modifies the Ruling.

"Supplemental Tax Opinion" means, with respect to a specified action, an opinion (other than the Tax Opinion) from Tax Counsel to the effect that such action will not preclude (i) the Reorganization from qualifying as a reorganization within the meaning of Section 368(a)(1)(F) of the Code, (ii) the Internal Contribution and the Internal Spin-Off from qualifying as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code and (iii) the

External Contribution and the External Spin-Off from qualifying as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code (except, in the case of holders of SLM stock, with respect to cash received in lieu of fractional shares). No opinion relied upon by Navient or SLM BankCo to satisfy the requirements of Section 8.2 or Section 8.3, respectively, shall be considered a "Supplemental Tax Opinion" unless such opinion is, in addition to the requirements above, reasonably satisfactory to SLM BankCo (in the case of an opinion provided under Section 8.2) or Navient (in the case of an opinion provided under Section 8.3), which opinion may rely upon a Supplemental Ruling and may rely upon, and may assume the accuracy of, any representations given in any Supplemental IRS Submission and any customary representations contained in an officer's certificate delivered by an officer of SLM BankCo or Navient to Tax Counsel.

"Tax Event" has the meaning set forth in Section 4.2(a).

"Tax" or "Taxes" means all forms of taxation imposed by a Governmental Entity, whenever created or imposed, and whether of the United States or elsewhere, and whether imposed by a local, municipal, state, national, federal, or other body, and without limiting the generality of the foregoing, shall include income, sales, use, ad valorem, gross receipts, value added, franchise, transfer, recording, withholding, payroll, employment, excise, occupation, premium, and property taxes, together with any related interest, penalties, additions to tax or additional amounts imposed (in any case) by any Governmental Entity.

"<u>Tax Authority</u>" means, with respect to any Tax, the Governmental Entity or political subdivision, agency, commission or authority thereof that imposes such Tax or that is charged with the assessment, determination or collection of such Tax.

"<u>Tax Benefit</u>" means a Tax Item that generally decreases taxable income or Taxes payable of a taxpayer, including a credit, deductible loss or other deduction.

"<u>Tax Contest</u>" means an audit, review or examination by a Governmental Entity or Tax Authority, or any other administrative or judicial proceeding, with the purpose or effect of examining, determining or redetermining Taxes (including any administrative or judicial review of any claim for Tax refund).

"<u>Tax Counsel</u>" means (i) with respect to the Tax Opinion, Baker Botts L.L.P., and (ii) with respect to a Supplemental Tax Opinion obtained by SLM BankCo or Navient, a nationally recognized law firm or accounting firm designated by the party that obtains such opinion.

"<u>Tax Item</u>" means any item of income, gain, loss, deduction, credit or other attribute that may have the effect of increasing or decreasing taxable income or Taxes payable.

"Tax Law" means the Law, including any controlling judicial or administrative interpretations of such Law, relating to any Tax.

"Tax Materials" means (i) the Ruling, (ii) the Ruling Request and any other submission to the IRS in connection with the Ruling, (iii) the representation letter delivered to Tax Counsel in connection with the delivery of the Tax Opinion, (iv) any other materials delivered or deliverable by SLM, SLM BankCo, Navient or others in connection with the

rendering by Tax Counsel of the Tax Opinion or the issuance by the IRS of the Ruling (or the request therefor), and (v) with respect to a party that requests or receives a Supplemental Ruling, or delivers a representation letter or other materials in connection with a Supplemental Ruling or Supplemental Tax Opinion, such Supplemental Ruling, representation letter or other materials, as the case may be.

"<u>Tax Opinion</u>" means the opinion to be delivered by Tax Counsel on or prior to the Spin-Off Date relating to the tax treatment of the Reorganization, Internal Contribution, Internal Spin-Off, External Contribution and External Spin-Off under Sections 368 and 355 of the Code.

"<u>Tax Records</u>" means Tax Returns, Tax Return work papers, documentation relating to any Tax Contests, and any other books of account or records required to be maintained under applicable Tax Laws (including but not limited to Section 6001 of the Code) or under any record retention agreement with any Tax Authority.

"<u>Tax Return</u>" means any return, information return, report, declaration, election, questionnaire, notice, form, claim for refund or other document filed or required to be filed with any Governmental Entity or Tax Authority with respect to Taxes (whether or not a payment is required to be made with respect to such filing), and any supplement, amendment, appendix, exhibit, schedule or attachment thereto.

"Tax Year" means with respect to any Tax, the year, or other period, if applicable, for which the Tax is reported as provided under applicable Tax Law.

"Treasury Regulations" means the regulations promulgated from time to time under the Code as in effect for the relevant Tax Year.

Article II

Allocation of Tax Liabilities and Tax Benefits

Liability for and Payment of Taxes. Except as otherwise provided in this Agreement:

Navient Liabilities and Payments. Navient shall be liable for and pay (i) all Navient Separate Taxes and (ii) the amount of all Navient Tax Obligations with respect to Consolidated Taxes, in each case either to the applicable Tax Authority and/or to SLM BankCo in accordance with the mechanics of (and without duplication of payments under) Article IV, and (iii) the Section 108(i) Quarterly Amounts as required under Section 2.2(c).

SLM BankCo Liabilities and Payments. SLM BankCo shall be liable for and pay (i) all SLM BankCo Separate Taxes and (ii) the amount of all SLM BankCo Tax Obligations with respect to Consolidated Taxes, in each case either to the applicable Tax Authority and/or to Navient in accordance with the mechanics of (and without duplication of payments under) Article IV.

Special Rules for Certain Tax Items.

- (a) Spin-Off Taxes and Spin-Off Tax Items.
- (i) Responsibility of SLM BankCo. To the extent that the imposition of any Spin-Off Taxes (other than as a result of the recognition of any Spin-Off Tax Items) or the recognition of any Spin-Off Tax Items is directly attributable to SLM BankCo's breach of any covenant or negative covenant in Article VIII, such Spin-Off Taxes or Spin-Off Tax Items, as the case may be, shall be treated as Taxes or Tax Items of, and allocated to, the SLM BankCo Group for purposes of applying the Separate Return Method.
- (ii) Responsibility of Navient. To the extent that the imposition of any Spin-Off Taxes (other than as a result of the recognition of any Spin-Off Tax Items) or the recognition of any Spin-Off Tax Items is directly attributable to Navient's breach of any covenant or negative covenant in Article VIII, such Spin-Off Taxes or Spin-Off Tax Items, as the case may be, shall be treated as Taxes or Tax Items of, and allocated to, the Navient Group (and not the SLM BankCo Group) for purposes of applying the Separate Return Method.
- (iii) Joint Responsibility. To the extent any Spin-Off Tax (that is not attributable to a Spin-Off Tax Item) or any Spin-Off Tax Item is not allocated under Section 2.2(a)(i) or Section 2.2(a)(ii), (x) fifty percent (50%) of such Spin-Off Tax or Spin-Off Tax Item, as the case may be, shall be treated as a Tax or Tax Item of, and allocated to, the SLM BankCo Group for purposes of applying the Separate Return Method, and (y) fifty percent (50%) of such Spin-Off Tax or Spin-Off Tax Item, as the case may be, shall be treated as a Tax or Tax Item of, and allocated to, the Navient Group (and not the SLM BankCo Group) for purposes of applying the Separate Return Method.

Bank Loan Sales. Notwithstanding anything to the contrary in this Agreement, for all purposes of this Agreement (including, without limitation, for purposes of applying the Separate Return Method and determining whether a Tax Return is a Joint Return), any Tax Items resulting from a Bank Loan Sale shall be treated as Tax Items of, and allocated to, SLM BankCo.

Section 108(i) Income. To reimburse SLM BankCo for income Taxes that may be imposed on the Section 108(i) Income, for each calendar quarter from and including second quarter 2014 through and including fourth quarter 2018 (i.e., a total of nineteen quarters), Navient shall pay to SLM BankCo an amount that, net of any payments for such quarter by SLM BankCo to Navient pursuant to this Section 2.2(c), is equal to the Section 108(i) Quarterly Amount for such quarter. Subject to the succeeding sentence, the due date for Navient to pay the Section 108(i) Quarterly Amount for an applicable quarter to SLM BankCo shall be the later of (i) three Business Days prior to the Section 108(i) Installment Date for such quarter and (ii) seven Business Days after SLM BankCo gives notice to Navient requesting such amount. If the Section 108(i) Quarterly Amount for a quarter increases or decreases due to a change in

applicable U.S. federal income tax rates, then (x) in the case of a decrease in the Section 108(i) Quarterly Amount for such quarter, SLM BankCo shall pay to Navient the excess (if any) of the net amount previously paid by Navient to SLM BankCo under this Section 2.2(c) for such quarter over the actual Section 108(i) Quarterly Amount for such quarter (revised to take into account such change in tax rates), within seven Business Days of the later of (1) the date that such change in tax rates is enacted and (2) the date that Navient gives notice to SLM BankCo requesting such amount, and (y) in the case of an increase in the Section 108(i) Quarterly Amount for such quarter, the due date for Navient to pay the amount of such increase to SLM BankCo shall be the later of (1) three Business Days prior to the Section 108(i) Installment Date for such quarter, (2) seven Business Days after the date that such change in tax rates is enacted, and (3) seven Business Days after the date that SLM BankCo gives notice to Navient requesting the amount of such increase. For the avoidance of doubt, if the Section 108(i) Tax Factor for all applicable quarters is always one (1), the aggregate amount required to be paid by Navient pursuant to this Section 2.2(c) is two hundred eighty-three million two hundred fifty-five thousand eight hundred dollars (\$283,255,800). Notwithstanding anything to the contrary in this Agreement, for all purposes of this Agreement (including, without limitation, for purposes of applying the Separate Return Method and determining whether a Tax Return is a Joint Return), all Section 108(i) Income shall be treated as income of, and allocated to, SLM BankCo.

Article III

Preparation and Filing of Tax Returns

Section 3.1 Joint Returns.

(a) Preparation and Filing. Navient shall be responsible for preparing (or causing to be prepared) any Joint Return for a taxable year (or other applicable taxable period) ending on or before December 31, 2013 ("Navient Joint Returns"). SLM BankCo shall be responsible for (i) preparing (or causing to be prepared) any Joint Return for a taxable year (or other applicable taxable period) ending after December 31, 2013 (an "SLM Bankco Joint Return"), and (ii) calculating the amount of U.S. federal estimated income taxes that are Consolidated Taxes for any quarter after the first quarter of 2014. The party responsible for signing any such Joint Return (or whose Group member is responsible for signing such Joint Return) under applicable Law (the "Return Signer") shall sign (or cause to be signed) such Joint Return and file (or cause to be filed) such Joint Return with the applicable Tax Authority. For the avoidance of doubt, SLM BankCo shall be responsible for signing and filing the U.S. federal income tax return for the SLM consolidated group for the taxable year ending December 31, 2013 and the U.S. federal income tax return for the SLM/SLM BankCo consolidated group for the taxable year ending December 31, 2014.

(b) Draft Joint Returns. The Preparer of a Joint Return shall provide the Non-Preparer of such Joint Return with a substantially final draft of such Joint Return (other than a Joint Return filed prior to the Spin-Off Date) at least 10 Business Days prior to the due date for such Joint Return (or, if there is no due date for such Joint Return, as soon as reasonably practical). Such Non-Preparer shall promptly notify such Preparer of any objections that it may have to any items or elections set forth in any such draft Joint Return. Notwithstanding anything to the contrary in Section 3.3(a), (i) Navient and SLM BankCo shall work together to resolve in

good faith any such objection and to mutually consent to the filing of such Joint Return (and no such Joint Return will be filed without such mutual consent), and (ii) after the filing of any original Joint Return, Navient and SLM BankCo shall work together, at the request of the other party, to determine whether to file any amended Joint Return (and no amended Joint Return will be filed on or after the Spin-Off Date without the consent of both Navient and SLM BankCo). If the Preparer fails to provide such draft Joint Return within the time period provided in this Section 3.1(b) or fails to properly file such Joint Return within two Business Days following the resolution of all such objections, then notwithstanding any other provision of this Agreement, such Preparer shall be liable for, and shall indemnify and hold harmless each member of the Non-Preparer's Group from and against, any penalties, interest, or other payment obligation imposed by reason of any resulting delay in filing such return.

- (c) Provision of Information. At the request of the Preparer of a Joint Return, the Non-Preparer of such Joint Return shall provide the Preparer with any information in its possession (or in the possession of any member of the Non-Preparer's Group) reasonably necessary for the Preparer to properly and timely prepare such Joint Return (including, without limitation, to the extent required, (i) the amount of taxable income of such Non-Preparer's Group for the period ending on the Spin-Off Date based on the closing of the books method, and (ii) the amount of taxable income recognized by such Non-Preparer's Group as a result of the External Spin-Off). Such Non-Preparer shall provide such information at least thirty days prior to the extended due date of such Joint Return or, if later, within fifteen Business Days of its receipt of such request. If such Non-Preparer (or any member of its Group) is in possession of information and fails to provide such information within the time period provided in this Section 3.1(c) or in the form reasonably requested by the Preparer to permit the timely filing of such Joint Return, then notwithstanding any other provision of this Agreement, such Non-Preparer shall be liable for, and shall indemnify and hold harmless each member of the Preparer's Group from and against, any penalties, interest, or other payment obligation imposed by reason of any resulting delay in filing such return.
- (d) Information with Respect to Estimated Payments and Extension Payments. At the request of the Preparer of a Joint Return, the Non-Preparer of such Joint Return shall provide the Preparer with all information in its possession (or in the possession of any member of its Group) that the Preparer needs to determine the amount of Taxes due on any Payment Date with respect such Joint Return. Such Non-Preparer shall provide such information at least thirty days prior to such Payment Date or, if later, within fifteen Business Days of its receipt of such request. In the event that such Non-Preparer fails to provide information within the time period provided in this Section 3.1(d) or in the form reasonably requested by the Preparer to permit the timely payment of such Taxes, the indemnification principles of Section 3.1(c) shall apply with respect to any penalties, interest, or other payments imposed by reason of any resulting delay in paying such Taxes. The Preparer shall promptly inform the Non-Preparer of its determination of the amount of Taxes due on any Payment Date with respect to such Joint Return and the parties shall reasonably cooperate to finalize such determination.
- (e) Provision of Assistance with Respect to Joint Returns. At the request of the Preparer of a Joint Return, the Non-Preparer of such Joint Return shall take (at its own cost and expense), and shall cause the members of its Group to take (at their own cost and expense), any reasonable action (e.g., executing a power of attorney) that is reasonably necessary in order

for such Preparer or any other member of the Preparer's Group to prepare, file, amend or take any other action with respect to such Joint Return (to the extent the Preparer's Group is permitted to do so under this Agreement). In the event that such Non-Preparer fails to take, or cause to be taken, any such requested action, the indemnification principles of Section 3.1(c) shall apply with respect to any penalties, interest, or other payments imposed by reason of a failure to timely take any such requested action.

(f) Engagement Letter for Certain 2014 Tax Returns. SLM Bankco shall enter into an engagement letter (the "2014 Engagement Letter") with a third-party accounting firm (the "2014 Accountant") pursuant to which such accounting firm shall agree to assist in (i) preparing certain U.S. federal, state and local income, franchise and similar Tax Returns for taxable periods ending during 2014 and (ii) calculating U.S. federal estimated income taxes for the second, third and fourth quarters of 2014 and the 2014 extension payments. SLM Bankco shall provide a draft of the 2014 Engagement Letter to Navient before the execution thereof and Navient shall notify SLM Bankco of any objections that it may have thereto within 10 Business Days after the receipt thereof. Failure of Navient to so object shall be deemed to constitute Navient's written consent to execute such 2014 Engagement Letter. SLM Bankco and Navient shall work together in good faith to resolve any such objections, and the 2014 Engagement Letter shall not be executed without the prior written consent of Navient, such consent not to be unreasonably withheld or delayed.

(g) Allocation of Third-Party Preparer Expenses. Any fees, costs or expenses payable to KPMG LLP (or its affiliates) under the 2013 Engagement Letter that have not been paid prior to the Spin-Off Date shall be allocated between Navient and SLM BankCo in proportion to the number of entities for which KPMG LLP (or its affiliates) prepares a pro forma separate company return pursuant to the 2013 Engagement Letter that are in the Navient Group and the SLM BankCo Group, respectively. Any fees, costs or expenses payable to the 2014 Accountant (or its affiliates) under the 2014 Engagement Letter shall be allocated between Navient and SLM BankCo in proportion to the number of entities for which the 2014 Accountant (or its affiliates) prepares a pro forma separate company return pursuant to the 2014 Engagement Letter that are in the Navient Group and the SLM BankCo Group, respectively. SLM BankCo shall be responsible for remitting such fees, costs and expenses to KPMG LLP or the 2014 Accountant, as the case may be. Navient shall remit to SLM BankCo the portion of any such fees, costs or expenses that are allocated to Navient pursuant to this Section 3.1(g) by the date that is the later of (i) five Business Days after the date that SLM BankCo remits such fees, costs or expenses to KPMG LLP or the 2014 Accountant, as applicable, and (ii) seven Business Days after SLM BankCo gives Navient notice requesting such amount.

Section 3.2 Separate Returns.

Separate Returns to be Prepared by SLM BankCo. SLM BankCo shall be responsible for preparing and filing (or causing to be prepared and filed) any Separate Return that includes Tax Items, operations, activities or assets of the SLM BankCo Operations, determined without regard to Tax Items carried forward to such Tax Year (an "SLM BankCo Separate Return").

Separate Returns to be Prepared by Navient. Navient shall be responsible for preparing and filing (or causing to be prepared and filed) any Separate Return that includes Tax Items, operations, activities or assets of the Navient Operations, determined without regard to Tax Items carried forward to such Tax Year (a "Navient Separate Return").

(a) Provision of Information and Assistance.

(i) Provision of Information. SLM BankCo shall provide to Navient, and Navient shall provide to SLM BankCo, any information in its possession (or in the possession of a member of its Group) requested by the other party that such party requesting such information reasonably needs to properly and timely file any Separate Returns pursuant to Section 3.2(a) or (b). Such information shall be provided within the time prescribed by Section 3.1(c) for the provision of information for Joint Returns. In the event that SLM BankCo or Navient fails to provide information within such time period or in the form reasonably requested by the other party to permit the timely filing of a Separate Return, the indemnification principles of Section 3.1(c) shall apply with respect to any penalties, interest, or other payments imposed against any member of the SLM BankCo Group or the Navient Group by reason of any resulting delay in filing such return.

(ii) Assistance. At the request of Navient, SLM BankCo shall take (at its own cost and expense), and shall cause the members of the SLM BankCo Group to take (at their own cost and expense), any reasonable action (e.g., executing a power of attorney) that is reasonably necessary in order for Navient or any other member of the Navient Group to prepare, file, amend or take any other action with respect to a Navient Separate Return. At the request of SLM BankCo, Navient shall take (at its own cost and expense), and shall cause the members of the Navient Group to take (at their own cost and expense), any reasonable action (e.g., executing a power of attorney) that is reasonably necessary in order for SLM BankCo or any other member of the SLM BankCo Group to prepare, file, amend or take any other action with respect to an SLM BankCo Separate Return. In the event that SLM BankCo or Navient, as the case may be, fails to take, or cause to be taken, any such requested action, the indemnification principles of Section 3.1(c) shall apply with respect to any penalties, interest, or other payments imposed against any member of either Group by reason of a failure to timely take any such requested action.

Section 3.3 Additional Rules Relating to the Preparation of Tax Returns.

General Rule. Except as otherwise provided in this Agreement (including, without limitation, under Section 3.3(b) and Section 3.3(c)), the party responsible for filing (or causing to be filed) a Tax Return pursuant to Section 3.2 shall have the exclusive right, in its sole discretion, with respect to such Tax Return to determine (i) the manner in which such Tax Return shall be prepared and filed, including the elections, methods of accounting, positions, conventions and principles of taxation to be used and the manner in which any Tax Item shall be reported, (ii) whether any extensions may be requested, (iii) whether an amended Tax Return shall be filed, (iv) whether any claims for refund shall be made, (v) whether any refunds shall be paid by way of refund or credited against any liability for the related Tax, and (vi) whether to retain outside firms to prepare or review such Tax Return.

Navient Separate Returns. With respect to any Navient Separate Return:

- (i) Navient may not take (and shall cause the members of the Navient Group not to take) any positions that it knows, or reasonably should know, would adversely affect any member of the SLM BankCo Group, without the prior written consent of SLM BankCo, except to the extent taking such position is consistent with the past practice of the SLM group or is required by applicable Law; and
- (ii) Navient and other members of the Navient Group must allocate Tax Items between such Navient Separate Return and any related Joint Return in a manner that is consistent with the reporting of such Tax Items on the related Joint Return.

SLM BankCo Separate Returns. With respect to any SLM BankCo Separate Return, SLM BankCo may not take (and shall cause the members of the SLM BankCo Group not to take) any positions that it knows, or reasonably should know, would adversely affect any member of the Navient Group, without the prior written consent of Navient, except to the extent taking such position is consistent with the past practice of the SLM group or is required by applicable Law.

Election to File Consolidated, Combined or Unitary Tax Returns. Navient and SLM BankCo shall cooperate to determine whether to file (or cause to be filed) any Tax Return on a consolidated, combined or unitary basis if such Tax Return would include at least one member of each Group and the filing of such Tax Return on such basis is elective under the relevant Tax Law.

Standard of Performance. Navient and SLM BankCo shall prepare Navient Joint Returns and SLM BankCo Joint Returns, respectively, with the same general degree of care as they use in preparing Separate Returns. Notwithstanding the preceding sentence, neither Navient nor SLM BankCo shall be liable for any additional Taxes for which the other is otherwise liable under Article II that result from a redetermination in a Tax Contest, regardless of whether such Taxes arise as a result of Navient's or SLM BankCo's failure to exercise such degree of care.

Reliance on Exchanged Information. If a member of the Navient Group supplies information to a member of the SLM BankCo Group, or a member of the SLM BankCo Group supplies information to a member of the Navient Group, in connection with the preparation of a Tax Return and an officer of the requesting member intends to sign a statement or other document under penalties of perjury in reliance upon the accuracy of such information, then a duly authorized officer of the member supplying such information shall certify (upon the request of such other member), to such officer's knowledge and belief, the accuracy and completeness of the information so supplied.

Allocation of Tax Items. Navient and SLM BankCo shall cooperate to determine the allocation of any applicable Tax Items and Tax attributes (e.g., NOLs, Tax credits and earnings and profits) as of the Effective Time, in accordance with applicable Tax Laws, among SLM BankCo, each other SLM BankCo Group member, Navient, and each other Navient Group member. SLM BankCo and Navient hereby agree that each such Tax Item shall be allocated in a manner consistent with allocations under the Separate Return Method to the extent permitted under applicable Law.

Article IV

Tax Payments

Payment of Taxes to Tax Authority. SLM BankCo shall be responsible for remitting to the proper Tax Authority the Tax shown on any Joint Return for which a member of the SLM BankCo Group is the Return Signer or any SLM BankCo Separate Return (including Taxes for which Navient is wholly or partially liable pursuant to Article II) and any estimated Taxes required to be paid with respect thereto, to the extent such amounts have not been remitted prior to the Spin-Off Date, and Navient shall be responsible for remitting to the proper Tax Authority the Tax shown on any Joint Return for which a member of the Navient Group is the Return Signer or any Navient Separate Return (including Taxes for which SLM BankCo is wholly or partially liable pursuant to Article II) and any estimated Taxes required to be paid with respect thereto, to the extent such amounts have not been remitted prior to the Spin-Off Date. For the avoidance of doubt, SLM BankCo shall be responsible for remitting to the U.S. government U.S. federal estimated income Taxes for the SLM consolidated group and the SLM BankCo consolidated group, as applicable, for each quarter in 2014 (to the extent such amounts have not been remitted prior to the Spin-Off Date).

Section 4.1 Indemnification Payments.

Tax Payments. In connection with any payment of, or any filing of a Consolidated Return with respect to, any Consolidated Tax for any Tax Year (or portion thereof) (a "Tax Event") on a date on or after the Spin-Off Date (the "Applicable Date"), the parties shall cooperate to determine the amount of the Navient Tax Amount and SLM BankCo Tax Amount with respect to such Consolidated Tax for such Tax Year (or portion thereof) as of the Applicable Date. If Navient is responsible for remitting such Consolidated Tax to, or filing (or causing to be filed) such Consolidated Return with, the applicable Tax Authority pursuant to Article III or Section 4.1, as applicable, Navient shall notify SLM BankCo prior to making such payment or filing, and (i) if the SLM BankCo Net Obligation with respect to such Consolidated Tax for such Tax Year (or portion thereof) is or will be positive as of the Applicable Date (not taking into account any payment made pursuant to this sentence with respect to such Tax Event), SLM BankCo shall remit to Navient the amount of such SLM BankCo Net Obligation, and (ii) if the SLM BankCo Net Obligation with respect to such Consolidated Tax for such Tax Year (or portion thereof) is or will be negative as of the Applicable Date (not taking into account any payment made pursuant to this sentence with respect to such Tax Event), Navient shall remit to SLM BankCo an amount equal to the absolute value of such SLM BankCo Net Obligation, in each case by the date that is the later of (x) three Business Days prior to the date that Navient remits such Tax payment to such Tax Authority or files such Consolidated Return, as applicable and (y) seven Business Days after the party entitled to payment gives notice to the other party requesting such amount. If SLM BankCo is responsible for remitting such Consolidated Tax to, or filing (or causing to be filed) such Consolidated Return with, the applicable Tax Authority pursuant to Article III or Section 4.1, as applicable, SLM BankCo shall notify Navient prior to making such payment or filing, and (i) if the Navient Net Obligation with respect to such Consolidated Tax for such Tax Year (or portion thereof) is or will be positive as of the Applicable Date (not taking into account any payment made pursuant to this sentence with respect to such Tax Event), Navient shall remit to SLM BankCo the amount of such Navient Net Obligation, and (ii) if the Navient Net

Obligation is or will be negative as of the Applicable Date (not taking into account any payment made pursuant to this sentence with respect to such Tax Event), SLM BankCo shall remit to Navient an amount equal to the absolute value of such Navient Tax Obligation, in each case by the date that is the later of (x) three Business Days prior to the date that SLM BankCo remits such Tax payment to such Tax Authority or files such Consolidated Return, as applicable and (y) seven Business Days after the party entitled to payment gives notice to the other party requesting such amount.

Credit for Prior Tax Payments. SLM BankCo shall be deemed to have paid any amount related to Taxes as of the Effective Time to the extent the SLM BankCo Group is treated as having paid such amount under the Separate Return Method applied as of the Effective Time. Any such amount treated as paid by SLM BankCo shall be treated as having been paid to either Navient or the applicable Tax Authority, as applicable, for purposes of this Agreement. Navient shall be deemed to have paid any applicable amount related to Taxes as of the Effective Time to the extent that such amount was actually paid to the applicable Tax Authority as of the Effective Time or (without duplication) treated as paid under the Separate Return Method as of the Effective Time and the SLM BankCo Group is not treated as having paid such amount under the Separate Return Method applied as of the Effective Time. Any such amount treated as paid by Navient shall be treated as having been paid to either SLM BankCo or the applicable Tax Authority, as applicable, for purposes of this Agreement.

Initial Determinations and Subsequent Adjustments. The initial determination of the amount of any payment that one party is required to make to another under this Agreement shall be made on the basis of the Tax Return as filed or, if the Tax to which the payment relates is not reported on a Tax Return, on the basis of the amount of Tax initially paid to the Tax Authority. The amounts paid under this Agreement will be redetermined, and additional payments relating to such redetermination will be made, as appropriate, if as a result of an audit by a Tax Authority or an amended Tax Return or for any other reason (w) additional Taxes are subsequently paid or required to be paid, (x) a refund of such Taxes is received, (y) the party using a Tax Benefit changes by reason of a Tax Contest or the filing of an amended Tax Return, or (z) the amount or character of any Tax Item is adjusted or redetermined by reason of a Tax Contest or the filing of an amended Tax Return. Each payment required by the immediately preceding sentence (i) as a result of a payment of additional Taxes will be due thirty days after the date on which the additional Taxes were paid or, if later, fifteen days after the party that made such payment of additional Taxes gives notice to the other party requesting reimbursement, (ii) as a result of a refund will be due thirty days after the refund was received, (iii) as a result of a change in use of a Tax Benefit (to which clauses (i) and (ii) do not apply) by reason of a Tax Contest or the filing of an amended Tax Return will be due thirty days after the date on which the final action resulting in such change is taken by a Tax Authority or either party or any of their Subsidiaries, or (iv) as a result of an adjustment or redetermination of the amount or character of a Tax Item (to which clauses (i) and (ii) do not apply) by reason of a Tax Contest or the filing of an amended Tax Return will be due thirty days after the date on which the final action resulting in such adjustment or redetermination is ta

Interest on Late Payments. Payments pursuant to this Agreement that are not made by the date prescribed in this Agreement or, if no such date is prescribed, within fifteen days after written demand for payment is made (the "<u>Due Date</u>") shall bear interest for the period from and including the date immediately following the Due Date through and including the date of payment at a per annum rate equal to the Prime Rate on the Due Date (or, if the Due Date is not a Business Day, as of 11:00 a.m. New York, NY time on the first Business Day following the Due Date) <u>plus</u> 2%, subject to any maximum amount permitted by applicable Law. Such rate shall be redetermined at the beginning of each calendar quarter following such Due Date. Such interest will be payable at the same time as the payment to which it relates and shall be calculated on the basis of a year of 365 days and the actual number of days for which due.

Payments by or to Other Group Members. When appropriate under the circumstances to reflect the underlying liability for a Tax or entitlement to a Tax refund or Tax Benefit, a payment which is required to be made by or to SLM BankCo or Navient under this Agreement may be made by or to another member of the SLM BankCo Group or the Navient Group, as appropriate, but nothing in this Section 4.5 shall relieve SLM BankCo or Navient of its obligations under this Agreement.

Procedural Matters. Any notice requesting payment delivered to the indemnifying party in accordance with Section 9.5 shall show the amount due and owing together with a schedule calculating in reasonable detail such amount (and shall include any relevant Tax Return, pro forma Tax Return, statement, bill or invoice related to such Taxes, costs, expenses or other amounts due and

owing). Payments shall be deemed made when received. If the indemnifying party fails to make a payment to the indemnified party within the time period set forth in this Article IV, the indemnifying party shall pay to the indemnified party, in addition to interest that accrues pursuant to Section 4.4, any costs or expenses incurred by the indemnified party to secure such payment or to satisfy the indemnifying party's portion of the obligation giving rise to the indemnification payment.

Tax Consequences of Payments; Tax Gross-Up. For all Tax purposes, to the extent permitted by applicable Tax Law, the parties will treat any payment made pursuant to this Agreement as a capital contribution made by SLM BankCo to Navient or as a distribution made by Navient to SLM BankCo, as the case may be, on the date recited above on which the parties entered into the Agreement. If any Spin-Off Tax Payment (or portion thereof) directly or indirectly causes an increase in Taxes for which the payee would otherwise be liable under Article II, the payer of such Spin-Off Tax Payment shall be required to pay the payee (either by means of direct payment or by means of offset against amounts otherwise due under this Agreement from such payee to such payer) an additional amount, such that the amount of any Taxes for which the payee is liable under Article II that result from either such Spin-Off Tax Payment or from the payment of such additional amount. Except to the extent described in the preceding sentence of this Section 4.7, under no circumstances shall any payment (or portion thereof) made pursuant to this Agreement be grossed up to take into account any additional Taxes that may be owed by the payee (or any of the members of its Group) as a result of such payment. In the event that a Tax Authority asserts that the treatment of a payment pursuant to this Agreement should be other than as determined pursuant to this Section 4.7, SLM BankCo or Navient, as appropriate, shall use its reasonable best efforts to contest such assertion. Section 4.4(c) of the Separation Agreement shall not apply to payments made pursuant to this Agreement.

Article V

Assistance and Cooperation

Cooperation. In addition to the obligations enumerated in Section 3.1(c), Section 3.1(e), Section 3.2(c) and Section 7.3, SLM BankCo and Navient will cooperate (and will cause their respective Groups to cooperate) with each other and with each other's agents, including accounting firms and legal counsel, in connection with Tax matters upon the reasonable request of the other party. Such cooperation will include Navient or SLM BankCo, as the case may be, (x) providing documents and information in its possession (or in the possession of another member of its Group) as reasonably requested, and (y) making available to the other, as reasonably requested and available, personnel (including officers, directors, employees and agents of either Group) responsible for preparing, maintaining, and interpreting information and documents relevant to Taxes and personnel reasonably required as witnesses or for purposes of providing information or documents in connection with any administrative or judicial proceedings relating to Taxes.

Supplemental Rulings and Supplemental Tax Opinions. Each of the parties agrees that, at the reasonable request of the other party (the "Requesting Party"), such party shall (and shall cause each member of its Group to) (i) cooperate and use reasonable efforts to seek to obtain, as expeditiously as reasonably practicable, a Supplemental Ruling from the IRS or (ii) cooperate

and use reasonable efforts to assist the Requesting Party in obtaining, as expeditiously as reasonably practicable, a Supplemental Tax Opinion from Tax Counsel. Within thirty days after receiving an invoice from the other party therefor, the Requesting Party shall reimburse such other party for all reasonable out-of-pocket costs and expenses incurred by such other party and the members of such other party's Group in connection with obtaining or requesting a Supplemental Ruling or in connection with assisting the Requesting Party in obtaining a Supplemental Tax Opinion. Notwithstanding the foregoing, a party shall not be required to file any Supplemental IRS Submission at the request of the Requesting Party unless the Requesting Party represents to such other party that (x) the Requesting Party has reviewed the Supplemental IRS Submission and (y) all information and representations, if any, relating to any member of the Requesting Party's Group contained in the Supplemental IRS Submissions are true, correct and complete in all material respects.

Article VI

Tax Records

Retention of Tax Records. Each of SLM BankCo and Navient shall preserve, and shall cause the members of their respective Groups to preserve, all Tax Records that are in their possession and that could reasonably be expected to affect the liability of any member of the other Group for Taxes, for as long as the contents thereof may become material in the administration of any matter under applicable Tax Law, but in any event until the later of (x) the expiration of any applicable statutes of limitation, as extended, and (y) seven years after the Spin-Off Date.

Access to Tax Records. Navient shall make available, and cause the members of the Navient Group to make available, to SLM BankCo for inspection and copying (x) all Tax Records in their possession at the time of any written request therefor that relate to Tax Years (or portions thereof) that end on or before the Spin-Off Date, and (y) the portion of any Tax Record in their possession at the time of any written request therefor that relates to Tax Years (or portions thereof) that end after the Spin-Off Date and that is reasonably necessary to have for the preparation of a Joint Return or an SLM BankCo Separate Return or in connection with a Tax Contest relating to such return. SLM BankCo shall make available, and cause the members of the SLM BankCo Group to make available, to Navient for inspection and copying (x) all Tax Records in their possession at the time of any written request therefor that relate to Tax Years (or portions thereof) that end on or before the Spin-Off Date, and (y) the portion of any Tax Record in their possession at the time of any written request therefor that relates to Tax Years (or portions thereof) that end after the Spin-Off Date and that is reasonably necessary to have for the preparation of a Joint Return or a Navient Separate Return or in connection with a Tax Contest relating to such return.

Confidentiality. The provisions of Sections 6.9 and 6.10 of the Separation Agreement shall apply with respect to any Information provided pursuant to this Agreement; provided that either party may disclose, or may permit disclosure of, Information to the extent necessary in connection with the filing of Tax Returns or any administrative or judicial proceedings relating to Taxes.

Article VII

Tax Contests

Notices. Each party shall provide prompt notice to the other party of any pending or threatened Tax audit, assessment or proceeding or other Tax Contest of which it becomes aware relating to (i) Taxes for which it is or may be indemnified by the other party hereunder, (ii) the qualification of the Reorganization as a tax-free transaction described under Section 368(a)(1)(F) of the Code, (iii) the qualification of the Internal Contribution and the Internal Spin-Off as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code, or (iv) the qualification of the External Contribution and the External Spin-Off as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code. Such notice shall contain factual information (to the extent known) describing any asserted Tax liability in reasonable detail and shall be accompanied by copies of any notice and other documents (or applicable portions thereof) received from any Tax Authority in respect of any such matters. If (i) an indemnified party has knowledge of an asserted Tax liability with respect to a matter for which it is to be indemnified hereunder, (ii) such party fails to give the indemnifying party prompt notice of such asserted Tax liability, and (iii) the indemnifying party has the right, pursuant to Section 7.2(a), to control the Tax Contest relating to such Tax liability, then (x) if the indemnifying party is precluded from contesting the asserted Tax liability as a result of such asserted Tax liability, and (y) if the indemnifying party is not precluded from contesting the asserted Tax liability to give prompt notice, the indemnifying party, then any amount which the indemnifying party is otherwise required to pay the indemnified party pursuant to this Agreement shall be reduced by the amount of such detriment.

Section 7.1 Control of Tax Contests.

General Rule. Except as provided in Section 7.2(b), (i) SLM BankCo shall have full responsibility, control and discretion in handling, settling or contesting any Tax Contest involving a Tax reported on a Joint Return (or which a Tax Authority asserts should have been reported on a Joint Return), and (ii) each party (or the appropriate member of its Group) shall have full responsibility, control and discretion in handling, settling or contesting any Tax Contest involving a Tax reported on a Separate Return of such party (or which a Tax Authority asserts should have been reported on a Separate Return).

(a) Participation and Settlement Rights.

(i) With respect to a Tax Contest which involves a Tax liability for which the Non-Controlling Party may be liable, or a Tax Benefit to which the Non-Controlling Party may be entitled, in whole or in part under this Agreement (in any case, a "Non-Prepare Item," which, to the extent Navient is the Non-Controlling Party, shall be deemed to include any Spin-Off Taxes and Spin-Off Tax Items and any issues related to the tax-free treatment of the Reorganization, Internal Contribution, External Contribution, Internal Spin-Off or External Spin-Off), subject to Section 7.2(b)(ii), (1) the Non-Controlling Party shall be entitled to participate, at its own cost and expense, in such Tax Contest to the extent it relates to a Non-Preparer Item, (2)

the Controlling Party shall keep the Non-Controlling Party reasonably informed and consult seriously and in good faith with the Non-Controlling Party and its Tax advisors with respect to any issue in such Tax Contest relating to a Non-Preparer Item, (3) the Controlling Party shall provide the Non-Controlling Party with copies of all correspondence, notices, and other written materials received from any Tax Authority and shall otherwise keep the Non-Controlling Party and its Tax advisors advised of significant developments in the Tax Contest and of significant communications involving representatives of the Tax Authority, to the extent (in any case) related to a Non-Preparer Item, (4) the Non-Controlling Party may request that the Controlling Party take a position related to a Non-Preparer Item in respect of such Tax Contest, and the Controlling Party shall do so provided that (A) there exists substantial authority for such position (within the meaning of the accuracy-related penalty provisions of Section 6662 of the Code) and (B) the adoption of such position could not reasonably be expected to increase the Taxes or reduce the Tax Benefits allocated to the Controlling Party pursuant to Article II of this Agreement (unless the Non-Controlling Party shall provide the Non-Controlling Party with a copy of any written submission to be sent to a Tax Authority to the extent related to a Non-Preparer Item prior to the submission thereof and shall give serious and good faith consideration to any comments or suggested revisions that the Non-Controlling Party or its Tax advisors may have with respect thereto, and (6) there will be no settlement, resolution (within the control of the Controlling Party or any member of the Controlling Party, which consent shall not be unreasonably withheld or delayed.

(ii) Notwithstanding anything to the contrary in this Agreement, if Navient and SLM BankCo cannot agree as to how to proceed (e.g., whether to propose or accept a settlement offer, file a protest or petition, etc.) with respect to a Tax Item or issue in a Tax Contest involving Consolidated Taxes after consulting with each other on such matter in good faith, the party with the most Taxes at risk with respect to such Tax Item or issue in such Tax Contest shall be entitled to decide how to proceed with respect to such Tax Item or issue, and the other party shall comply and reasonably cooperate with such decision (including by signing a power of attorney with respect to such issue, to the extent necessary). For purposes of this Section 7.2(b)(ii), the party with the most Taxes at risk with respect to a Tax Item or issue shall be the party whose liability for Taxes under this Agreement would increase the most (in terms of absolute dollar amount) if the position of the applicable Governmental Entity with respect to such Tax Item or issue were upheld in full (treating a loss of any claimed refund as an increase in Tax liability).

Cooperation. The Non-Controlling Party shall promptly provide the Controlling Party with all documents and information in its possession (or in the possession of its Group) which the Controlling Party reasonably needs to handle, settle or contest the Tax Contest (including copies of all correspondence, notices, and other written materials with respect to such Tax Contest sent by a Tax Authority to the Non-Controlling Party (or a member of its Group) but not to any member of the Controlling Party's Group). Subject to Section 7.2(b), at the reasonable request of the Controlling Party, the Non-Controlling Party shall take any action (e.g., executing a power of attorney) that is reasonably necessary in order for the Controlling Party to handle, settle or contest the Tax Contest. The Controlling Party shall reimburse the Non-Controlling Party for

any reasonable out-of-pocket costs and expenses incurred in complying with the first two sentences of this Section 7.3. The Controlling Party shall have no obligation to indemnify the Non-Controlling Party for any additional Taxes resulting from the Tax Contest if such Non-Controlling Party fails to reasonably cooperate in accordance with the first two sentences of Section 7.3.

Article VIII

Restriction on Certain Actions of SLM BankCo and Navient

General Restrictions. Navient and SLM BankCo shall not (and shall cause the members of their respective Groups to not) take any action or fail to take any action if such action or failure to take action, as the case may be, would (i) be inconsistent with the Reorganization qualifying, or preclude the Reorganization from qualifying, as a tax-free transaction described under Section 368(a)(1)(F) of the Code, (ii) be inconsistent with the Internal Contribution and Internal Spin-Off qualifying, as tax-free transactions described under Sections 368(a)(1)(D) and 355 of the Code, (iii) be inconsistent with the External Contribution and External Spin-Off qualifying, or preclude the External Contribution and External Spin-Off from qualifying, as tax-free transactions (except with respect to cash received in lieu of fractional shares) described under Sections 368(a)(1)(D) and 355 of the Code, or (iv) be reasonably likely to be inconsistent with, or cause any Person to be in breach of, any representation, covenant or material statement made in the Tax Materials.

Certain Navient Actions Following the Effective Time. Without limiting the other provisions of this Article VIII, during the two-year period following the Effective Time, Navient shall not take (and shall cause the members of the Navient Group to not take), and shall not negotiate or enter into (and shall cause the members of the Navient Group to not negotiate or enter into) a binding agreement to take, any of the following actions: (i) liquidate, sell all or substantially all of its assets or sell or transfer fifty percent (50%) or more of (1) the assets that constitute the Education Loan Management Business as of the Effective Time to any Person other than Navient or an entity which is and will be wholly-owned, directly or indirectly, by Navient or (2) the assets that constitute the Internal Distributing Business as of the Effective Time to any Person other than Internal Distributing or an entity which is and will be wholly-owned, directly or indirectly, by Internal Distributing, (ii) issue stock of Navient or any Navient Affiliate (or any instrument that is convertible or exchangeable into any such stock), other than an issuance to which Treasury Regulations Section 1.355-7(d)(8) or (9) applies, equal to or exceeding twenty-five percent (25%) (by vote or value) of the stock of Navient or of such Navient Affiliate that was issued and outstanding immediately following the Effective Time, (iii) facilitate or otherwise participate in any acquisition (or deemed acquisition) or other transfer of stock of Navient or Internal Distributing that would result in (1) any shareholder owning (or being deemed to own after applying the rules of Sections 355(e)(3)(B) of the Code) forty percent (40%) or more (by vote or value) of the outstanding stock of Navient or (2) any shareholder other than Navient owning (or being deemed to own after applying the rules of Sections 355(e)(3)(B) of the Code) forty percent (40%) or more (by vote or value) of the outstanding stock of Internal Distributing, (iv) redeem or otherwise repurchas

Navient Group of the Education Loan Management Business or the Internal Distributing Business, in each case, without first obtaining and delivering to SLM BankCo at Navient's own expense a Supplemental Tax Opinion or Permitted Supplemental Ruling reasonably satisfactory to SLM BankCo with respect to such action.

Certain SLM BankCo Actions Following the Effective Time. Without limiting the other provisions of this Article VIII, during the two-year period following the Effective Time, SLM BankCo shall not take (and shall cause the members of the SLM BankCo Group to not take), and shall not negotiate or enter into (and shall cause the members of the SLM BankCo Group to not negotiate or enter into) a binding agreement to take, any of the following actions: (i) liquidate, sell all or substantially all of its assets or sell or transfer fifty percent (50%) or more of (1) the assets that constitute the Consumer Banking Business as of the Effective Time to any Person other than SLM BankCo or an entity which is and will be wholly-owned, directly or indirectly, by SLM BankCo or (2) the assets that constitute the Internal Controlled Business as of the Effective Time to any Person other than Internal Controlled or an entity which is and will be wholly-owned, directly or indirectly, by Internal Controlled, (ii) issue stock of SLM BankCo or any SLM BankCo Affiliate (or any instrument that is convertible or exchangeable into any such stock), other than an issuance to which Treasury Regulations Section 1.355-7(d)(8) or (9) applies, equal to or exceeding twenty-five percent (25%) (by vote or value) of the stock of SLM BankCo or of such SLM BankCo Affiliate that was issued and outstanding immediately following the Effective Time, (iii) facilitate or otherwise participate in any acquisition (or deemed acquisition) or other transfer of stock of SLM BankCo or Internal Controlled that would result in (1) any shareholder owning (or being deemed to own after applying the rules of Sections 355(e)(4)(C) and 355(e)(3)(B) of the Code) forty percent (40%) or more (by vote or value) of the outstanding stock of SLM BankCo or (2) any shareholder other than SLM BankCo owning (or being deemed to own after applying the rules of Sections 355(e)(4)(C) and 355(e)(3)(B) of the Code) forty percent (40%) or more (by vote or value) of the outstanding stock of Internal Controlled, (iv) redeem or otherwise repurchase any stock of SLM BankCo other than pursuant to open market stock repurchase programs meeting the requirements of Section 4.05(1)(b) of Rev. Proc. 96-30, 1996-1 C.B. 696, or (v) terminate the active conduct by the SLM BankCo Group of the Consumer Banking Business or the Internal Controlled Business, in each case, without first obtaining and delivering to Navient at SLM BankCo's own expense a Supplemental Tax Opinion or Permitted Supplemental Ruling reasonably satisfactory to Navient with respect to such action.

Article IX

General Provisions

Section 9.1 Counterparts; Corporate Power.

(a) This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties.

- (b) SLM BankCo represents on behalf of itself, and Navient represents on behalf of itself, as follows:
- (i) it has the requisite corporate or other power and authority and has taken all corporate or other action necessary in order to execute, deliver and perform this Agreement to which it is a party and to consummate the transactions contemplated hereby; and
- (ii) this Agreement has been duly executed and delivered by it and constitutes a valid and binding agreement of it enforceable in accordance with the terms hereof.
- (c) Each party acknowledges that it and the other party may execute this Agreement by facsimile, stamp or mechanical signature. Each party expressly adopts and confirms each such facsimile, stamp or mechanical signature made in its respective name as if it were a manual signature, agrees that it will not assert that any such signature is not adequate to bind such party to the same extent as if it were signed manually and agrees that at the reasonable request of the other party at any time it will as promptly as reasonably practicable cause this Agreement to be manually executed (any such execution to be as of the date of the initial date thereof).
- (d) Notwithstanding any provision of this Agreement, neither SLM BankCo nor Navient shall be required to take or omit to take any act that would violate its fiduciary duties to any minority shareholders of any non-wholly owned Subsidiary of SLM BankCo or Navient, as the case may be (it being understood that directors' qualifying shares or similar interests will be disregarded for purposes of determining whether a Subsidiary is wholly owned).

Governing Law. This Agreement (and any claims or disputes arising out of or related hereto or to the transactions contemplated hereby or to the inducement of any party to enter herein, whether for breach of contract, tortious conduct or otherwise and whether predicated on common law, statute or otherwise) shall be governed by and construed and interpreted in accordance with the Laws of the State of Delaware irrespective of the choice of laws principles of the State of Delaware including all matters of validity, construction, effect, enforceability, performance and remedies.

Assignability. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns; provided, however, that no party hereto may assign its respective rights or delegate its respective obligations under this Agreement, directly or indirectly (including by sale of assets or stock or by means of a merger or consolidation) without the express prior written consent of the other party hereto. Notwithstanding the foregoing, no such consent shall be required for the assignment of a party's rights and obligations under this Agreement in whole in connection with a Change of Control of a party so long as the resulting, surviving or transferee Person assumes all the obligations of the relevant party thereto by operation of Law or pursuant to an agreement in form and substance reasonably satisfactory to the other party. Nothing in this Section 9.3 is intended to, or shall be construed to, prohibit either party or any of its Subsidiaries from being party to or undertaking a Change of Control.

Third-Party Beneficiaries. Except for the indemnification rights under this Agreement of any members of the SLM BankCo Group or the Navient Group in their respective capacities as such, (a) the provisions of this Agreement are solely for the benefit of the parties hereto and are not intended to confer upon any Person except the parties hereto any rights or remedies hereunder, and (b) there are no third party beneficiaries of this Agreement and this Agreement shall not provide any third Person with any remedy, claim, liability, reimbursement, claim of action or other right in excess of those existing without reference to this Agreement.

Notices. The provisions of Section 10.5 of the Separation Agreement shall apply to all notices, requests, claims, demands or other communications under this Agreement.

Severability. If any provision of this Agreement or the application thereof to any Person or circumstance is determined by a court of competent jurisdiction to be invalid, void or unenforceable, the remaining provisions hereof, or the application of such provision to Persons or circumstances or in jurisdictions other than those as to which it has been held invalid or unenforceable, shall remain in full force and effect and shall in no way be affected, impaired or invalidated thereby. Upon such determination, the parties shall negotiate in good faith in an effort to agree upon a suitable and equitable provision to effect the original intent of the parties.

Section 9.2 <u>Force Majeure</u>. No party shall be deemed in default of this Agreement to the extent that any delay or failure in the performance of its obligations under this Agreement results from any Force Majeure. In the event of any such excused delay, the time for performance shall be extended for a period equal to the time lost by reason of the delay.

Section 9.3 <u>Expenses</u>. Except as otherwise expressly set forth in this Agreement, each Group shall bear its own expenses incurred after the Effective Time in connection with the preparation of Tax Returns and other matters related to Taxes under the provisions of this Agreement for which it is liable.

Headings. The article, section and paragraph headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.

Section 9.4 Termination.

- (a) This Agreement may be terminated by the Sallie Mae Board at any time, in its sole and absolute discretion, prior to the Separation Effective Time. After the Separation Effective Time, this Agreement may not be terminated except by an agreement in writing signed by each of the parties.
- (b) In the event of any termination of this Agreement prior to the Separation Effective Time, no party (or any of its directors or officers) shall have any liability or further obligation to any other party.

Waiver of Default. Waiver by any party of any default by the other party of any provision of this Agreement shall not be deemed a waiver by the waiving party of any subsequent or other default, nor shall it prejudice the rights of the other party. No failure or delay by any party in exercising any right, power or privilege under this Agreement shall operate as a waiver thereof, nor shall a single or partial exercise thereof prejudice any other or further exercise thereof or the exercise of any other right, power or privilege.

Amendments. No provisions of this Agreement shall be deemed waived, amended, supplemented or modified by any party, unless such waiver, amendment, supplement or modification is in writing and signed by the authorized representative of the party against whom it is sought to enforce such waiver, amendment, supplement or modification.

Interpretation. In this Agreement, (a) words in the singular shall be held to include the plural and vice versa and words of one gender shall be held to include the other genders as the context requires; (b) the terms "hereof," "herein," and "herewith" and words of similar import shall, unless otherwise stated, be construed to refer to this Agreement as a whole and not to any particular provision of this Agreement; (c) Article and Section references are to the Articles and Sections to this Agreement unless otherwise specified; (d) the word "including" and words of similar import when used in this Agreement shall mean "including, without limitation,"; (e) the word "or" shall not be exclusive; (f) unless expressly stated to the contrary in this Agreement, all references to "the date hereof," "the date of this Agreement," "hereby" and "hereupon" and words of similar import shall all be references to April 29, 2014, regardless of any amendment or restatement hereof; and (g) references to any agreement, instrument or other document shall mean such agreement, instrument or other document as amended, supplemented or modified from time to time to the extent permitted by the provisions thereof and not prohibited by this Agreement.

Limitation of Liability. Except as expressly set out in this Agreement, neither Navient or its Affiliates, on the one hand, nor SLM BankCo or its Affiliates, on the other hand, shall be liable under this Agreement to the other for any special, indirect, punitive, exemplary, remote, speculative or similar damages in excess of compensatory damages of the other arising in connection with the transactions contemplated hereby (other than any such liability included in the definition of Taxes or with respect to a Third Party Claim).

Performance. SLM BankCo will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by any member of the SLM BankCo Group. Navient will cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth in this Agreement to be performed by any member of the Navient Group. Each party (including its permitted successors and assigns) further agrees that it will (a) give timely notice of the terms, conditions and continuing obligations contained in this Agreement to all of the other members of its Group and (b) cause all of the other members of its Group not to take any action or fail to take any such action inconsistent with such party's obligations under this Agreement or the transactions contemplated hereby.

Predecessors or Successors. Any reference to SLM BankCo, Navient, a Person, or a Subsidiary in this Agreement shall include any predecessors or successors (e.g., by merger or other reorganization, liquidation, conversion, or election under Treasury Regulations Section 301.7701-3) of SLM BankCo, Navient, such Person, or such Subsidiary, respectively.

Effective Time. This Agreement shall become effective on the date recited above on which the parties entered into this Agreement.

Change in Law. Any reference to a provision of the Code or any other Tax Law shall include a reference to any applicable successor provision or Law.

Disputes. Except as otherwise provided under this Agreement (*e.g.*, under Section 7.2(b)(ii)), the procedures for resolving Disputes set forth in Article VII of the Separation Agreement shall apply to all Disputes (whether sounding in contract, tort or otherwise) arising out of or related to this Agreement or the transactions contemplated hereby between or among any members of the Navient Group and the SLM BankCo Group.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by the respective officers as of the date set forth above.

NAVIENT CORPORATION

By: /s/ John F. Remondi
Name: John F. Remondi
Title: Chief Executive Officer

NEW BLC CORPORATION

By: /s/ Raymond Quinlan
Name: Raymond Quinlan
Title: Chief Executive Officer

AMENDED AND RESTATED LOAN SERVICING AND ADMINISTRATION AGREEMENT

This AMENDED AND RESTATED LOAN SERVICING AND ADMINISTRATION AGREEMENT (together with all schedules and attachments hereto, the "Servicing Agreement"), effective as of April 30, 2014 (the "Effective Date"), between **SALLIE MAE BANK** (the "Bank") and **NAVIENT SOLUTIONS, INC.** (herein called the "Servicer").

WITNESSETH:

WHEREAS, the Bank holds certain government-guaranteed education loans and certain Private Loans (as such term is defined below); and

WHEREAS, the Servicer is in the business of servicing education loans and providing related services; and

WHEREAS, the Bank and the Servicer are currently parties to that certain Amended and Restated Loan Origination and Management Services Agreement dated as of February 1, 2010, as amended (the "Prior LOMSA"), pursuant to which the Servicer is currently servicing such education loans on behalf of the Bank; and

WHEREAS, in conjunction with the corporate restructuring of the Bank's and the Servicer's common parent company, SLM Corporation, the Bank and the Servicer desire to amend and restate their agreement pursuant to which the Servicer will continue the servicing of such loans, pursuant to the terms of this Servicing Agreement;

NOW THEREFORE, in consideration of the mutual covenants herein contained and of other valuable consideration, the parties hereto agree as follows:

ARTICLE ONE DEFINITIONS

SECTION 1.1 Definitions. Wherever used in this Servicing Agreement, unless the context clearly indicates a contrary intent, the following terms will have the meanings indicated:

- (a) "Act" means, with respect to FFELP loans, Part B of Title IV of the Higher Education Act of 1965, as amended (20 U.S.C. section 1071 *et seq.*), and in effect from time to time, or any successor enactment thereto.
- (b) "Affiliate" means, with respect to a party hereto, an entity or person who, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such party.
- (c) "Amount That Guarantor Would Have Paid" means the amount of money that the Guarantor would have paid the Bank if the claim in question had been paid by the Guarantor in accordance with, for government-guaranteed loans, the Act and the guarantee agreement between the Bank and the Guarantor.
- (d) "Application" means an application, in any form authorized by the applicable Program Rules, from a prospective Borrower for a Private Loan.

- (e) "AUPs" has the meaning set forth in Section 5.1(e)(i).
- (f) "Bank" has the meaning set forth in the preamble of this Servicing Agreement.
- (g) "Bank Customer Information" has the meaning set forth in Section 5.1(f)(x)(A).
- (h) "Bank Default" has the meaning set forth in Section 8.2.
- (i) "Bank Security Requirements" has the meaning set forth in Section 5.1(f).
- (j) "Borrower" means the obligor on a Loan, and includes any Cosigner.
- (k) "Borrower Benefits" are those benefits granted to certain borrowers as more fully set forth in Attachment 8.
- (1) "Career Training Loan" means a loan for career training offered under the Program Rules of a specific program approved by the Servicer for servicing hereunder, as the same may be amended from time to time by written agreement of the Bank and the Servicer.
- (m) "Collection Costs" mean collection costs that are authorized to be charged to the Borrower by the Act, Regulations, promissory notes, the applicable Program Rules or applicable law.
- (n) "Confidential Information" means all information (including, without limitation, Bank Customer Information), technical data or know-how, in whatever form, which the Bank transmits to the Servicer in any manner, even if disclosed to the Servicer prior to the Effective Date, relating to the properties, customers, business activities or operations of the Bank, including, without limitation, (i) all confidential or "proprietary" information of the Bank disclosed in writing or other tangible form, including, without limitation, all customer data and information, trade secrets, patents, specifications, know-how, designs, drawings, sketches, models, notes, documents, samples, reports, plans, forecasts, methods of doing business, current or historical data, computer software and programs, codes and all other technical, financial or business information; (ii) all information of the Bank that is not known to the public, without regard to the form in which such information is disclosed; (iii) any and all of the foregoing which is prepared by the Servicer or its employees or agents that contains, reflects or is based upon, in whole or in part, any of the Confidential Information; and (iv) the contents of discussions and negotiations between the parties concerning this Agreement. Confidential Information shall not include information that the Servicer can demonstrate: (A) is or (through no improper action or inaction by the Servicer or any affiliate, agent, consultant or employee of the Servicer) becomes part of the public domain; (B) was lawfully in the possession of or known by the Servicer on a non-confidential basis prior to the time of disclosure by the Bank (and such prior possession or knowledge can be demonstrated by written documentation); (C) was received on a non-confidential basis from a third party having a lawful right to disclose such information; or (D) was independently developed by the Servicer without reference to or use of the Confidential Information as demonstrated by the Servicer's records.

- (o) "Consolidation Loan" means a Loan made pursuant to and in full compliance with Section 428C of the Act, which is insured or guaranteed pursuant to Subsection 428C(a)(2) thereof to the maximum extent such Loans are insurable by the Secretary as of the disbursement date.
- (p) "Cosigner" includes any person whose signature is requested as a condition to granting credit to a Borrower, or as a condition for forbearance on collection of a Borrower's obligation that is in default.
- (q) "Depository Account" means an account or accounts owned or maintained by the Servicer into which all funds collected on behalf of the Bank shall be deposited in the first instance. The Depository Account may be a pooled account in which funds collected for the purposes of this Servicing Agreement are commingled with other funds.
- (r) "EFT" means electronic funds transfer processes.
- (s) "FDR" means the First Data Resources servicing platform of First Data Corporation, a subcontractor of both the Servicer and the Bank.
- (t) "FFELP" means the Federal Family Education Loan Program authorized pursuant to the Act.
- (u) "FFELP Loan" means any loan that is made pursuant to the FFELP program authorized by the Act, including, without limitation, Consolidation Loans made under Section 428C of the Act.
- (v) "GLB Requirements" means the Federal Trade Commission's Privacy of Consumer Financial Information; Final Rule (16 CFR 313) (the "FTC Final Rule"), the Joint Banking Agencies' Privacy of Consumer Financial Information; Final Rule (12 CFR Parts 40, 216, 332 and 573) (the "Banking Agency Regulation P"), or the Consumer Financial Protection Bureau's Privacy of Consumer Financial Information (Regulation P) (12 CFR 1016) (the "CFPB Regulation P"), as applicable, each implementing Title V of the Gramm-Leach-Bliley Act, Public Law 106-102.
- (w) "Guarantee Agency" or "Guarantor" means (i) for FFELP Loans, an entity authorized to guarantee FFELP Loans under the provisions of the Act and (ii) for Private Loans, an entity that enters into a guarantee or insurance agreement with the Bank.
- (x) "Indemnified Party" has the meaning set forth in Section 10.4.
- (y) "Indemnifying Party" has the meaning set forth in Section 10.4.
- (z) "Identified Funds" means a payment or payments received by Servicer from any third party and which is identified by Servicer as a payment or payments made on account of a specific Loan, including, without limitation: (i) a payment of principal, interest, or other charges or fees; (ii) a guarantee claim payment; (iii) the return of previously

negotiated proceeds of a wholly or partially canceled Loan; (iv) the return of a guarantee/insurance premium on account of a wholly or partially canceled Loan; (v) Returned Check Charges, Late Fees, and Collection Costs (as defined herein); and (vi) other fees and charges that are permitted by the Act or the Regulations or the applicable promissory note.

- (aa) "Information Security Program Requirements" has the meaning set forth in Section 5.1(f)(x)(B).
- (bb) "Insolvency Event" shall mean with respect to a party, the existence of any of the following conditions: (i) a petition in bankruptcy or insolvency regarding the party shall be filed; (ii) the party shall admit in writing its inability to pay its debts generally; (iii) the party shall make a general assignment for the benefit of creditors; (iv) any proceeding shall be instituted by the party seeking relief by such party, as debtor, under any bankruptcy or insolvency legislation, or seeking liquidation, winding up, reorganization, arrangement, adjustment, protection, relief, or composition of it or its debts, or seeking the entry of an order for the appointment of a receiver, trustee, or other similar official for it or for any substantial part of its property; or (v) any of the foregoing actions shall be instituted against a party and not dismissed within sixty (60) days or a party shall take any corporate action to authorize any of the actions set forth herein. "Bank Insolvency Event" means an Insolvency Event with respect to the Bank or an entity of which the Bank is a wholly-owned subsidiary. "Servicer Insolvency Event" means an Insolvency Event with respect to the Servicer or an entity of which it is a wholly-owned subsidiary.
- (cc) "LARS" means the Lender's Request for Payment of Interest and Special Allowance (LARS), formerly known as ED Form 799. This is applicable to loans made under the FFELP.
- (dd) "Late Fees" means late fees that are permitted by the Act or the Regulations or the applicable promissory notes.
- (ee) "Loan" means any of, and "Loans" means all of, the FFELP Loans (including, without limitation, Consolidation Loans), and Private Loans (including, without limitation, Career Training Loans) serviced pursuant to the provisions of this Servicing Agreement.
- (ff) "Loan Documentation" means all attachments and other documentation relating to a Loan that are in the possession of the holder, or an agent of the holder. Without limiting the generality of the foregoing, the Loan Documentation for any Loan may include, as applicable:
 - i. A copy of the Borrower's Application;
 - ii. The notice of guarantee or other proof of guarantee of the Loan by a Guarantor;
 - iii. Evidence of all other documentation necessary to establish the diligent origination of the Loan in accordance with the Act and the Regulations (for

FFELP Loans) or the applicable Program Rules (for Private Loans), including but not limited to Customer Identification Program (CIP) documentary verification evidence, FICO score or credit bureau reports and record of any denials;

- iv. The Note or a true copy thereof;
- v. Evidence of full disbursement;
- vi. Evidence of all other documentation necessary to establish the diligent servicing of the Loan in accordance with the Act and the Regulations (for FFELP Loans) or the applicable Program Rules (for Private Loans), including CIP and e-sign verification documentation;
- vii. An assignment or endorsement of the Note;
- viii. Loan denial documentation;
- ix. For purposes of sale of a Loan to the Servicer's designee, Loan Documentation shall also include:
 - (A) An assignment or endorsement of the Note to Servicer's designee; and
 - (B) Evidence of authority of the Bank to convey the Loan and the rights thereto.
- (bb) "Material Guarantee Breach" has the meaning set forth in Section 9.1(a).
- (cc) "Monetary Breach" has the meaning set forth in Section 12.4.
- (dd) "Note" or "Promissory Note" means (i) for a FFELP Loan, the form promissory note (in a form approved under the Act) pursuant to which a Borrower obtains a FFELP Loan, and (ii) for a Private Loan, the form promissory note applicable to the particular loan program pursuant to which a Borrower obtains such Private Loan. A "Master Promissory Note" or "MPN" denotes the form promissory note that allows Borrowers to receive, in addition to initial loans, additional loans for the same or subsequent periods of enrollment through a student confirmation process approved under the Act.
- (ee) "Originate," "originated," "originating," or "origination," used in connection with prospective or approved Loans includes any or all of the following: application processing, guarantee processing, credit analysis (when applicable), and disbursement processing (including multiple disbursements).
- (ff) "Other Agreements" has the meaning set forth in Section 10.1(a).
- (gg) "Payment Processing Day" means any day other than a Saturday, Sunday, or a day on which the banking institutions in the city in which a payment is received or a disbursement hereunder must be initiated are authorized or required by law or executive order to close.

- (hh) "PCI" and "DSS" have the respective meanings set forth in Section 5.1(k).
- (ii) "Preferred Stock Period" has the same meaning ascribed to such term in the Separation and Distribution Agreement.
- (jj) "Previously Serviced Loan" shall mean any Private Loan that was serviced by the Servicer under the Prior LOMSA and/or under this Servicing Agreement the servicing of which is subsequently deconverted to the Bank or its Affiliate for servicing on or after the Effective Date, but for which the Servicer remains in possession of relevant historical data, including but not limited to Borrower servicing histories and the form promissory notes applicable to the particular Private Loan program pursuant to which the Borrower obtained such Private Loan.
- (kk) "Prior LOMSA" has the meaning set forth in the recitals to this Servicing Agreement.
- (ll) "Private Loan" means an education loan not guaranteed by the government and offered by a lender under the terms of a specific program approved by the Servicer for servicing hereunder, as described in Attachment 5, as the same may be amended from time to time by written agreement of the Bank and the Servicer.
- (mm) "Program Rules" means the regulations, and rules, documents or manuals adopted by the Bank that govern any Private Loan program (including Career Training Loans), as amended from time to time, provided such materials have been delivered to and approved in writing by the Servicer.
- (nn) "Receipts Account" means an account beneficially owned by the Bank and designated by the Bank as the account to which the Servicer shall transfer funds collected on behalf of the Bank from Borrowers, Guarantee Agencies, and the Secretary.
- (00) "Refund Account" means an account or accounts owned or maintained by the Servicer from which refunds are made to Borrowers and others entitled thereto in connection with payments related to Loans.
- (pp) "Regulation" or "Regulations" means any rule, regulation, interpretation, instruction or procedure issued and in effect from time to time by the Secretary under the Act or by a Guarantee Agency with respect to FFELP Loans guaranteed by it, and, with regard to Private Loans, and as applicable to FFELP Loans, all state and federal consumer credit laws and their respective interpretive and implementing rules and regulations.
- (qq) "Returned Check Charges" means charges for the processing of a Borrower's check or other payment that is returned because of insufficient funds.
- (rr) "Secretary" means the U.S. Secretary of Education or his or her delegate or successor under the Act.
- (ss) "Security Audit" has the meaning set forth in Section 5.1(h).

- (tt) "Security Incident" has the meaning set forth in Section 5.1(g).
- (uu) "Separation and Distribution Agreement" means that certain Separation and Distribution Agreement dated as of April , 2014 among SLM Corporation, New BLC Corporation, and Navient Corporation.
- (vv) "Service," "services," "serviced," or "servicing" used in connection with a Loan or Loans means the process and procedures required of the Servicer or a lender with respect to such Loan or Loans pursuant to the terms of the Act or the Regulations (for a FFELP Loan) or the applicable Program Rules that are delivered to and approved by the Servicer (for a Private Loan).
- (ww) "Servicer" has the meaning set forth in the preamble of this Servicing Agreement.
- (xx) "Servicer Default" has the meaning set forth in Section 9.2.
- (yy) "Service Performance Measures" means the specific activity-related measures set forth in Attachment 2 hereto.
- (zz) "Servicing Agreement" means this Amended and Restated Loan Servicing Agreement.
- (aaa) "Special Allowance" means amounts paid by the Secretary to the holder of a Loan as authorized and calculated under Section 438 of the Act.
- (bbb) "Split-Account Loan" means a Private Loan held by the Bank where the same Borrower account also has at least one private loan held by the Servicer or an Affiliate of the Servicer.
- (ccc) "Subcontractors" means all agents, contractors, subcontractors, and vendors that perform services that are used by the Servicer to service the Loans pursuant to the terms of this Servicing Agreement.
- (ddd) "Term" has the meaning set forth in Section 7.1(a).
- (eee) "Transition Services Agreement" means that certain Transition Services Agreement dated as of April , 2014 between the Servicer and SLM Corporation.
- (fff) "Transition Services Period" means the period beginning on the Effective Date and ending on the date upon which the parties to the Transition Services Agreement agree that the separation of loan servicing systems covered by Schedule 2 thereof has been completed.

ARTICLE TWO AMENDMENT AND RESTATEMENT; CUSTODY AND TITLE

SECTION 2.1 <u>Amendment and Restatement.</u> As of the Effective Date, the Prior LOMSA is hereby amended and restated in its entirety with the terms and conditions of this Servicing Agreement.

SECTION 2.2 Custody and Title.

- (a) <u>Loans Previously Serviced by Servicer</u>. Except as set forth in subsection (d) below, the Servicer, as agent for the Bank, shall maintain custody of the Loan Documentation relating to those Loans serviced by Servicer on behalf of the Bank on or before the Effective Date, where the Loan Documentation was delivered to (or already in the possession of) the Servicer.
- (b) <u>Custody of Loan Documentation by Guarantee Agency</u>. The covenants set forth in (a) and (b) of this Section 2.2 are qualified to the extent any Guarantee Agency, as a part of its program, maintains custody of promissory notes or other Loan Documentation. The Servicer makes no representation or warranty with respect to custody and maintenance of promissory notes or other Loan Documentation held by the Guarantee Agency as of the Effective Date or thereafter.
- (c) Collateral Identification. The Servicer, as agent for the Bank, shall ensure that all loans identified by the Bank to be used as pledged collateral to secure federal funding, and for which the Servicer maintains the Loan Documentation on behalf of the Bank, are clearly identified as collateral by all of the following: (i) labeling file cabinet(s) which contain the pledged loan documents, (ii) posting a highly visible sign in the area where the pledged loans are located, with wording prescribed and approved by the Bank, and (iii) electronic notation to flag pledged loans on the Servicer's loan system.
- (d) <u>Title to Loans</u>. Delivery of Loans or Loan Documentation to the Servicer hereunder and/or custody of the Loans, and/or servicing activities contemplated hereby, shall not be deemed to convey to the Servicer any of the Bank's beneficial or legal ownership interest in the Loans. The Servicer recognizes that the Bank has beneficial and legal ownership of the Loans, the Loan Documentation therefore, and the rights and benefits pertaining thereto.
- (e) <u>Custody of Loan Documentation for Previously Serviced Loans</u>. The Servicer shall maintain custody of any Loan Documentation in its possession as of the Effective Date pertaining to the Previously Serviced Loans. From time to time upon request from the Bank, the Servicer will deliver to the Bank physical original copies, or if such physical original copies do not exist, imaged copies of Loan Documentation, including but not limited to Borrower servicing histories and the form promissory notes applicable to the particular loan programs pursuant to which the Borrowers obtained such Private Loans. With the exception of documentation requested relating to the US Patriot Act or Bank Secrecy Act ("BSA"), the Servicer will make a commercially reasonable effort to provide the requested documentation to the Bank within seven (7) business days after receiving a request from the Bank; provided, however, that, if the volume of documentation requested pursuant to this Section 2.2(e) shall cause the Servicer to be unable to comply with such seven (7) business day delivery obligation without significant additional burden or expense, such delivery obligation may be modified or extended by consent of the parties hereto, with such consent not to be unreasonably withheld. With regard to documentation requested relating to the US Patriot Act or BSA, Servicer shall provide those documents to the Bank within 96 hours. Requests for information pertaining to ten (10) or more Borrower accounts in any given thirty (30) day period will be provided to the Bank for a fee equal to the Servicer's cost to comply with the request, plus twenty percent (20%).

ARTICLE THREE SERVICING

SECTION 3.1 General Obligations of the Servicer. From and after the Effective Date, the Servicer will service: (a) all FFELP Loans that are serviced by the Servicer or its Affiliate on the Bank's behalf immediately prior to the Effective Date; (b) all Split-Account Loans that are serviced by the Servicer or its Affiliate on the Bank's behalf immediately prior to the Effective Date (provided, however, in accordance with Section 3.6 hereof, that during the Transition Services Period, the Transition Services Agreement shall govern the servicing of such Split-Account Loans); and (c) such other Loans that the parties mutually agree are to be serviced hereunder. The parties acknowledge that all Private Loans held by the Bank that are not Split-Account Loans and that were previously serviced by the Servicer or its Affiliate under the Prior LOMSA shall not be serviced under this Servicing Agreement; provided, however, that the Servicer's obligations under Section 2.2(e) shall apply to all such Loans in their capacity as Previously Serviced Loans. The Servicer, as agent for the Bank, shall maintain custody of the Loan Documentation unless custody shall be maintained by the Guarantee Agency; shall manage, service, administer and make collections on the Loans; shall maintain records with respect to all Loans; and shall make reports as specified in Attachment 3 hereto (Servicer Reporting), except as may be otherwise agreed upon by the Bank. The Servicer shall (i) attempt to collect payments from the obligors of the Loans as and when the same shall become due and payable (including Returned Check Charges, Late Fees, and Collection Costs, but the Servicer is authorized to waive such items in its reasonable discretion in accordance with applicable Lending Policies); (ii) submit, pursue and collect guarantee claims with respect to the Loans from the Guarantee Agencies as and when the same shall become due and payable; (iii) take all other steps mandated by the Act or the Regulations (for FFELP Loans) or applicable Program Rules (for Private Loans) in connection with third-party servicing and necessary to preserve the guarantee of a Loan under the Act; (iv) process all deferrals and forbearances; (v) collect on Loans in a diligent manner and also as required under the Act or Regulations or the applicable Program Rules (as applicable); (vi) at the request of the Bank and based upon procedures agreed to by the parties, provide cure servicing, that is, where, without the fault of the Servicer, the guarantee of a Loan has been lost or impaired, or the Servicer has determined that the guarantee of a Loan has been lost or impaired, take reasonable steps for the purpose of reinstating the guarantee; (vii) prepare and transmit to the Bank all necessary information for interest subsidy and Special Allowance billing applicable to the Loans; and (viii) perform fraud monitoring, investigation, and reporting, including the preparation of Suspicious Activity Reports (SARs). Servicer will also administer the Borrower Benefits as set forth in Attachment 8.

SECTION 3.2 Establishment and Authorization for Use of the Receipts Account. The Bank shall provide the Servicer with all authorizations and information, and shall take all such further steps as may be necessary, in order to authorize and enable the Servicer to initiate the movement of funds to and from the Receipts Account by EFT. The Servicer shall not use its access to the Receipts Account except for the purposes expressly authorized in this Servicing Agreement.

SECTION 3.3 Collections. The Servicer will use reasonable efforts to collect, on behalf of the Bank, pursuant to the Act and the Program Rules standards, all payments with respect to any Loan from the Borrower, any party responsible for payments as a Guarantor or Cosigner of the Loan, school (refunds, voids) and the Guarantee Agency. Should Servicer collect from a Bank customer who is also a Servicer customer, Servicer will treat the Bank's account(s) with the same rigor it employs for its own, and shall not treat Servicer's accounts in a manner more favorably than the Bank's.

- (a) <u>Identified Funds: Deposit and Transfer.</u> Not later than the first Payment Processing Day that any payment received from a third party becomes Identified Funds, the Servicer will deposit the payment into the Depository Account. The Servicer will post the payment and initiate EFT transfer of Identified Funds to the Receipts Account within one Payment Processing Day after deposit to the Depository Account; provided, however, that Servicer may withhold for their proper purpose amounts needed to offset previous overpayments to the Receipts Account and to effect any other adjustment reasonably deemed necessary by the Servicer to resolve an account balance discrepancy, including, by way of example, but without limitation, (i) overpayments resulting from a Borrower payment received after a guarantee claim has been processed for the same account, (ii) overpayments resulting from overestimated balances due to retire or consolidate Loans, (iii) payments mistakenly recognized as Identified Funds, wholly or in part, (iv) overdue servicing fees and charges as authorized in Section 8.2, below, and (v) Returned Check Charges and Collection Costs pursuant to subsection (c), below. Monies so withheld may be transferred to the Refund Account or otherwise transferred to the proper party.
- (b) <u>Unidentified Funds: Processing</u>. Payments received by the Servicer, which are not readily identified as payments on account of a particular Loan because of the omission or absence of a scannable coupon or billing statement, or other reason, will be further processed manually. Payments that are ultimately recognized as payments on account of a particular Loan become Identified Funds and will be posted and transferred in accordance with the provisions of subparagraph (a), above. The Servicer will be responsible for escheating funds which, after reasonable due diligence, remain unidentifiable.
- (c) Charges to Borrower for Servicer's Account. The Servicer is authorized to charge to and collect from any Borrower, for the Servicer's account, such reasonable sums as agreed to by the Bank and the Borrower in appropriate disclosures to compensate it for (i) activities that would not be necessary except for a special request made by the Borrower or conduct of the Borrower inconsistent with the terms of the contract between the Bank and the Borrower (for example, but without limitation, Returned Check Charges, Collection Costs, and retrieval and transmittal to Borrower of Borrower's payment history or amortization schedule). To the extent of any Identified Funds attributable to a Loan, and from the first monies collected, the Servicer shall be entitled to retain Returned Check Charges and Collection Costs applicable to that Loan and, solely with respect to FFELP Loans, fifty percent (50%) of the Late Fees applicable to that Loan. Late Fees applicable to Private Loans shall be the property of the Bank.

SECTION 3.4 Reports. The Servicer will make available to the Bank on a periodic basis as agreed to by the parties, reports, data extracts, and data tapes substantially in the form prepared as of the date hereof (or as modified if necessary to comply with the Act, the Regulations, the applicable Program Rules or the federal laws or as otherwise permitted by this Servicing Agreement) by the Servicer and generated by the Servicer's servicing system. Except as otherwise provided herein, the Servicer shall also, for and on behalf of the Bank, file all statements, reports and bills required to be submitted to the Secretary, the Guarantee Agency, any other governmental agency or any other entity pursuant to the Act or the Regulations. In addition, the Servicer will provide the Bank with necessary documentation and information and assist the Bank in the Bank's response to any inquiry or investigation by any governmental agency.

SECTION 3.5 LARS Information. With respect to all FFELP Loans serviced hereunder, the Servicer shall, within twenty (20) days after the end of each calendar quarter, prepare and (i) deliver to the Secretary, in such manner as the Secretary requires, all necessary information for interest subsidy and Special Allowance billing, and (ii) make available to the Bank, a summary report showing the interest and Special Allowance information that was reported to the Secretary on the Lender's behalf. In addition, the Servicer will provide the Bank with a copy of its annual Servicer Audit Report (as set forth on Attachment 1) to assist the Bank in its preparation and filing of the Bank's Lender Compliance Attestation Examination reports.

SECTION 3.6. <u>Transition Services Period</u>. During the Transition Services Period, certain servicing activities that the Servicer might otherwise be required to provide pursuant to this Servicing Agreement may be covered by the Transition Services Agreement. In such event, the applicable provisions of the Transition Services Agreement shall control, and the affected provisions of this Servicing Agreement shall be inoperative with respect to such servicing activities. Following the termination of the Transition Services Period, any provisions of this Servicing Agreement that may have been inoperative during such time, pursuant to the immediately preceding sentence, shall again become operative in accordance with their terms and conditions under this Servicing Agreement for the remainder of the Term.

SECTION 3.7 Obligations of the Bank.

- (a) <u>Forwarding Communications and Payments</u>. If the Bank receives any communications regarding a Loan, it will promptly forward those communications to the Servicer in the manner in which the Servicer reasonably designates. The Bank will also promptly forward to the Servicer any payments the Bank receives on account of a Loan.
- (b) <u>Authorization and Power of Attorney</u>. The Bank hereby authorizes the Servicer on behalf of the Bank to communicate as the Bank's agent with the school, Borrower, Guarantee Agency, and the United States Department of Education by electronic means or otherwise, to issue checks and electronically disburse funds as contemplated herein, to exchange information pursuant to the Act and Regulations with credit bureau organizations selected by the Servicer, and to do or perform any other act for the purpose of carrying out its obligations of servicing Loans. The Bank hereby appoints the Servicer as its lawful attorney-in-fact to sign in the name of the Bank such documents as are necessary for the Servicer to perform its obligations as contemplated under this Servicing Agreement. Each party will provide to the other specimen authorized signatures upon request. The Bank will also execute a specific power-of-attorney covering the foregoing if required by a Guarantee Agency, or otherwise as the Servicer may reasonably request.
- (c) <u>Guarantee Agency Agreements</u>. The Bank will furnish the Servicer with copies of all agreements between the Bank and any Guarantee Agency or other party affecting the Loans serviced under this Servicing Agreement.
- (d) <u>Bank Identification Numbers</u>. The Bank shall secure, provide and maintain all necessary U.S. Department of Education lender identification numbers sufficient to support the duties and obligations of the Bank and the Servicer under the terms of this Servicing Agreement and Guarantor requirements.

SECTION 3.8 Inquiries and Complaints. Each party will cooperate with the other as necessary to respond promptly and appropriately to any inquiries and complaints from any source, including social media, e-mail, letter, etc., from the Borrower, the Guarantee Agency, the Secretary, the Borrower's school, or any government or regulatory agency regarding Loans that the Servicer is servicing for the Bank. It is anticipated that all complaint responses shall be drafted by the Servicer, subject to the Bank's approval (provided, however, that if the Servicer does not receive comment back from the Bank within one (1) business day after the Servicer has delivered the draft response to the Bank, the Bank's approval shall be deemed to have been given and the Servicer may proceed with submitting the response). However, the Bank reserves the right to respond to complaints (with information in order to so respond to be supplied by the Servicer) if required by the Bank's regulators. To the extent allowed by law, each party will promptly notify the other if it receives any inquiries or complaints from the Borrower, the Guarantee Agency, the Secretary, the Borrower's school, or any government or regulatory agency regarding Loans that the Servicer is servicing for the Bank. Each party will advise the other within five (5) business days after such party's receipt of any preliminary finding or final report related to a review done by the Secretary or a Guaranty Agency pertaining to Loans serviced under this Servicing Agreement.

ARTICLE FOUR REPRESENTATIONS AND WARRANTIES

SECTION 4.1 Servicer's Representations and Warranties. The Servicer represents and warrants the following to the Bank:

- (a) The Servicer is duly organized and qualified to do business in each jurisdiction in which qualification is required for the activities contemplated by this Servicing Agreement. The execution and performance of this Servicing Agreement has been duly authorized by all necessary action and does not and will not contravene any provision of law, rule or regulation applicable to the Servicer. The Servicer is eligible under the Act and the Regulations to service the Loans.
- (b) This Servicing Agreement constitutes a legal, valid and binding obligation of the Servicer enforceable in accordance with its terms subject to bankruptcy, insolvency and other similar laws affecting creditors' rights generally and subject to equitable principles.

SECTION 4.2 Bank's Representations and Warranties. The Bank represents and warrants the following to the Servicer:

- (a) The Bank is duly organized and qualified to do business in each jurisdiction in which qualification is required for the activities contemplated by this Servicing Agreement. The execution and performance of this Servicing Agreement has been duly authorized by all necessary action and does not and will not contravene any provision of law, rule or regulation applicable to the Bank.
- (b) This Servicing Agreement constitutes a legal, valid and binding obligation of the Bank, enforceable in accordance with its terms subject to bankruptcy, insolvency and other similar laws affecting creditors' rights generally and subject to equitable principles.

ARTICLE FIVE COVENANTS

SECTION 5.1 Servicer's Covenants. The Servicer covenants and agrees that:

- (a) The Servicer will (or will cause its subservicer or Subcontractor to) furnish specific Loan information to the Bank on request, will retain the Loan documents and all computer and manual records, receipts, tapes and other papers delivered to, acquired by or generated by the Servicer with respect to the Loans, and will not destroy or dispose of them, or use the information contained therein or disclose the contents thereof to others except as required or permitted under this Servicing Agreement or as required by law. All such documents, records, receipts, tapes and papers must be retained by the Servicer until the end of any period required by law, including the Act, the Regulations, the US Patriot Act and the Internal Revenue Code and any regulations issued thereunder. The Loan documents, records, receipts, tapes and papers must be kept in a manner to enable delivery to the Bank upon request on a timely basis; however, unless Loan documents, records, receipts, tapes and papers relating to the Loans are required to be maintained in paper form pursuant to the Act, Regulations, or the applicable Program Rules (as applicable), or other applicable state or federal law, the Servicer is authorized to destroy and dispose of any of them, provided that the Servicer has retained them in retrievable imaged form or has recorded their contents electronically and integrated such contents into the Servicer's data base pertaining to the Loans such that they are easily retrievable. In the event that the Bank requests that the Servicer retain the aforementioned Loan documents, records, receipts, tapes and papers for a period longer than is set forth above, the Servicer shall do so at the Bank's expense, billed as a special service in accordance with Attachment 1.
- (b) The Servicer will (or will cause its subservicer or Subcontractor to) furnish call recordings with respect to the Loans to the Bank on request, will retain the call recordings with respect to the Loans in accordance with the Servicer's call retention policy (which as of the Effective Date is for a period of thirty (30) days) or as otherwise required by law or as otherwise required by regulators (provided, however, that if the longer period is due to a regulatory requirement placed on the Bank and not the Servicer, then the Servicer shall do so at the Bank's expense, billed as a special service in accordance with Attachment 1), and will not use the information contained therein or disclose the contents thereof to others except as required or permitted under this Servicing Agreement or as required by law.
- (c) The Servicer will permit the Bank during normal business hours to gain real-time access to electronic information and data relating to the Loans, in accordance with the applicable provisions of Attachment 4 (Data Access for Owners and Borrowers).
- (d) The Servicer will make appropriate provision for physical security of its data processing and record retention systems. The physical security provisions will, at a minimum, restrict physical access to any and all Bank data to only those individuals who have a legitimate business need to access the data. Physical security controls will be tested on a regular basis to make sure that only authorized individuals have physical access to the Bank's data.
- (e) The Servicer will make appropriate provision for data back-up and off-premises storage of Bank data. The Servicer will promptly regenerate any lost, erased or destroyed data. In the event of a disaster, the Servicer agrees to implement a disaster recovery plan, in

conformance with Attachment 6, that will enable it to conduct due diligence (except to the extent that any applicable Guarantor waives due diligence as to it) and payment processing within fifteen (15) Payment Processing Days and to regain all other aspects of its ability to service the Loans within thirty (30) days after the occurrence of a disaster.

(f) The Servicer will maintain appropriate information security controls as more particularly set forth in Attachment 7 (the "Bank Security Requirements"), including:

- (i) Access controls that restrict access to system resources and data to only to those users who require access as part of their job responsibilities. The Servicer will update access rights based on personnel or system changes and periodically review users' access rights at an appropriate frequency based on the risk to the application or system. The Servicer will implement appropriate Acceptable-Use Policies ("AUPs") for each system or application and will require users to adhere to the AUPs as part of the Servicer's Information Security Policy.
- (ii) Authentication controls appropriate to the level of risk in order to protective electronic Bank data and information.
- (iii) Appropriate network access controls that secure the computer networks to protect against unauthorized access.
- (iv) Operating systems access controls that secure access to the operating systems of all system components by: securing access to system utilities; restricting and monitoring privileged access; logging and monitoring user or program access to sensitive resources and alerting on security events; updating the operating systems with security patches, and securing the devices that can access the operating system through physical and logical means.
- (v) Application access controls that control access to applications by using authentication and authorization controls appropriately for the risk level of the application, and monitoring of access rights to ensure they are the minimum required for the user's current business needs.
- (vi) Remote access controls that provide secure remote access connectivity and appropriate user validation and authentication, and which can be deactivated on an individual user basis as may be required; controlling remote access through management approvals and subsequent audits, logging and monitoring of remote access, and using strong authentication and encryption to secure communications.
- (vii) Use of encryption to protect information and mitigate the risk of disclosure or alteration of sensitive information stored on backup tapes or while being shipped or transported via courier service.
- (viii) Malicious code controls that protect against the risk of malicious code, including but not limited to using anti-virus products on clients and

servers; using appropriate blocking strategy on the network perimeter; filtering input to applications; and creating, implementing, and training staff in appropriate computing policies and practices.

- (ix) Intrusion detection and monitoring controls that allows the Servicer to detect and respond to information system intrusions.
- (x) The Servicer shall comply, and shall cause all of its employees and Subcontractors to comply, to the extent they have access to the Bank Customer Information (as defined below), with all reuse, redisclosure and other customer information handling, processing, security, and protection requirements that are specifically required of a non-affiliated third-party processor or servicer (or Subcontractor) under the GLB Requirements and other applicable federal and state consumer privacy laws, rules, and regulations. Without limiting the foregoing, the Servicer agrees that:
 - (A) it is prohibited from disclosing or using any nonpublic personal information (as defined in the GLB Requirements) that it may obtain, maintain, process, or otherwise receive from, through or on behalf of Bank in connection with this Servicing Agreement (the "Bank Customer Information"), except solely to carry out the purposes for which it was disclosed, including use under an applicable exception contained in Section 313.14 or 313.15 of the FTC Final Rule, Section 332.14 or 332.15 of the Banking Agency Regulation P, or Section 1016.14 or 1016.15 of the CFPB Regulation P, as applicable, in the ordinary course of business to carry out those purposes; and
 - (B) it has implemented and will maintain a written information security program that complies with applicable state laws and regulations pertaining to the protection of personal information (e.g., MA 201 CMR 17.00) and that is designed to meet the following objectives as set forth in the Interagency Guidelines Establishing Information Security Standards; Final Rule (12 CFR Part 30, et al.) (the "Information Security Program Requirements"):
 - (1) Ensure the security and confidentiality of the Bank Customer Information;
 - (2) Protect against any anticipated threats or hazards to the security or integrity of such information;
 - (3) Protect against unauthorized access to or use of such information that could result in substantial harm or inconvenience to any customer; and

(4) Ensure the proper disposal of Bank Customer Information and "consumer information" (as that term is defined in the Interagency Guidelines Establishing Information Security Standards; Final Rule (12 CFR Part 30, et al.)).

(g) Unless specifically requested by law enforcement not to communicate with the Bank, the Servicer shall report to the Bank all known Security Incidents. "Security Incident" means any unauthorized action by a known or unknown person or automated program (e.g., worm, virus, web crawler, malware, etc.) that leads to one of the following: unintended disclosure of confidential, customer, employee or other sensitive information, denial of service, misuse of system access, or unauthorized access or intrusion, all to the extent they affect the security, confidentiality or integrity of the Bank's Confidential Information received, stored, processed, or maintained by the Servicer. "Security Incident" shall also include any contact by a law enforcement agency with the Servicer regarding any of the Bank's Confidential Information, unless specifically mandated by law enforcement not to communicate with the Bank. To the extent the Servicer becomes aware of any Security Incidents occurring with respect to its Subcontractors that have access (either authorized or unauthorized) to the Bank's Confidential Information, the Servicer shall be required to report such Security Incidents in accordance with the provisions of this Section. If a Security Incident occurs, the Servicer shall notify the Bank within 24 hours at telephone number (855) 756-2677 and by email to SOC@salliemae.com (with a follow-up written notification sent immediately via overnight mail to Sallie Mae Bank, Attention: Chief Security Officer and Chief Privacy Officer, 300 Continental Dive, Newark, DE 19713), and provide the following information, to the extent known at such time: nature and impact of the Security Incident; actions already taken by the Servicer; the Servicer's assessment of immediate risk; and corrective measures to be taken, evaluation of alternatives, and next steps. The Servicer shall continue providing (i) appropriate status reports to the Bank regarding the resolution of the Security Incident and prevention of future such Security Incidents, and (ii) cooperation, as reasonably requested by the Bank, in order to further investigate and resolve the Security Incident. The Bank may require that the services provided by the Servicer to the Bank be suspended, connectivity with the Servicer be terminated, or that other appropriate action be taken pending such resolution. The Servicer shall preserve evidence of all Security Incidents and allow external forensic analysis either onsite or through shipment of components.

(h) During the Term and thereafter for as long as the Servicer retains the Bank's Confidential Information, the Bank, its representatives and agents will be entitled to conduct audits of the Servicer's relevant operations, facilities, and systems to confirm that the Servicer has complied with the Bank Security Requirements and the Information Security Program Requirements (each, a "Security Audit"). Any Security Audit shall be scheduled with reasonable prior notice and conducted during normal business hours and shall not unreasonably interfere with the Servicer's business activities. The Servicer may require any such auditor to sign a customary confidentiality agreement. In the event that any Security Audit results in the discovery of material security risks to the Bank's Confidential Information, the Servicer shall respond to the Bank in writing with the Servicer's plan to promptly take reasonable measures and corrective actions necessary to effectively eliminate the risk, at no cost to the Bank. The Servicer shall have fifteen (15) business days to cure such security risk, unless the parties mutually agree in writing to a longer period of time for such cure. The Bank's right, and the right of its representatives and agents, to conduct Security Audits, and any exercise of such right, shall not in any way diminish or affect the Servicer's duties and liabilities under this Servicing Agreement.

- (i) The Servicer will, at its sole cost, conduct its own audits pertaining to the services hereunder consistent with the audit practices of well managed companies that perform similar loan servicing services. Annually, during the Term and for so long as the Servicer continues to maintain any of the Bank's Confidential Information, the Servicer shall, at its sole cost: (i) have a nationally recognized accounting firm conduct an audit in accordance with the Statement on Standards for Attestation Engagements No. 16 (SSAE 16) developed by the American Institute of Certified Public Accountants, and have such accounting firm issue a Service Organization Control 1 (SOC-1) Type 2 report (or substantially similar report in the event that the SSAE 16 auditing standard, SOC-1 Type 2 report is no longer an industry standard) which shall cover the environment maintained by the Servicer to provide the services hereunder; and (ii) provide the Bank and its independent auditors with a copy of the SOC-1 Type 2 report promptly upon the completion thereof, together with plans for addressing any deficiencies identified in the report. The foregoing pertains to the Servicer's SSAE 16 report for FFELP Loans; in the event that the Bank desires that the Servicer also produce an annual SSAE 16 report for Private Loans, the Servicer will make a copy of such report available to the Bank at the Bank's expense, billed as a special service in accordance with Attachment 1.
- (j) The Servicer shall be responsible for the acts and omissions of its Subcontractors as if they were the acts and omissions of its employees. Without limiting the foregoing, the Servicer (i) shall oversee any such Subcontractors that obtain, maintain, process, receive, or otherwise are permitted access to Bank's Confidential Information (including, without limitation, all Bank Customer Information) by taking reasonable steps to select and retain Subcontractors that are capable of maintaining appropriate safeguards to protect the security and confidentiality of the Bank's Confidential Information, (b) shall require Subcontractors to comply with (i) confidentiality provisions substantially similar to those set forth in this Servicing Agreement, (ii) to the extent the Subcontractor has access to Bank Customer Information, privacy and security provisions substantially similar to those set forth in the Customer Information Handling Requirements paragraphs of this Servicing Agreement, and (iii) to the extent the Subcontractor has access to, stores, or processes the Bank's customers' cardholder information, the PCI DSS (defined immediately below) in effect from time to time, and (c) shall take commercially reasonable steps to require Subcontractors to adhere to the Bank Security Requirements set forth in Attachment 7.
- (k) If the Servicer has access to, stores, or processes the Bank's customers' cardholder information, the Servicer hereby confirms that it has on file a current Report on Compliance, evidencing that it is in compliance with the payment card industry ("PCI") data security standard ("DSS"). The Servicer shall provide the Bank with a copy of the PCI DSS Attestation of Compliance Letter upon request. In addition to the foregoing, to the extent applicable to the services being provided to the Bank hereunder, the Servicer will comply with and adhere to the PCI DSS in effect from time to time. Each party shall be responsible for the implementation, testing, and compliance with respect to PCI data security controls within their respective PCI DSS boundaries. These requirements are applicable to all infrastructure and systems processing or storing any cardholder information as defined by the PCI DSS. Any change in the Servicer's PCI compliance and/or certification status shall be promptly communicated to the Bank.

(l) The Servicer will comply with all audit activities required under the Act, Regulations, and the applicable Program Rules (as applicable) and will permit and assist any Guaranty Agency, Bank official or authorized agent, or governmental agency, during the Servicer's regular business hours, to examine and audit the books and records of the Servicer applicable to the Loans and to inspect the facilities, operations, training materials, and operating policies and procedures used or available for use to service the Loans, as well as a summary of the disaster recovery and business continuity plan set forth in Attachment 6 and the testing results required thereunder. The Bank shall reimburse the Servicer for the Servicer's actual and reasonable costs and expenses (including reasonable attorneys' fees) in responding to inquiries from or audits or examinations by the Bank's regulatory authorities or agencies in their review of the Bank. In addition, should the Servicer ever cease to be a publicly traded company, the Bank shall be entitled to receive periodic audited financials of the Servicer.

(m) To the extent allowed by applicable law, the Servicer shall route Bank Loan telephone calls into recording queues on a monthly basis such that all Bank-owned FFELP and Private Loan servicing and collections calls subject to this Servicing Agreement are recorded, can be segmented and identified via reasonable search methods and are available for random and unlimited listening by the Bank from the date the call is put into the queue and extending 30 days after the end of the month of the call. Until such time as the aforesaid recording queues are effectuated, the Servicer will allow the Bank to listen to Bank-owned FFELP and Private Loan servicing and collections calls subject to this Servicing Agreement by placing a minimum of 20 randomly selected calls per month in folders within the NICE listening system. There are no fees associated with this monthly listening activity. While listening to a call, if the Bank employee or representative believes the customer or the Servicer agent is disclosing non-Bank / non-affiliate data, the Bank employee or representative will immediately terminate the listening activity. If, prior to terminating the listening activity, the Bank employee or representative believes there is reason for concern about the quality of the call, the Bank employee or representative should forward the associated call identification information and the Servicer will produce a write-up concerning the call. Beginning 180 days prior to the end of the Preferred Stock Period, the parties will seek to implement a method as to how the Bank may be legally able to access its FFELP and Private Loan servicing and collections calls following the end of the Preferred Stock Period. If the parties determine that they will be unable to implement such a method to the Bank's reasonable satisfaction prior to the end of the Preferred Stock Period, then the Bank shall have the right to deconvert its Loans from the Servicer's system, in which case (i) the Bank and the Servicer will in good faith agree upon a business plan and schedule for the smooth transition of the servicing of such Loans to the Bank's servicing platform, and (ii) the Bank shall be obligated to pay only 50% of the deconversion fees that would otherwise be payable under Attachment 1. In the event the Bank is subject to a regulatory or litigation requirement to retain calls, the Servicer will, at the Bank's expense and billed in accordance with Attachment 1, provide a file of the recorded calls for delivery to the Bank in an agreed upon format. Any on-going call retention will be delivered to the Bank within 10 business days from the end of the previous month. The Bank will store the calls and retain them per the requirements of the regulatory or litigation request.

(n) The Servicer agrees to advise the Bank within five (5) business days of the Servicer's receipt of any final report issued by the Secretary, Guarantee Agency, the Federal Financial Institutions Examination Council (to the extent that the Servicer is permitted to share any such report), or other state or federal governmental entity of the Servicer's material noncompliance with the Act or the Regulations or applicable laws and that affects the Loans serviced hereunder.

- (o) The Servicer has the right to change any part or all of its equipment, its servicing system, computer or data processing programs and any procedures, forms, reports and methods in connection with the services performed hereunder so long as the Servicer continues to service the Loans in conformance with the requirements herein. Material changes to services provided (including, but not limited to, material changes in off-shore servicing) and controls must be approved by the Bank wherein such approval shall not be unreasonably withheld. It is specifically understood that the intent of this paragraph is to allow the Servicer flexibility over the methods and techniques of servicing subject to full compliance with substantive terms of this Servicing Agreement. The Servicer agrees to seek the Bank's approval of any change that persons similarly situated to the Servicer would in their reasonable judgment deem to be a material change made pursuant to this paragraph.
- (p) During the term of this Servicing Agreement, the Servicer shall maintain at least the following insurance with financially sound and reputable insurers:
 - (i) Liability Insurance:
 - (A) Commercial General Liability insurance in an amount not less than \$1,000,000 per occurrence, \$2,000,000 aggregate for claims arising out of Bodily Injury, Property Damage and Personal Injury, and broad form contractual liability insurance to insure against any Bodily Injury, Property Damage or Personal Injury arising out of this Servicing Agreement.
 - (B) Umbrella Excess Liability in an amount of at least \$10,000,000 each occurrence/aggregate excess of above.
 - (ii) Financial Institution Blanket Bond including Fidelity, Premises, Transit, Computer Crime (Funds Transfer) in an amount not less than \$15,000,000 per loss and aggregate which shall respond to loss involving assets of the Bank under the care, custody, and control of the Servicer.
 - (iii) Errors and Omissions/Professional Liability insurance in an amount not less than \$2,500,000 each claim and aggregate which shall respond to claims for wrongful acts, errors, and omissions arising out of the Servicer's performance of or failure to perform its obligations to the Bank under this Servicing Agreement, including technology and cyber risk coverage if required by law.

On request of the Bank, made not more often than annually, the Servicer shall deliver to the Bank official certificates issued by the insurer to the Servicer or an applicable Affiliate and evidencing the coverage specified above.

(q) The Bank will have the right to conduct internal control and/or audit reviews of the Servicer's provision for physical security, information security, compliance regulations, adherence to this Servicing Agreement, and other issues after giving the Servicer appropriate notice.

ARTICLE SIX SERVICE PERFORMANCE MEASURES

SECTION 6.1 Service Performance Measures. Except as may be otherwise agreed upon by the Bank, the Servicer shall perform its duties as outlined in the attached Service Performance Measures (Attachment 2) using reasonable care, applying that degree of skill, attention, and technological innovation that the Servicer exercises generally with respect to comparable student loans that the Servicer services on behalf of its Affiliates, all in accordance with the Act, Regulations, and the applicable Program Rules (as applicable), and other applicable federal or state law.

The Bank will monitor compliance with the Act, Regulations, and applicable Program Rules. If issues related to Servicer non-compliance arise, as reported in Bank monitoring reports or reviews or in regulatory examination reports, the Bank will notify the Servicer of the specifics. The Servicer will have thirty (30) days, following receipt of notice to respond to the issues and establish action plans to correct the issues, as more particularly set forth in Attachment 2.

SECTION 6.2 Compliance with Service Performance Measures. The Servicer shall comply with the specific time frames, reliability rates, and procedures set forth in the Service Performance Measures (Attachment 2 hereto), as the same may be modified in accordance with this Section 6.2. Both parties acknowledge that industry norms and marketplace preferences dictate changes to performance service levels from time to time. The Servicer retains the right to modify Service Performance Measures relating to servicing so long as (a) the Bank is given thirty (30) days' prior notice and the right to reasonably negotiate pricing based on the changes in the servicing structure, (b) the modifications apply to all FFELP Loans or Private Loans of the same types as the Loans serviced by the Servicer hereunder, regardless of the identity of the owner/holder, and (c) the modifications do not: (i) cause a violation of applicable state or federal law or the Act, Regulations, or the applicable Program Rules (as applicable); (ii) materially and adversely affect the Servicer's or the Bank's ability to perform Loan servicing in a professional, cost-efficient, and customer-oriented manner, or (iii) cause such servicing to fall below industry standards as demonstrated in benchmarking studies conducted by or for Servicer.

ARTICLE SEVEN TERM, TERMINATION, AND TRANSFER

SECTION 7.1 Term. The term of this Servicing Agreement (the "Term") commences as of the Effective Date and shall remain in full force and effect for an initial period of four (4) years, unless sooner terminated in accordance with its terms and except as provided in this Article Seven for termination with respect to specific Loans, and shall renew automatically for additional one (1) year terms unless the Bank or the Servicer notifies the other party of its intention not to renew the Servicing Agreement at least ninety (90) days before the end of the then current term.

SECTION 7.2 <u>Termination as to Specific Loans</u>. This Servicing Agreement shall terminate as to a specific Loan upon (i) write-off of such Loan pursuant to the policies described in Section 7.5 below, (ii) the payment in full of all principal and interest by or on behalf of the

Borrower, the Secretary, or the Guarantee Agency, or (iii) the purchase or other acquisition of such Loan by the Servicer or its Affiliates in accordance with the terms hereof, or (iv) the sale of such Loan by the Bank to a third party, provided that the Servicer and such third party have entered into a loan servicing agreement with respect to such Loan in accordance with Section 11.3 below, or (v) with respect solely to the Private Loans, and at the Bank's discretion, upon deconversion of such Private Loans following no less than 60 days' prior written notice to the Servicer setting forth the Bank's intention to deconvert such Private Loans from the Servicing system. Notwithstanding the foregoing, the Servicer's responsibilities with respect to the Bank's documentation, as set forth herein, shall remain active. Whether or not this Servicing Agreement is terminated with respect to specific Loans pursuant to any of clauses (i) – (v) above, it shall remain in full force and effect with respect to all other Loans.

SECTION 7.3 Procedures for Deconversion of Loans. Within thirty (30) days after delivery of a proper notice of termination or non-renewal of this Servicing Agreement pursuant to this Article Seven, Section 8.2, Section 9.2 or Section 10.5, a schedule and procedure for the orderly deconversion of Loans will be agreed to by the Servicer and the Bank (and by FDR, to the extent applicable). The parties will cooperate with each other to resolve problems that arise during or as a result of the deconversion process. Except for deconversions of Loans pursuant to Section 9.2, below or due to sale of the Loan(s) to the Servicer, the Bank shall pay the deconversion fees specified in Attachment 1 (and such deconversion fees will be payable, without limitation, upon any deconversion of the Loans that occurs upon the expiration of the term of this Servicing Agreement). In addition, the Bank shall pay the removal fees specified in Attachment 1 with respect to other removals of Loans, as set forth in Attachment 1 (and described in Note 4 of Attachment 1).

SECTION 7.4 Agreement Extension. If, despite the reasonable efforts of the parties, deconversion (i.e., transfer or removal) cannot be completed by the effective date of termination, then, as to all Loans not removed, this Servicing Agreement will extend automatically through the date upon which removal procedures are completed. Provided the Servicer has acted in good faith, if the failure to effect the removal of all such Loans continues for four months or more beyond the effective date of termination and is not due to the fault or negligence of the Servicer, then in addition to any increases already provided for in Attachment 1, the fees payable by the Bank shall be increased by an amount equal to 10.0% of the fees that are then in effect, effective as of the end of such four-month period.

SECTION 7.5 Policies Regarding Loan Balance Adjustments Through Write-Offs, Write-Downs, and Write-Ups. Except with the prior consent of the Bank or as authorized herein, the Servicer shall not write off or write down any Loan balance. In the course of normal operations, and in accordance with the requirements of the Act, Regulations, and the Program Rules (as applicable), and the same customs, policies, and procedures followed by the Servicer with respect to similar loans serviced by it on behalf of its Affiliates, as the same may be modified from time to time, the Servicer may systematically and routinely adjust individual Loan accounts and Borrower accounts by writing off, writing down, or writing up balances deemed uncollectible or cost-ineffective to rectify.

SECTION 7.6 <u>Survival of Terms</u>. Notwithstanding the termination of this Servicing Agreement, the provisions of Article Three with respect to remission of payments belonging to the Bank will survive and continue in full force and effect. Any liabilities of either party to the other party arising from acts or omissions occurring prior to termination of this Servicing

Agreement shall survive the termination of this Servicing Agreement. Moreover, with regard to any rejection from a Guarantor with regard to its guarantee of a FFELP Loan, the Servicer shall have an additional 90 days after termination of this Servicing Agreement to secure reinstatement of the Guarantor's guarantee. In the event such guarantee cannot be reinstated within that time frame, the Servicer shall purchase the applicable Loan(s) from the Bank pursuant to Section 9 below.

ARTICLE EIGHT COMPENSATION

SECTION 8.1 Compensation. The Bank will compensate the Servicer for its services in accordance with the schedule of fees set forth in Attachment 1 to this Servicing Agreement. Such fees listed in Attachment 1 may be subject to increase to the extent that a demonstrable increase occurs in the cost incurred by the Servicer in providing the services to be provided hereunder due to changes in applicable governmental regulations or Guarantee Agency program requirements, Program Rules or increases in U.S. Postal Service rates. To the extent practicable, the Servicer shall provide the Bank with thirty (30) days' notice prior to implementing any such fee increase.

SECTION 8.2 Time of Payment; Interest Accrual; Set-Off; Bank Default. Fees shall be invoiced monthly in arrears and shall be paid within 30 days of receipt of invoice. If the bill is not paid when due, interest on the unpaid balance shall accrue at the rate of the prime interest rate plus 2% per annum. If the bill (including accrued interest) is not paid within thirty (30) days after the due date, and the bill is not disputed by the Bank (provided, however, that the Bank shall remain obligated to pay all of those charges in the bill that are not in dispute), the Servicer shall be entitled to a set-off against funds within Servicer's possession or control including, without limitation, funds representing payments collected on behalf of the Bank. If the bill (including accrued interest) is not paid by the Bank within forty-five (45) days of delivery to the Bank of notice of nonpayment from the Servicer given in accordance with the notice provisions hereof (in such case, a "Bank Default"), the Servicer may terminate its obligations hereunder, subject only to its duty to cooperate with the Bank pursuant to Article Seven. The Servicer shall not be liable to the Bank for any loss, cost, or expense suffered by the Bank and arising from the proper exercise of the right of set-off herein granted or the proper termination of Servicer's obligations because of a Bank Default. Servicer's remedies are intended to be cumulative and the exercise or failure to exercise any one such remedy on a particular occasion shall not affect Servicer's right to exercise any other remedy on that occasion or Servicer's right to exercise any one or more such remedies on any other occasion.

SECTION 8.3 Special Services. If special services are requested by the Bank, the Bank and the Servicer shall agree on pricing in advance. The Bank shall reimburse the Servicer for all special services agreed to be performed by the Servicer in accordance with the applicable terms, if any, of Attachment 1 hereto. In addition, and to avoid ambiguity, the Bank shall also reimburse the Servicer for any charges assessed by FDR to the Servicer for any services requested by the Bank.

ARTICLE NINE BREACH AND DEFAULT

SECTION 9.1 Material Breach.

(a) In the event of a "Material Guarantee Breach" (as hereinafter defined) that is not curable by reinstatement of the Guarantee Agency's guarantee, the Servicer shall purchase the affected Loan(s) as soon as practicable but not later than 90 days following the earlier of (i) the date of discovery of such Material Guarantee Breach, and (ii) the date of receipt of the Guarantee Agency reject transmittal form with respect to such Loan. If a Servicer Default has occurred and is continuing, the Servicer shall purchase the Loan(s) affected by the Material Guarantee Breach not later than thirty (30) days after receipt of a written demand from the Bank. The purchase price hereunder will be the sum of the unpaid principal amount plus accrued interest (calculated using the Amount That Guarantor Would Have Paid), the amount of any unpayable interest subsidy payments and Special Allowance payments through the date of payment of the purchase price, plus, for FFELP Loans, any amounts the Bank must pay the Secretary in respect of such Loan (including interest subsidy payments and Special Allowance payments which must be refunded to the Secretary). A "Material Guarantee Breach" is (1) a servicing breach that causes a full loss of a Guarantee Agency's guarantee of a Loan serviced pursuant hereto, or (2) a failure of a subservicer or Subcontractor to comply with a servicing requirement set forth in this Servicing Agreement and the inability of the Servicer to comply on its own with such requirement, or to force compliance by such subservicer or Subcontractor. Any servicing breach that relates to compliance with the requirements of the Act, Program Rules, or the applicable Guarantee Agency but that does not cause a full loss of such Guarantee Agency's obligation to guarantee payments of a Loan is not a Material Guarantee Breach for purposes of this Section 9.1.

(b) In addition, if any servicing breach by the Servicer is not a Material Guarantee Breach as defined in Section 9.1(a), and does not trigger such purchase obligation but does result in the refusal by a Guarantee Agency to guarantee or reimburse all or a portion of the accrued interest (or any obligation of the Bank to repay such interest to a Guarantee Agency), or the loss (including any obligation to repay to the Secretary) of interest subsidy and Special Allowance payments with respect to any FFELP Loan affected by such breach, then the Servicer shall reimburse the Bank in an amount equal to the sum of all such non-guaranteed interest amounts and such forfeited interest subsidy and Special Allowance payments by remitting any amounts that would have been owed to the Bank by the Guarantee Agency but for such breach not later than (i) 60 days from the date of the Guarantee Agency's refusal to guarantee all or a portion of accrued interest or loss of interest subsidy payments or Special Allowance payments, or (ii) in any case in which the Servicer reasonably believes such amounts are likely to be collected, not later than 360 days from the date of the Guarantee Agency's refusal to guarantee all or a portion of accrued interest subsidy payments or Special Allowance payments.

SECTION 9.2 Servicer Default. If any one of the following events (a "Servicer Default") shall occur and be continuing:

- (a) with respect to FFELP Loans, any determination by the Secretary of the loss of the Servicer's eligibility as a third-party servicer;
- (b) a Servicer Insolvency Event;
- (c) chronic, sustained failures to meet the Service Performance Measures set forth on Attachment 2, as described below; or

(d) with respect to Private Loans, the Bank's regulators have determined that there has been a default and/or the Bank's regulators effectively require that the Agreement be terminated;

then, and in each and every case, so long as the Servicer Default shall not have been remedied in accordance with Section 12.4 (or in the case of subsection (d) above, not remedied to the satisfaction of the Bank's regulators), the Bank may terminate this Servicing Agreement upon thirty (30) days' prior written notice to the Servicer (or in the case of a Servicer Insolvency Event immediately upon written notice) and may initiate the transfer process pursuant to the applicable termination and transfer provisions of this Servicing Agreement. For the purposes of subsection (c) above, the failure by the Servicer to meet the Service Performance Measures shall not rise to the level of a chronic, sustained failure unless: (i) at least two such particular Service Performance Measures are missed by at least 10.0% during at least three consecutive months, or (ii) any one particular Service Performance Measure is missed by at least 10.0% during any consecutive twelve-month period during the Term.

ARTICLE TEN LIABILITY AND INDEMNIFICATION

SECTION 10.1 (a) <u>Conflicts With Other Agreements</u>. Notwithstanding anything to the contrary in this Article Ten or anywhere else in this Agreement, each party hereto acknowledges and agrees that this Agreement shall not negate, override, or otherwise alter indemnities agreed to by and between the parties and/or their Affiliates in the Separation and Distribution Agreement and the Transition Services Agreement (collectively, the "Other Agreements").

(b) Servicer's Liability. The Servicer shall be liable to the Bank for any claim, loss, cost, damage, liability or expense, including any claim, loss, cost (including, but not limited to remediation costs in case of data breach), damage, liability or expense that may be imposed on, incurred by or asserted against the Bank, resulting from Servicer's negligence, willful misfeasance, bad faith, or breach of any obligation, representation, warranty, or covenant under this Servicing Agreement. Other than in the event of the Servicer's willful misfeasance or bad faith, the Servicer's liability under this Servicing Agreement shall be limited to the lesser of (a) the total amount of the compensation theretofore paid to the Servicer under Article Eight of this Servicing Agreement or \$2.5 million, whichever is greater or (b) \$10 million. In no event shall the Servicer be liable to the Bank for consequential, exemplary or punitive damages. Anything to the contrary contained herein notwithstanding, no additional liability shall be imposed on the Servicer pursuant to this Section 10.1(b) with respect to losses that are attributable to breaches that result in the Servicer's purchase of a Loan or reimbursement to the Bank pursuant to Section 9.1 of this Servicing Agreement except for the payment of third-party claims related thereto.

The Servicer shall not be liable to the Bank pursuant to this Section 10.1(b) or otherwise for any loss or damage incurred by the Bank arising from any incorrect or incomplete information provided by the Bank, any Guarantee Agency, or the Secretary, or for the failure of the Bank to comply with its obligations. The Servicer may rely in good faith on any information or document of any kind reasonably believed by the Servicer to be properly executed and

submitted by an authorized person respecting any matters arising under this Servicing Agreement. In addition, by way of further limitation, (i) where the exercise of discretion by the Servicer in order to carry out its duties is required or permitted hereunder, the Servicer shall not be liable for any error of judgment made in the good faith exercise of such discretion, and (ii) the Servicer shall not be liable for any breach of a Service Performance Measure that is not a material breach. No breach of a Service Performance Measure shall be deemed material unless it is the sole proximate cause of a clearly demonstrable loss to the Bank in excess of \$10,000,00 or is part of a persistent pattern of non-compliance with the Service Performance Measures.

Except as otherwise provided in this Servicing Agreement, the Servicer shall not be under any obligation to appear in, prosecute or defend any legal action where it is not named as a party; provided, however, that the Servicer may undertake any reasonable action that it may deem necessary or desirable in respect of this Servicing Agreement and the rights and duties of the parties hereunder.

In addition, anything to the contrary contained herein notwithstanding, the Servicer shall not have any liability to the Bank with respect to any losses, claims, costs or expenses arising after the Effective Date from interpretations of any law or Regulation by any applicable governmental or regulatory authority after the Effective Date that are different from interpretations previously taken by such governmental or regulatory authority, provided, however, that the Servicer shall promptly take the necessary corrective action(s) to change the applicable practice, system or procedure on a going-forward basis. Notwithstanding the forgoing, all indemnities relating to governmental or regulatory actions contained in the Other Agreements shall continue in full force and effect to the extent set forth in such Other Agreements.

SECTION 10.2 Indemnification by Servicer. The Servicer, after notice and opportunity to cure as provided in Section 12.4, below, will indemnify and hold the Bank harmless from any claim, loss, cost, or expense, including reasonable attorneys' fees, and that results from (i) any material failure of the Servicer to comply with the terms of this Servicing Agreement, or (ii) the failure of any warranty given or representation made by the Servicer in this Servicing Agreement. Anything to the contrary contained herein notwithstanding, no additional liability shall be imposed on the Servicer pursuant to this Section 10.2 with respect to losses that are attributable to breaches that result in the Servicer's purchase of a Loan or reimbursement to the Bank pursuant to Section 9.1 of this Servicing Agreement except for the payment of third-party claims related thereto.

SECTION 10.3 Indemnification by Bank. The Bank will indemnify and hold Servicer harmless from any loss, cost, or expense suffered or incurred by Servicer, including reasonable attorneys' fees, that results from any action of the Servicer authorized by this Servicing Agreement or otherwise authorized by the Bank, or that results from any material failure of the Bank to comply with the terms of this Servicing Agreement, or the failure of any warranty given or representation made by the Bank in this Servicing Agreement. In particular, but without limitation, to the extent that the Servicer is required pursuant to Section 10.1 to appear in or is made a defendant in any legal action or other proceeding relating to the servicing of the Loans, the Bank shall indemnify and hold the Servicer harmless from all cost, liability or expense of the Servicer (including, without limitation, reasonable attorney fees) not arising out of or relating to the failure of the Servicer to comply with the terms of this Servicing Agreement.

SECTION 10.4 Certain Rights of the Indemnifying Party. A party with the responsibility to indemnify hereunder (an "Indemnifying Party") shall control the defense or settlement of any third party claim. The other party (the "Indemnified Party") may participate in, but not control, any defense or settlement of any third party claim to be indemnified by the Indemnifying Party, in which event the Indemnified Party will bear its own costs and expenses with respect to such participation, except as set forth in the following sentence. However, an indemnifying party may not admit liability or agree to a settlement that would impose an obligation upon an indemnified party without the latter's prior written consent. Notwithstanding the foregoing, the Indemnified Party may take over the control of the defense or settlement of a third party claim if (x) the Indemnifying Party fails to appear in or to defend such claim, in which event the Indemnifying Party shall indemnify the Indemnified Party for the Indemnified Party's costs and expenses in connection with such defense or settlement or (y) the Indemnified Party irrevocably waives its right to indemnity with respect to such third party claim.

SECTION 10.5 Force Majeure. Notwithstanding the foregoing, if either party is rendered unable, in whole or in part, by a force not reasonably within the control of that party (including acts of God, acts of war, riots, insurrections, illegality of performance, strikes or other industrial disturbances, fires, earthquakes, hurricanes, floods and other disasters) to satisfy its obligations under this Servicing Agreement, such party shall not be deemed to have breached any such obligation upon delivery of written notice of such event to the other party hereto, for so long as such party remains unable to perform such obligation as a result of such event. The following events shall not constitute force majeure events: (a) acts or omissions of Subcontractors, or (b) delays or failures to the extent they could have been avoided or their impact mitigated through the use of commercially reasonable business continuity measures. If either party is unable to materially perform its obligations under this Servicing Agreement and its performance is excused pursuant to this paragraph for a period of at least thirty (30) days, the other party may terminate this Servicing Agreement upon at least three (3) business days' prior written notice.

ARTICLE ELEVEN ASSIGNMENTS AND SUBCONTRACTS

SECTION 11.1 General Rule. To the fullest extent permitted by law, and except as provided in this Article Eleven, no party may assign its rights or delegate its duties under this Servicing Agreement without the prior written consent of the other parties, and any such assignment shall be void and without effect.

SECTION 11.2 Mergers and Reorganizations. Each party agrees that upon (a) any merger or consolidation of such party into another person or entity, (b) any merger or consolidation to which such party is a party that results in the creation of another person or entity, or (c) any person or entity succeeding to the properties and assets of such party substantially as a whole, such party shall cause such person or entity (if other than such party) to execute an agreement of assumption to perform every obligation of such party hereunder.

SECTION 11.3 Sale of Loans. In the event that during the Term of this Servicing Agreement, as the same may be extended, the Bank proposes to sell any Loans serviced hereunder, unless the proposed purchaser is the Servicer or one of the Servicer's Affiliates, the Bank shall provide to the Servicer at least thirty (30) days' prior written notice of the proposed sale, including the name of the purchaser, the name of the responsible contact at the purchaser

and the date of the proposed sale. Any such sale shall be subject to (i) the Servicer's approval of the financial condition of the purchaser, which approval shall not be unreasonably withheld, and (ii) the purchaser's entering into a new loan servicing agreement with the Servicer for the servicing of the Loans being sold, which the Servicer and the purchaser shall negotiate in good faith, and that is either (a) substantially in the form of the private loan servicing agreement attached hereto as Attachment 9, if the sale is to occur on or before the first anniversary of the Effective Date, or (b) on market-based terms and conditions, if the sale is to occur after the first anniversary of the Effective Date. Any such purchaser or subsequent purchaser must pay to the Servicer a reasonable transfer fee in connection with such sale. Any such purchaser or subsequent purchaser of FFELP Loans must be an "eligible lender" as defined under the Act. The Bank shall indemnify the Servicer and hold the Servicer harmless from any loss, cost, or expense arising from the sale of Loans serviced hereunder and from any claim by the purchaser or purchaser's successors and assigns in conflict with the Servicer's rights hereunder.

SECTION 11.4 Subcontracts; Assignment to Servicer's Legal Successor. The Servicer has the right to employ Subcontractors to perform activities associated with or in support of the services provided hereunder. Notwithstanding any provision of this Servicing Agreement that requires Servicer to perform any particular service for the Bank, or to comply with any particular servicer performance measures, or to perform any other obligation hereunder, or to accept liability for any aspect of servicing such Loans, if servicing of the applicable Loans is performed by any subservicer or Subcontractor pursuant to a subservicing agreement or subcontracting agreement executed prior to the Effective Date and the subservicing agreement or subcontracting agreement does not require the subservicer or Subcontractor to comply with the particular servicer performance measures set forth in this Servicing Agreement, or to perform any one or more of the other obligations hereunder with respect to such Loans, then if the Servicer is unable to force compliance by such subservicer or Subcontractor with the requirements set forth in this Servicing Agreement, the Bank's remedy shall be limited to treating the failure as a Material Guarantee Breach under Section 9.1 above and requiring the Servicer to purchase the affected Loan(s) pursuant to the provisions of Section 9.1. With regard to all Subcontractors with which Servicer contracts after the Effective Date, Servicer will ensure that any new Subcontractor complies with the terms of this Agreement. Except to the extent set forth earlier in this Section 11.4, the Servicer shall remain fully obligated hereunder for the performance by any Subcontractor of services hereunder. In connection with any audit of the Servicer pursuant to Section 5.1(l), the Servicer shall make available to the Bank, onsite at the Servicer's facilities, all policies and procedures and available audit reports related to the Servicer's audit of any such subservicer or Subcontractor or the Servicer's vendor management of any such subservicer or Subcontractor applicable to the Loans being serviced under this Servicing Agreement, including, without limitation, any Security Assessment Framework evaluations (SAFe assessments) of such subservicer or Subcontractor. The Servicer may assign its rights and delegate its duties hereunder to any legal successor of the Servicer. Upon the written request of the Bank, the Servicer shall provide information regarding the name and qualifications of any significant Subcontractor. With respect to all such subcontracts entered into after the Effective Date of this Servicing Agreement, the Servicer will obtain the prior approval of the Bank of any offshore entity that is to serve as a Subcontractor if such entity is to have access to the non-public personal information of the Bank's customers, which approval will not be unreasonably withheld or delayed; provided, however, that if the Bank approves such an offshore Subcontractor for the purposes of this Section 11.4 for a specific service, such Subcontractor shall be deemed approved to perform other services that are reasonably defined as being within the same category of

services. For example, if a Subcontractor is approved to perform in-bound call center activity where the Subcontractor will have access to the non-public personal information of the Bank's customers, such Subcontractor shall be deemed approved to perform outbound call center activity if the scope of the Servicer's subcontract with such Subcontractor is subsequently expanded. Similarly, if a Subcontractor is approved to perform specific account processing activities where the Subcontractor will have access to the non-public personal information of the Bank's customers, such Subcontractor shall be deemed approved to perform other account processing activities if the scope of the Servicer's subcontract with such Subcontractor is subsequently expanded.

ARTICLE TWELVE NOTICES AND REPRESENTATIVES; DISPUTE RESOLUTION

SECTION 12.1 Notices. All notices, requests and other communications hereunder shall be in writing and will be deemed to have been duly given only if delivered personally, sent by nationally recognized overnight courier, or mailed (first class postage prepaid or electronically) to the parties at the following addresses:

If to the Bank:

Sallie Mae Bank 175 S. West Temple, Suite 600 Salt Lake City, UT 84101 Attention: President

with a copy to:

Sallie Mae Bank 300 Continental Drive Newark, DE 19713 Attention: General Counsel

If to the Servicer:

Navient Solutions, Inc. 11100 USA Parkway Fishers, IN 46037

Attention: Senior Vice President, Financial Institution Sales

with a copy to:

Navient Solutions, Inc. 2001 Edmund Halley Drive Reston, VA 20191 Attention: Legal Department

All such notices, requests and other communications shall (i) if delivered personally to the address as provided in this Section, be deemed given upon delivery, (ii) if delivered by nationally

recognized overnight courier to the address as provided in this Section, be deemed given upon receipt, and (iii) if delivered by mail in the manner described above to the address as provided in this Section, be deemed given upon receipt (in each case regardless of whether such notice, request or other communication is received by any other person to whom a copy of such notice, request or other communication is to be delivered pursuant to this Section). Any party from time to time may change its address or other information for the purpose of notices to that party by giving notice specifying such change to the other party hereto.

SECTION 12.2 Representatives. Each of the Bank and the Servicer will designate one or more persons to consult with the other party to resolve day-to-day operating questions that arise in the performance of this Servicing Agreement and will notify the other party of the identity of the representatives so designated. The designation of such representatives may specify reasonable limitations upon their authority, but each of the representatives will have the authority, on behalf of the party represented, to resolve questions arising under the Service Performance Measures.

SECTION 12.3 Administrative Dispute Resolution. The parties will endeavor in good faith to achieve amicable resolution of all disputes and disagreements arising hereunder, if any, with a view to the avoidance of litigation. Each party will continue to perform its respective obligations under the Agreement during this process. In the event that a question arises under this Servicing Agreement the resolution of which is beyond the authority of one or both of the parties' representatives or upon which the representatives cannot agree despite good faith consultation, the parties shall make diligent and prompt effort to resolve the question through the intervention of senior officers with plenary authority to waive an interpretive stance taken by the party represented and/or, on behalf of such party, to negotiate and agree to any amendment to this Servicing Agreement. Each party may rely on the representation of any officer of the other party of the rank of Vice President or higher (other than a representative appointed pursuant to Section 12.2) who asserts in writing that he or she is acting on behalf of that party under the provisions of this Section 12.3. In aid of this Section, whenever reasonably requested by the other party, each party will provide a current list identifying its officers by name and title. Notwithstanding, where irreparable harm to a party's interest is imminent, either party may circumvent this time line and seek injunctive relief in a court of law.

SECTION 12.4 Opportunity to Cure. In the event of a failure by the Servicer to comply with any provision of this Servicing Agreement, no action may be commenced by the Bank to enforce such provision unless the Bank has first given written notice of such non-compliance to the Servicer in accordance with the notice provisions of this Servicing Agreement and has afforded the Servicer an opportunity to cure as specified in this Section. If the failure to comply is a failure to make a required deposit, purchase a Loan, reimburse lost interest or Special Allowance payments, or otherwise expressly to pay money to or for the Bank (a "Monetary Breach"), the Servicer shall be accorded a period of ten days after such notice to make the deposit, purchase the Loan, reimburse the interest or Special Allowance payment, or otherwise pay the money that is expressly required to be paid. If the failure to comply is not a Monetary Breach, the Servicer shall be entitled to a reasonable period of time after such notice in order to remedy the failure. As used in the preceding sentence, the term "reasonable period of time" means a period of time that is reasonable under the circumstances, taking into account the nature of the failure, the ease or difficulty of an appropriate remedy, and the harm likely to be suffered with the passage of time, but in any event not less than thirty (30) days and not more than ninety (90) days after such notice. If the failure asserted by the Bank is disputed by the Servicer, the parties will use such reasonable period of time to attempt to resolve the dispute through the administrative dispute resolution procedure described in Section 12.3, above.

ARTICLE THIRTEEN MISCELLANEOUS

SECTION 13.1 Interpretation in Accordance with the Act. The servicing of FFELP Loans under this Servicing Agreement will be subject to all applicable provisions of the Act and the Regulations, as the same may be amended from time to time. In the event of a conflict between any provision of the Act or a Regulation and this Servicing Agreement, the Act or Regulation will govern with respect to the servicing of such FFELP Loans.

SECTION 13.2 Captions, Attachments and Schedules. The captions used in this Servicing Agreement are for convenient reference only and do not define, limit, describe or modify the meaning of the text of the Servicing Agreement. The attachments and schedules hereto are an integral part hereof, and all references herein to this Servicing Agreement shall be deemed to refer as well to such attachments and schedules.

SECTION 13.3 Confidentiality; Servicer's Press Releases and other Public Announcements. The Bank's Loan Documentation, and the contents of this Servicing Agreement, together with all supporting documentation, exhibits, attachments, schedules, and any amendments thereto which form the basis of the business relationship between the Bank and the Servicer, together with all negotiating, explanatory and interpretive materials pertaining thereto including, without limitation, all related correspondence between the parties (together, the "Agreement and Materials"), will be held in the strictest confidence by the Bank and the Servicer. Each party will make appropriate efforts to ensure that the Agreement and Materials is not disclosed or otherwise discussed with any third parties except as hereunder provided, or as required for financing or acquisition purposes, audit purposes, legal advice, as otherwise required by law, regulations, or as agreed to by the parties. Prior to distributing any press release pertaining to this Servicing Agreement or the relationship between the parties, the party wishing to share the information will disclose the proposed contents thereof to the other and will consult with the other party as to the accuracy and advisability of such press release or other public announcement.

SECTION 13.4 <u>Liability of Individuals</u>. No director, officer, or employee of any party (or of the general partners in such party) shall be subject to any liability for errors in judgment or for any action taken, or for refraining from taking any action, provided such error, action or inaction was made in good faith.

SECTION 13.5 <u>Relationship of Parties</u>. The Servicer is an independent contractor and, except for the services to be performed as set forth herein, is not and will not hold itself out as an agent of the Bank. Nothing herein contained shall create or imply a general agency relationship between the Servicer and the Bank, nor shall this Servicing Agreement be deemed to constitute a joint venture or partnership between the parties.

SECTION 13.6 Counterparts. This Servicing Agreement may be executed in counterparts each of which shall be an original and all of which shall constitute but one and the same instrument.

SECTION 13.7 Severability. If a court of competent jurisdiction holds any provision of this Servicing Agreement inoperative or unenforceable either generally or as applicable in any particular situation, it is the intention of the parties that such invalid or unenforceable provisions be reduced in scope or eliminated by the court, but only to the extent deemed necessary by the court to render the provisions of this Servicing Agreement reasonable and enforceable.

SECTION 13.8 Remedies Not Exclusive. Except for the specific breaches and their respective remedies described in Section 9.1, above, no remedy by the terms of this Servicing Agreement conferred upon or reserved to either party hereto is intended to be exclusive of any other remedy existing at law or in equity.

SECTION 13.9 Non-Discrimination. In servicing Loans pursuant to this Servicing Agreement, the Servicer will not discriminate on the basis of race, color, religion, national origin, sex or marital status, or age (provided the applicant has the capacity to contract), because all or part of the applicant's income derives from any public assistance program, or because the applicant has in good faith exercised any right under the Consumer Credit Protection Act.

SECTION 13.10 Lending Policy. The Lending Policies attached hereto as Attachment 5 are hereby incorporated herein as the applicable Lending Policies as of the date hereof, and the Lending Policies, as amended from time to time, is also hereby incorporated as part of this Servicing Agreement. The Lending Policies must be able to be reasonably administered by the Servicer, be consistent with the provisions of this Servicing Agreement, the Act, Regulations, and the Program Rules (as applicable) and cannot be amended without sufficient prior notice to the Servicer to allow implementation of any servicing modifications. The Lending Policies will be reviewed and updated by the Bank if deemed necessary on at least an annual basis or as required by the Act or the Regulations or as any Program Rule changes may occur. To the extent that a change in the Bank's Lending Policies require the Servicer to implement material operational changes or systems modifications, the cost of such changes and modifications may be borne by the Bank.

SECTION 13.11 Governing Law. This Servicing Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without regard for principles of conflicts of law that would require the application of the law of a jurisdiction other than the State of Delaware.

SECTION 13.12 Entire Agreement; No Amendment Except in Writing. This Servicing Agreement and the agreements referred to herein shall constitute the entire agreement between the parties with respect to the subject matter hereof and supersede all prior agreements between the parties with respect to such subject matter. This Servicing Agreement may not be amended except by a subsequent written agreement signed by the parties.

SECTION 13.13 Waiver. The waiver or failure of a party to exercise in any respect any right provided for herein shall not be deemed a waiver of any further right herein.

ARTICLE FOURTEEN SUPPLEMENTARY PRODUCTS AND SERVICES

SECTION 14.1 Supplementary Products and Services. The Servicer and the Bank may agree from time to time, in writing, for the Servicer to support supplemental lending and servicing products and services for the benefit of the Bank and the Bank's borrowers. Except as the parties may subsequently agree in writing, however, no other supplemental product or service will be supported by the Servicer.

[SIGNATURE PAGE FOLLOWS.]

IN WITNESS WHEREOF, the parties hereto have caused this Servicing Agreement to be duly executed and delivered as of the Effective Date.

SALLIE MAE BANK

NAVIENT SOLUTIONS, INC.

By: /s/ Raymond Quinlan By: /s/ John F. Remondi

Name: Raymond Quinlan Name: John F. Remondi

Title: Chief Executive Officer Title: Chief Executive Officer

- 33 -

LIST OF ATTACHMENTS

ATTACHMENT 1: Monthly Fees and Activity Based Charges Schedule

ATTACHMENT 2: Service Performance Measures

ATTACHMENT 3: Servicer Reporting

ATTACHMENT 4: Data Access for Owners and Borrowers

ATTACHMENT 5: Lending Policy

ATTACHMENT 6: Disaster Recovery and Business Continuity Planning

ATTACHMENT 7: Sallie Mae Bank's Confidential Information Security Requirements

ATTACHMENT 8: Sallie Mae Bank Borrower Benefits

ATTACHMENT 9: Form of Servicing Agreement for Third-Party Purchaser

ATTACHMENT 3: Servicer Reporting

Standard Monthly Reporting Applicable to the Serviced Loans

Servicer will make available to the Bank on a monthly basis a standard set of reports. Reports will be available in the format described in the table below. Reconciliation reporting will be available by the 15th business day of the month. Most other reports will be available by the 6th business day of the month.

Report Number	Report Name	Media	Report Description
800	Monthly Activity Summary Report	On-line access	Summary of program balance and accounting transactions at month-end.
802	Monthly Activity Summary Detail Report	On-line access	Summary of outstanding principal by status, weighted interest rate and terms of maturity.
N/A	Servicing Invoice	• Hardcopy	Monthly invoice detailing originations, premium billing, post sale cancellations and servicing fees for the prior month.
N/A	UFLM017201	• Electronic	Unused Commitments
N/A	EIVDPSLR01	• Electronic	Sallie Mae Post Sale Cancellation Report
830	LSF030D\$01	• Electronic	Daily Cash Transactions Posted
831	LSF030D\$02	• Electronic	Daily Advice Transactions Posted
N/A	Daily FFELP Pledge Monitoring	• Electronic	Daily report of outstanding principal balance of all FFELP Loans owned by the Bank
N/A	Escalated inquiries, complaints	• Electronic	Monthly reporting on escalated inquiries and complaints (by category type)
N/A	OFAC reports	• Electronic	Monthly reporting on OFAC
OPSF0002	Daily Activity Summary Report	On-line access	Daily summary of program balance and accounting transactions for FDR serviced loans
OPSF0003	Monthly Activity Summary Report	On-line access	Summary of program balance and accounting transactions at month-end for FDR serviced loans
OPSF0004	Activity Detail Report	On-line access	Daily detail of program balance and accounting transactions for FDR serviced loans
OPSF0005	Monthly Activity Detail Report	On-line access	Monthly detail of program balance and accounting transactions for FDR serviced loans
OPSF0006	Monthly Delinquency Summary Report	On-line access	Summary of outstanding principal and borrower interest by delinquency or claim status for FDR serviced loans
OPSF0007	Monthly Characteristics Summary Report	On-line access	Summary of outstanding principal by status and terms of maturity for FDR serviced loans
OPSF0008	Monthly Borrower Detail Report	On-line access	Detail of borrower/loan activity at account level, including loan balance for FDR serviced loans
OPSF0009	Monthly Delinquency Detail Report	On-line access	Detail of outstanding principal and borrower interest by delinquency or claim status for FDR serviced loans
LFHOW01	Owner Daily Cash Transactions Posted	On-line access	Daily cash totals and details for payment, refund, and reversal transactions for FDR serviced loans

Quarterly Reports

The Servicer will prepare and make available to the Bank (by on-line access), within twenty (20) calendar days after the end of each calendar quarter, a summary report showing the interest and Special Allowance information that was reported to the Secretary on the Lender's behalf.

Ad Hoc Reports

Ad hoc reporting may be provided upon agreement with Servicer. The Bank will be charged for such reports in accordance with the schedule of fees in Attachment 1.

ATTACHMENT 4: Data Access for Owners and Borrowers

Owners can access borrower and loan information on Servicer's Consolidated Loan Administration Servicing System (CLASS) and FDR System through the following means:

- (1) Internet access using an internet browser
- (2) Data updates or extract files delivered in a machine-readable format

In addition, Borrowers may have limited internet access to certain account information.

1. Internet Account Access for Owners

Owners have the option of using a standard Internet browser to access individual borrower account detail by logging on to the Servicer's web site—www.navient.com. Owners desiring this type of access must first establish a user ID and password. Unlike CLASS Administrative Terminal access, a single user ID/password combination can be configured for access to any combination of Owner OE numbers the Owner chooses. The following describes the types of data available through our Internet Account Access for Owners:

- · Loan Balance and Status Summary
- · Payment Summary
- Interest Summary
- · Application and Disbursement Summary
- · Payment History

2. Data Updates/Extract Files

The Servicer can provide monthly extracts of CLASS servicing data for transfer to the Owner in machine-readable format. The following fields represent a small portion of the detail available on each loan.

- · Social Security Number
- Full Name
- · Date of Birth
- · Address and Telephone Number
- School
- Guarantor
- · Total Disbursed Principal
- Payment Amount and Payment Due Date
- Current Principal Outstanding
- Current Accrued Interest
- · Deferment History
- Cosigner/Co-maker/Reference
- Claim Detail

3. Direct Access to CLASS, FDR and Associated Data Warehouses:

The Servicer will provide direct access to CLASS to access individual borrower account detail, allowing the Bank to monitor its portfolio of loans.

4. Internet Account Access for Borrowers

Borrowers can also establish a user ID and password to access their individual account information.

ATTACHMENT 5: Lending Policy

SECTION 1 - FFELP

As agreed to by the Bank and the Servicer, in accordance with the guidelines set forth in the applicable FFELP and Consolidation Loan Program documents developed by the Bank.

SECTION 2 – PRIVATE LOANS

As agreed to by the Bank and the Servicer, in accordance with the guidelines set forth in the applicable Private Loan Program document developed by the Bank.

ATTACHMENT 6: Disaster Recovery and Business Continuity Planning

A. Disaster Recovery and Business Continuity. The Servicer will be responsible for disaster recovery and business continuity planning, testing, implementation, and execution related to all locations from which services are provided under this Servicing Agreement (including locations of the Servicer's Subcontractors, "Service Locations"). The Servicer shall provide the services described in subparagraphs (i) through (iv) below without extra charge to the Bank:

- (i) maintaining and testing the Disaster Recovery/Business Continuity Plan (as defined below);
- implementing the Disaster Recovery/Business Continuity Plan in the event of a disaster affecting Service Locations within the specified timeframes;
- (iii) carrying out disaster recovery, business continuity or redundancy procedures related to the services in accordance with such Disaster Recovery/Business Continuity Plan; and
- (iv) designing and implementing a Disaster Recovery/Business Continuity plan to minimize disruption to the business of Bank.
- B. Disaster Recovery/Business Continuity Plan. The Servicer shall maintain a Disaster Recovery/Business Continuity plan (the "Disaster Recovery/Business Continuity Plan") that covers services that are provided to Bank and each of the proposed Service Locations. The Servicer will provide a target RTO and RPO of 48 hours and 2 hours, respectively, unless a lower minimum is provided in the Disaster Recovery/Business Continuity Plan in respect of any particular application.
- C. Testing. The Servicer will test the Disaster Recovery/Business Continuity Plan on at least an annual basis or more often as required by any regulatory authority having authority over Servicer.
- D. *Priority*. If a disaster or major site impact occurs involving the need to recover operations at a Service Location or to operate from a fallback site and, as a result, it is necessary for the Servicer or its Subcontractors to allocate limited resources between or among several organizations, the Servicer shall not treat the Bank less favorably than any other customer for purposes of allocation of such resources.

ATTACHMENT 7: Sallie Mae Bank's Confidential Information Security Requirements

Access to Bank Confidential Information (including, without limitation, any non-public personal information or personally identifiable information of customers or consumers for which the Bank is the custodian) shall, subject to the other applicable provisions of the Servicing Agreement, be restricted to the Servicer with whom the Bank is contracting under the Servicing Agreement and the employees and, to the extent permitted by the Servicing Agreement, the permitted Subcontractors of the Servicer (and their respective personnel) (each, including the Servicer, an "Accessing Party") who (1) shall provide the services under the Servicing Agreement and (2) have been properly trained and instructed as to all obligations set forth in the Servicing Agreement and this Attachment 7 with respect to the access and use of the Bank's Confidential Information. Any Accessing Party hardware or software used to accesses, process, store or transmit any Bank Confidential Information shall be referred to herein as an "Affected Computing Device." In order to prevent unauthorized use, access, modification, or disruption to Bank Confidential Information, information security measures shall be implemented with respect to any Affected Computing Device.

In addition to the security of Affected Computing Devices, this Attachment must take a broader view of security. The security of these devices is dependent upon the Accessing Party maintaining a secure infrastructure that implements an array of industry standard controls. Therefore, this Attachment covers the infrastructure necessary to protect the Affected Computing Device. This does not cover the Accessing Party's entire infrastructure and controls regime, but only the slice which assures the security of the Affected Computing Devices.

Access Control

- 1. Access to Affected Computing Devices shall be configured with the concept of "Least Privilege" enforced. Access to Affected Computing Devices shall be granted only to those individuals who shall have it in order to provide the Services.
- Access to Affected Computing Devices shall be granted in line with formally documented user administration processes that are designed to ensure:
 - All users granted access to Affected Computing Devices shall be explicitly approved by appropriate Accessing Party management;
 and
 - b. Provisioning and de-provisioning of access to critical Affected Computing Devices is reviewed at regularly scheduled intervals by appropriate Accessing Party management.
- 3. All Affected Computing Devices shall be configured to require users to periodically change their passwords. Password length, complexity and reuse shall adhere to industry standard practices.
- 4. Accessing Party shall limit access by means of wireless networking to Bank Confidential Information to the minimum required situations. Where such wireless access is necessary, such Accessing Party shall require that authentication, encryption, and security for the wireless network meets generally accepted industry standard practices.
- 5. Accessing Party authentication credentials that are no longer required or that are no longer authorized to obtain access to Bank Confidential Information from Affected Computing Devices shall be (promptly disabled.

- 6. In the event that Accessing Party is provided with Bank administered credentials or devices to permit access to Bank Confidential Information, Accessing Party shall promptly notify the Bank, within 24 hours in a manner that is mutually agreed upon by the parties, of a personnel change whereby an individual no longer requires Bank administered authentication credentials to Bank Confidential Information.
- 7. Network protection mechanisms meeting industry standard practices shall be implemented for all Accessing Party company networks accessing Bank Confidential Information.
- 8. Accessing Party's personnel shall exercise restraint with their access privileges to Bank Confidential Information, shall not attempt to circumvent or subvert any security measures and not use their access for anything other than the provision of Services.
- 9. Access to Bank Confidential Information over the public Internet shall be restricted.

Logging & Monitoring

- 10. All Affected Computing Devices shall be configured to audit relevant security events using industry standard practices.
- 11. Audit and event logs for Affected Computing Devices shall be adequately protected to prevent contents from being modified or deleted in an unauthorized fashion.
- 12. System logs are to be analyzed for suspicious activity.

Patching & Configuration Management

- 13. Accessing Party shall maintain a security patching/update validation process that is designed to ensure that all Affected Computing Devices are patched in a timely manner consistent with industry standard practices.
- 14. Accessing Party shall follow industry standard practices to implement secure system configurations.
- 15. Accessing Party change control management procedures shall be documented and followed.

Storage & Media Protection

- 16. Servers, enterprise data storage devices, backup tapes and media, and other Affected Computing Devices used to support network communications shall be located in a secure and restricted access location within Accessing Party facilities, or approved Subcontractor facilities,
- 17. All backup media, including tapes, sent off site that contains Bank Confidential Information shall be encrypted using industry standard practices.
- 18. When Bank Confidential Information is required to be destroyed or disposed per the terms of the Servicing Agreement, such Bank Confidential Information, whether in paper, electronic or other form requires secure disposal or destruction. These measures shall be in line with industry standard practices.

Physical & Environmental Protection

- 19. Accessing Party Facilities at which Affected Computing Devices are stored or maintained shall have appropriate controls implemented that restrict physical access to only authorized personnel. Visitor access to such facilities shall be based on authentication of visitor identity and visitors shall be escorted.
- Data centers housing Bank Confidential Information are required to have controlled access with working security cameras. Data center access control lists shall be kept current.
- 21. Accessing Party shall provide at least 90 days' advance notification to Bank prior to relocation or transfer of Accessing Party's datacenter housing Bank Confidential Information.

Vulnerability Assessments

- 22. Accessing Party shall maintain a vulnerability management and assessment program based on industry standard practices that frequently assesses Affected Computing Devices and mitigates or eliminates vulnerabilities.
- 23. Routine network and database scans shall be scheduled. The scan results shall be analyzed and vulnerabilities identified and remediated within a timeframe commensurate with the relative risk.
- 24. Accessing Party shall engage, at its expense, an unrelated security firm to perform an annual penetration test of Affected Computing devices. Accessing Party shall implement security controls and practices to mitigate risks identified during such an assessment.

Encryption

25. When Bank Confidential Information is being accessed, or transmitted, over the Internet or via a public switched network, the communications session shall utilize a secure transport mechanism meeting industry standard for encryption. All laptops, hand-held devices and removable storage devices, shall utilize full disk encryption meeting industry standards for encryption.

Malware Protection

26. Affected Computing Devices shall be appropriately and reasonably protected against malicious software. All such Affected Computing Devices shall be configured with up-to-date anti-virus.

Security Awareness & Training

27. All Accessing Party personnel who have access to Bank Confidential Information are required to successfully complete initial security awareness training, including affirmative acknowledgement of their security and privacy responsibilities, and annual refresher training thereafter.

Host-Based Security – End Point Computing

28. Only devices provided by or under the management of Accessing Party personnel or by Bank may be used to access Bank Confidential Information. Public resources such as hotel PC kiosks, or other public-access terminals such as those available in malls and airports, may not be used for this purpose.

Policies and Procedures

- 29. Upon request, but no more frequently than annually, Accessing Party shall permit Bank to review its information security, physical security, and privacy policy at the Accessing Party's facility.
- 30. Accessing Party shall develop, implement, and maintain logical network diagrams and security documentation for the Affected Computing Devices to provide an overview of the security requirements and a description of the security controls in place.

Change Management

31. Accessing Party shall utilize a change management process based on industry standards that is designed to ensure that Accessing Party information security personnel have insight and approval over changes affecting security devices (e.g., firewalls, VPNs, IPS) and systems handling authentication, authorization, and auditing.

Data Loss Prevention

32. Accessing Party shall maintain an approach to data loss prevention in-line with industry standards to mitigate the risk of unauthorized data disclosure from applications and network infrastructure.

Asset Inventory

33. Accessing Party shall maintain an asset inventory of all critical Affected Computing Devices that access, store, process or transmit Bank Confidential Information.

Remote Access

- 34. In the event Accessing Party requires remote access to Bank's network:
 - a. Accessing Party's use of individually assigned Bank remote access credentials and authentication devices are for the sole use of that individual and not to be shared with anyone else for any reason.
 - b. Accessing Party remote access to Bank's network will require a Bank approved or Bank-provided multi-factor authentication.

Background Investigations

- 35. Accessing Party on-boarding process for personnel that will have access to Bank Confidential Information shall include, at its expense, an industry standard background investigation.
- 36. Accessing Party shall not assign any individual to Bank who (1) has been convicted of a felony of any nature or a misdemeanor of violence, theft or fraud or a crime involving dishonesty or breach of trust; (2) has an unverifiable SSN, address, employment, or education; or (3) appears on the OFAC list.
- 37. Accessing Party shall promptly notify Bank in the event Accessing Party becomes aware that any individual assigned to Bank is under investigation or arrested for, or convicted of, a felony of any nature or a misdemeanor of violence, theft or fraud or a crime involving dishonesty or breach of trust.

Systems Development and Maintenance

38. Accessing Party will annually conduct appropriate application security reviews for critical applications and prior to the promotion of production changes to critical application.

Segregation of Duties

 Accessing Party shall safeguard Bank Confidential Information by employing commercially reasonable practices, techniques and/or technologies, to implement segregation of duties.

Cloud Service Delivery*

40. Accessing Party shall document controls used to maintain logical separation of data (prevent inadvertent release of data) in multi-tenant environments.

^{*} Adapted from the Cloud Security Alliance Cloud Controls Matrix Version 1.1 (2010), http://www.cloudsecurityalliance.org/cm.html

ATTACHMENT 8: Sallie Mae Bank Borrower Benefits

BENEFIT PACKAGE	BENEFIT NAME BENEFIT TYPE: AUTO	BENEFIT DESCRIPTION DEBIT (DDD) INTEREST RATE REDUCTION - FFELP	APPLICABLE LOAN <u>PROGRAMS</u>
9999	ER/ACH INTEREST RATE REDUCTION	AN INTEREST RATE DISCOUNT EARNED WHEN MONTHLY PAYMENTS ARE MADE BY AUTOMATIC DEBIT ON YOUR ELIGIBLE FEDERAL STAFFORD AND PLUS LOANS IN AN EXTENDED REPAYMENT PLAN FOR THE MAXIMUM REPAYMENT TERM AVAILABLE	GS, PL, PL/GP
ADDM	SMS HELP CONSOLIDATED ACH RATE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
ADSM	SMART LOAN DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
ADSN	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS, PL, PL/GP
CARA	AUTOMATIC REPAYMENT ADVANTAGE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CBAA	AUTOADVANTAGE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS
CBAC	CONSOLIDATION BORR INCENTIVE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CBAP	1/4% AUTOPAY REDUCTION	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM

BENEFIT PACKAGE CBAU	BENEFIT NAME AUTOCHEK DISCOUNT	BENEFIT DESCRIPTION DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 %	APPLICABLE LOAN <u>PROGRAMS</u> GS
		INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	
CBDD	NATL STUDENT CONSOLIDATION CORP	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CITA	INTEREST RATE REDUCTION FOR ACH	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	PL, PL/GP
CNS2	AUTO DEBIT	DURING ACTIVE PERIODS OF REPAYMENT, A 0.375 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CNSL	AUTOMATIC DEBIT	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CRAD	AUTO ADVANTAGE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
CSBA	STUDENT BREAK	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS, PL
CT23	AUTOMATIC DEBIT BENEFIT	A 0.50% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER SUCCESSFULLY ACTIVATES ACH PAYMENTS.	PL, PL/GP
CT25	AUTOMATIC DEBIT BENEFIT	A 0.25% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER SUCCESSFULLY ACTIVATES ACH PAYMENTS.	GS, PL, PL/GP, SM

BENEFIT PACKAGE CT26	BENEFIT NAME AUTOMATIC DEBIT BENEFIT	BENEFIT DESCRIPTION A 0.25% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER SUCCESSFULLY ACTIVATES ACH PAYMENTS.	APPLICABLE LOAN <u>PROGRAMS</u> GS, PL, PL/GP, SM
CT27	AUTOMATIC DEBIT BENEFIT	A 0.50% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER SUCCESSFULLY ACTIVATES ACH PAYMENTS.	PL, PL/GP
CTAR	1/4% AUTOPAY REDUCTION	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS
CTDD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
DDFG	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	PLGP
DIRP	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS
FPAD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	PL
PRDD	PREMIER	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	SM
SMRH	HELP RATE	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	GS, PL

BENEFIT PACKAGE STTA	BENEFIT NAME TOP PERFORMER-AUTO DEBIT	BENEFIT DESCRIPTION DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	APPLICABLE LOAN <u>PROGRAMS</u> GS
	BENEFIT TYPE: DISCOUNT (DSC) INTEREST I	RATE REDUCTION, TYPICALLY REQUIRING ON-TIME PAYMENT	TS - FFELP
CBBO	CITIBONUS	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 60 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT AND FORBEARANCE.	SM
CBCI	CONSOLIDATION INCENTIVE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM
CBDC	NATL STUDENT CONSOLIDATION CORP	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM
CBDM	DEBT MINDER REPAY ADVANTAGE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM

BENEFIT PACKAGE CBDS	BENEFIT NAME DISCOUNT STUDENT LOAN	BENEFIT DESCRIPTION A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	GS	APPLICABLE LOAN PROGRAMS
CBRT	CONSOLIDATION BORROWER INCENTIVE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM	
CBSM	CONSOLIDATED BORROWER INCENTIVE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT.	SM	
СВТМ	CONSOLIDATION BORROWER INCENTIVE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM	
СВТР	CONSOLIDATION BORR INCENTIVE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM	

BENEFIT PACKAGE CETP	BENEFIT NAME CITIEXTRA\$	BENEFIT DESCRIPTION A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	GS	APPLICABLE LOAN PROGRAMS
CIT3	INTEREST RATE REDUCTION	A 0.75% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 60 ON-TIME PAYMENTS.	SM	
CITD	CONSOLIDATION SAVINGS	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE. MINIMUM ORIGINAL LOAN BALANCE OF \$ 5,000.00 IS REQUIRED.	SM	
CITI	STAFFORD 06/07 BORR INCENTIVES	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE. ADDITIONAL 1.00 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	GS	
CONS	SMS HAWAII CONSOL HELP BENEFIT	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 24 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM	

BENEFIT PACKAGE CSST	BENEFIT NAME TOP PERFORMER - STAFFORD	BENEFIT DESCRIPTION A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	GS	APPLICABLE LOAN PROGRAMS
CT05	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS.	SM	
CT07	INTEREST RATE REDUCTION	A 2.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS.	GS	
CT08	INTEREST RATE REDUCTION	A 2.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 48 ON-TIME PAYMENTS.	GS	
CT12	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS.	SM	
CT18	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS.	SM	
CT20	INTEREST RATE REDUCTION	A 2.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 48 ON-TIME PAYMENTS.	GS	
CT22	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 48 ON-TIME PAYMENTS.	SM	
CT45	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS.	SM	
CT51	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS. AND ADDITIONAL 1.00% INTEREST RATE REDUCTION AFTER THE BORROWER MAKES 48 ON-TIME PAYMENTS.	GS	

BENEFIT PACKAGE CT52	BENEFIT NAME INTEREST RATE REDUCTION	BENEFIT DESCRIPTION A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS. AN ADDITIONAL 1.00% INTEREST RATE REDUCTION AFTER THE BORROWER MAKES 48 ON-TIME PAYMENTS. AFTER 60 ON-TIME PAYMENTS, THE BORROWER WILL RECEIVE AN ADDITIONAL 2.00% INTEREST RATE REDUCTION.	APPLICABLE LOAN <u>PROGRAMS</u> GS
CT54	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED AFTER THE BORROWER MAKES 36 ON-TIME PAYMENTS. AND ADDITIONAL 1.25% INTEREST RATE REDUCTION AFTER THE BORROWERS MAKES 48 ON-TIME PAYMENTS.	GS
СТТР	2% RATE REDUCTION	A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	GS
DMRA	DEBT MINDER REPAYADVANTAGE	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	SM
GF02	INTEREST RATE DISCOUNT	A 0.750 % INTEREST RATE DISCOUNT.	PL/GP
NCBD	DISCOUNT STUDENT LOAN	A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT AND FORBEARANCE.	GS

BENEFIT PACKAGE SLGR	BENEFIT NAME GREAT REWARDS - 2% AFT 48 MTH	BENEFIT DESCRIPTION A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS	GS	APPLICABLE LOAN PROGRAMS
		INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.		
SMHH	HELP REPAY	A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT.	GS	
SMSM	SLMA CONSOLIDATION SAVINGS	A 1.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 36 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE. MINIMUM ORIGINAL LOAN BALANCE OF \$10,000.00 IS REQUIRED.	SM	
STTS	TOP PERFORMER-STAFFORD	A 2.000 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT, DEFERMENT, FORBEARANCE AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	GS	
	BENEFIT TYPE: IMMEDIATE	E DISCOUNT (IMP) INTEREST RATE REDUCTION - FFELP		
CBAR	DIVIDENDS AUTO RATE REDUCTION	A 0.500 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, FORBEARANCE THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	GS	

BENEFIT PACKAGE CBCB	BENEFIT NAME CONSOLIDATION BENEFIT	BENEFIT DESCRIPTION A 0.500 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, FORBEARANCE THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN <u>PROGRAMS</u> SM
CBHL	HEAL BORROWER INCENTIVE	A 1.500 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, AND FORBEARANCE.	SM
CBIM	NATL STUDENT CONSOLIDATION CORP	A 0.500 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, FORBEARANCE THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	SM
CI14	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	GS
C115	INTEREST RATE REDUCTION	A 1.50% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	GS
CIRR	.25% PLUS REPAYMENT INCENTIVE	A 0.250 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, AND FORBEARANCE.	PL

BENEFIT PACKAGE CITP	BENEFIT NAME INTEREST RATE REDUCTION	BENEFIT DESCRIPTION A 0.500 % INTEREST RATE DISCOUNT DURING REPAYMENT. DISCOUNT BECOMES PERMANENT AFTER THE FIRST 0 PAYMENT IS MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN <u>PROGRAMS</u> PL, PL/GP
CORS	IMMEDIATE REPAYMENT DISCOUNT	A 0.250 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, FORBEARANCE THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	GS, PL
CT00	INTEREST RATE REDUCTION	A 0.25% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT.	GS, PL
CT28	INTEREST RATE REDUCTION	A 1.00% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	GS, PL, PL/GP
CT30	INTEREST RATE REDUCTION	A 0.50% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	GS, PL, PL/GP
CT31	INTEREST RATE REDUCTION	A 0.50% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT.	PL

BENEFIT PACKAGE CT38	BENEFIT NAME INTEREST RATE REDUCTION	BENEFIT DESCRIPTION A 0.75% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	APPLICABLE LOAN <u>PROGRAMS</u> PL, PL/GP	
CT39	INTEREST RATE REDUCTION	A 1.25% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT. TO RETAIN THE BENEFIT THE BORROWER MUST CONTINUE TO MAKE ON-TIME PAYMENTS. IF THIS INTEREST RATE REDUCTION IS LOST DUE TO A MISSED PAYMENT, THE BORROWER CAN RE-EARN THE INTEREST RATE REDUCTION BY MAKING 24 CONSECUTIVE ON-TIME PAYMENTS.	PL, PL/GP	
CT41	INTEREST RATE REDUCTION	A 1.25% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT.	PL/GP	
CT44	INTEREST RATE REDUCTION	A 0.75% INTEREST RATE REDUCTION APPLIED WHEN THE BORROWER BEGINS REPAYMENT.	PL	
GPF1	INTERST RATE REDUCTION	A 0.250 % INTEREST RATE DISCOUNT DURING REPAYMENT, DEFERMENT, AND FORBEARANCE.	PL/GP	
BENEFIT TYPE: REBATE 1 (RBT) REFUND OR LOAN CREDIT - FFELP				
6ATU	LOAN CREDIT	A 1.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT AFTER THE FIRST 12 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. ADDITIONAL 1.000 % LOAN CREDIT AFTER THE FIRST 30 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. TO EARN THE BENEFIT, WITHIN 1 DAYS AFTER THE LOAN'S FIRST PAYMENT DUE DATE, SIGN UP FOR EMAIL COMMUNICATION AND MAINTAIN A VALID EMAIL ADDRESS.	PL/GP	

BENEFIT PACKAGE 6BOS	BENEFIT NAME SALLIE MAE CASH BACK	BENEFIT DESCRIPTION A 3.300 % LOAN CREDIT OR CASH AWARD BASED ON ORIGINAL LOAN AMOUNT AFTER THE FIRST 33 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. TO EARN THE BENEFIT, PRIOR TO THE LOAN'S REPAYMENT BEGIN DATE, SIGN UP FOR EMAIL COMMUNICATION AND MAINTAIN A VALID EMAIL ADDRESS.	APPLICABLE LOAN PROGRAMS GS
CBPB	PREMIER BENEFIT	A 1.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT.	GS
СВРР	PREMIER BENEFIT	A 1.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT.	PL
CPRR	2% PRINCIPAL BALANCE REDUCTION	A 2.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT.	GS
CT35	PRINCIPAL REDUCTION	A 0.50% PRINCIPAL REDUCTION APPLIED AFTER THE BORROWER MAKES 24 ON-TIME PAYMENTS.	GS
RSUN	PRINCIPAL REDUCTION	A 1.500 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT AFTER THE FIRST 12 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	GS, PL, PL/GP
SMCB	SALLIE MAE CASH BACK	A 3.300 % LOAN CREDIT OR CASH AWARD BASED ON ORIGINAL LOAN AMOUNT AFTER THE FIRST 33 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. TO EARN THE BENEFIT, PRIOR TO THE LOAN'S REPAYMENT BEGIN DATE, SIGN UP FOR EMAIL COMMUNICATION AND MAINTAIN A VALID EMAIL ADDRESS.	GS
SMSH	HELP REWARD	A 2.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT AFTER THE FIRST 12 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	GS
SMSP	PLUS PRINCIPAL REBATE	A 3.000 % LOAN CREDIT BASED ON ORIGINAL LOAN AMOUNT.	PL

BENEFIT PACKAGE STTR	BENEFIT NAME TOP PERFORMER-STAFFORD RBT	BENEFIT DESCRIPTION A LOAN CREDIT BASED ON ORIGINATION FEES PAID IN EXCESS OF \$250.00 AFTER THE FIRST 24 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN <u>PROGRAMS</u> GS
	BENEFIT TYPE: REB	PATE 2 (RB2) REFUND OR LOAN CREDIT - FFELP	
CT33	WAIVED 9 PAYMENTS	THIS BENEFIT WILL PAY THE LAST 9 MONTHS OF PAYMENTS ON THE BORROWER'S LOAN AFTER 111 ONTIME PAYMENTS ARE MADE.	GS
CT47	PRINCIPAL REDUCTION	A 1.00% PRINCIPAL REDUCTION APPLIED TO THE LOAN WHEN THE BORROWER BEGINS REPAYMENT.	GS
CT48	PRINCIPAL REDUCTION	A 2.00% PRINCIPAL REDUCTION APPLIED TO THE LOAN WHEN THE BORROWER BEGINS REPAYMENT.	GS
CT53	WAIVED 6 PAYMENTS	THIS BENEFIT WILL PAY THE LAST 6 MONTHS OF PAYMENTS ON THE BORROWER'S LOAN AFTER 114 ONTIME PAYMENTS ARE MADE.	GS, PL, PL/GP
	BENEFIT TYPE: AUTO D	EBIT (DDD) INTEREST RATE REDUCTION - PRIVATE	
AEXS	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
CXXS	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
DELA	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	MX/DL XS/DL, XS/SD
MLBD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	BS, LS, MT

BENEFIT PACKAGE	BENEFIT NAME	BENEFIT DESCRIPTION	APPLICABLE LOAN PROGRAMS
MXIH	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	MX/IH
MXMD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	MD/02
NUDD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
PHXS	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
UC45	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
UTXS	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.250 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
XSAD	AUTOMATIC DEBIT BENEFIT	A 0.25 % INTEREST RATE REDUCTION IS AVAILABLE FOR ENROLLMENT IN THE AUTOMATIC DEBIT PROGRAM. THE RECURRING AUTOMATIC MONTHLY PAYMENTS MUST BE SUCCESSFULLY DEDUCTED FROM A DESIGNATED ACCOUNT FOR RATE REDUCTION TO APPLY. THIS BENEFIT IS SUSPENDED DURING PERIODS OF FORBEARANCE AND CERTAIN DEFERMENTS.	BS, MX/DL, MX/IH, MX/02, XS, XS/PP

BENEFIT PACKAGE XSAP	BENEFIT NAME AUTOMATIC DEBIT BENEFIT WITH TUITION INSURANCE	BENEFIT DESCRIPTION A 0.25 % INTEREST RATE REDUCTION IS AVAILABLE FOR ENROLLMENT IN OUR AUTOMATIC DEBIT PROGRAM. THE RECURRING AUTOMATIC MONTHLY PAYMENTS MUST BE SUCCESSFULLY DEDUCTED FROM A DESIGNATED BANK ACCOUNT FOR THE RATE REDUCTION TO APPLY. THIS BENEFIT IS SUSPENDED DURING PERIODS OF FORBEARANCE AND CERTAIN DEFERMENTS. THE TUITION INSURANCE COVERS UP TO \$5000 (\$2500 PER SEMESTER) OF TUITION, ROOM, BOARD AND OTHER FEES NOT REFUNDED BY THE SCHOOL IF A STUDENT IS FORCED TO WITHDRAW FOR A COVERED MEDICAL REASON. THIS BENEFIT IS PROVIDED TO STUDENTS WITH SMART OPTION STUDENT LOANS AND REQUIRES ACTIVATION TO RECEIVE A FULL 12 MONTHS OF COVERAGE.	APPLICABLE LOAN PROGRAMS XS/PP
XSIU	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
XSRV	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
XSSD	DIRECT REPAY	DURING ACTIVE PERIODS OF REPAYMENT, A 0.500 % INTEREST RATE DISCOUNT. FOR SCHEDULED PAYMENTS MADE THROUGH AUTOMATIC DEBIT PARTICIPATION.	XS
BENEFIT TYPE: DISCOUNT (DSC) INTEREST RATE REDUCTION, TYPICALLY REQUIRING ON-TIME PAYMENTS - PRIVATE			
6AFR	INTEREST RATE REDUCTION	A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 24 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT. TO EARN AND	ES/TA

BENEFIT PACKAGE	BENEFIT NAME	BENEFIT DESCRIPTION CONTINUE RECEIVING THE BENEFIT, PRIOR TO THE LOAN'S THIRD PAYMENT DUE DATE, SIGN UP FOR EMAIL COMMUNICATION, MAINTAIN A VALID EMAIL ADDRESS AND MAKE PAYMENTS BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN PROGRAMS
6AHP	ELECTRONIC COMMUNICATIONS BNFT	A 0.25 % INTEREST RATE REDUCTION IS AVAILABLE IF THE BORROWER PROVIDES A VALID EMAIL ADDRESS AND ELECTS TO RECEIVE ALL SERVICING COMMUNICATIONS VIA EMAIL WITHIN 90 DAYS OF THE LOAN'S FIRST PAYMENT DUE DATE. THE ELECTRONIC COMMUNICATIONS BENEFIT IS AVAILABLE FOR AS LONG AS THE BORROWER MAINTAINS A VALID EMAIL ADDRESS, CONTINUES TO CONSENT TO RECEIVE ALL SERVICING COMMUNICATIONS VIA EMAIL AND MAKES ON-TIME PAYMENTS OF AMOUNTS SCHEDULED WITH NO INTERRUPTION. IF THE BORROWER FAILS TO MAKE A PAYMENT BY THE INITIALLY SCHEDULED DUE DATE, THE BENEFIT CAN BE REINSTATED ONE TIME BY (1) REQUESTING REINSTATEMENT OF THE BENEFIT THROUGH YOUR ONLINE ACCOUNT AND (2) AFTER REQUESTING REINSTATEMENT, MAKING THE NEXT SCHEDULED PAYMENT BY THE INITIALLY SCHEDULED DUE DATE. THIS BENEFIT LOWERS THE BORROWER'S INTEREST RATE, RESULTING IN A SHORTER TERM, BUT DOES NOT CHANGE THE AMOUNT OF THE MONTHLY PAYMENT.	XS/PP

BENEFIT PACKAGE DEXS	BENEFIT NAME INTEREST RATE REDUCTION	BENEFIT DESCRIPTION A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT. TO EARN AND CONTINUE RECEIVING THE BENEFIT, PRIOR TO THE LOAN'S REPAYMENT BEGIN DATE, SIGN UP FOR EMAIL COMMUNICATION, MAINTAIN A VALID EMAIL ADDRESS AND MAKE PAYMENTS BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN <u>PROGRAMS</u> XS
NULR	STUDENT LOAN REWARDS	A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	XS
SGDC	SIGNATURE REWARDS .5% AFT 48	A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 48 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	XS
TUA9	INTEREST RATE REDUCTION	A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 24 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING REPAYMENT AND AS LONG AS PAYMENTS CONTINUE TO BE MADE ACCORDING TO THE INITIAL SCHEDULE.	ES/TA
TUAN	INTEREST RATE REDUCTION	A 0.500 % INTEREST RATE DISCOUNT AFTER THE FIRST 24 PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED. BENEFIT IS AVAILABLE DURING	ES/TA

BENEFIT PACKAGE	BENEFIT NAME	BENEFIT DESCRIPTION REPAYMENT. TO EARN AND CONTINUE RECEIVING THE BENEFIT, PRIOR TO THE LOAN'S THIRD PAYMENT DUE DATE, SIGN UP FOR EMAIL COMMUNICATION, MAINTAIN A VALID EMAIL ADDRESS AND MAKE PAYMENTS BY THE DUE DATES AS INITIALLY SCHEDULED.	APPLICABLE LOAN PROGRAMS
XNEA	INTEREST RATE REDUCTION	A 0.25 PERCENTAGE POINT INTEREST RATE REDUCTION AFTER MAKING 12 MONTHLY CONSECUTIVE ON-TIME PAYMENTS ON OR BEFORE THE DUE DATES AS SHOWN ON THE BILLING STATEMENTS. IF BORROWER MISSES OR IS LATE WITH ANY OF THE INITIAL 12 PAYMENTS, THE BENEFIT WILL NOT BE EARNED. IF BORROWER EARNS THE BENEFIT, THEN MISSES ANY PAYMENT, THE BENEFIT WILL BE SUSPENDED UNTIL AFTER REINSTATEMENT IS REQUESTED AND 3 CONSECUTIVE SCHEDULED PAYMENTS ARE MADE ON-TIME. THE BENEFIT IS SUSPENDED DURING PERIODS OF FORBEARANCE AND CERTAIN DEFERMENTS.	XS/PP
XSPP	ELECTRONIC COMMUNICATIONS BNFT	A 0.25 % INTEREST RATE REDUCTION IS AVAILABLE IF THE BORROWER PROVIDES A VALID EMAIL ADDRESS AND ELECTS TO RECEIVE ALL SERVICING COMMUNICATIONS VIA EMAIL WITHIN 90 DAYS OF THE LOAN FIRST PAYMENT DUE DATE. THE ELECTRONIC COMMUNICATIONS BENEFIT IS AVAILABLE FOR AS LONG AS THE BORROWER MAINTAINS A VALID EMAIL ADDRESS, CONTINUES TO CONSENT TO RECEIVE ALL SERVICING COMMUNICATIONS VIA EMAIL AND MAKES ON-TIME PAYMENTS OF AMOUNTS SCHEDULED WITH NO INTERRUPTION. IF THE	XS/PP

BENEFIT PACKAGE	BENEFIT NAME	BENEFIT DESCRIPTION BORROWER FAILS TO MAKE A PAYMENT BY THE INITIALLY SCHEDULED DUE DATE THE BENEFIT CAN BE REINSTATED ONE TIME BY (1) REQUESTING REINSTATEMENT OF THE BENEFIT THROUGH YOUR ONLINE ACCOUNT AND (2) AFTER REQUESTING REINSTATEMENT, MAKING THE NEXT SCHEDULED PAYMENT BY THE INITIALLY SCHEDULED DUE DATE.	APPLICABLE LOAN <u>PROGRAMS</u>
	BENEFII I I PE: IMMEI	DIATE (IMP) INTEREST RATE REDUCTION - PRIVATE	
MDMX	INTEREST RATE REDUCTION	A 0.500 % INTEREST RATE DISCOUNT DURING REPAYMENT THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	MD/02
MXRR	INTEREST RATE REDUCTION	A 1.000 % INTEREST RATE DISCOUNT DURING REPAYMENT THAT CONTINUES AS LONG AS PAYMENTS ARE MADE BY THE DUE DATES AS INITIALLY SCHEDULED.	MX/02
	BENEFIT TYPE: RE	EBATE (RBT) TO UPROMISE ACCOUNT - PRIVATE	
SRCT	SMART REWARD (K-12 Loans)	PRIMARY BORROWER CAN EARN A REWARD INTO HIS OR HER ACTIVE UPROMISE ACCOUNT OF 2 % OF THE SCHEDULED LOAN PAYMENT AMOUNT FOR EACH ON TIME PAYMENT DURING THE IN-SCHOOL PERIOD. LOAN PAYMENTS MUST REMAIN CURRENT TO BE ELIGIBLE FOR THE REWARD. BENEFIT AND UPROMISE MEMBERSHIP ARE SUBJECT TO THE TERMS AND CONDITIONS OF THE UPROMISE SERVICE.	СТ
XSSR	SMART REWARD	THE 2 % SMART REWARD BENEFIT IS AVAILABLE ON SMART OPTION STUDENT LOANS FIRST DISBURSED ON OR AFTER JUNE 1, 2010. PRIMARY BORROWER CAN EARN A REWARD INTO HIS OR HER ACTIVE UPROMISE ACCOUNT OF 2 % OF THE SCHEDULED LOAN PAYMENT AMOUNT FOR EACH ON TIME PAYMENT DURING THE	XS/PP
		20	

SALLIE MAE BANK - BORROWER BENEFITS

BENEFIT PACKAGE BENEFIT NAME

BENEFIT DESCRIPTION

APPLICABLE LOAN PROGRAMS

IN-SCHOOL AND SEPARATION PERIODS. LOAN PAYMENTS MUST REMAIN CURRENT TO BE ELIGIBLE FOR THE REWARD. BENEFIT AND UPROMISE MEMBERSHIP ARE SUBJECT TO THE TERMS AND CONDITIONS OF THE UPROMISE SERVICE.

ATTACHMENT 9: Form of Servicing Agreement for Third-Party Purchaser

PRIVATE LOAN SERVICING AGREEMENT

This PRIVATE LOAN SERVICING AGREEMENT (together with all schedules and attachments hereto, the "Servicing Agreement") is effective as of , 201 (the "Effective Date"), between [] (the "Holder") and NAVIENT SOLUTIONS, INC. (the "Servicer").

WITNESSETH:

WHEREAS, the Servicer is in the business of servicing student loans and providing related services; and

WHEREAS, the Holder has acquired certain private student loans from Sallie Mae Bank (the "Acquisition") that were being serviced by the Servicer immediately prior to the Acquisition; and

WHEREAS, the Holder desires the Servicer to continue to service such private student loans on behalf of the Holder pursuant to the terms of this Servicing Agreement;

NOW THEREFORE, in consideration of the mutual covenants herein contained and of other valuable consideration, the parties hereto agree as follows:

ARTICLE ONE DEFINITIONS

SECTION 1.1 Definitions. Wherever used in this Servicing Agreement, unless the context clearly indicates a contrary intent, the following terms will have the meanings indicated:

- (gg) "Acquisition" has the meaning set forth in the recitals of this Servicing Agreement.
- (hh) "Affiliate" means, with respect to a party hereto, an entity or person who, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such party.
- (ii) "Application" means an application, in any form authorized by the applicable Program Rules, from a prospective Borrower for a Private Loan.
- (jj) "AUPs" has the meaning set forth in Section 5.1 (e)(i).
- (kk) "Borrower" means the obligor on a Loan, and includes any Cosigner.
- (II) "Collection Costs" mean collection costs that are authorized to be charged to the Borrower by Regulations, the promissory notes, the applicable Program Rules or applicable law.
- (mm) "Cosigner" includes any person whose signature is requested as a condition to granting credit to a Borrower, or as a condition for forbearance on collection of a Borrower's obligation that is in default.

- (nn) "Depository Account" means an account or accounts owned or maintained by the Servicer into which all funds collected on behalf of the Holder shall be deposited in the first instance. The Depository Account may be a pooled account in which funds collected for the purposes of this Servicing Agreement are commingled with other funds.
- (00) "Effective Date" has the meaning set forth in the preamble of this Servicing Agreement.
- (pp) "EFT" means electronic funds transfer processes.
- (qq) "GLB Regulations" means the Joint Banking Agencies' Privacy of Consumer Financial Information, Final Rule (12 CFR Parts 40, 216, 332 and 573), the Federal Trade Commission's Privacy of Consumer Financial Information, Final Rule (16 CFR Part 313), or the Securities and Exchange Commission's Regulation S-P (17 CFR 248), as applicable, each implementing Title V of the Gramm-Leach-Bliley Act, Public Law 106-102, as amended.
- (rr) "Holder" has the meaning set forth in the preamble of this Servicing Agreement.
- (ss) "Holder Default" has the meaning set forth in Section 8.2.
- (tt) "Indemnified Party" has the meaning set forth in Section 10.4.
- (uu) "Indemnifying Party" has the meaning set forth in Section 10.4.
- (vv) "Identified Funds" means a payment or payments received by Servicer from any third party and which is identified by Servicer as a payment or payments made on account of a specific Loan, including, without limitation: (i) a payment of principal, interest, or other charges or fees; (ii) the return of previously negotiated proceeds of a wholly or partially canceled Loan; (iii) Returned Check Charges, Late Fees, and Collection Costs (as defined herein); and (iv) other fees and charges that are permitted by Regulations or the applicable promissory note.
- (ww) "Insolvency Event" shall mean with respect to a party, the existence of any of the following conditions: (i) a petition in bankruptcy or insolvency regarding the party shall be approved; (ii) the party shall admit in writing its inability to pay its debts generally; (iii) the party shall make a general assignment for the benefit of creditors; (iv) any proceeding shall be instituted by the party seeking relief by such party, as debtor, under any bankruptcy or insolvency legislation, or seeking liquidation, winding up, reorganization, arrangement, adjustment, protection, relief, or composition of it or its debts, or seeking the entry of an order for the appointment of a receiver, trustee, or other similar official for it or for any substantial part of its property; or (v) any of the foregoing actions shall be instituted against a party and not dismissed within sixty (60) days or a party shall take any corporate action to authorize any of the actions set forth herein. "Holder Insolvency Event" means an Insolvency Event with respect to the Holder or an entity of which the Holder is a wholly-owned subsidiary. "Servicer Insolvency Event" means an Insolvency Event with respect to the Servicer or an entity of which it is a wholly-owned subsidiary.

- (xx) "Late Fees" means late fees that are permitted by the applicable promissory notes.
- (yy) "Loan" means any of, and "Loans" means all of, the Private Loans serviced pursuant to the provisions of this Servicing Agreement.
- "Loan Documentation" means all attachments and other documentation relating to a Loan that are in the possession of the holder, or an agent of the holder. Without limiting the generality of the foregoing, the Loan Documentation for any Loan may include, as applicable:
 - x. A copy of the Borrower's Application;
 - xi. Evidence of all other documentation necessary to establish the diligent origination of the Loan in accordance with the applicable Program Rules, including but not limited to FICO score or credit bureau reports and record of any denials;
 - xii. The Note or a true copy thereof;
 - xiii. Evidence of full disbursement;
 - xiv. Evidence of all other documentation necessary to establish the diligent servicing of the Loan in accordance with the applicable Program Rules;
 - xv. An assignment or endorsement of the Note;
 - xvi. For purposes of sale of a Loan to the Servicer's designee, Loan Documentation shall also include:
 - (C) An assignment or endorsement of the Note to Servicer's designee; and
 - (D) Evidence of authority of the Holder to convey the Loan and the rights thereto.
- (aaa) "Monetary Breach" has the meaning set forth in Section 12.4.
- (bbb) "Note" or "Promissory Note" means the form promissory note applicable to the particular loan program pursuant to which a Borrower obtains such Private Loan.
- (ccc) "Payment Processing Day" means any day other than a Saturday, Sunday, or a day on which the banking institutions in the city in which a payment is received or a disbursement hereunder must be initiated are authorized or required by law or executive order to close.
- (ddd) "Principal Balance" means with respect to any Loan, the original principal amount of such Loan, plus any capitalized interest, minus any payments received that are allocated to principal.

- (eee) "Private Loan" means an education loan not guaranteed by the government and offered by a lender under the terms of a specific program approved by the Servicer for servicing hereunder, as described in Attachment 5, as the same may be amended from time to time by written agreement of the Holder and the Servicer.
- (fff) "Program Rules" means the regulations, and rules, documents or manuals adopted by the Holder that govern any Private Loan program, provided such materials have been delivered to and approved in writing by the Servicer.
- (ggg) "Receipts Account" means an account beneficially owned by the Holder and designated by the Holder as the account to which the Servicer shall transfer funds collected on behalf of the Holder from Borrowers.
- (hhh) "Refund Account" means an account or accounts owned or maintained by the Servicer from which refunds are made to Borrowers and others entitled thereto in connection with payments related to Loans.
- (iii) "Regulation" or "Regulations" means any rule, regulation, interpretation, instruction or procedure issued and in effect from time to time under all state and federal consumer credit laws.
- (jjj) "Returned Check Charges" means charges for the processing of a Borrower's check or other payment that is returned because of insufficient funds.
- (kkk) "Service," "services," "serviced," or "servicing" used in connection with a Loan or Loans means the process and procedures required of the Servicer or a holder with respect to such Loan or Loans pursuant to the terms of the applicable Program Rules that are delivered to and approved by the Servicer.
- (Ill) "Servicer" has the meaning set forth in the preamble of this Servicing Agreement.
- (mmm) "Servicer Default" has the meaning set forth in Section 9.2.
- (nnn) "Service Performance Measures" means the specific activity-related measures set forth in Attachment 2 hereto.
- (000) "Servicing Agreement" means this Private Loan Servicing Agreement.
- (ppp) "Term" has the meaning set forth in Section 7.1(a).

ARTICLE TWO ORIGINATIONS [Intentionally Omitted]

ARTICLE THREE SERVICING

SECTION 3.1 General Obligations of the Servicer. The Servicer will service: (a) all Loans that were purchased by the Holder from Sallie Mae Bank pursuant to the Acquisition and that were serviced by the Servicer or its Affiliate immediately prior to the Effective Date, and (b) such other Loans that the parties hereto mutually agree are to be serviced hereunder. The Servicer, as agent for the Holder, shall maintain custody of the Loan Documentation; shall manage, service, administer and make collections on the Loans; shall maintain records with respect to all Loans; and shall make reports as specified in Attachment 3 hereto, except as may be otherwise agreed upon by the Holder. The Servicer shall (i) attempt to collect payments from the obligors of the Loans as and when the same shall become due and payable (including Returned Check Charges, Late Fees, and Collection Costs, but the Servicer is authorized to waive such items in its discretion in accordance with applicable Lending Policies); (ii) take all other steps mandated by the applicable Program Rules in connection with third-party servicing; (iii) process all deferrals and forbearances; (iv) collect on Loans in a diligent manner and also as required under the applicable Program Rules; and (v) perform fraud monitoring, investigation, and reporting, including the preparation of Suspicious Activity Reports (SARs).

SECTION 3.2 <u>Title to Loans</u>. Neither custody of Loans or Loan Documentation by the Servicer hereunder, nor the servicing activities contemplated hereby, shall be deemed to convey to the Servicer any of the Holder's beneficial or legal ownership interest in the Loans. The Servicer recognizes that the Holder has beneficial and legal ownership of the Loans, the Loan Documentation therefor, and the rights and benefits pertaining thereto.

SECTION 3.3 Establishment and Authorization for Use of the Receipts Account. The Holder shall provide the Servicer with all authorizations and information, and shall take all such further steps as may be necessary, in order to authorize and enable the Servicer to initiate the movement of funds to and from the Receipts Account by EFT. The Servicer shall not use its access to the Receipts Account except for the purposes expressly authorized in this Servicing Agreement.

SECTION 3.4 Collections. The Servicer will use reasonable efforts to collect, on behalf of the Holder, all payments with respect to any Loan from the Borrower, any party responsible for payments as a Cosigner of the Loan, and the school (refunds, voids).

(a) <u>Identified Funds: Deposit and Transfer.</u> Not later than the first Payment Processing Day that any payment received from a third party becomes Identified Funds, the Servicer will deposit the payment into the Depository Account. The Servicer will post the payment and initiate EFT transfer of Identified Funds to the Receipts Account within one Payment Processing Day after deposit to the Depository Account; provided, however, that Servicer may withhold for their proper purpose amounts needed to offset previous overpayments to the Receipts Account and to effect any other adjustment reasonably deemed necessary by the Servicer to resolve an account balance discrepancy, including, by way of example, but without limitation, (i) overpayments resulting from overestimated balances due to retire or consolidate Loans, (ii) payments mistakenly recognized as Identified Funds, wholly or in part, (iii) overdue servicing fees and charges as authorized in Section 8.2, below, and (iv) Returned Check Charges, Late Fees and Collection Costs pursuant to subsection (c), below. Monies so withheld may be transferred to the Refund Account or otherwise transferred to the proper party.

- (b) <u>Unidentified Funds: Processing</u>. Payments received by the Servicer, which are not readily identified as payments on account of a particular Loan because of the omission or absence of a scannable coupon or billing statement, or other reason, will be further processed manually. Payments that are ultimately recognized as payments on account of a particular Loan become Identified Funds and will be posted and transferred in accordance with the provisions of subparagraph (a), above.
- (c) Charges to Borrower for Servicer's Account. The Servicer is authorized to charge to and collect from any Borrower, for the Servicer's account, such reasonable sums as agreed to by the Holder and the Borrower in appropriate disclosures to compensate it for (i) activities that would not be necessary except for a special request made by the Borrower or conduct of the Borrower inconsistent with the terms of the contract between the Holder and the Borrower (for example, but without limitation, Returned Check Charges, Late Fees, Collection Costs, and retrieval and transmittal to Borrower of Borrower's payment history or amortization schedule). To the extent of any Identified Funds attributable to a Loan, and from the first monies collected, the Servicer shall be entitled to retain Returned Check Charges and Collection Costs. Late Fees shall be the property of the Holder.
- SECTION 3.5 Inquiries, Complaints and Reports. The Servicer will respond promptly to any inquiries and complaints from the Holder, the Borrower, or the Borrower's school regarding Loans that the Servicer is servicing for the Holder. The Servicer will make available to the Holder on a periodic basis, as agreed to by the parties, reports, data extracts, and data tapes substantially in the form prepared as of the date hereof (or as modified if necessary to comply with the Regulations, the applicable Program Rules or applicable law or as otherwise permitted by this Servicing Agreement) by the Servicer and generated by the Servicer's servicing system.

SECTION 3.6 <u>Loan Acquisition</u>. The Holder may from time to time acquire portfolios of loans that it desires the Servicer to service on its behalf. If the Holder and the Servicer mutually agree upon a conversion plan (including the Servicer's fees) for any such portfolio, the Holder may transfer to the Servicer such portfolio for servicing hereunder. Such transferred portfolio of loans will become Loans subject to this Servicing Agreement upon conversion to the Servicer's system.

SECTION 3.7 Obligations of the Holder.

- (a) <u>Forwarding Communications and Payments</u>. If the Holder receives any communications regarding a Loan, it will promptly forward those communications to the Servicer in the manner in which the Servicer reasonably designates. The Holder will also promptly forward to the Servicer any payments the Holder receives on account of a Loan.
- (b) <u>Authorization and Power of Attorney</u>. The Holder hereby authorizes the Servicer on behalf of the Holder to communicate as the Holder's agent with the school and the Borrower by electronic means or otherwise, to issue checks and electronically disburse funds as contemplated herein, to exchange information pursuant to Regulations and applicable law with credit bureau organizations selected by the Servicer, and to do or perform any other act for the purpose of carrying out its obligations of servicing Loans. The Holder hereby appoints the

Servicer as its lawful attorney-in-fact to sign in the name of the Holder such documents as are necessary for the Servicer to perform its obligations as contemplated under this Servicing Agreement. Each party will provide to the other specimen authorized signatures upon request. The Holder will also execute a specific power-of-attorney covering the foregoing as the Servicer may reasonably request.

(c) <u>Inquiries, Complaints and Reports</u>. The Holder will cooperate with the Servicer as necessary to respond promptly and appropriately to complaints and inquiries concerning Applications and Loans.

ARTICLE FOUR REPRESENTATIONS AND WARRANTIES

SECTION 4.1 Servicer's Representations and Warranties. The Servicer represents and warrants the following to the Holder:

- (a) The Servicer is duly organized and qualified to do business in each jurisdiction in which qualification is required for the activities contemplated by this Servicing Agreement. The execution and performance of this Servicing Agreement has been duly authorized by all necessary action and does not and will not contravene any provision of law, rule or regulation applicable to the Servicer.
- (b) This Servicing Agreement constitutes a legal, valid and binding obligation of the Servicer enforceable in accordance with its terms subject to bankruptcy, insolvency and other similar laws affecting creditors' rights generally and subject to equitable principles.

SECTION 4.2 Holder's Representations and Warranties. The Holder represents and warrants the following to the Servicer:

- (a) The Holder is duly organized and qualified to do business in each jurisdiction in which qualification is required for the activities contemplated by this Servicing Agreement. The execution and performance of this Servicing Agreement has been duly authorized by all necessary action and does not and will not contravene any provision of law, rule or regulation applicable to the Holder.
- (b) This Servicing Agreement constitutes a legal, valid and binding obligation of the Holder, enforceable in accordance with its terms subject to bankruptcy, insolvency and other similar laws affecting creditors' rights generally and subject to equitable principles.

ARTICLE FIVE COVENANTS

SECTION 5.1 Servicer's Covenants. The Servicer covenants and agrees that:

(a) The Servicer will (or will cause its subservicer or subcontractor to) furnish specific Loan information to the Holder on request, will retain the Loan documents and all computer and manual records, receipts, tapes and other papers delivered to, acquired by or generated by the Servicer with respect to the Loans, and will not destroy or dispose of them, or

use the information contained therein or disclose the contents thereof to others except as required or permitted under this Servicing Agreement or as required by law. All such documents, records, receipts, tapes and papers must be retained by the Servicer until the end of any period required by applicable law, including the Regulations, and the Internal Revenue Code and any regulations issued thereunder. The Loan documents, records, receipts, tapes and papers must be kept in a manner to enable delivery to the Holder upon request on a timely basis; however, unless Loan documents, records, receipts, tapes and papers relating to the Loans are required to be maintained in paper form pursuant to Regulations or the applicable Program Rules, or other applicable law, the Servicer is authorized to destroy and dispose of any of them, provided that the Servicer has retained them in retrievable imaged form or has recorded their contents electronically and integrated such contents into the Servicer's data base pertaining to the Loans.

- (b) The Servicer will permit the Holder during normal business hours to gain real-time access to electronic information and data relating to the Loans, in accordance with the applicable provisions of Attachment 4.
- (c) The Servicer will make provision for physical security of its data processing and record retention systems. The physical security provisions will, at a minimum, restrict physical access to any and all Holder data to only those individuals who have a legitimate business need to access the data. Physical security controls will be tested on a regular basis to make sure that only authorized individuals have physical access to the Holder's data.
- (d) The Servicer will make provision for data back-up and off-premises storage of Holder data. The Servicer will promptly regenerate any lost, erased or destroyed data. In the event of a disaster, the Servicer agrees to implement a disaster recovery plan that will enable it to conduct due diligence and payment processing within fifteen (15) Payment Processing Days and to regain all other aspects of its ability to service the Loans within thirty (30) days after the occurrence of a disaster.
 - (e) The Servicer will maintain appropriate information security controls including:
 - (xi) Access controls that restrict access to system resources and data to only to those users who require access as part of their job responsibilities. The Servicer will update access rights based on personnel or system changes and periodically review users' access rights at an appropriate frequency based on the risk to the application or system. The Servicer will implement appropriate Acceptable-Use Policies ("AUPs") for each system or application and will require users to adhere to the AUPs as part of the Servicer's Information Security Policy.
 - (xii) Authentication controls appropriate to the level of risk in order to protective sensitive electronic Holder data and information.
 - (xiii) Appropriate network access controls that secure the computer networks to protect against unauthorized access.
 - (xiv) Operating systems access controls that secure access to the operating systems of all system components by: securing access to system utilities; restricting and monitoring privileged access; logging and monitoring user

- or program access to sensitive resources and alerting on security events; updating the operating systems with security patches, and securing the devices that can access the operating system through physical and logical means.
- (xv) Application access controls that control access to applications by using authentication and authorization controls appropriately for the risk level of the application, and monitoring of access rights to ensure they are the minimum required for the user's current business needs.
- (xvi) Remote access controls that provide secure remote access connectivity and appropriate user validation and authentication, and which can be deactivated on an individual user basis as may be required; controlling remote access through management approvals and subsequent audits, logging and monitoring of remote access, and using strong authentication and encryption to secure communications.
- (xvii) Use of encryption to protect information and mitigate the risk of disclosure or alteration of sensitive information stored on backup tapes or while being shipped or transported via courier service.
- (xviii) Malicious code controls that protect against the risk of malicious code, including but not limited to using anti-virus products on clients and servers; using appropriate blocking strategy on the network perimeter; filtering input to applications; and creating, implementing, and training staff in appropriate computing policies and practices.
- (xix) Intrusion detection and monitoring controls that allows the Servicer to detect and respond to information system intrusions.
- (xx) With respect to information that is "non-public personal information" (as defined in the GLB Regulations) that is disclosed or provided by the Holder to the Servicer in connection with this Servicing Agreement, the Servicer agrees, subject to the terms hereof and the limitations of liability set forth herein, that in performing its obligations under this Servicing Agreement, the Servicer shall comply with all reuse, redisclosure, or other customer information handling, processing, security, and protection requirements that are specifically required of a non-affiliated third-party processor or servicer (or subcontractor) under the GLB Regulations and other applicable federal consumer privacy laws, rules, and regulations. Without limiting the foregoing, the Servicer agrees that:
 - (A) The Servicer is prohibited from disclosing or using any "nonpublic personal information" (as defined in the GLB Regulations) disclosed or provided by the Holder's behalf to the Servicer, except solely to carry out the purposes for which it was disclosed, including use under an exception contained in 12 CFR sections 40.14 or 40.15 or 16 CFR sections 313.14 or 313.15, as applicable, of the GLB

Regulations in the ordinary course of business to carry out those purposes; the Servicer will maintain records in accordance with the Servicer's Customer Identification Program guidelines in accordance with 31 CFR 103.121; and

- (B) The Servicer has implemented and will maintain an information security program designed to meet the objectives of the Interagency Guidelines Establishing Standards for Safeguarding Customer Information, Final Rule (12 CFR Part 30, Appendix B) and the Federal Trade Commission's Standards for Safeguarding Customer Information (16 CFR Part 314).
- (f) The Servicer will comply with all audit activities required under the Regulations, and the applicable Program Rules and will permit and assist any Holder official or authorized agent, or governmental agency having regulatory authority over the Holder, during the Servicer's regular business hours, to examine and audit the books and records of the Servicer applicable to the Loans and to inspect the facilities, operations and operating procedures used or available for use to service the Loans.
- (g) The Servicer agrees to advise the Holder within five (5) business days of the Servicer's receipt of any final report issued by any state or federal governmental entity of the Servicer's material noncompliance with the Regulations or applicable laws affecting the Loans.
- (h) The Servicer has the right to change any part or all of its equipment, its servicing system, computer or data processing programs and any procedures, forms, reports and methods in connection with the services performed hereunder so long as the Servicer continues to service the Loans in conformance with the requirements herein. Material changes to services provided and controls must be approved by the Holder wherein such approval shall not be unreasonably withheld. It is specifically understood that the intent of this paragraph is to allow the Servicer flexibility over the methods and techniques of servicing subject to full compliance with substantive terms of this Servicing Agreement. The Servicer agrees to seek the Holder's approval of any change that persons similarly situated to the Servicer would in their reasonable judgment deem to be a material change made pursuant to this paragraph.
- (i) During the term of this Servicing Agreement, the Servicer shall maintain at least the following insurance with financially sound and reputable insurers:

(A) Liability Insurance:

- (1) Commercial General Liability insurance in an amount not less than \$1,000,000 per occurrence, \$2,000,000 aggregate for claims arising out of Bodily Injury, Property Damage and Personal Injury, and broad form contractual liability insurance to insure against any Bodily Injury, Property Damage or Personal Injury arising out of this Servicing Agreement.
- (2) Umbrella Excess Liability in an amount of at least \$10,000,000 each occurrence/aggregate excess of above.

- (B) Financial Institution Blanket Bond including Fidelity, Premises, Transit, Computer Crime (Funds Transfer) in an amount not less than \$15,000,000 per loss and aggregate which shall respond to loss involving assets of the Holder under the care, custody, and control of the Servicer.
- (C) Errors and Omissions/Professional Liability insurance in an amount not less than \$2,500,000 each claim and aggregate which shall respond to claims for wrongful acts, errors, and omissions arising out of the Servicer's performance of or failure to perform its obligations to the Holder under this Servicing Agreement.

On request of the Holder, made not more often than annually, the Servicer shall deliver to the Holder, official certificates issued by the insurer to the Servicer or an applicable Affiliate and evidencing the coverage specified above.

(j) The Holder will have the right to conduct internal control and/or audit reviews of the Servicer's provision for physical security, information security, compliance regulations, adherence to this Servicing Agreement, and other issues after giving the Servicer appropriate notice.

ARTICLE SIX SERVICE PERFORMANCE MEASURES

SECTION 6.1 Service Performance Measures. Except as may be otherwise agreed upon by the Holder, the Servicer shall perform its duties as outlined in the attached Service Performance Measures (Attachment 2) using reasonable care, applying that degree of skill, attention, and technological innovation that the Servicer exercises generally with respect to comparable student loans that the Servicer services on behalf of its Affiliates, all in accordance with Regulations, the applicable Program Rules, and other applicable law.

SECTION 6.2 Compliance with Service Performance Measures. The Servicer shall comply with the specific time frames, reliability rates, and procedures set forth in the Service Performance Measures (Attachment 2 hereto), as the same may be modified in accordance with this Section 6.2. Both parties acknowledge that industry norms and marketplace preferences dictate changes to performance service levels from time to time. The Servicer retains the right to modify Service Performance Measures relating to servicing so long as (a) the modifications apply to all Private Loans of the same type as the Loans serviced by the Servicer hereunder, regardless of the identity of the owner/holder, and (b) the modifications do not: (i) cause a violation of applicable law, Regulations, or the applicable Program Rules, (ii) materially and adversely affect the Servicer's or the Holder's ability to perform Loan servicing in a professional, cost-efficient, and customer-oriented manner, or (iii) cause such servicing to fall below industry standards as demonstrated in benchmarking studies conducted by or for the Servicer.

ARTICLE SEVEN TERM, TERMINATION, AND TRANSFER

SECTION 7.1 Term.

(a) Except as provided in Section 7.2, below, with respect to specific Loans, the term of this Servicing Agreement (the "Term") commences as of the Effective Date and continues in full force and effect for so long as any of the Loans remain outstanding, unless sooner terminated in accordance with its terms.

SECTION 7.2 <u>Termination as to Specific Loans</u>. This Servicing Agreement shall terminate as to a specific Loan upon (i) write-off of such Loan pursuant to the policies described in Section 7.5 below, (ii) the payment in full of all principal and interest by or on behalf of the Borrower, (iii) the purchase or other acquisition of such Loan by the Servicer or its Affiliates in accordance with the terms hereof, or (iv) the sale of such Loan by the Holder in accordance with Section 11.3 below. Whether or not this Servicing Agreement is terminated with respect to specific Loans pursuant to any of clauses (i) – (iv) above, it shall remain in full force and effect with respect to all other Loans.

SECTION 7.3 <u>Procedures for Deconversion of Loans</u>. Within thirty (30) days after delivery of a proper notice of termination of this Servicing Agreement pursuant to this Article Seven, Section 8.2, Section 9.2 or Section 10.5, a schedule and procedure for the orderly deconversion of Loans will be agreed to by the Servicer and the Holder. The parties will cooperate with each other to resolve problems that arise during or as a result of the deconversion process. Except for deconversions of Loans pursuant to Section 9.2, below, the Holder shall pay the deconversion fees specified in Attachment 1 (and such deconversion fees will be payable, without limitation, upon any deconversion of the Loans that occurs upon the expiration of the Term of this Servicing Agreement). In addition, the Holder shall pay the removal fees specified in Attachment 1 with respect to other removals of Loans, as set forth in Attachment 1 (and described in Note 2 of Attachment 1).

SECTION 7.4 Agreement Extension. If, despite the reasonable efforts of the parties, deconversion (i.e., transfer or removal) cannot be completed by the effective date of termination, then, as to all Loans not removed, this Servicing Agreement will extend automatically through the date upon which removal procedures are completed. Provided the Servicer has acted in good faith, if the failure to effect the removal of all such Loans continues for four months or more beyond the effective date of termination, then in addition to any increases already provided for in Attachment 1, the fees payable by the Holder shall be increased by an amount equal to 10.0% of the fees that are then in effect, effective as of the end of such four-month period.

SECTION 7.5 Policies Regarding Loan Balance Adjustments Through Write-Offs, Write-Downs, and Write-Ups. Except with the prior consent of the Holder or as authorized herein, the Servicer shall not write off or write down any Loan balance. In the course of normal operations, and in accordance with the requirements of the Regulations, the applicable Program Rules, and the same customs, policies, and procedures followed by the Servicer with respect to similar loans serviced by it on behalf of its Affiliates, as the same may be modified from time to time, the Servicer may systematically and routinely adjust individual Loan accounts and Borrower accounts by writing off, writing down, or writing up balances deemed uncollectible or cost-ineffective to rectify.

SECTION 7.6 <u>Survival of Terms</u>. Notwithstanding the termination of this Servicing Agreement, the provisions of Article Three with respect to remission of payments belonging to the Holder will survive and continue in full force and effect. Any liabilities of either party to the other party incurred prior to termination of this Servicing Agreement shall survive the termination of this Servicing Agreement.

ARTICLE EIGHT COMPENSATION

SECTION 8.1 Compensation. The Holder will compensate the Servicer for its services in accordance with the schedule of fees set forth in Attachment 1 to this Servicing Agreement. Such fees listed in Attachment 1 may be subject to increase to the extent that a demonstrable increase occurs in the cost incurred by the Servicer in providing the services to be provided hereunder due to changes in applicable governmental regulations or Program Rules or increases in U.S. Postal Service rates.

SECTION 8.2 Time of Payment; Interest Accrual; Set-Off; Holder Default. Fees shall be invoiced monthly in arrears and shall be paid within thirty (30) days of receipt of invoice. In any event, Servicer is required to provide not less than seven (7) days' notice of a request for payment. If the bill is not paid when due, interest on the unpaid balance shall accrue at the rate of the prime interest rate plus 5% per annum. If the bill (including accrued interest) is not paid within thirty (30) days after the due date, the Servicer shall be entitled to a set-off against funds within Servicer's possession or control including, without limitation, funds representing payments collected on behalf of the Holder. If the bill (including accrued interest) is not paid by the Holder within forty-five (45) days of delivery to the Holder of notice of nonpayment from the Servicer given in accordance with the notice provisions hereof, or immediately upon the occurrence of a Holder Insolvency Event (in such case, a "Holder Default"), the Servicer may terminate its obligations hereunder, subject only to its duty to cooperate with the Holder pursuant to Article Seven. The Servicer shall not be liable to the Holder for any loss, cost, or expense suffered by the Holder and arising from the proper exercise of the right of set-off herein granted or the proper termination of Servicer's obligations because of a Holder Default. The Servicer's remedies are intended to be cumulative and the exercise or failure to exercise any one such remedy on a particular occasion shall not affect the Servicer's right to exercise any other remedy on that occasion or the Servicer's right to exercise any one or more such remedies on any other occasion.

SECTION 8.3 <u>Special Services</u>. If special services are requested by the Holder, the Holder and the Servicer shall agree on pricing in advance. The Holder shall reimburse the Servicer for all special services agreed to be performed by the Servicer in accordance with the applicable terms, if any, of Attachment 1 hereto.

ARTICLE NINE BREACH AND DEFAULT

SECTION 9.1 Material Breach. For the purposes of this Servicing Agreement, a "Material Breach" is (i) a failure by the Servicer to materially perform any of its obligations hereunder, or (ii) a failure by a subservicer or subcontractor to materially comply with a servicing requirement set forth in this Servicing Agreement, and the corresponding inability of

the Servicer to comply on its own with such requirement or to force compliance by such subservicer or subcontractor. With respect to any Service Performance Measure, no breach of a Service Performance Measure shall be deemed a Material Breach unless it is the sole proximate cause of a clearly demonstrable loss to the Holder in excess of \$10,000.00 or is part of a persistent pattern of non-compliance with the Service Performance Measures such that the Holder is deprived of the benefit of its bargain. The Servicer shall be afforded in all cases notice and the opportunity to cure as provided in Section 12.4 below.

SECTION 9.2 Servicer Default. If any one of the following events (a "Servicer Default") shall occur and be continuing:

- (a) a Material Breach by the Servicer that has not been cured in accordance with Section 12.4; or
- (b) a Servicer Insolvency Event;

then, so long as the Servicer Default shall not have been remedied, the Holder may terminate this Servicing Agreement upon thirty (30) days' written notice to the Servicer (or in the case of a Servicer Insolvency Event, immediately upon written notice) and may initiate the transfer process pursuant to the applicable termination and transfer provisions of this Servicing Agreement.

ARTICLE TEN LIABILITY AND INDEMNIFICATION

SECTION 10.1 Servicer's Liability. The Servicer shall be liable to the Holder for any claim, loss, cost, damage, liability or expense, including any claim, loss, cost, damage, liability or expense that may be imposed on, incurred by or asserted against the Holder resulting from Servicer's negligence, willful misfeasance, bad faith, or breach of any obligation, representation, warranty, or covenant under this Servicing Agreement. Other than in the event of the Servicer's willful misfeasance or bad faith, the Servicer's liability under this Servicing Agreement shall be limited to the lesser of (a) the total amount of the compensation theretofore paid to the Servicer under Article Eight of this Servicing Agreement or (b) \$10 million. In no event shall the Servicer be liable to the Holder for consequential, exemplary or punitive damages.

The Servicer shall not be liable to the Holder pursuant to this Section 10.1 or otherwise for any loss or damage incurred by the Holder arising from any incorrect or incomplete information provided by the Holder, or for the failure of the Holder to comply with its obligations. The Servicer may rely in good faith on any information or document of any kind reasonably believed by the Servicer to be properly executed and submitted by an authorized person respecting any matters arising under this Servicing Agreement. In addition, by way of further limitation, where the exercise of discretion by the Servicer in order to carry out its duties is required or permitted hereunder, the Servicer shall not be liable for any error of judgment made in the good faith exercise of such discretion.

Except as otherwise provided in this Servicing Agreement, the Servicer shall not be under any obligation to appear in, prosecute or defend any legal action where it is not named as a party; provided, however, that the Servicer may undertake any reasonable action that it may deem necessary or desirable in respect of this Servicing Agreement and the rights and duties of the parties hereunder.

SECTION 10.2 <u>Indemnification by Servicer</u>. The Servicer, after notice and opportunity to cure as provided in Section 12.4, below, will indemnify and hold the Holder harmless from any claim, loss, cost, or expense, including reasonable attorneys' fees, and that results from (i) any material failure of the Servicer to comply with the terms of this Servicing Agreement, or (ii) the failure of any warranty given or representation made by the Servicer in this Servicing Agreement.

SECTION 10.3 Indemnification by Holder. The Holder will indemnify and hold the Servicer harmless from any loss, cost, or expense suffered or incurred by the Servicer, including reasonable attorneys' fees, that results from (i) any action of the Servicer authorized by this Servicing Agreement or otherwise authorized by the Holder, (ii) any material failure of the Holder to comply with the terms of this Servicing Agreement, or (iii) the failure of any warranty given or representation made by the Holder in this Servicing Agreement. In particular, but without limitation, to the extent that the Servicer is required pursuant to Section 10.1 to appear in or is made a defendant in any legal action or other proceeding relating to the servicing of the Loans, the Holder shall indemnify and hold the Servicer harmless from all cost, liability or expense of the Servicer (including, without limitation, reasonable attorney fees) not arising out of or relating to the failure of the Servicer to comply with the terms of this Servicing Agreement.

SECTION 10.4 Certain Rights of the Indemnifying Party. A party with the responsibility to indemnify hereunder (an "Indemnifying Party") shall control the defense or settlement of any third-party claim. The other party (the "Indemnified Party") may participate in, but not control, any defense or settlement of any third-party claim to be indemnified by the Indemnifying Party, in which event the Indemnified Party will bear its own costs and expenses with respect to such participation, except as set forth in the following sentence. Notwithstanding the foregoing, the Indemnified Party may take over the control of the defense or settlement of a third party claim if (i) the Indemnifying Party fails to appear in or to defend such claim, in which event the Indemnifying Party shall indemnify the Indemnified Party for the Indemnified Party's costs and expenses in connection with such defense or settlement or (ii) the Indemnified Party irrevocably waives its right to indemnity with respect to such third-party claim.

SECTION 10.5 Force Majeure. Notwithstanding the foregoing, if either party is rendered unable, in whole or in part, by a force not reasonably within the control of that party (including acts of God, acts of war, riots, insurrections, illegality of performance, strikes or other industrial disturbances, fires, earthquakes, hurricanes, floods and other disasters) to satisfy its obligations under this Servicing Agreement, such party shall not be deemed to have breached any such obligation upon delivery of written notice of such event to the other party hereto, for so long as such party remains unable to perform such obligation as a result of such event. If either party is unable to materially perform its obligations under this Servicing Agreement and its performance is excused pursuant to this paragraph for a period of at least thirty (30) days, the other party may terminate this Servicing Agreement upon at least three (3) business days' prior written notice.

ARTICLE ELEVEN ASSIGNMENTS AND SUBCONTRACTS

SECTION 11.1 General Rule. Except as provided in this Article Eleven, no party may assign its rights or delegate its duties under this Servicing Agreement without the prior written consent of the other party, and any such purported assignment shall be void and without effect.

SECTION 11.2 Mergers and Reorganizations. Each party agrees that upon (a) any merger or consolidation of such party into another person or entity, (b) any merger or consolidation to which such party is a party that results in the creation of another person or entity, or (c) any person or entity succeeding to the properties and assets of such party substantially as a whole, such party shall cause such person or entity (if other than such party) to execute an agreement of assumption to perform every obligation of such party hereunder.

SECTION 11.3 Sale of Loans. In the event that during the Term of this Servicing Agreement, as the same may be extended, the Holder proposes to sell any Loans serviced hereunder, the Holder shall provide to the Servicer at least thirty (30) days' prior written notice of the proposed sale, including the name of the proposed purchaser, the name of the responsible contact at the proposed purchaser, and the date of the proposed sale.

Any such sale shall be subject to:

- (i) the purchaser's having a servicing agreement with the Servicer covering the type of Loan being sold, and
- (ii) the Servicer's approval of the financial condition of the purchaser, which approval shall not be unreasonably withheld.

The Holder must pay, or cause such purchaser to pay, to Servicer a reasonable transfer fee in connection with such sale, as set forth in Attachment 1 hereto. The Holder shall indemnify the Servicer and hold the Servicer harmless from any loss, cost, or expense arising from the sale of Loans serviced hereunder and from any claim by the purchaser or purchaser's successors and assigns in conflict with the Servicer's right hereunder.

SECTION 11.4 Subcontracts; Assignment to Servicer's Legal Successor. The Servicer has the right to employ subcontractors to perform activities associated with or in support of the services provided hereunder. Notwithstanding any provision of this Servicing Agreement that requires Servicer to perform any particular service for the Holder, or to comply with any particular Service Performance Measures, or to perform any other obligation hereunder, or to accept liability for any aspect of servicing such Loans, if servicing of the applicable Loans is performed by any subservicer or subcontractor pursuant to a subservicing agreement or subcontracting agreement and the subservicing agreement or subcontracting agreement does not require the subservicer or subcontractor to comply with the particular Service Performance Measures set forth in this Servicing Agreement, or to perform any one or more of the other obligations hereunder with respect to such Loans, then if the Servicer is unable to force compliance by such subservicer or subcontractor with the requirements set forth in this Servicing Agreement, the Holder's remedy shall be limited to treating the failure as a Material Breach under Section 9.1. Except to the extent set forth earlier in this Section 11.4, the Servicer shall remain fully obligated hereunder for the performance by any subcontractor of services hereunder. The Servicer may assign its rights and delegate its duties hereunder to any legal successor of the Servicer. Upon the written request of the Holder, the Servicer shall provide

information regarding the name and qualifications of any significant subcontractor. With respect to all such subcontracts entered into after the Effective Date of this Servicing Agreement, the Servicer will obtain the prior approval of the Holder of any offshore entity that is to serve as a subcontractor if such entity is to have access to the non-public personal information of the Holder's customers, which approval will not be unreasonably withheld or delayed; *provided*, *however*, that if the Holder approves such an offshore subcontractor for the purposes of this Section 11.4 for a specific service, such subcontractor shall be deemed approved to perform other services that are reasonably defined as being within the same category of services. For example, if a subcontractor is approved to perform in-bound call center activity where the subcontractor will have access to the non-public personal information of the Holder's customers, such subcontractor shall be deemed approved to perform outbound call center activity if the scope of the Servicer's subcontractor will have access to the non-public personal information of the Holder's customers, such subcontractor shall be deemed approved to perform other account processing activities where the subcontractor will have access to the non-public personal information of the Holder's customers, such subcontractor shall be deemed approved to perform other account processing activities if the scope of the Servicer's subcontractor with such subcontractor is subsequently expanded.

ARTICLE TWELVE NOTICES AND REPRESENTATIVES; DISPUTE RESOLUTION

SECTION 12.1 <u>Notices</u>. All notices, requests and other communications hereunder shall be in writing and will be deemed to have been duly given only if delivered personally, sent by nationally recognized overnight courier, or mailed by certified mail, return receipt requested, to the parties at the following addresses:

If to the	Holder:
	-
	Attention:
	with a copy to:
	Attention:

If to the Servicer:

Navient Solutions, Inc. 11100 USA Parkway Fishers, IN 46037

Attention: Senior Vice President, Financial Institution Sales

with a copy to:

Navient Solutions, Inc. 2001 Edmund Halley Drive Reston, VA 20191 Attention: Legal Department

All such notices, requests and other communications shall (i) if delivered personally to the address as provided in this Section, be deemed given upon delivery, (ii) if delivered by nationally recognized overnight courier to the address as provided in this Section, be deemed given upon receipt, and (iii) if delivered by mail in the manner described above to the address as provided in this Section, be deemed given upon receipt (in each case regardless of whether such notice, request or other communication is received by any other person to whom a copy of such notice, request or other communication is to be delivered pursuant to this Section). Any party from time to time may change its address or other information for the purpose of notices to that party by giving notice specifying such change to the other party hereto.

SECTION 12.2 Representatives. Each of the Holder and the Servicer will designate one or more persons to consult with the other party to resolve day-to-day operating questions that arise in the performance of this Servicing Agreement and will notify the other party of the identity of the representatives so designated. The designation of such representatives may specify reasonable limitations upon their authority, but each of the representatives will have the authority, on behalf of the party represented, to resolve questions arising under the Service Performance Measures.

SECTION 12.3 Administrative Dispute Resolution. The parties will endeavor in good faith to achieve amicable resolution of all disputes and disagreements arising hereunder, if any, with a view to the avoidance of litigation. In the event that a question arises under this Servicing Agreement the resolution of which is beyond the authority of one or both of the parties' representatives designated under Section 12.2 or upon which such representatives cannot agree despite good faith consultation, the parties shall make diligent and prompt effort to resolve the question through the intervention of senior officers with plenary authority to waive an interpretive stance taken by the party represented and/or, on behalf of such party, to negotiate and agree to any amendment to this Servicing Agreement. Each party may rely on the representation of any officer of the other party of the rank of Vice President or higher (other than a representative appointed pursuant to Section 12.2) who asserts in writing that he or she is acting on behalf of that party under the provisions of this Section 12.3. In aid of this Section 12.3, whenever reasonably requested by the other party, each party will provide a current list identifying its officers by name and title.

SECTION 12.4 Opportunity to Cure. In the event of a failure by the Servicer to comply with any provision of this Servicing Agreement, no action may be commenced by the Holder to enforce such provision unless the Holder has first given written notice of such non-compliance to the Servicer in accordance with the notice provisions of this Servicing Agreement and has afforded the Servicer an opportunity to cure as specified in this Section 12.4. If the failure to comply is a failure to make a required deposit or otherwise expressly to pay money to or for the Holder (a "Monetary Breach"), the Servicer shall be accorded a period of ten (10) days after such notice to make the deposit or otherwise pay the money that is expressly required to be paid. If the failure to comply is not a Monetary Breach, the Servicer shall be entitled to a reasonable period of time after such notice in order to remedy the failure. As used in the preceding sentence, the term "reasonable period of time" means a period of time that is reasonable under

the circumstances, taking into account the nature of the failure, the ease or difficulty of an appropriate remedy, and the harm likely to be suffered with the passage of time, but in any event not less than thirty (30) days and not more than one hundred twenty (120) days after such notice. If the failure asserted by the Holder is disputed by the Servicer, the parties will use such reasonable period of time to attempt to resolve the dispute through the administrative dispute resolution procedure described in Section 12.3, above.

ARTICLE THIRTEEN MISCELLANEOUS

SECTION 13.1 Captions, Attachments and Schedules. The captions used in this Servicing Agreement are for convenient reference only and do not define, limit, describe or modify the meaning of the text of the Servicing Agreement. The attachments and schedules hereto are an integral part hereof, and all references herein to this Servicing Agreement shall be deemed to refer as well to such attachments and schedules.

SECTION 13.2 Confidentiality; Servicer's Press Releases and other Public Announcements. The contents of this Servicing Agreement, together with all supporting documentation, exhibits, attachments, schedules, and any amendments thereto which form the basis of the business relationship between the Holder and the Servicer, together with all negotiating, explanatory and interpretive materials pertaining thereto including, without limitation, all related correspondence between the parties, will be held in the strictest confidence by the Holder and the Servicer. Each party will make appropriate efforts to ensure that such information is not disclosed or otherwise discussed with any third parties except as hereunder provided, or as required for financing purposes, audit purposes, legal advice, as otherwise required by law, regulations, or as agreed to by the parties. Prior to distributing any press release pertaining to this Servicing Agreement or the relationship between the parties, the Holder will disclose the proposed contents thereof to the Servicer and will consult with the Servicer as to the accuracy and advisability of such press release or other public announcement.

SECTION 13.3 <u>Liability of Individuals</u>. No director, officer, or employee of any party (or of the general partners in such party) shall be subject to any liability for errors in judgment or for any action taken, or for refraining from taking any action, provided such error, action or inaction was made in good faith.

SECTION 13.4 <u>Relationship of Parties</u>. The Servicer is an independent contractor and, except for the services to be performed as set forth herein, is not and will not hold itself out as an agent of the Holder. Nothing herein contained shall create or imply a general agency relationship between the Servicer and the Holder, nor shall this Servicing Agreement be deemed to constitute a joint venture or partnership between the parties.

SECTION 13.5 Counterparts. This Servicing Agreement may be executed in counterparts each of which shall be an original and all of which shall constitute but one and the same instrument.

SECTION 13.6 Severability. If a court of competent jurisdiction holds any provision of this Servicing Agreement inoperative or unenforceable either generally or as applicable in any particular situation, it is the intention of the parties that such invalid or unenforceable provisions be reduced in scope or eliminated by the court, but only to the extent deemed necessary by the court to render the provisions of this Servicing Agreement reasonable and enforceable.

SECTION 13.7 <u>Remedies Not Exclusive</u>. No remedy by the terms of this Servicing Agreement conferred upon or reserved to either party hereto is intended to be exclusive of any other remedy existing at law or in equity.

SECTION 13.8 Non-Discrimination. In servicing Loans pursuant to this Servicing Agreement, the Servicer will not discriminate on the basis of race, color, religion, national origin, sex or marital status, or age (provided the applicant has the capacity to contract), or because all or part of the applicant's income derives from any public assistance program, or because the applicant has in good faith exercised any right under the Consumer Credit Protection Act.

SECTION 13.9 Lending Policy. The Lending Policies attached hereto as Attachment 5 are hereby incorporated herein as the applicable Lending Policies as of the Effective Date, and the Lending Policies, as amended from time to time, are also hereby incorporated as part of this Servicing Agreement. The Lending Policies must be able to be reasonably administered by the Servicer, be consistent with the provisions of this Servicing Agreement, Regulations, the applicable Program Rules and applicable law, and cannot be amended without sufficient prior notice to the Servicer to allow implementation of any servicing modifications. The Lending Policies will be reviewed and updated by the Holder if deemed necessary on at least an annual basis or as required by Regulations or applicable law or as any Program Rule changes may occur. To the extent that a change in the Holder's Lending Policies require the Servicer to implement operational changes or systems modifications, the cost of such changes and modifications may be borne by the Holder.

SECTION 13.10 Governing Law. This Servicing Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without regard for principles of conflicts of law that would require the application of the law of a jurisdiction other than the State of Delaware.

SECTION 13.11 Entire Agreement; No Amendment Except in Writing. This Servicing Agreement and the agreements referred to herein shall constitute the entire agreement between the parties with respect to the subject matter hereof and supersede all prior agreements between the parties with respect to such subject matter. This Servicing Agreement may not be amended except by a subsequent written agreement signed by the parties.

SECTION 13.12 Waiver. The waiver or failure of a party to exercise in any respect any right provided for herein shall not be deemed a waiver of any further right herein.

ARTICLE FOURTEEN SUPPLEMENTARY PRODUCTS AND SERVICES

SECTION 14.1 Supplementary Products and Services. The Servicer and the Holder may agree from time to time, in writing, for the Servicer to support supplemental lending and servicing products and services for the benefit of the Holder and the Holder's borrowers. Except as the parties may subsequently agree in writing, however, no other supplemental product or service will be supported by the Servicer.

[SIGNATURE PAGE FOLLOWS.]

IN WITNESS WHEREOF, the parties hereto have caused this Servicing Agreement to be duly executed and delivered as of the Effective Date.

[HOLDER NAME]	NAVIENT SOLUTIONS, INC.
By:	Ву:
Name:	Name:
Title:	Title:

LIST OF ATTACHMENTS

ATTACHMENT 1: Monthly Fees and Activity Based Charges Schedule

ATTACHMENT 2: Service Performance Measures

ATTACHMENT 3: Servicer Reporting

ATTACHMENT 4: Data Access for Owners and Borrowers

ATTACHMENT 5: Lending Policy

ATTACHMENT 2: Service Performance Measures

As provided for in Section 6.2 of the Servicing Agreement, the following outlines quantifiable and minimum allowable Service Performance Measures to be provided by Servicer. Generally, the Servicer agrees to report performance to the Holder by the end of the month immediately following the reported period's end. For example, an annual report for period ending 12/31/14 will be provided by 01/31/15.

Activity	Performance Goal
Forms Processing Time - Hardcopy	95% within 5 days 100% within 10 days
Borrower Inquires Processing Time - Hardcopy	95% within 5 days 100% within 10 days
Borrower Inquiries Processing Time - Internet	95% within 3 days 100% within 10 days
Enrollment Updates Processing Time ¹	85% within 30 days 100% within 45 days
Borrower Payments Deposited	99% of all identifiable borrower payments deposited/posted within 2 business days; 100% within 3 days
Servicing CCO 50/20 Service Level ²	50% within 20 seconds

¹ Enrollment updates are prioritized by loan status; items processed beyond 30 days are those in, and remaining in, an interim status.

² Call Center 50/20 Service Level defined as "Percent of Operator Calls Answered Within First 20 Seconds".

ATTACHMENT 3: Servicer Reporting

Standard Monthly Reporting Applicable to the Serviced Loans

The Servicer will make available to the Holder on a monthly basis a standard set of reports. Reports will be available in the format described in the table below. Reconciliation reporting will be available by the 15th business day of the month. Most other reports will be available by the 6th business day of the month.

Report Number	Report Name	Media	Report Description
OPSF0002	Daily Activity Summary Report	On line access	Daily summary of program balance and accounting transactions
OPSF0003	Monthly Activity Summary Report	On line access	Summary of program balance and accounting transactions at month-end
OPSF0004	Activity Detail Report	On line access	Daily detail of program balance and accounting transactions
OPSF0005	Monthly Activity Detail Report	On line access	Monthly detail of program balance and accounting transactions
OPSF0006	Monthly Delinquency Summary Report	On line access	Summary of outstanding principal and borrower interest by delinquency or claim status
OPSF0007	Monthly Characteristics Summary Report	On line access	Summary of outstanding principal by status and terms of maturity
OPSF0008	Monthly Borrower Detail Report	On line access	Detail of borrower/loan activity at account level, including loan balance
OPSF0009	Monthly Delinquency Detail Report	On line access	Detail of outstanding principal and borrower interest by delinquency or claim status
LFHOW01	Owner Daily Cash Transactions Posted	On line access	Daily cash totals and details for payment, refund, and reversal transactions
N/A	Servicing Invoice	Hardcopy	Monthly invoice detailing servicing fees for the prior month.

Ad Hoc Reports

Ad hoc reporting may be provided upon agreement with Servicer. The Holder will be charged for such reports in accordance with the schedule of fees in Attachment 1.

ATTACHMENT 4: Data Access for Owners and Borrowers

The Holder can access borrower and loan information on the Servicer's FDR System through the following means:

- (1) Internet access using an internet browser
- (2) Data updates or extract files delivered in a machine-readable format

In addition, Borrowers have limited internet access to certain account information.

5. Internet Account Access for Holder

The Holder has the option of using a standard Internet browser to access individual borrower account detail by logging on to the Servicer's—www.navient.com. If the Holder desires this type of access, it must first establish a user ID and password.

6. Data Updates/Extract Files

The Servicer can provide monthly extracts of FDR servicing data for transfer to the Holder in machine-readable format. The following fields represent a small portion of the detail available on each loan.

- · Social Security Number
- Full Name
- · Date of Birth
- · Address and Telephone Number
- · School
- · Total Disbursed Principal
- Payment Amount and Payment Due Date
- Current Principal Outstanding
- Current Accrued Interest
- Deferment History
- Cosigner/Co-maker/Reference
- Claim Detail

7. Internet Account Access for Borrowers

Borrowers can also establish a user ID and password to access their individual account information.

ATTACHMENT 5: Lending Policy

As agreed to by the Holder and the Servicer, in accordance with the guidelines set forth in the applicable Private Loan Program document developed by the Holder.



NEWS RELEASE

FOR IMMEDIATE RELEASE

Sallie Mae Completes Strategic Separation, Launches New Sallie Mae As Standalone Consumer Banking Business

Company to Carry Forward Mission of Helping Families Save, Plan and Pay for College With Emphasis on Responsible Financial Practices

NEWARK, Del., May 1, 2014—Sallie Mae (NASDAQ: SLM), formally SLM Corporation, announced today that it has completed its strategic separation and is now a standalone consumer banking business with a mission of helping families save, plan and pay responsibly for college.

The company offers responsible private education loans, saving products and insurance options to students and families to facilitate their investment in higher education. Its Smart Option Student Loan rewards customers for minimizing loan costs and includes consumer safeguards such as cosigner participation, disciplined underwriting, school certification, in-school payments and proactive customer contact.

"With a 40-year foundation of experience, today Sallie Mae begins anew as a bank that emphasizes responsible consumer financial practices to benefit our customers," said Raymond Quinlan, Chief Executive Officer. "The new Sallie Mae is squarely oriented toward meeting our customers' needs and building long-term relationships. Our leadership includes a mix of tenured education lending and banking talent who share my vision for growing the franchise."

Sallie Mae maintains a strong capital position and funding capabilities, and the company now manages a portfolio of approximately \$6.5 billion high-quality, private education loans.

Customers with student loans serviced by Sallie Mae will continue to conduct business as they do today, with no changes until the fall of 2014. This spring and summer, they will receive personalized information about their account and advance notice of any changes needed to manage their account.

Forward-Looking Statements

This press release contains statements that constitute "forward-looking statements" within the meaning of the Securities Act of 1933 and the Securities Exchange Act of 1934 as amended by the Private Securities Litigation Reform Act of 1995. These forward-looking statements include, among others, statements about the risk factors related to the spin-off of Navient, and the development of trading markets after the spin. These statements are based on the current expectations of Sallie Mae management and are subject to known and unknown risks and uncertainties. There are a number of risks and uncertainties that could cause actual results to differ materially from those contemplated by the forward-looking statements, including, but not limited to, uncertainties and risks related to the spin-off. For further information regarding the risks associated with Salle Mae's businesses and the spin-off

of Navient, please refer to the Sallie Mae's filings with the SEC, including its most recent Annual Report on Form 10-K. Sallie Mae assumes no obligation to update any forward-looking statements as a result of new information, future developments or otherwise.

Sallie Mae (NASDAQ: SLM) is the nation's No. 1 financial services company specializing in education. Whether college is a long way off or just around the corner, Sallie Mae turns education dreams into reality for American families. With products and services that include Upromise rewards, scholarship search and planning tools, private education loans, insurance, and online banking, Sallie Mae offers solutions that help families save, plan, and pay for college. Learn more at SallieMae.com. Commonly known as Sallie Mae, SLM Corporation and its subsidiaries are not sponsored by or agencies of the United States of America.

Contacts:

Journalists and members of the media may contact: Martha Holler, 302-283-4036, <u>Martha.Holler@SallieMae.com</u>

Investors may contact: Brian Cronin, 302-283-8435, <u>Brian.Cronin@SallieMae.com</u>

Customers may contact: 888-2-SALLIE (888-272-5543)

###